

Friedrich-Alexander-Universität Professorship for Open-Source Software

Friedrich-Alexander-Universität Erlangen-Nürnberg Technische Fakultät, Department Informatik

Betül Durmuş Bachelor Thesis

# A Literature Review of Open-Source Foundations and Open-Source Projects in Healthcare

Submitted on 13.01.2025

Supervisors: Prof. Dr. Dirk Riehle, M.B.A.; Elçin Yenişen Yavuz, M.Sc. Professur für Open-Source-Software Department Informatik, Technische Fakultät Friedrich-Alexander University Erlangen-Nürnberg

### Versicherung

Ich versichere, dass ich die Arbeit ohne fremde Hilfe und ohne Benutzung anderer als der angegebenen Quellen angefertigt habe und dass die Arbeit in gleicher oder ähnlicher Form noch keiner anderen Prüfungsbehörde vorgelegen hat und von dieser als Teil einer Prüfungsleistung angenommen wurde. Alle Ausführungen, die wörtlich oder sinngemäß übernommen wurden, sind als solche gekennzeichnet.

Erlangen, 13.01.2025

### License

This work is licensed under the Creative Commons Attribution 4.0 International license (CC BY 4.0), see <u>https://creativecommons.org/licenses/by/4.0/</u>

Erlangen, 13.01.2025

## Abstract

Open-Source Software has been increasingly implemented in the healthcare industry, providing advantages such as interoperability, cost-efficiency and adaptability. This study conducts a systematic literature review (SLR) to identify Open-Source Projects and Foundations in healthcare. It provides an overview of research trends, projects and foundations related to Open-Source Software in the healthcare sector from 2016 to 2023. The review examines 217 papers collected from IEEE Xplore, ACM Digital Library, Scopus and Springer. The findings show that while Open-Source Projects like "OpenMRS", "3D Slicer" and "VistA" are well researched, Open-Source Foundations remain mainly unexplored. Key topics of the investigated literature include "Health Data Management", "Medical Imaging" and "Digital Health". There is a gap of research on niche topics like "Genomic Medicine" and "Diabetes Care". The study provides a general understanding of the current landscape, showing directions for future research.

# Contents

1.	Introduction	1
2.	Related Work	2
	2.1 Open-Source Software Projects in Healthcare	2
	2.2 Open-Source Foundations in Healthcare	3
3.	Research Questions	4
4.	Research Design	5
	4.1 Planning the Review	5
	4.2 Conducting the Review	5
	4.2.1 Identification of Research	6
	4.2.2 Selection of Studies	6
	4.2.3 Data Extraction	8
	4.3 Reporting the Review	8
5.	Research Results	9
	5.1 Research Trends Regarding Open-Source Foundations and Projects in Healthcare Between 2016 and 2023	9
	5.2 Open-Source Foundations Explored in the Literature Between 2016 and 2023	11
	5.3 Open-Source Projects Explored in the Literature Between 2016 and 2023	14
6.	Discussion	21
	6.1 Research Trends	21
	6.2 Open-Source Foundations	21
	6.3 Open-Source Projects	22
7.	Limitations	23
8.	Conclusion	24
Re	eferences	25
Aj	ppendix A-1	29
Aj	ppendix A-2	33
Aj	opendix B	36
Aj	ppendix C	78

# **List of Figures**

Figure 1 Search and Filtering Process for Research Papers	.7
Figure 2 Annual Distribution of Included Papers	.9
Figure 3 Research Topic Distribution in Relevant Papers	10
Figure 4 Comparison of Open-Source Topics in Healthcare (2016-2019 and 2020-2023)1	11
Figure 5 Open-Source Foundations in Included Papers	11
Figure 6 Open-Source Projects in Included Papers	15
Figure 7 Research Topics within 3D Slicer Projects	17
Figure 8 Research Topics within OpenMRS Projects	19
Figure 9 Research Topics within VistA Projects	20

# List of Tables

Table 1 SLR Stages and Specific Steps	5
Table 2 Search Engine Results	6
Table 3 Distribution of Journals and Conferences	10
Table 4 Overview of Related Studies for Open-Source Foundations	12
Table 5 Overview of Related Studies for Open-Source Healthcare Projects	15

# 1. Introduction

Open-Source Software (OSS) has existed since the 1950s, primarily used for software and computer applications and has grown in popularity ever since (Waring & Maddocks, 2005). In the late 1990s, when healthcare providers began focusing on creating integrated-care delivery systems, conventional software failed to meet these demands. OSS posed as an alternative, offering system compatibility and cost reduction (Goulde & Brown, 2006; Bagayoko et al., 2010; Syzdykova et al., 2017).

The growing significance of open-source software projects in the healthcare industry led to increased academic research in diverse areas including electronic health records (EHR), medical 3D printing and data management tools (Iribarren et al., 2017; Mohamad et al., 2022; Sobral et al., 2021; Van Biesen et al., 2022). Despite the increased research and growing number of OSS projects in the healthcare sector, there is a lack of literature reviews or systematic mapping studies that analyze the scope of existing research in this domain.

This article reviews the current state of research on OSS in healthcare. Our research questions are:

- **RQ1:** What are the **research trends** about open-source foundations and projects in healthcare between 2016 and 2023?
- **RQ2:** Which **open-source foundations** are investigated in the literature between 2016 and 2023?
- **RQ3:** Which **open-source projects** are investigated in the literature between 2016 and 2023?

To address these research questions, a systematic literature review was conducted across four databases, covering the period from 2016 to 2023. This review aims to provide a comprehensive overview of the current state of the literature, offering valuable insights to guide researchers in identifying promising areas for future investigation. Furthermore, medical professionals may utilize this review to explore available tools and projects that could be integrated into their clinical practice.

The rest of this paper is organized as follows: Section 2 provides an overview of related work. Section 3 explains the research questions that guided our study. Section 4 outlines the methodology used for the systematic literature review and Section 5 presents the results of our research. In Section 6, we discuss the implications of these findings and in Section 7 the limitations of this review are discussed. Finally, Section 8 concludes the paper.

# 2. Related Work

This section provides an overview of prior research relevant to the topic. Subsection 2.1 focuses on open-source software (OSS) projects in the healthcare sector and section 2.2 examines open-source foundations and their role in supporting OSS development within healthcare.

#### 2.1 Open-Source Software Projects in Healthcare

One of the earliest open-source software (OSS) projects in healthcare was COSTAR, introduced in the late 1970s when Octo Barnett freely distributed the source code for the COSTAR ambulatory medical record. This groundbreaking initiative marked the beginning of a series of OSS projects in the healthcare sector (McDonald et al., 2003). Open-source projects have since significantly contributed to the healthcare field by enabling different systems to connect seamlessly, fostering an online community of informaticians, and advancing the development of interoperable and standardized electronic health records (Frade et al., 2013; Nagy, 2007; Oyri & Murray, 2005).

Researchers have mostly been focusing on open-source electronic health records (OS-EHR). Purkayastha et al. (2019) compared five OS-EHR systems with each other. These five systems were *GNUHealth*, *OpenEMR*, *OpenEHR-based EHR*, *OpenMRS* and *VistA EHR*. The authors evaluated these systems based on 32 criteria, including interoperability, privacy and security, and whether they meet essential healthcare standards outlined by the Institute of Medicine, such as ensuring patient safety and providing effective care. OpenEMR performed the best, meeting all 32 criteria, followed by VistA EHR meeting 28 criteria. OpenMRS met 12 criteria fully and 11 partially, OpenEHR-based EHR met 10 fully and 3 partially, and GNUHealth met the least with only 10 criteria fully and 2 partially. Despite this, OpenMRS stood out for its better user performance, such as faster patient searches and navigation. Additionally, the study identified geographic trends, with OpenEHR gaining popularity in Northern Europe and Australia, OpenMRS in Africa and South Asia, and OpenEMR in the United States and Brazil (Purkayastha et al., 2019).

Aminpour et al. (2014) conducted a systematic review to examine the utilization of open-source electronic health records (OS-EHR) worldwide. The study identified 13 OS-EHR systems: *AMRS, Androbase, Hospital OS, iSante, iTrust, OpenEHR, OpenMRS, OpenVistA, OSCAR, OSCC, PING, PropeRweb,* and *WorldVistA*. They evaluated the countries where these systems were utilized, such as AMRS in Kenya and the USA, OpenMRS across 19 countries primarily in Africa and South Asia, and OpenEHR in Australia, Sweden, and Uruguay. The primary reasons for their adoption varied, including cost-efficiency (e.g., Hospital OS, iSante), flexibility to modify and improve (e.g., OpenEHR, OpenVistA), improving workflows and usability (e.g., Androbase), and enhancing data integration and communication (e.g., OSCC). These findings highlight how OS-EHR systems address diverse needs, particularly in resource-limited regions (Aminpour et al., 2014).

Syzdykova et al. (2017) conducted a systematic review of open-source electronic health record (OS-EHR) systems, focusing on their implementation in low-resource settings. The study evaluated five OS-EHR systems—*GNU Health, OpenEMR, FreeMED, OpenMRS,* and *Bahmni*—using 21 key features such as interoperability, modularity, development activity, and community support. Bahmni stood out for its comprehensive integration of different systems, while OpenMRS excelled in modularity and customization for resource-constrained

environments. OpenEMR offered strong interoperability and multilingual support, making it suitable for diverse settings. FreeMED was praised for its simplicity but lacked offline support, and GNU Health emphasized social determinants of health with its patient and health center management features. Despite their benefits, the study highlighted common challenges such as lack of offline functionality and limited training resources. This study's findings aim to guide the selection and implementation of OS-EHR solutions in resource-constrained environments (Syzdykova et al., 2017).

Iribarren et al. (2017) conducted a scoping review to evaluate SMS/text messaging platforms for mobile health (mHealth) interventions. The study initially identified 27 platforms but included only 22 in the final evaluation, focusing on features such as interoperability, ease of setup, data security, and suitability for low-resource settings. Of these platforms, 14 were open source, 10 were healthcare-specific, and 16 targeted low-resource environments. 15 platforms supported a do-it-yourself setup, while the remainder required programming or customization by vendors. Frequently described features included data security and access via cloud-based systems. Notable platforms included CommCare, FrontlineSMS, and RapidSMS. Challenges included limited offline functionality, high costs for scaling, and insufficient documentation on platform-specific strengths and weaknesses. This review offers valuable insights for researchers and healthcare professionals aiming to integrate SMS-based interventions into healthcare programs effectively (Iribarren et al., 2017).

The OSS community in the healthcare sector is thriving, contributing to diverse research areas. However, the adoption and implementation of OSS in healthcare still face challenges, including ensuring interoperability between systems, compliance with healthcare standards, and integration into clinical workflows (Janamanchi et al., 2009). Despite these obstacles, OSS shows significant potential for addressing key healthcare needs and continues to be a focus of research and innovation. Nonetheless, the limited availability of systematic literature reviews in this field underlines the need for a comprehensive overview of existing research.

#### 2.2 Open-Source Foundations in Healthcare

Open-source software foundations (OSSFs) are non-profit organizations that provide governance structures, financial resources, and legal frameworks for open-source projects. Their responsibilities include organizing the developer community, managing intellectual property rights, setting strategic directions, and ensuring the long-term viability of their projects. These structures ensure their sustainability and independence from individual contributors or single companies (Riehle, 2010; Riehle & Berschneider, 2012; Prattico, 2012).

Despite their widespread application in technology sectors, the role of OSSFs in healthcare remains underexplored. Further exploration of OSSFs in healthcare could reveal new opportunities for innovation, collaboration, and accessibility.

# **3. Research Questions**

This review aims to evaluate the current state of research of open-source software in the healthcare sector. To achieve this, the review is guided by the following three research questions.

**RQ1:** What are the research trends regarding open-source foundations and projects in healthcare between 2016 and 2023?

This question aims to reveal key research topics. By analyzing these trends, we aimed to provide insights into the development of open-source healthcare research, highlighting trends and identifying areas that have not been researched yet.

**RQ2:** Which open-source foundations have been explored in the literature between 2016 and 2023?

Examining the foundations that support open-source projects is key to understanding the support structure behind those projects. This evaluation seeks to assess the role of open-source foundations in fostering healthcare innovation and explore their impact on research.

**RQ3:** Which open-source projects have been studied in the literature between 2016 and 2023? Identifying the open-source projects evaluated in the literature provides insights into the available technology. By addressing this question, we aimed to show medical professionals available open-source solutions for specific needs and to provide an understanding about the state of research to researchers which projects have been investigated in the literature.

# 4. Research Design

We followed the guideline of Kitchenham (2004) for our systematic literature review (SLR). The systematic review by Kitchenham (2004) consists of three main stages: *Planning, Conducting* and *Reporting* the study. Each stage is structured with specific steps to ensure rigor and transparency (Table 1).

SLR Stages	Specific Steps		
Planning	<ul> <li>→ Justifying the need for the review</li> <li>→ Developing a protocol, including research questions and search strategies</li> </ul>		
Conducting	<ul> <li>→ Identifying and selecting relevant studies using defined inclusion and exclusion criteria</li> <li>→ Reviewing the relevance of selected studies and extracting data systematically</li> <li>→ Synthesizing data to answer research questions</li> </ul>		
Reporting	→ Presenting the methodology and findings transparently in a structured format.		

Table 1 SLR Stages	and Specific	Steps
--------------------	--------------	-------

#### 4.1 Planning the Review

The planning phase began with identifying the need for the review, as detailed in *Section 1: Introduction* and *Section 2: Related Work.* Given the growing application of open-source software in healthcare, this review aimed to address gaps in understanding research trends, projects, and foundations in this domain.

A review protocol was developed to guide the process. This included Research Questions (Section 3: Research Questions) and a search strategy with the specific terms "open source software" AND "healthcare". The full protocol can be found in Appendix C.

#### 4.2 Conducting the Review

This phase includes identifying and selecting the primary studies, applying inclusion and exclusion criteria, and systematically extracting and synthesizing data to address the research questions.

#### 4.2.1 Identification of Research

The identification of primary studies was conducted in two stages. In May 2024, we performed the first search in the databases ACM Digital Library, IEEE Xplore, and Scopus using the search string "open source software" AND "healthcare". This search was applied to titles, keywords, and abstracts, but conference papers were excluded at this stage to focus on journal articles. This search resulted in a total of 968 papers.

In October 2024, a second search was conducted to expand the scope of the review. This search included the Springer database and incorporated conference proceedings that were previously excluded. By broadening the sources and including additional publication types, this stage ensured a more comprehensive coverage of the literature.

The overall search covered publications from 2016 to 2023, a time frame chosen to reflect the most current state of research while maintaining a manageable scope. Across both searches, a total of 2,202 papers were identified (Table 2).

Database	Number of Primary Search Results
ACM Digital Library	557
IEEE Xplore	157
Scopus	256
Springer	1232
total	2202

Table 2	Search	Engine	Results
---------	--------	--------	---------

#### 4.2.2 Selection of Studies

To identify the publications relevant to our research questions, firstly we defined our inclusion and exclusion selection criteria.

Our inclusion criteria are:

- Papers that are written in English
- Papers that are peer-reviewed
- Papers that focus on open-source software in the healthcare sector

Our exclusion criteria are:

- Papers that are duplicate
- Papers that are inaccessible
- Paper that are flagged as "not reliable" by the database
- Studies that are not peer-reviewed research papers

Articles that did not meet the inclusion criteria or met the exclusion criteria were excluded.

The filtering process was conducted in three systematic steps to ensure a transparent selection of relevant studies (Figure 1). In *Step 1*, duplicate entries (273) and papers not written in English (7) were excluded. In *Step 2*, further exclusions were applied. Papers that were inaccessible or flagged as "unreliable" by the database (20), papers that did not meet the type criteria (199), and papers with mismatching keywords (1,437) were removed. This step resulted in a reduced dataset of 266 papers. In *Step 3*, a full-text review was performed on the selected papers to evaluate their relevance in greater detail. During this step, papers not focusing on OSS (49) were excluded, leaving 217 papers for inclusion in the data extraction process.



Figure 1 Search and Filtering Process for Research Papers

#### 4.2.3 Data Extraction

We extracted key information from each paper. These information are the

- Search Venue (one of the four databases)
- Number of citations
- Author(s)
- Title of the paper
- Type of content (conference paper or article)
- Year of publication
- Article URL

For each paper, the status (excluded or included) was documented. For excluded papers, the specific reason for exclusion was noted to ensure transparency of the selection process. For included papers, the following information was systematically extracted:

- Short summary of the paper
- Publication Venue
- Key healthcare area researched
- (If mentioned) open-source project investigated
- (If mentioned) open-source foundation evaluated

To ensure clarity and reproducibility, "query date" was recorded. The extracted data was organized in a data extraction table using Google Sheets, providing a clear and structured overview of the included studies. This approach facilitated the systematic documentation of findings and supported the synthesis process.

#### 4.3 Reporting the Review

The reporting is presented in this thesis, with further detailed documentation available in Appendix A-1, A-2, and B. The full list of reviewed papers is divided into two appendices: Appendix A-1, which presents the Search Venue, Query Date, Author, Title, Type, Status, Journal, and Year of Publication of the included papers; and Appendix A-2, which provides the Title, Topic (a short summary of the paper), Topic Keywords, and Article URL of the included publications. Appendix B contains the full list of excluded papers.

## 5. Research Results

By extracting and analyzing the data, we addressed our research questions. In Section 5.1, we present the research trends observed in the literature and provide answers to RQ1. In Section 5.2, we present the open-source foundations to answer RQ2. Finally, in Section 5.3, we evaluate the identified open-source projects and address RQ3.

# 5.1 Research Trends Regarding Open-Source Foundations and Projects in Healthcare Between 2016 and 2023

Figure 2 illustrates the annual distribution of academic papers included in this literature review from 2016 to 2023. The number of included papers started at 17 in 2016 and increased to 19 in 2017, followed by a further rise to 22 in 2018. In 2019, the number reached 28 papers and slightly decreased to 27 in 2020.





Figure 2 Annual Distribution of Included Papers

Table 3 shows the distribution of included papers across various journals and conferences. The venues are listed alongside the number of papers published in each. The most frequent publication venue is the *Journal of the American Medical Informatics Association*, with 7 relevant articles, followed by the *Conference on Human Factors in Computing Systems*, which accounts for 6 papers. Other notable venues include the *Journal of Digital Imaging* with 4 papers and *Human-Computer Interaction*, *IEEE Access, International Journal of Medical Informatics and Decision Making*, each with 3 publications. Journals and conferences with only one relevant article were excluded from this table. In total, 157 different journals and conferences were included in our research which published papers related to OSS in healthcare. Out of the 217 collected papers, 98 are research articles and 119 conference papers.

Venues	Count	Venues	Count
Journal of the American Medical Informatics Association	7	ACM International Conference on Bioinformatics, Computational Biology and Health Informatics (BCB)	2
Conference on Human Factors in Computing Systems	6	Winter Simulation Conference	2
Journal of Digital Imaging	4	Neumann Colloquium (NC)	2
Human-Computer Interaction	3	IEEE Global Humanitarian Technology Conference (GHTC)	2
IEEE Access 3		IEEE International Conference on Big Data	2
International Journal of Medical Informatics	3	International Conference on Informatics, Electronics & Vision	2
Lecture Notes in Computer Science	3	IEEE Computer Software and Applications Conference	2
BMC Medical Informatics and Decision Making	3	(COMPSAC)	2
International Symposium on Open Collaboration Companion	2	Sensors	2
(OpenSym)		Yearbook of Medical Informatics	2
International Conference on Digital Health	2	Diagnostics, MDPI	2
International Conference on Information and Communication Technologies and Development (ICTD)	2	Computer Methods and Programs in BioMedicine	2

Table 3	3 Distribution	of Journals	and Conferences
10010 .	Distriction	0,00000000	and conjerences

We also analyzed the different research topics to discover trends. Figure 3 presents the research topics of the included papers with the counts of published papers about this topic. *Health Data Management* is the most researched topic with 30 papers, followed by *Medical Imaging* with 24 and *Digital Health* with 23 papers. Other significant topics are *IT Infrastructure* with 22 published papers and *EHR* with 16 papers. In total, 20 different topics are investigated in our research scope.



Figure 3 Research Topic Distribution in Relevant Papers

We analyzed the distribution of research topics across two time periods: 2016–2019 and 2020–2023 and presented it in Figure 4. The charts on the left display the topics studied between 2016 and 2019, with "*Health Data Management*" leading at 16 papers, followed by "*IT Infrastructure*" (13 papers) and "*Electronic Health Records*" (12 papers). During this period, a total of 19 distinct research areas were explored. In the subsequent period, 2020–2023, "*Medical Imaging*" emerged as a key focus with 15 research papers, closely followed by "*Health Data Management*" (14 papers) and "*Digital Health*" (13 papers). Additional prominent topics included "*COVID-19*" and "*Public Health*." Overall, 20 unique topics were examined during the second time frame.



Figure 4 Comparison of Open-Source Topics in Healthcare (2016-2019 and 2020-2023)

# 5.2 Open-Source Foundations Explored in the Literature Between 2016 and 2023

We observed six open-source foundations explored in the literature. These are: openEHR Foundation, Open Health Imaging Foundation, LibreHealth, OpenHIE, IMIA Open Source Working Group and HL7 FHIR Foundation.

Figure 5 showcases the different open-source foundations in healthcare that were evaluated in the included papers and Table 4 provides an overview of the literature mentioning each of these foundations.



Figure 5 Open-Source Foundations in Included Papers

Foundation Name	Related studies found in the literature			
openEHR Foundation	Hak et al. (2020), Min et al. (2018), Oliveira et al. (2016), Paton & Karopka (2017)			
Open Health Imaging Foundation (OHIF)	Chang et al. (2016), Escudero Sanchez et al. (2023), Wadali et al. (2020)			
LibreHealth	Merine et al. (2022), Purkayastha et al. (2020)			
OpenHIE	Nguyen et al. (2017), Nguyen et al. (2019)			
IMIA Open Source Working Group	Warner et al. (2016)			
HL7 FHIR Foundation	Paton et al. (2022)			

Table 4 Overview of Related Studies for Open-Source Foundations

The "openEHR Foundation" was mentioned the most, with four mentions. The "openEHR" Foundation is an international, non-profit community aiming to improve and help build reusable and interoperable electronic health records of high quality, supporting the needs of patients and clinicians all around the world (Ingram, 2002). Oliveira et al. (2016) discussed the use of openEHR compositions, supported by the openEHR Foundation, to improve the interoperability of healthcare data. The study focused on the project "Clinical Document Online Translator (CDOT)". CDOT converts HL7v2.X messages, which are standardized formats used to exchange healthcare information between systems like hospital records and laboratory systems, into openEHR compositions. This project emphasized the importance of open standards, like the OpenEHR standard provided by the openEHR foundation, in addressing fragmented hospital information systems and improving data interoperability (Oliveira et al., 2016). Hak et al. (2020) highlighted the role of the openEHR Foundation in providing a framework to support interoperability and standardized clinical data management. The authors emphasized the openEHR Clinical Knowledge Manager (CKM), an open repository provided by the foundation. CKM facilitates the creation, management, and sharing of archetypes and templates. This infrastructure supports healthcare facilities in achieving data consistency, improving clinical decision-making, and fostering collaboration among healthcare professionals (Hak et al., 2020). Min et al. (2018) conducted a study on the feasibility of using the openEHR approach to model complete electronic health records (EHRs) in a Chinese hospital. The paper focused on creating archetypes and templates to represent clinical data requirements. Most archetypes were reused from the openEHR Clinical Knowledge Manager (CKM), with some modified or newly developed to meet local needs. The study emphasized the role of the openEHR Foundation in enabling standardized and interoperable EHRs while highlighting challenges like localization, tool support, and governance in implementing openEHR solutions in emerging markets like China (Min et al., 2018). Paton and Karopka (2017) explored the role of OSS in building Learning Health Systems (LHSs), focusing on its potential to improve data sharing and decision-making in healthcare. They briefly mentioned the openEHR foundation as an international community that has also developed open source tools for modeling clinical data, supporting interoperability and secondary data use (Paton & Karopka, 2017).

The "Open Health Imaging Foundation (OHIF)" follows with three mentions. OHIF was launched in 2015 through a partnership between a Hospital's Department of Radiology and Radical Imaging. It is a non-profit organization that focuses on a web-based open-source medical imaging platform for the global community (Open Health Imaging Foundation, 2015). Wadali et al. (2020) evaluated several free, open-source, web-based DICOM viewers for their integration into the Indian National Telemedicine Service, eSanjeevani. A DICOM viewer is a software application that enables healthcare professionals to view, analyze, and manipulate medical images stored in the DICOM format, which is the global standard for medical imaging. Among these, the OHIF DICOM Viewer was mentioned as a solution for displaying medical images in web browsers. The study noted that OHIF provides general-purpose and specialized tools, such as the Lesion Tracker for oncology, showcasing its flexibility in supporting clinical imaging workflows. However, the need for a DICOM server for its full functionality posed a challenge for the targeted use case (Wadali et al., 2020). Chang et al. (2016) explained how image informatics can be used in personalized medicine and introduced open-source tools for working with medical images. They mention OHIF as a nonprofit organization focused on developing open-source clinical radiology tools, including a web-based viewer that supports displaying medical images on various platforms (Chang et al., 2016). Escudero Sanchez et al. (2023) presented a prototype pipeline for integrating artificial intelligence (AI) tools into clinical research settings, focusing on ovarian cancer as a use case. They utilized the OHIF Viewer, a web-based tool for visualizing and interacting with medical images, as part of their workflow. The OHIF Viewer was integrated with XNAT, providing a zero-footprint, userfriendly interface that enables radiologists to view and edit AI-generated image segmentations directly in their browser (Escudero Sanchez et al., 2023).

"LibreHealth" was mentioned twice in our included papers. It is a collaborative community supporting free and open-source software projects in Health IT. All discussion, project plans, and designs are open to everyone, giving the same opportunity to all. Main focuses of projects include EHR and Radiology Information Systems (RIS) (LibreHealth, n.d.). The paper by Merine et al. (2022) describes the development of the LibreHealth *Cost-of-Care Explorer*, a mobile application aimed at improving *price transparency in healthcare*. This app helps patients access hospital chargemasters in a consumer-friendly format, enabling comparison of medical procedure costs across hospitals to promote informed decision-making and affordability (Merine et al., 2022). Purkayastha et al. (2020) described the integration of the *CheXNet deep learning algorithm* into the *LibreHealth Radiology Information System (RIS)* to facilitate the *detection of thoracic diseases from chest X-rays*. The study highlighted how LibreHealth enables the use of advanced AI tools within clinical workflows, emphasizing its modular architecture and open-source accessibility. The integration of AI into LibreHealth highlights its potential to enhance diagnostics and address radiologist shortages in resource-limited settings (Purkayastha et al., 2020).

**"OpenHIE**" was also mentioned twice. It is a community, formed in 2013, that builds an open framework to support nations with the development of health information exchanges (HIEs). This enables data to be shared between databases, facilities and across countries (OpenHIE, n.d.). Nguyen et al. (2019) examined the role of OpenHIE in promoting *interoperability* and *reducing fragmentation in health information systems* in Tanzania and the Philippines. In Tanzania, it guided the development of a national Health Information Exchange (HIE), integrating shared records and facility registries. In the Philippines, OpenHIE inspired the design of the Philippine Health Information Exchange (PHIE), influencing foundational registries and governance (Nguyen et al., 2019). Nguyen et al. (2017) explored how OpenHIE helps *improve national health information systems in developing countries*. The study

highlighted OpenHIE's work in addressing fragmented health systems by creating a reusable framework based on international health data standards. This framework includes components like shared health records and facility registries, connected through an interoperability layer. OpenHIE also provides advisory services and advocates for underserved regions, promoting better integration and person-focused healthcare systems (Nguyen et al., 2017).

The "HL7 FHIR Foundation" has one mention and is a non-profit open-source community that supports global adoption and implementation of the FHIR platform standard. The foundation provides information, tools, websites and project support. The aim is to improve interoperability of health data and quality, efficiency and effectiveness of the healthcare sector (HL7 FHIR Foundation, n.d.). Warner et al. (2016) described the development of an application called *SMART Precision Cancer Medicine (SMART PCM)*, which uses the HL7 FHIR standard to integrate genomic data into clinical workflows. They highlighted how the HL7 FHIR Foundation supports interoperability by providing a framework to standardize genomic data. The app extends FHIR resources to help oncologists visualize and use genomic information in real time, demonstrating how FHIR can advance precision medicine by connecting genomic data with electronic health records (Warner et al., 2016).

**"IMIA Open Source Working Group (IMIA OS WG)"** was evaluated once. The "IMIA OS WG" was founded in 2002 as a working group of the International Medical Informatics Association. The objective of the group is raising awareness of open-source software in healthcare. They disseminate knowledge, provide a neutral collaboration platform, foster collaborations between open-source healthcare projects and lower the perceived barriers to the adoption of OSS in healthcare (IMIA, n.d.). Paton et al. (2022) provided an overview of the contributions of the IMIA OS WG to improving healthcare systems through open-source digital health tools. The study highlighted key projects like *DHIS2, OpenMRS, and openIMIS*, showcasing their role in making healthcare more resilient, accessible, and equitable. The IMIA OSWG was described as a collaborative global community promoting the adoption of open-source solutions to enhance interoperability, usability, and health equity in diverse contexts (Paton et al., 2022).

# 5.3 Open-Source Projects Explored in the Literature Between 2016 and 2023

A total of 86 open-source projects were analyzed in the collected papers, with the top 10 most frequently mentioned projects displayed in Figure 6. "3D Slicer" leads with mentions in 10 papers, followed by "OpenMRS" with 8 mentions. "VistA" ranks third with 4 mentions, while "Synthea," "Open Data Kit (ODK)," "Apache Hadoop," "Blender," "Dicoogle," and "XNAT" each appear in 3 papers. Additionally, 10 other projects were examined in 2 papers each. Notably, 96 of the collected papers did not reference any specific open-source project.



Figure 6 Open-Source Projects in Included Papers

Table 5 provides a summary of the literature that analyzed the projects 3DSlicer, OpenMRS, and VistA.

Project Name	Related studies found in the literature
3D Slicer	Bianchi et al. (2020), Fichtinger et al. (2021), Geerlings-Batt et al. (2022), Lemarteleur et al. (2020), Mandolini et al. (2022), Pakvasa et al. (2022), Quiñones Rodríguez et al. (2023), Ruiz-Alzola et al. (2018), Sharma, Vilarrubias, and Verschure (2023), Syamlan et al. (2022)
OpenMRS	Eapen et al. (2019), Goel et al. (2017), Hossain et al. (2019), Jawhari et al. (2016), Martsenyuk et al. (2019), Paton et al. (2022), Paton and Karopka (2017), Shaikh et al. (2022)
VistA	Inoue and Zhou (2016), Jones, Smith, and Patel (2020), Rajeevan et al. (2017), Shaikh et al. (2022)

Table 5 Overview of Related Studies for Open-Source Healthcare Projects

**"3D Slicer"** is an open-source desktop software used to solve advanced image computing challenges and focuses on clinical and biomedical applications. It provides users tools for visualizing, processing and analyzing medical images and encourages collaboration among developers and users worldwide (Pieper et al., 2004; Slicer, n.d.). 3D Slicer was mentioned in four different research areas in our collected papers (Figure 7). These are: medical imaging, 3D printing, education and training, and cancer care.

"Medical Imaging" makes up nearly half of the research regarding "3D Slicer" with 4 papers [Ruiz-Alzola et al., 2018; Bianchi et al., 2020; Sharma, Vilarrubias, and Verschure, 2023; Mandolini et al., 2022]. Ruiz-Alzola et al. (2018) examined the use of "3D Slicer" for developing *advanced segmentation techniques in radiology*. The study highlighted the platform's ability to integrate different imaging modalities, such as CT and MRI, into a single workflow, enabling detailed anatomical modeling. However, challenges such as the steep learning curve and the need for user training were noted as areas for improvement, especially for non-technical users (Ruiz-Alzola et al., 2018). Bianchi et al. (2020) explored 3DSlicer for healthcare in *dental and craniofacial research*. 3DSlicer was used to support decision-making

by integrating and visualizing multimodal imaging data. Notable applications included treatment evaluations and surgical planning. Challenges such as ensuring user-friendly interfaces and optimizing workflows for clinical translation were identified as areas for improvement (Bianchi et al., 2020). Sharma, Vilarrubias, and Verschure (2023) researched the use of 3D Slicer in *neuroimaging data*. Specifically, the software was used to visualize and analyze structural brain data, such as MRI and CT scans. While 3D Slicer provided essential tools for visualization and segmentation, the study noted the challenges of managing data across multiple modalities and integrating them into unified workflows (Sharma, Vilarrubias, and Verschure, 2023). Mandolini et al. (2022) examined the application of 3D Slicer in *hip surgical planning*. The study highlighted 3D Slicer's capabilities in reconstructing 3D anatomical models of femoral heads from CT images. It demonstrated the platform's effectiveness in achieving dimensional accuracy and geometric precision. Challenges mentioned were in usability and longer segmentation times compared to Syngo.via Frontier, a more user-friendly alternative (Mandolini et al., 2022).

It is followed by "3D Printing", investigated in 3 different papers [Geerlings-Batt et al., 2022; Quiñones Rodríguez et al., 2023; Pakvasa et al., 2022], accounting for 27.3% of research. Geerlings-Batt et al. (2022) investigated the use of 3D Slicer for segmenting and reconstructing MRI datasets to create 3D-printed anatomical models. It specifically focuses on the foot and ankle. While the models were detailed, the process was time-intensive and required manual effort due to a lack of automated tools (Geerlings-Batt et al., 2022). Quiñones Rodríguez et al. (2023) describes the use of 3D Slicer in designing 3D-printed moulds for personalized cancer treatment, specifically brachytherapy. The software was used to process CT images of patients, generating files to create customized molds for these patients. While the approach proved effective in providing tailored treatment solutions, the researchers noted that it required significant time and expertise due to the need for extensive manual adjustments and refinements (Quiñones Rodríguez et al., 2023). Pakvasa et al. (2022) developed a protocol for using 3D Slicer to segment CT images and create 3D-printed-models. The protocol aims to make 3D printing accessible to users without prior experience. It was successfully tested by inexperienced users, who were able to create accurate models comparable to commercial alternatives (Pakvasa et al., 2022).

3D Slicer is mentioned twice [Syamlan et al., 2022; Lemarteleur et al., 2020] in the research topic "Education and Training". Syamlan et al. (2022) evaluated the use of 3D Slicer to construct *virtual spine models in a surgical simulator*. 3D Slicer helped segment CT images. The data then was refined with other tools and algorithms and exported as 3D files. The 3D models were used in a virtual reality simulator to help surgeons practice placing pedicle screws (Syamlan et al., 2022). Lemarteleur et al. (2020) used 3D Slicer to segment CT images and *create 3D models of the kidney and ureter for surgical training*. While effective in replicating surgical sensations such as needle penetration and cutting strength, the elasticity of the materials differed from biological tissues, highlighting an area for improvement (Lemarteleur et al., 2020).

Fichtinger et al. (2021) researched 3D Slicer in "**Cancer Care**" by developing "*NaviPBx*", a prostate biopsy system. 3D Slicer was incorporated for image analysis and visualization. The modular design of 3D Slicer allowed integration of spatial tracking and real-time guidance, enhancing biopsy accuracy (Fichtinger et al., 2021).



Figure 7 Research Topics within 3D Slicer Projects

**"OpenMRS"** project started in 2004 to facilitate the development of electronic medical record (EMR) systems in developing countries. It is the world's leading open source EMR platform as of 2024 being used in more than 80 countries and has helped over 22 million patients so far (Mamlin et al., 2006; OpenMRS, n.d.). The system was designed for a limited-resource installation, using tools that are mostly freely available and open-source, ensuring a low-cost and easily updatable system (Wolfe et al., 2006). Figure 8 displays the distribution of research topics within OpenMRS projects in our included papers.

"Electronic Health Records" is one of the most common topics, accounting for 25% of the research analyzing OpenMRS projects with 2 papers [Shaikh et al., 2022; Martsenyuk et al., 2019]. Shaikh et al. (2022) assess the alignment of OpenMRS with healthcare standards. The findings highlight that OpenMRS provides a comprehensive framework for customization, supports multiple clients within intranet and internet architectures, and demonstrates strong potential for integration with third-party modules, such as Clinical Decision Support (CDS) and Computerized Practitioner Order Entry (CPOE). However, the challenges identified include the need for improved architecture and layout in the pharmacy, laboratory, and radiology modules, as well as the absence of a Picture Archiving and Communication System (PACS). The authors suggest enhancing advanced CDS functionalities, integrating AI and data analytics tools, and extending interoperability with external data sources and FHIR standards to address these limitations and advance the system's capabilities (Shaikh et al., 2022). Martsenyuk et al. (2019) focus on the integration of EHR, specifically OpenMRS, into machine learning workflows for medical research. It explores how OpenMRS is used to export structured datasets for tasks such as classification, regression, and prediction. The study highlights the challenges of data interoperability and the reliance on custom modules for HL7based data exchange. Despite these limitations, the authors demonstrate OpenMRS's potential in facilitating predictive healthcare applications, such as fracture risk modeling, by providing accessible clinical data and supporting the development of machine learning models (Martsenyuk et al., 2019).

**"Digital Health"** also contributes to 25% of the research on OpenMRS with two research papers [Paton et al., 2022; Goel et al., 2017]. Paton et al. (2022) researched open-source digital health software and evaluated the use of OpenMRS as a *digital electronic health record (EHR)* 

*system in low-resource settings*. Their findings highlight that OpenMRS is deployed in at least 44 countries to support care for HIV, tuberculosis and maternal-child health (Paton et al., 2022). The use of OpenMRS as the backend electronic health record (EHR) system for the *Intelehealth telemedicine platform* was researched by Goel et al. (2017). The platform is designed to improve healthcare access for underserved populations. OpenMRS facilitates the collection, storage, and management of patient data, enabling community health workers (CHWs) to connect with remote doctors for diagnosis and treatment (Goel et al., 2017).

One paper [Paton and Karopka, 2017] focuses on the implementation of OpenMRS in the research area "Medical Education and Training". OpenMRS was examined in the *development of Learning Health Systems (LHS)*. Its modular and collaborative design makes it a key component in creating scalable healthcare infrastructures. However, the authors highlight challenges such as the need for broader standardization, improved data privacy, and enhanced scalability to fully realize its potential in supporting (Paton and Karopka, 2017).

**"Patient-Care"** was researched once [Eapen et al., 2019)] with OpenMRS, focusing on its role in *consolidating patient data from various sources*. It was integrated into the Drishti framework. The system offers features like hGraph, providing visual summaries of patient health to assist clinicians in decision-making. However, limitations were noted in handling data types such as images and ECG recordings, as well as in privacy, security, and patient safety measures (Eapen et al., 2019).

**"Health IT Infrastructure"** was examined in Jawhari et al. (2016), which evaluates the implementation of *OpenMRS in two urban slum clinics in Nairobi, Kenya*. The study focuses on how OpenMRS supports patient data management, care continuity, and system efficiency in resource-constrained environments. It highlights challenges such as unreliable infrastructure, limited interoperability with other health systems, and difficulties with staff training and resistance to unique personal identifiers. The authors emphasize the importance of tailored implementation strategies to address these barriers and ensure sustainable use of OpenMRS in such settings (Jawhari et al., 2016).

Hossain et al. (2019) explored the application of OpenMRS in "Health Data Management". The study evaluated *OpenMRS as a tool for process modeling and improvement in clinical workflows*. It highlighted the use of Petri Net diagrams to document workflow patterns, such as patient registration and vitals collection, and identified areas for optimization, including reducing bottlenecks and inefficiencies. OpenMRS was noted for its modular design, which enables customization for specific clinical settings. However, challenges such as the need for standardized workflows and data integration across systems were emphasized as areas for further development (Hossain et al., 2019).



Figure 8 Research Topics within OpenMRS Projects

"VistA" (Veterans Health Information System and Technology Architecture) was first deployed under the name "Decentralized Hospital Computer Program (DHCP) in 1985 by the Veterans Health Administration (VA) of the U.S. Veterans Administration. In 1996 DHCP was revised and named "VistA" (West & O'Mahony, 2003). It is used to create a robust national healthcare information infrastructure to improve healthcare quality and patient safety. Figure 9 illustrates the different research topics within "VistA".

**"Electronic Health Records"** is the leading topic with 3 papers [Jones, Smith, and Patel, 2020; Inoue and Zhou, 2016; Shaikh et al., 2022]. Jones, Smith, and Patel (2020) examined how VistA is used as an *EHR system*, highlighting its ability to improve patient care by organizing and managing medical data effectively. Key features included its flexible design, which can be adjusted to fit different healthcare settings, and tools like automated order entry and patient scheduling. However, the research also pointed out issues such as outdated technology and difficulties in connecting with other systems, suggesting a need for updates and better data sharing (Jones, Smith, and Patel, 2020). Inoue and Zhou (2016) examined "*Electronic Health Record*" systems in the United States. The study identified VistA as a cost-effective alternative to proprietary systems. However, challenges such as limited specialist availability and interoperability with modern systems were noted, underscoring the need for ongoing updates (Inoue and Zhou, 2016). Shaikh et al. (2022) examines the *advancements in open-source EHR systems*. The study highlights VistA as a robust and adaptable system widely used in healthcare facilities, particularly in the United States (Shaikh et al., 2022).

**"Decision Support"** was examined in relation to VistA in one study [Rajeevan et al., 2017]. The paper explored the *integration of web-based Clinical Decision Support (CDS) systems with VistA*. The real-time patient data was provided by VistA and then used to deliver personalized recommendations and clinical insights. However, the study identified challenges related to data access and standardization within the VistA infrastructure (Rajeevan et al., 2017).



Figure 9 Research Topics within VistA Projects

# 6. Discussion

In this discussion, we analyze and interpret our research results. Subsection 6.1 interprets Research Trends, section 6.2 reflects Open-Source Foundations and section 6.3 evaluates Open-Source Projects.

#### 6.1 Research Trends

Figure 2 reveals the number of papers published per year in our research scope. 2016 starts with 17 publications rising steadily to 28 papers in 2019. This slow, but steady increase might indicate the growing interest and advancements in the research of open-source software in healthcare. In 2021, we witnessed a significant peak with 35 research papers published. This could be associated with the global COVID-19 pandemic. The pandemic urged for scalable and cost-effective solutions during the crisis, making open-source software research in this area very important. After the peak the number of research decreases in 2022 and 2023. This might indicate a shift from the high-intensive research driven by the pandemic to a more sustainable approach. Also, economic factors, such as inflation and supply-chain issues like shortages in the healthcare sector, might have impacted research.

In Table 3, we can observe a wide variety of venues publishing papers in our research area. This indicates that open-source healthcare research is important for diverse facets of healthcare and spans across multiple fields, from clinical science to data analytics and decision support systems. However, there are core journals and conferences, such as the "Journal of the American Medical Informatics Association" and the "Conference on Human Factors in Computing Systems" that serve as key platforms in this research scope.

Our review identified 20 different topics, showing the wide range of healthcare areas where open-source solutions can make an impact. Key areas like Health Data Management, Medical Imaging, Digital Health, and IT Infrastructure were the most studied, suggesting that open-source software (OSS) is especially effective in these fields. In contrast, smaller topics like "Genomic Medicine" and "Rural Healthcare Access" received less attention, highlighting opportunities for further research.

In Figure 4, we can see a shift of priorities in research. While the research on "Electronic Health Records" and "IT Infrastructure" diminished in 2020-2023, "Medical Imaging" gained significance, overtaking "Health Data Management". "Digital Health" also was researched more, probably due to the pandemic's impact on remote healthcare. The pandemic also led to the new topic "COVID-19" and the interest in the areas "Public Health" and "Emergency Healthcare" grew, again reflecting the influence of the global health crisis on OSS research.

#### 6.2 Open-Source Foundations

From 217 papers, only 13 evaluated an open-source healthcare foundation. The limited number of foundations could be implying a potential gap in the support structure of healthcare projects using open-source. This gap could also affect the ability to scale open-source solutions effectively. However, many papers mentioned projects supported by a foundation without mentioning or analyzing the foundation. To overcome this gap, future research could focus more on how foundations contribute and support open-source projects in healthcare. Understanding community structures, funding processes and governance strategies behind OSS projects could provide valuable insights.

#### 6.3 Open-Source Projects

In the collected papers, 86 different projects were mentioned. The diversity of projects underlines the broad applicability and impact of OSS in healthcare.

Figure 7 illustrates the topics in which the leading project, "3D Slicer," has been utilized. Nearly half of the research papers focus on "Medical Imaging" (40%), followed by "3D Printing" (30%), reflecting its primary purpose as a medical imaging tool. Its use in "Education and Training" (20%) highlights its role as a learning resource for medical professionals. While research on "Cancer Care" (10%) is limited, it indicates potential for future exploration in this area.

Figure 8 presents the topics in which "OpenMRS," has been applied. The focus on "Electronic Health Records" and "Digital Health" aligns with its core mission of delivering affordable, scalable, and digital EHR solutions. Mentions of "Health IT Infrastructure" and "Health Data Management" underline its technical capabilities in managing healthcare systems. Its mentions in "Patient Care" and "Education and Training" highlight its adaptability to diverse healthcare needs and contexts.

"VistA" has been primarily researched in the context of "Electronic Health Records," accounting for 75% of the studies. This highlights its foundational role in managing patient records securely. Its application in "Decision Support," explored in one study (25%), further emphasizes its value in enhancing clinical decision-making.

# 7. Limitations

To address the limitations of our research, we applied Guba's (1981) trustworthiness criteria including credibility, transferability, confirmability and dependability.

**Credibility** refers to the confidence in the truth of the findings. The credibility of this study was established by thoroughly selecting relevant studies. By using four databases, we aimed to capture a diverse and comprehensive set of studies relevant to our research questions. Each identified study was carefully reviewed twice to ensure its validity and alignment with the research scope. Additionally, we addressed the risk of including opinion-based articles by excluding non-research-based articles, non-peer-reviewed publications, and papers flagged as unreliable.

**Transferability** addresses the applicability of findings to other contexts. The findings of this study reveal a high degree of diversity, with numerous projects being applied in varying contexts and for different purposes. Even the same projects were utilized in distinct areas, highlighting the flexibility and adaptability of open-source solutions in healthcare. This diversity indicates a broad applicability of the findings, suggesting their potential relevance across a wide range of contexts. This criterion was additionally addressed by providing detailed descriptions of the context and scope of the included studies, enabling readers to assess whether the findings could be applied to other contexts.

**Confirmability** concerns the extent to which findings are shaped by the data rather than researcher bias. This was achieved by carefully documenting the selection process, exclusion criteria, and data extraction methods. The results are firmly rooted in the reviewed evidence. During the research period we further ensured accuracy by evaluating the relevance of each study at least twice and thoroughly reviewing the data analysis results.

**Dependability** emphasizes the consistency and traceability of research processes. This was ensured through a systematic methodology following the guideline outlined by Kitchenham (2004). We present the data of included and excluded literature with details in Appendix A-1, Appendix A-2 and Appendix B.

# 8. Conclusion

Open-source software plays an important role in healthcare, offering interoperability and costreduction of implementation and maintenance. However, the current research on different projects and foundations is limited. To provide an overview of the state-of-the-art, we conducted a systematic literature review to investigate open-source projects and foundations in healthcare. The review covered the period from 2016 to 2023 and analyzed 217 papers collected from the venues ACM Digital Library, IEEE Xplore, Scopus and Springer.

The findings reveal an increase in research activity, particularly during 2021 and 2022, coinciding with the COVID-19 pandemic, highlighting the growing importance of OSS in healthcare. The diversity of projects and publication venues further underscores the broad applicability and significance of OSS across various domains within the industry. Key research areas include "Health Data Management," "Medical Imaging," and "Digital Health," which dominate the literature. Conversely, niche fields such as "Genomic Medicine" and "Diabetes Care" remain largely underexplored, presenting opportunities for future investigation.

Additionally, the review identified a relatively limited focus on open-source foundations, pointing to a significant research gap. Future studies could prioritize examining the role and impact of OSS foundations while exploring underrepresented areas of healthcare to fully leverage the potential of OSS in addressing emerging challenges and advancing the field.

# Acknowledgement

During the writing process of this work the author used Artificial Intelligence tools to check the grammar structure of written paragraphs. After using this tool, the author reviewed and edited the content as needed and takes full responsibility for the content of the paper.

### References

Aminpour, F., Sadoughi, F., & Ahamdi, M. (2014). Utilization of open source electronic health record around the world: A systematic review. *Journal of Research in Medical Sciences*, 19(1), 57–64.

Bagayoko, C. O., Dufour, J. C., Bouhaddou, O., et al. (2010). Open source challenges for hospital information system (HIS) in developing countries: A pilot project in Mali. *BMC Medical Informatics and Decision Making*, 10(22).

Bianchi, J., Paniagua, B., De Oliveira Ruellas, A. C., Fillion-Robin, J. C., Prietro, J. C., Gonçalves, J. R., ... & Cevidanes, L. (2020). 3D Slicer Craniomaxillofacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research. In *Machine Learning in Clinical Decision Support* (pp. 44–53).

Chang, Y. H., Foley, P., Azimi, V., Borkar, R., & Lefman, J. (2016). Primer for image informatics in personalized medicine. *Proceedia Engineering*, 159, 58–65.

Escudero Sanchez, L., Buddenkotte, T., Al Sa'd, M., McCague, C., Darcy, J., Rundo, L., Samoshkin, A., Graves, M. J., Hollamby, V., Browne, P., et al. (2023). Integrating artificial intelligence tools in the clinical research setting: The ovarian cancer use case. *Diagnostics*, 13(17), 2813.

Frade, S., Freire, S. M., Sundvall, E., Patriarca-Almeida, J. H., & Cruz-Correia, R. (2013). Survey of openEHR storage implementations. *Proceedings of the IEEE 26th International Symposium on Computer-Based Medical Systems (CBMS)*, 303–307.

Geerlings-Batt, J., Tillett, C., Gupta, A., & Sun, Z. (2022). Enhanced visualisation of normal anatomy with potential use of augmented reality superimposed on three-dimensional printed models. *Micromachines*, 13(10), 1701.

Goel, N. A., Alam, A. A., Eggert, E. M. R., & Acharya, S. (2017). Design and development of a customizable telemedicine platform for improving access to healthcare for underserved populations. 2017 IEEE Global Humanitarian Technology Conference (GHTC), 2658–2661.

Goulde, M., & Brown, E. (2006). Open source software: A primer for health care leaders. *Forrester Consulting*.

Guba, E. G. (1981). Criteria for assessing the trustworthiness of naturalistic inquiries. *Educational Communication and Technology Journal*, 29(2), 75–91.

Hak, F., Oliveira, D., Abreu, N., Leuschner, P., Abelha, A., & Santos, M. (2020). An OpenEHR adoption in a Portuguese healthcare facility. *Procedia Computer Science*, 170, 1047–1052.

HL7 FHIR Foundation. (n.d.). About the HL7 FHIR Foundation. FHIR Foundation. Retrieved December 19, 2024, from https://fhir.org/about.html

IMIA. (n.d.). Open source health informatics. *International Medical Informatics Association*. Retrieved December 19, 2024, from https://imia-medinfo.org/wp/open-source-health-informatics/

Ingram, D. (2002). Origins of openEHR. *openEHR*. Retrieved December 19, 2024, from https://openehr.org/about/origins

Iribarren, S. J., Brown, W. III, Giguere, R., Stone, P., Schnall, R., Staggers, N., & Carballo-Diéguez, A. (2017). Scoping review and evaluation of SMS/text messaging platforms for mHealth projects or clinical interventions. *International Journal of Medical Informatics*, 101, 28–40.

Jawhari, B., Keenan, L., Zakus, D., Ludwick, D., Isaac, A., Saleh, A., & Hayward, R. (2016). Barriers and facilitators to electronic medical record (EMR) use in an urban slum. *International Journal of Medical Informatics*, 94, 246–254.

Janamanchi, B., Katsamakas, E., Raghupathi, W., & Gao, W. (2009). The state and profile of open source software projects in health and medical informatics. *International Journal of Medical Informatics*, 78(6), 457–472.

Kitchenham, B. (2004). Procedures for performing systematic reviews. *Keele University Technical Report TR/SE-0401*. Keele University & National ICT Australia Ltd.

Lemarteleur, V., Peycelon, M., Sablayrolles, J. L., Plaisance, P., El-Ghoneimi, A., & Ceccaldi, P. F. (2020). Realization of open software chain for 3D modeling and printing of organs in simulation centers: Example of renal pelvis reconstruction. *Journal of Surgical Education*, 78(1), 232–244.

LibreHealth. (n.d.). About LibreHealth. Retrieved December 19, 2024, from https://librehealth.io

Mamlin, B. W., Biondich, P. G., Wolfe, B. A., Fraser, H., Jazayeri, D., Allen, C., Miranda, J., & Tierney, W. M. (2006). Cooking up an open-source EMR for developing countries: OpenMRS—A recipe for successful collaboration. *AMIA Annual Symposium Proceedings*, 2006, 529–533.

McDonald, C. J., Schadow, G., Barnes, M., Dexter, P., Overhage, J. M., Mamlin, B., & McCoy, J. M. (2003). Open source software in medical informatics—why, how and what. *International Journal of Medical Informatics*, 69(3), 175–184.

Merine, R., Pinnamraju, J., Singh, D., Gichoya, J. W., & Purkayastha, S. (2022). LibreHealth cost-ofcare explorer: Mobile application for patient-friendly access to hospital chargemasters. 2022 IEEE Healthcare Innovations and Point of Care Technologies (HI-POCT), 26–29.

Min, L., Tian, Q., Lu, X., & Duan, H. (2018). Modeling EHR with the openEHR approach: An exploratory study in China. *BMC Medical Informatics and Decision Making*, 18(75).

Mohamad, Y., Gabber, A., Heidenblut, S., Zenz, D., Siddiqi, A., & Gappa, H. (2022). How to overcome lack of health record data and privacy obstacles in initial phases of medical data analysis projects. *Computing and Informatics*, 41(1), 233–252.

Nagy, P. (2007). Open source in imaging informatics. Journal of Digital Imaging, 20(Suppl 1), 1-10.

Nguyen, S. P., Nielsen, P., & Sæbø, J. I. (2017). The role of global standardization communities in shaping national health information architectures. In J. Choudrie et al. (Eds.), *IFIP Advances in Information and Communication Technology: ICT4D 2017* (pp. 93–103).

Nguyen, S. P., Nielsen, P., & Sæbø, J. I. (2019). Global standards and local development. In K. Krauss et al. (Eds.), *IDIA 2018: International Development Informatics Association* (pp. 141–155).

Oliveira, R., Ferreira, D., Ferreira, R., & Cruz-Correia, R. (2016). Open-source based integration solution for hospitals. 2016 IEEE 29th International Symposium on Computer-Based Medical Systems (CBMS), 294–299.

Open Health Imaging Foundation. (2015). About. Open Health Imaging Foundation. Retrieved December 19, 2024, from https://ohif.org/about

OpenHIE. (n.d.). Community. *OpenHIE*. Retrieved December 19, 2024, from https://ohie.org/community/

OpenMRS. (n.d.). About OpenMRS. Retrieved December 19, 2024, from https://openmrs.org

Oyri, K., & Murray, P. J. (2005). osni.info—Using free/libre/open source software to build a virtual international community for open source nursing informatics. *International Journal of Medical Informatics*, 74(11–12), 937–945.

Paton, C., Braa, J., Muhire, A., Marco-Ruiz, L., Kobayashi, S., Fraser, H., Falcón, L., & Marcelo, A. (2022). Open source digital health software for resilient, accessible, and equitable healthcare systems. *IMIA Yearbook of Medical Informatics*, 31(1), 67–73.

Paton, C., & Karopka, T. (2017). The role of Free/Libre and Open Source Software in Learning Health Systems. *Yearbook of Medical Informatics*, 26(1), 53–58.

Pieper, S., Halle, M., & Kikinis, R. (2004). 3D Slicer. *IEEE International Symposium on Signal Processing and Applications*, 2004, 632–635.

Prattico, L. (2012). Governance of Open Source Software Foundations: Who Holds the Power? *Technology Innovation Management Review, 2012*(December), 37–42.

Purkayastha, S., Allam, R., Maity, P., & Gichoya, J. W. (2019). Comparison of open-source electronic health record systems based on functional and user performance criteria. *Healthcare Informatics Research*, 25(2), 89–98.

Purkayastha, S., Buddi, S. B., Nuthakki, S., Yadav, B., & Gichoya, J. W. (2020). Evaluating the implementation of deep learning in LibreHealth Radiology on chest X-rays. *Advances in Intelligent Systems and Computing*, 943, 648–657.

Rajeevan, N., Niehoff, K. M., Charpentier, P., Levin, F. L., Justice, A. C., Brandt, C. A., Fried, T. R., & Miller, P. L. (2017). Utilizing patient data from the Veterans Administration electronic health record to support web-based clinical decision support: Informatics challenges and issues from three clinical domains. *BMC Medical Informatics and Decision Making*, 17(1), 111.

Riehle, D. (2010). The Economic Case for Open Source Foundations. Computer, 43(1), 86–90.

Riehle, D., & Berschneider, S. (2012). A Model of Open Source Developer Foundations. *IFIP Advances in Information and Communication Technology*, 378, 15–28.

Ruiz-Alzola, J., Skareb, A., Idriss, A. M., Diao, B., Da Rosa De Pina Ferreira, J., Fernandes Rodrigues, M. A., López Rivero, L., & Kikinis, R. (2018). Train the Trainers: Medical Technology for the Sustainable Development of Africa. *Proceedings of the IEEE Symposium on Biomedical Imaging*, 978-1-5386-5566-5.

Shaikh, S., Hussain, F., Masood, I., Jamil, A., & Ahmad, M. (2022). Open-source electronic health record systems: A systematic review of most recent advances. *Journal of Biomedical Informatics*, 138, 104212.

Sharma, V., Vilarrubias, R. B., & Verschure, P. F. M. J. (2023). BrainX3: A Neuroinformatic Tool for Interactive Exploration of Multimodal Brain Datasets. In *Living Machines 2023* (pp. 157–177).

Sobral, D. S., Duarte, D. W., Dornelles, R. F. V., & Moraes, C. A. C. (2021). 3D virtual planning for rhinoplasty using a free add-on for open-source software. *Aesthetic Surgery Journal*, 41(8).

Syzdykova, A., Malta, A., Zolfo, M., Diro, E., & Oliveira, J. L. (2017). Open-source electronic health record systems for low-resource settings: Systematic review. *JMIR Medical Informatics*, 5(4).

Van Biesen, W., Van Cauwenberge, D., Decruyenaere, J., et al. (2022). An exploration of expectations and perceptions of practicing physicians on the implementation of computerized clinical decision support systems using a Q-sort approach. *BMC Medical Informatics and Decision Making*, 22, 185.

Wadali, J. S., Sood, S. P., Kaushish, R., Syed-Abdul, S., Khosla, P. K., & Bhatia, M. (2020). Evaluation of free, open-source, web-based DICOM viewers for the Indian National Telemedicine Service (eSanjeevani). *Journal of Digital Imaging*, 33(6), 1499–1513.

Warner, J. L., Rioth, M. J., Mandl, K. D., Mandel, J. C., Kreda, D. A., Kohane, I. S., Carbone, D., Oreto, R., Wang, L., Zhu, S., & Yao, H. (2016). SMART precision cancer medicine: A FHIR-based app to provide genomic information at the point of care. *Journal of the American Medical Informatics Association*, 23(4), 701–710.

Waring, T., & Maddocks, P. (2005). Open Source Software implementation in the UK public sector: Evidence from the field and implications for the future. *International Journal of Information Management*, 25(5), 411–428.

Wolfe, B. A., Mamlin, B. W., Biondich, P. G., Fraser, H. S. F., Jazayeri, D., Allen, C., Miranda, J., & Tierney, W. M. (2006). The OpenMRS system: Collaborating toward an open source EMR for developing countries. *AMIA Annual Symposium Proceedings*, 2006, 1146–1150.

# Appendix A-1

Search venue	Query date	Author	Title	Туре	Status	Journal	Year
Scopus	29.06.2024	Angulo D.; Cortes M.F.; Mura I.; Akhavan-Tabatabaei R.	Fine-grained mathematical modeling for cost-effectiveness evaluation of public health policies for cervical cancer, with application to a Colombian case study	Article	included	BMC Public Health	2023
Coonico	01.07.0004	Carpentier P.	Open source hardware, exploring how industry regulation affects knowledge	Article	Induded	International Journal of the Commons	2021
Scopus	01.07.2024	Fichtinger G.; Mousavi P.; Ungi T.; Fenster A.; Abolmaesumi P.; Kronreif G.;	Design of an ultrasound-navigated prostate cancer biopsy system for nationwide	Article	Induded	International Journal of the Commons	2021
Scopus	28.06.2024	Ruiz-Alzola J.; Ndoye A.; Diao B.; Kikinis R. Oasim M.: Puigianer D.: Herrero J.: López J.M.: Olivé C.: Fortuny G.: Garcia-	implementation in senegal Biomechanical modelling of the pelvic system: improving the accuracy of the	Article	included	Journal of Imaging	2022
Scopus	28.06.2024	Bennett J. Escudero Sanchez L.; Buddenkotte T.; Al Sa'd M.; McCague C.; Darcy J.;	location of neoplasms in MRI-TRUS fusion prostate biopsy Integrating Artificial Intelligence Tools in the Clinical Research Setting: The	Article	included	BMC Cancer	2023
Scopus	28.06.2024	Rundo L.; Samoshkin A.; Graves M.J.; Hollamby V.; Browne P.; Crispin- Ortuzar M.; Woltek R.; Sala E.; Schönlieb CB.; Doran S.J.; Öktern O. Pati S.; Baid U.; Edwards B.; Sheller M.J.; Foley P.; Anthony Reina G.;	Ovarian Cancer Use Case The federated tumor segmentation (FeTS) tool: an open-source solution to	Article	included	Diagnostics, MDPI	2022
Scopus	27.06.2024	Thakur S.; Sako C.; Bilelio M.; Davatzikos C.; Martin J.; Shah P.; Menze B.; Bakas S. Manara I. L.: Bieth M. L. Mandil K. D.: Mandel J. C.: Krada D. A.; Kahana J. S.;	further solid tumor research	Articlo	included	Physics in Medicine and Biology	2016
Scopus	02.07.2024	Carbone D.; Oreto R.; Wang L.; Zhu S.; Yao H.; Alterovitz G.	information at the point of care	Anuce	included	Association	2010
IEEE Xplore	05.12.2024	A. Brezulianu; I. Popa; O. Geman; I. Chiuchisan; C. Nastase	Integrated Extended SIR-type Epidemiological Modeling Platform for Covid-19	Conference paper	included	IEEE EHB 2022	2022
ACM Digital	10 11 2024	Pohot Viceur Paranage Fally and Amal Chadoux	How makers responded to the PPE shortage during the COVID-19 pandemic:	Conference	indudad	17th International Symposium on Open	2021
ACM Digital	10.11.2024	Hubert viseur, bereingere nany, and Anner Chaneux	The power of open-source hardware in crisis: perspectives from the Wilson	Article	maudeu	XRDS: Crossroads, The ACM Magazine for	2021
Library	22.06.2024	Alex Long Sun Sun LimBoland Bouffanais	Center's open hardware and COVID-19 roundtable	Article	included	Students	2022
ACM Digital	00.12.2024	Kelly Mack, Megan Hofmann, Udaya Lakshmi, Jerry Cao, Nayha Auradkar,	Example and a service behavior by the mining of the mo	Article	maddeu	ACM Transactions on Computer-Human	LOLL
Library	28.05.2024	Rosa Arriaga, Scott Hudson, and Jen Mankoff Roston R.D.: Macgregor L.: Vass L.: Locker K.L.: Hyame C.: Bright P.D.:	Rapid Convergence: The Outcomes of Making PPE During a Healthcare Crisis	Article	included	Interaction	2023
Sconus	01 07 2024	Harding I.; Lazarus R.; Hamilton F.; Lawson D.; Danon L.; Pratt A.; Wood R.; Procke: Pollock E: Turner K M E	requirements in South West England: A mathematical modelling framework	Ande	included	PMI Onon	2021
000000	01.07.2024	Bloosen Jacob R. E., Bors K. Lis, J.; Feldman T.C.; Sung FC.; Hall R.B.; Nguyen V.; Link N.; Zwolinski R.; Dipietro S.; Miller S.J.; Aleksanyan A.; Goryachev S.D.; Corcoran P.; Bergstrom S.J.; Parenteau MA.; Sprague R.S.; Thornton D.J.; Driver J.A.; Strymish J.M.; Evans S.; Colonna B.; Brophy M.T.;	An application to support COVID-19 occupational health and patient tracking at a Veterans Affairs medical center	Artide	maddad	Journal of the American Medical Informatics	2020
Scopus	01.07.2024	Do N.V. Scharu W - Backali E - Bentechler S - Ruppert C - Wagner C E - Steinmetz L	Open-Source Adentable All-in Ope Smartnhone Record System for	Articlo	included	Association	2022
Scopus	27.06.2024	Deigner HP.; Kohl M. Deigner HP.; Kohl M. Cartis H.J.: MacKenna B.; Corker R.; Inglesby P.; Walker A.J.; Mortey J.; Markar A.; Morton C.E.; Bacos S.; Hickman G.; Bates C.; Evans D.; Ward Warkar A.; Morton C.E.; Bacos S.; Hickman G.; Bates C.; Evans D.; Ward Warkar A.; Mortano E.; Hauson S.; Hickman G.; Bates C.; Evans D.; Ward T.; Devertale H.: Evans P. W.; McDonald H.J.; Tominishe L.; M.G.; Divertale H.: Evans P. M.; With V. Wonz A.Y.S.; Corbes H.: Parru I.: Heater	Open-Sould's Adaptatic Anim Colle Sinia (Jubite Dased System to Quantitative Analysis of Point-Care Diagnostics 1: Primary care clinical OpenSAFELY NHS Service Restoration Observatory 1: Primary care clinical activity in England during the first wave of COVID-19	Article	induded	Diagnostics, MDPI	2022
Scopus	28.06.2024	F; Harper S.; Evans S.J.W.; Douglas I.J.; Smeeth L.; Goldacre B. Carla I. M. Bezerra, José Cezar de Souza Filho, Emanuel F. Coutinho, Alice		Conference	included	British Journal of General Practice	
ACM Digital Library	10.11.2024	Gama, Ana Lívia Ferreira, Gabriel Leitão de Andrade, and Carlos Eduardo Feitosa	How Human and Organizational Factors Influence Software Teams Productivity in COVID-19 Pandemic: A Brazilian Survey	paper	included	XXXIV Brazilian Symposium on Software Engineering (SBES '20)	2020
Scopus	29.06.2024	Lenert L.A.; Ilatovskiy A.V.; Agnew J.; Rudisill P.; Jacobs J.; Weatherston D.; Deans K.R., Jr.	Automated production of research data marts from a canonical fast healthcare interoperability resource data repository; Applications to COVID-19 research	Article	included	Journal of the American Medical Informatics Association	2021
Scopus	27.06.2024	Wilkins-Caruana A.; Bandara M.; Musial K.; Catchpoole D.; Kennedy P.J.	Inferring actual treatment pathways from patient records	Article	included	Journal of Biomedical Informatics	2023
Scopus	27.06.2024	Murray A.T.; Baik J.	Opensource spatial optimization in GIScience for strategic positioning Utilizing patient data from the Veterans Administration electronic health record	Article	included	Transactions in GIS	2023 2017
Springer	07.12.2024	Rajeevan, N., Niehoff, K. M., Charpentier, P., Levin, F. L., Justice, A. C., Brandt, C. A., Fried, T. R., & Miller, P. L.	to support web-based clinical decision support: Informatics challenges and issues from three clinical domains		included	BMC Medical Informatics and Decision Making	
Scopus	01.07.2024	Erdoğan G.; Stylianou N.; Vasilakis C.	An open source decision support system for facility location analysis	Article	included	Decision Support Systems	2019
Scopus	01.07.2024	Volanski W.: Do Prado A.L.: Al-Lahham Y.: Teleginski A.: Pereira F.S.:	D-GDM: A mobile diagnostic decision support system for gestational diabetes	Article	included	npj Digital Medicine	2019
Scopus	01.07.2024	Alberton D.; Rego F.G.M.; Valdameri G.; Picheth G. Tuli S.; Basumatary N.; Gill S.S.; Kahani M.; Arya R.C.; Wander G.S.; Buyya	HealthFog: An ensemble deep learning based Smart Healthcare System for	Article	included	Archives of Endocrinology and Metabolism	2020
Scopus	01.07.2024	R.	Automatic Diagnosis of Heart Diseases in integrated IoT and fog computing environments		included	Future Generation Computer Systems	
ACM Digital	00.44.0004		Health & amp; Play: Addressing together the challenges of creating an open	Conference	to de de d	12th International Symposium on Open	0040
ACM Digital	09.11.2024	Junhao Li, Miikka Kuutila, Eetu Huusko, Nimantha Kariyakarawana, Marko	Assessing Credibility Factors of Short-Form Social Media Posts: A	Conference	included	15th Biannual Conference of the Italian	2016
Library	04.11.2024	Savic, Nazanin Nakhaie Ahooie, Simo Hosio, and Mika Mäntylä Ionut Cristian Canciu, Constantinos S. Pattichis, Marios Christodoulou,	Crowdsourced Online Experiment	paper Conference	included	SIGCHI Chapter (CHitaly '23)	2023
ACM Digital Library	09.11.2024	Louiza Agroti, Maria Papaioannou, Andreas Neocleous, Panayiotis Savva, Constantinos Yiasemi, Marios Neofytou, Andreas Panayides, Zinonas Antoniou, and Ioannis Constantinou	Enhancing Healthcare Accessibility: A Teleconsultation Mobile Application for Patient-Centric Care in Cyprus: MYeHealthAppCY: Patient-Centered Teleconsultation	paper	included	24th International Conference on Computer Systems and Technologies (CompSysTech '23)	2023
ACM Digital	20.11.2024	Dhiman Sengupta, Tamara Zubatiy, Sean K. Hamilton, Arthur Boothroyd,	Open Speech Blatform: Democratizing Hearing Aid Research	Conference	indudad	PerveringHealth 120	2021
ACM Digital	23.11.2024	oligin ratoin, bezhintong, hajesin olipita, and namaan olaideaan	The Capabilities and Future Use of Facial Recognition Software on the	Conference	maddeu		2021
Library	30.11.2024	Alicia Kim Regina Merine, Jahnavi Pinnamraiu, Darshpreet Singh, Judy W. Gichova, Santarshi	Raspberry Pi – Exploratory System for Adolescent Mental Health Screening LibreHealth Cost-of-Care Explorer: Mobile Application for Patient-friendly	Conference	included	ICMHI 2021 IEEE Healthcare Innovations and Point of Care	2021
IEEE Xplore	17.12.2024	Purkayastha	Access to Hospital Chargemasters	paper	included	Technologies (HI-POCT)	2022
Scopus	05.12.2024	Nathanisillian S., Stechel Z., El di D.G.		paper	included	Conference	2010
ACM Digital Library	29.11.2024	Jochen Meyer and Parisa Eslambolchilar	Hesearch Challenges of Emerging Technologies Supporting Life-Long Health and Wellbeing Browser-based Informatic Tailoring Self-service Interface (BITSI)	conference paper	included	MMHealth'17	2017
IEEE Xplore	05.12.2024			paper	included	VAHC 2022	0047
Scopus	06.12.2024	Baker C.; Blackwood J.; Hartless C.; Pirro J.; Flower A.A.	Healthcare analytics and visualization using SEMantic Open Source Software (SEMOSS)	paper	included	2017 Systems and Information Engineering Design Symposium, SIEDS 2017	2017
Scopus	30.06.2024	Thimbleby H.	NHS Number Open Source Software: Implications for Digital Health Regulation and Development	Article	included	ACM Transactions on Computing for Healthcare	2022
ACM Digital	09.11.2024	Xi Wang, Zoya Bylinskii, Monica Castelhano, James Hillis, and Andrew Duchowski	EMICS'21: Eye Movements as an Interface to Cognitive State	Conference paper	included	2021 CHI Conference on Human Factors in Computing Systems (CHI EA '21)	2021
ACM Digital Library	04.11.2024	Mallory Helms, Shaun V. Ault, Guifen Mao, and Jin Wang	An Overview of Google Brain and Its Applications	Conference paper	included	2018 International Conference on Big Data and Education (ICBDE '18)	2018
ACM Digital			8-year Evaluation of GameBus: Status quo in Aiming for an Open Access	Conference		Human-Computer Interaction, Volume 7,	
Library	25.10.2024	Lamalva G.; Schmeelk S.	Platform to Prototype and Test Digital Health Apps MobSF: Mobile Health Care Android Applications Through the Lens of Open	Conference	included	2020 IEEE MIT Undergraduate Research	2023 2020
Scopus	06.12.2024	Paton C · Bras J · Muhire A · Marco Duiz L · Kohavachi S · Frasar H · Falcón	Source Static Analysis	paper	included	Technology Conference	2022
Scopus	11.11.2024	L.; Marcelo A.	Healthcare Systems	Aluce	included	Yearbook of Medical Informatics	LULL
ACM Digital Library	10.11.2024	Oliver L. Haimson, Kai Nham, Hibby Thach, and Aloe DeGuia	How Transgender People and Communities Were Involved in Trans Technology Design Processes	Conference paper	included	2023 CHI Conference on Human Factors in Computing Systems (CHI '23)	2023
Sconus	05 12 2024	Mohsin Alabassby B.F.N.; Mahdi J.F.; Kadhim M.A.	Design and Implementation WSN Based on Raspberry Pi for Medical Application	Conference	included	IOP Conference Series: Materials Science and Engineering	2019
oupus	00.12.2024	Vijay Arya, Rachel K. E. Bellamy, Pin-Yu Chen, Amit Dhurandhar, Michael Hind, Samuel C. Hoffman, Stephanie Houde, Q. Vera Liao, Ronny Luss, Aleksandra Mojsilović, Sami Mourad, Pablo Pedemonte, Ramya		Conference paper	madada	3rd ACM India Joint International Conference on Data Science & amp; Management of Data	
ACM Digital Library	04.11.2024	Hagnavenora, Jonn Hicharos, Prasanna Satugeri, Karmikeyan Sharimugari, Moninder Singh, Kush R. Varshney, Dennis Wei, and Yunfeng Zhang	Al Explainability 360 Toolkit	Conference paper	included	(CODS-COMAD '21) 2017 ACM International Joint Conference on Pervasive and Ubiquitous Computing and	2021
ACM Digital Library	11.11.2024	Amy M. Bauer, Sarah Hodsdon, Suzanne Hunter, Youlim Choi, Jared Bechtel, and John C. Fortney	Lessons from the deployment of the SPIRIT app to support collaborative care for rural patients with complex psychiatric conditions	0.1	included	Proceedings of the 2017 ACM International Symposium on Wearable Computers (UbiComp 17)	2017
ACM Digital		Aisling Ann O'Kane, Amy Hurst, Gerrit Niezen, Nicolai Marquardt, Jon Bird,		paper		Human Factors in Computing Systems (CHI	
Library	04.11.2024	and Gregory Abowd N. A. Goel; A. A. Alam; E. M. R. Eggert; S. Acharya	Advances in DIY Health and Wellbeing Design and development of a customizable telemedicine platform for improving	Conference	included	EA '16) 2017 39th Annual International Conference of	2016 2017
IEEE Xplore	05.12.2024	kiborron S. L. Brown W. III: Ciguoro B. Clene D. Calcall D. Classer M.	access to healthcare for underserved populations	paper	included	the IEEE Engineering in Medicine and Biology Society (EMBC)	2017
Scopus	02.07.2024	Carballo-Diéguez A.	projects or clinical interventions	Article	included	International Journal of Medical Informatics	2017
IEEE Xplore	05.12.2024	T. Lam; H. Yuan; M. C. de Oliveira	Low-Cost Open-Source Solution to Optimize Emergency Medical Services in Developing Communities by Tracking, Dispatching, and Simulating	Conference paper	included	IEEE Global Humanitarian Technology Conference 2019	2019
ACM Digital	20.44.0007	Albo Catalino Marelao Tirado, Envico Dano, and Envico Matta	Reasoning on Health Condition Evolution for Enhanced Detection of Vulnerable	Conference	Indu day	Knowledge Century Conference W. Conference	0004
Library	29.11.2024	nina Gatainia morares Firado, Enrico Daga, and Enrico Motta	Data Sharing and Information Platforms in Crisis Response and Preparedness:	Article	included	NIUWIE0ge Capture Conterence (K-CAP 2021)	2021
ACM Digital Library	23.05.2024	Angela Oduor Lungati	Exploring the Role of Open Data Sharing Platforms and Collective Intelligence in COVID-19 Response Efforts, and Preparedness for Future Pandemics		included	Digital Government: Research and Practice	2022
IEEE Xplore	05.12.2024	D. Perdanakusuma; W. Puspitasari; M. Saputra	Utilizing Open ERP for Creating Medical Record Management System in Smart Hospital : A Case Study	Conference paper	included	IEEE IAICT 2020	2020

		Bomharda A : Bonfanti S : Galhiati C : Gargantini A : Pelliccione P :	Guidelines for the development of a critical software under emergency	Article			2022
Scopus	29.06.2024	Riccobene E.; Wada M.	Guidennes for the development of a childar software under emergency	Aluce	included	Information and Software Technology	2022
ACM Digital	08 11 2024	Simon J. E. Taylor, Tillal Eldabi, Thomas Monks, Markus Rabe, and Adelinde M. Uhrmacher	Crisis, what crisis: does reproducibility in modeling & amp; simulation really matter? In Proceedings of the 2018 Winter Simulation Conference (WSC '18)	Conference paper	included	20XX Winter Simulation Conference	2018
e and a second		Quiñones Rodríguez LÁ.; Pavón J.P.; Ramírez I.C.; Bayard L.G.; Oquendo	Clinical implementation of 3D printed plesiotherapy moulds using free open	Article			2023
Scopus	28.06.2024	M.A.I. Sassaki Y.K.; de Azevedo Costa A.L.F.; Yamanaka P.G.; Chrispin T.T.B.;	source software Three-dimensional printing of orbital computed tomography scan images for use	Article	included	Annals of 3D Printed Medicine	2022
Scopus	27.06.2024	Daros K.A.C.; Choi S.N.J.H.; dos Santos V.R. Baluras M.; Brancher H.; Handres Santiaga B.; Da Lamba T.; MaKanzia N.;	in ophthalmology teaching	Article	included	Revista Brasileira de Oftalmologia	2022
Scopus	28.06.2024	Orsbon C.; Collier Z.; Ramirez-Garcia R.; Gomez I.; Reid R.R.	Images Using Open-Source Software	Anide	included	SAGE Publications Inc.	2022
ACM Digital	00 44 0004	Katie McFarlane, Thanapong Chaichana, Zhonghua Sun, Joseph Neil	Femoral Neck Angle Impacts Hip Disorder and Surgical Intervention: A Patient-	Conference	in alcode al	3rd International Conference on Digital	0000
ACM Digital	09.11.2024	Dentar, and Thip blown	3D medical objects processing and retrieval using spherical harmonics: a case	Conference	maudea	33rd Annual ACM Symposium on Applied	2020
Library	24.10.2024	Leila Cristina C. Bergamasco, Carlos E. Rochitte, and Fátima L. S. Nunes	study with congestive heart failure MRI exams	paper	included	Computing (SAC '18) Sixth International Conference on	2018
ACM Digital		Santiago González Izard, Juan A. Juanes Méndez, Pablo Ruisoto, and	NextMed: How to enhance 3D radiological images with Augmented and Virtual	paper		Technological Ecosystems for Enhancing	
Library	11.11.2024	Francisco José García-Peñalvo	Reality	Article	included	Multiculturality (TEEM'18)	2018
Scopus	29.06.2024	Geenings-Batt 5., There C., Gupta A., Gurt 2.	Reality Superimposed on Three-Dimensional Printed Models	Aluce	included	Micromachines	2022
Scopus	28.06.2024	Sobral D.S.; Duarte D.W.; Dornelles R.F.V.; Moraes C.A.C.	3D Virtual Planning for Rhinoplasty Using a Free Add-On for Open-Source Software	Article	included	Aesthetic Surgery Journal	2021
ACM Digital	24 06 2024	Alexander Mood, Kassen Najarian, and Dalamm Kabrahasi	Homomorphic Economics for Machine Learning in Madiaina and Disinformation	Article	included	ACM Computing Suprans (CSLIP)	2020
ACM Digital	24.00.2024	Alexander wood, Rayvan Najanan, and Delaram Ramobaer	Homomorphic Encryption for Machine Learning in Medicine and Biomornatics	Conference	maudea	13th International Conference on Availability,	2020
Library	04.11.2024	Priya Anand and Jungwoo Ryoo	Architectural Solutions to Mitigate Security Vulnerabilities in Software Systems	paper	included	Reliability and Security (ARES '18)	2018
Scopus	06.12.2024	Copponito L., D'Antonio S., Mazzeo G., Komano L., Sgagione L.	Level	paper	included	European Dependable Computing Conference	2010
		Preston C.G.; Wright M.W.; Madhavrao R.; Harrison S.M.; Goldstein J.L.; Luo X.; Wand H.; Wulf B.; Cheung G.; Mandell M.E.; Tong H.; Cheng S.; lacocca	ClinGen Variant Curation Interface: a variant classification platform for the application of evidence criteria from ACMG/AMP guidelines	Article			2022
		M.A.; Pineda A.L.; Popejoy A.B.; Dalton K.; Zhen J.; Dwight S.S.; Babb L.; Disterance M.: O'Daniel L.M.; Leo K.; Piners F.B.; Zastraw D.B.; Master L.L.;					
		Ritter D.I.; Patel R.Y.; Subramanian S.L.; Milosavljevic A.; Berg J.S.; Rehm H.					
Scopus	28.06.2024	L.; Plon S.E.; Cherry J.M.; Bustamante C.D.; Costa H.A. C. Weir: A. Dyson: D. Prince	A Lot Less Likely Than I Thought: Introducing Evidence-Based Security Risk	Conference	included	Genome Medicine	2023
IEEE Xplore	30.11.2024		Assessment for Healthcare Software	paper	included	IEEE Secure Development Conference	
Springer	17.12.2024	Lingtong Min, Qi Tian, Xudong Lu, and Huilong Duan Francini Hak. Daniela Oliveira. Nuno Abreu. Pedro Leuschner, António Abelha.	Modeling EHR with the openEHR approach: an exploratory study in China	Article	included	BMC Medical Informatics and Decision Making	2018
Scopus	17.12.2024	Manuel Santos	An OpenEHR Adoption in a Portuguese Healthcare Facility	Article	included	Procedia Computer Science	2020
ACM Digital Library	04.11.2024	Fodil Belghait, Beatriz Kanzki, and Alain April	ADAM Genomics Schema - Extension for Precision Medicine Research*	paper	included	Health (DH '18)	2018
IEEE Volum	02.07.000	C. Ma; Q. Zhang; Y. Ji; L. Hu; Y. Liu; K. Chen; J. Zhao; Y. Li	IGZO-TFT-PDK: Thin-Film Flexible Electronics Design Kit, Standard Cell and	Article	included	IFEE Open Journal of Orouting and Outland	2021
ACM Digital	02.07.2024	Hossain Shahriar, Chi Zhang, Md Arabin Islam Talukder, and Sheikh Iqbal	Design mentodology	Conference	madded	incre open Journal of Offcuits and Systems	
Library	29.11.2024	Ahamed	Resources for healthcare workflow modeling and analysis	paper	included	ACM Symposium	2019
Scopus	06.12.2024	Hossain M.S.A.; Ashraiuzzaman M.	Analysis and Cloud Storage	paper	included	Springer IFMBE Proceedings	2021
10110: 111			Big Data Framework for Coolable and Efficient Dismediaal Literature Mining in	Conference		2019 3rd International Conference on Natural	
Library	04.11.2024	Zhengru Shen, Xi Wang, and Marco Spruit	the Cloud	paper	included	Retrieval (NLPIR '19)	2019
		Estemph Routhon Aponth Grama Paul Griffin and Mohammad	Collaborative Cloud Computing Framework for Health Data with Open Source	Conference		11th ACM International Conference on Riginformation, Computational Rights, and	
Library	08.11.2024	Adibuzzaman	Technologies	paper	included	Health Informatics (BCB '20)	2020
IEEE Xolore	30 11 2024	M. K. Salloway; X. Deng; Y. Ning; S. L. Kao; Y. Chen; G. O. Schaefer; J. J L. Chin: E. S. Tai: C. S. Tan	A de-identification tool for users in medical operations and public health	Conference	included	Global Healthcare Conference	2016
ACM Digital	00.11.2024		Democracy beyond projects: how users may achieve lasting influence on it	Article	maaaaa		
Library ACM Digital	21.06.2024	Morten Kyng	systems	Conference	included	Interactions 2017 International Conference on Digital	2016
Library	08.11.2024	Keyuan Jiang, Yongbing Tang, G. Elliott Cook, and Michael M. Madden	Discovering Potential Effects of Dietary Supplements from Twitter Data	paper	included	Health (DH '17)	2017
Scopus	30.06.2024	Huang H.; Aschettino S.; Lari N.; Lee TH.; Rosenberg S.S.; Ng X.; Muthuri S.; Bakshi A.; Bishop K.; Ezzeldin H.	A Versatile and Scalable Platform That Streamlines Data Collection for Patient- Centered Studies: Usability and Feasibility Study	Article	included	JMIR Formative Research	2022
Samue	27.06.2024	Mohamad Y.; Gabber A.; Heidenblut S.; Zenz D.; Siddiqi A.; Gappa H.	How to Overcome Lack of Health Record Data and Privacy Obstacles in Initial	Article	included	Computing and Information	2022
Scopus	27.00.2024	Sharma P.; Montgomery R.N.; Graves R.S.; Meyer K.; Hunt S.L.; Vidoni E.D.;	CONSENSUS: a Shiny application of dementia evaluation and reporting for the	Article	Included	Computing and informatics	2021
Scopus	01.07.2024	Mahnken J.D.; Swerdlow R.H.; Burns J.M.; Mudaranthakam D.P.	KU ADC longitudinal Clinical Cohort database	Conference	included	JAMIA Open 2018 IEEE/ACM International Conference on	
ACM Digital		Reza M. Parizi, Liang Guo, Yao Bian, Ali Dehghantanha, Amin Azmoodeh,	CyberPDF: smart and secure coordinate-based automated health PDF data	paper		Connected Health: Applications, Systems and	
Library	08.11.2024	and Kim-Kwang Raymond Choo	batch extraction	Conference	included	Engineering Technologies (CHASE '18) 12th ACM Conference on Bioinformatics	2020
ACM Digital		Aishwarya Mandyam, Elizabeth C. Yoo, Jeff Soules, Krzysztof Laudanski,	COP-E-CAT: cleaning and organization pipeline for EHR computational and	paper		Computational Biology, and Health	
Library	08.11.2024	and Barbara E. Engelhardt J. Sivaraniani: A. N. Madheswari	A novel technique of motif discovery for medical big data using hadoop	Conference	included	Informatics (BCB '21)	2021
IEEE Xplore	30.11.2024	Ciapponetes TV : Fordinand A O : Illanamon O : Pagen E : Neurall W/P :	Identifying and manifiliaing honofits and viete of using privacy ophonoing	paper	included	ICEDSS	2024
and the second second		Kum HC.; Schmit C.D.	software through participatory design: A nominal group technique study with	Anide		Journal of the American Medical Informatics	2021
Scopus ACM Digital	29.06.2024	Isaac Holeman, Amanda Yembric, David Brown, Dianna Kane, Jane Katanu,	patients living with chronic conditions Design and Implementation of an Open Source 'Thin SIM' System for	Conference	included	Association 7th Annual Symposium on Computing for	
Library	08.11.2024	Marc Abbyad, and Ranju Sharma	Collecting Data & amp; Supporting Global Health Care	paper	included	Development (ACM DEV 16)	2016
Library	24.06.2024	Pedro Sanches and Barry Brown	Data Bites Man: The Production of Malaria by Technology	Article	included	Human-Computer Interaction	2018
Scopus	06 12 2024	Mande R.; Jayalakshmi G.; Yelavarti K.C.	Leveraging Distributed Data over Big Data Analytics Platform for Healthcare Services	Conference	included	2nd International Conference on Trends in Electronics and Informatics. ICOEL 2018	2018
-	CO.TELECET	Eapen B.R.; Costa A.; Archer N.; Sartipi K.	FHIRForm: An open-source framework for the management of electronic forms	Article			2019
Scopus	01.07.2024	D. R. Harris	In nealthcare Leveraging Differential Privacy in Geospatial Analyses of Standardized	Conference	included	Studies in Health Technology and Informatics	2020
IEEE Xplore	05.12.2024	Process E - Gounn J - Wan Z - Xia W - Vorobeuchik V - Kontercioglu M - Kuhn	Healthcare Data	paper	included	IEEE Big Data 2020	2017
Scopus	02.07.2024	K; Malin B.		A GUIG	included	Symposium	2017
Scopus	01.07.2024	Ray P.P.; Dash D.; De D.	real-time event-driven sensor data analytics at the edge-Internet of Things for smart personal healthcare	Article	included	Journal of Supercomputing	2020
Sconue	01.07.2024	Xu H.; Cox S.; Stillwell L.; Pfaff E.; Champion J.; Ahalt S.C.; Fecho K.	FHIR PIT: An open software application for spatiotemporal integration of clinical data and environmental exposures data	Article	included	BMC Medical Information and Dusinian Malein-	2020
O	00.000	Silva S.; Altmann A.; Gutman B.; Lorenzi M.	Fed-BioMed: A General Open-Source Frontend Framework for Federated	Conference	manuada	LUCO	2020
Scopus	06.12.2024	Chrimes D.; Zamani H.	Learning in Healthcare Using Distributed Data over HBase in Big Data Analytics Platform for Clinical	Article	included	Computational and Mathematical Methods in	2017
Scopus	02.07.2024	Rao A.R.: Clarke D	Services	Conformer	included	Medicine	2040
Scopus	05.12.2024	NOV SAN, WEINE D.	Architecture and applications	paper	included	Informatics	2016
		Bahmani A.; Alavi A.; Buergel T.; Upadhyayula S.; Wang Q.; Apanthakrishnan S.K.; Alavi A.; Celis D.; Gillespie D.; Young G.; Xing Z.;	A scalable, secure, and interoperable platform for deep data-driven health management	Article			2021
		Nguyen M.H.H.; Haque A.; Mathur A.; Payne J.; Mazaheri G.; Li J.K.; Kotipalli					
	100.00000000000000000000000000000000000	E.; Zhou W.; Berry C.L.; Van Winkle K.G.; Contrepois K.; Ray U.; Bettinger					
Scopus	27.06.2024	K.; Datta S.; Li X.; Snyder M.P. Bent B.; Wang K.; Grzesiak E.; Jiang C.: Qi Y.: Jiang Y.: Cho P.: Zingler K.:	The digital biomarker discovery pipeline: An open-source software platform for	Article	included	Nature Communications	2021
Scopus	01.07.2024	Ogbeide F.I.; Zhao A.; Runge R.; Sim I.; Dunn J.	the development of digital biomarkers using mHealth and wearables data		included	Journal of Clinical and Translational Science	
Scopus	02.07.2024	Blum M.; Adi S.	interoperability in type 1 diabetes management	Article	included	Association American Medical Informatics	2016
Scopus	02.07.2024	Memon M.A.; Soomro S.; Jumani A.K.; Kartio M.A.	Big data analytics and its applications	Article	included	Annals of Emerging Technologies in Computing	2017
Scopus	02.07.2024	Nabian M.; Yin Y.; Wormwood J.; Quigley K.S.; Barrett L.F.; Ostadabbas S.	An open-source teature extraction tool for the analysis of peripheral physiological data	Article	included	HELE Journal of Translational Engineering in Health and Medicine	2018
Sconue	05 12 2024	Harper A.; Monks T.	A FRAMEWORK TO SHARE HEALTHCARE SIMULATIONS ON THE WEB	Conference	induded	Operational Research Society Simulation	2023
ocopus	33.12.2024		SUNS THE AND OF EN SOUNCE FOULS AND PTIMON	Conference	manuded	2nd International Conference on Software	
ACM Digital	04 11 2024	Suganya Selvarai and Eunmi Choi	A Study on Traditional Medicine Ontology	paper	induded	Engineering and Information Management (ICSIM '19)	2010
Liorary	04.11.2024	Stefan Wagner	CARIOT+ Care Coach – An Ambient Assisted Living Ecosystem for Supporting	Conference	mauded	International Conference on ICT for Health.	2019
Springer	07.12.2024	Simon Patterson NauvanPatter Nielson Jahon Ivan Contra	Open Data and Open Science Projects	paper	included	Accessibility and Wellbeing	0010
Springer	08.12.2024	Sinon Fettersen inguyeriPetter NielsenJohan IVar Sæbø	Giouai Grandaros ano Local Development	paper	included	IDIA 2018	2019
ACM Digital	11 11 2024	Miltos D. Grammatikakis, Voula Pineraki, and Antonie Panagrigoriou	Multilaver NoC firewall services: case-study on e-health	Conference paper	included	15th IEEE/ACM International Symposium on Networks-on-Chin (NOCS '21)	2024
ACM Digital			the start of the s		mudued	2000 OIII Conference on United Enters in	2021
ACIVI Digital	11.11.2024		MyCareBudget: Co-creating a Healthcare Digital Commons with and for	Conference		2023 GHI Conierence on Human Factors in	
Library	11.11.2024	Peter Glick and Clara Crivellaro	MyCareBudget: Co-creating a Healthcare Digital Commons with and for Disabled Citizens and their Unpaid Carers Badillating Healthcare IoT Standardization with Open Source: A Case Study on	Conference paper	included	Computing Systems (CHI '23) 2019 ITU Kaleidoscope: ICT for Health:	2023
Library IEEE Xplore	11.11.2024 11.11.2024 05.12.2024	Peter Glick and Clara Crivellaro H. Cha; Y. Choi; K. Lee	MyCareBudget: Co-creating a Healthcare Digital Commons with and for Disabled Citizens and their Unpaid Carers Facilitating Healthcare IoT Standardization with Open Source: A Case Study on OCF and IoTivity	Conference paper Conference paper	included included	2023 Chi Conterence of Human Pactors in Computing Systems (CHI 23) 2019 ITU Kaleidoscope: ICT for Health: Networks, Standards and Innovation (ITU K)	2023 2019
	05 40 0004	Q. H. Nguyen; Q. Dang; C. Pham; TD. Nguyen; H. Nguyen; A. Setty; T. Q.	Developing an Architecture for IoT Interoperability in Healthcare: A Case Study	Conference	in all rate of	IEEE Internetional Conference on Ris Data	
---	--	---	--	--	---	--	
IEEE Xplore	05.12.2024	Le Dunbar E.; Elizabeth Olsen H.; Salomon E.; Bhatt S.; Mutuku R.; Wasunna	of Real-time SpO2 Signal Monitoring and Analysis Towards Responsible Data Practices in Digital Health: A case study of an open	Conference	included	CHI Conference on Human Factors in	
Scopus	06.12.2024	B.; Edwards J.; Kolko B.; Holeman I.	source community's journey	paper	included	Computing Systems	
		Alkarkoukiy S.; Kamal M.M.; Beyan O.	Transformation Using Open-Source Tools	paper		33rd Medical Informatics Europe Conference: Caring is Sharing - Exploiting the Value in Data for	
Scopus	05.12.2024	Khan M.T. Camar C. Sinka D.D.	Adaptation of every setting to be lithered in larger Cone of CAREOV	Antinta	included	Health and Innovation	
Scopus	02.07.2024	Khair W. I., Qaimar S., Sinna K.K.	Adaptation of open source soliware for nearlingare in India. Case of CAREZX	Antice	included	Medwell Journals	
Sconue	06 12 2024	Cha H.; Jeon J.	OCF healthcare proof of concept (PoC) on libelium MySignals	Conference	induded	2017 European Conference on Electrical	
ocopus	00.12.2024	Kotoulas A.; Stratis I.; Goumenidis T.; Lambrou G.; Koutsouris DD.	Short-term adoption rates for a web-based portal within the intranet of a hospital	Article	madded	ragineering and compliter screace, taxes 2017	
Scopus	01.07.2024		information system	Conference	included	BMJ Health and Care Informatics 26th International Conference on Evaluation	
ACM Digital				paper		and Assessment in Software Engineering	
Library	04.11.2024	Ajoy Das, Gias Uddin, and Guenther Ruhe	An Empirical Study of Blockchain Repositories in GitHub		included	(EASE '22)	
Springer	07.12.2024	A. Ravishankar RaoDaniel ClarkeMaryelena Vargas	Building an Open Health Data Analytics Platform: a Case Study Examining Relationships and Trends in Seniority and Performance in Healthcare Providers	Article	included	Journal of Healthcare Informatics Research	
0	05 40 0004	Janki Z.R.; Szabo Z.; Bilicki V.; Fidrich M.	Authorization solution for full stack FHIR HAPI access	Conference	Sector de d		
Scopus	05.12.2024	L. D. Pietro; C. De Maria; A. Ravizza; A. Ahluwalia	Co-design open-source medical devices: how to minimize the human error	Conference	included	IEEE 30th Jubliee Neumann Colloquium	
IEEE Xplore	05.12.2024	On the Ohida Mahad Fadal Union Law Tran Line Adapt Autors and	using UBORA e-infrastructure	paper	included	Horizon 2020	
ACM Digital Library	25.10.2024	Saurabh Shukla, Mohd Fadzil Hassan, Low Tang Jung, Azlan Awang, and Muhammad Khalid Khan	A 3-Tier Architecture for Network Latency Reduction in Healthcare Internet-of- Things Using Fog Computing and Machine Learning	paper	induded	8th International Conference on Software and Computer Applications (ICSCA '19)	
Lioidiy		R. Oliveira; D. Ferreira; R. Ferreira; R. Cruz-Correia	Open-Source Based Integration Solution for Hospitals	Conference		IEEE 29th International Symposium on	
ACM Digital	05.12.2024		A Systematic Review of Open Source Clinical Software on GitHub for Improving	Article	included	Computer-Based Medical Systems	
Library	13.05.2024	Zhengru Shen and Marco Spruit	Software Reuse in Smart Healthcare	6 -1 -1 -	included	Data Analytics in Smart Healthcare	
Scopus	02.07.2024	Jawnan B.; Keenan L.; Zakus D.; Ludwick D.; Isaac A.; Salen A.; Hayward R.	slum	Article	included	International Journal of Medical Informatics	
Connun	11 11 2024	Oyarzún-Diaz P.; Orellana-Cortés A.; Segura H.; Vidal-Silva C.; Sánchez- Ortiz A.; Sarzana Malabráa I.	Sustainable Health Education Simulator Using Open-Source Technology	Article	indudad	MORI	
ACM Digital	11.11.2024	Oniz A., Senano-Malebran J.		Article	maddea	MDFI	
Library	11.11.2024	Gregory W. Hislop, Heidi J. C. Ellis, and Herman Jackson	Student contribution to HFOSS: challenges and opportunities	Conformaco	included	Journal of Computing Sciences in Colleges	
IEEE Xplore	30.11.2024	Tjahjowidodo	Surgical Simulators	paper	included	SSCI	
IEEE Xolore	05 12 2024	C. Silva; J. C. Campos	Towards a Simulation-Based Medical Education Platform for PVSio-Web	Conference	included	2018 International Conference on Graphics	
Scopus	11.11.2024	Nohria R.; Kang J.; Belen J.M.	Reflection on the development of a patient case in twine	Article	included	Currents in Pharmacy Teaching and Learning	
0		Lemarteleur V.; Peycelon M.; Sablayrolles JL.; Plaisance P.; El-Ghoneimi	Realization of Open Software Chain for 3D Modeling and Printing of Organs in	Article	to do do d		
Scopus	02.07.2024	Paton C.; Karopka T.	The Role of Free/Libre and Open Source Software in Learning Health Systems	Article	included	Yearbook of medical information	
Soopaa		A. Munawar; J. Y. Wu; G. S. Fischer; R. H. Taylor; P. Kazanzides	Open Simulation Environment for Learning and Practice of Robot-Assisted	Article			
IEEE Xplore	02.07.2024	Y Feliz: P. Jethani: P. Jastrzebska-Perfect: H. Linson	Surgical Suturing The Case for a Portable Open-Source 3D Littrasound: Issues Reports and	Conference	included	IEEE Robotics and Automation Letters	
IEEE Xplore	05.12.2024		Challenges	paper	included	Symposium (IUS)	
Springer	08 12 2024	Timo BaumgärtnerBenjamin J. MittmannTill MalzacherJohannes BoßkopfMichael BraunBernd SchmitzAlfred M. Franz	Towards Clinical Translation of Deep Learning-based Classification of DSA Image Sequences for Stroke Treatment	Conference	included	Bildverarbeitung für die Medizin 2023	
opinigei	00.12.2024	Andrés Díaz Lantada	Reinventing Biomedical Engineering Education Working Towards the 2030	Conference	maaaaa	Didectarbeitang für die Medizin 2020	
Springer	08.12.2024	Mark Charge Daires D. Web Mary Mar David F. M. J. Maryakawa	Agenda for Sustainable Development	paper	included	BIOSTEC	
Springer	07.12.2024	vivek SnarmaRaimon Bullich VilarrubiasPaul F. M. J. Verschure	BrainX3: A Neuroinformatic Tool for Interactive Exploration of Multimodal Brain Datasets	paper	included	Conference on Biomimetic and Biohybrid Systems	
		George A.; Bharat C.R.; Singh T.; Sahil C.S.	Digital Twin of a Musculoskeletal System	Conference		2023 International Conference on Artificial	
Scopus	06.12.2024			paper	included	Intelligence and Applications, ICAIA 2023 and Alliance Technology Conference, ATCON-1 2023	
		Daniele PerloEnzo TartaglioneUmberto GavaFederico D'AgataEdwin	UniToBrain Dataset: A Brain Perfusion Dataset	Conference			
Springer	08.12.2024	G. Esposito: G. Pagliari: G. Coppola: M. Aiello: M. Salvatore: G. Aceto: A	A network performance view of a biobanking system for diagnostic images	Conference	included	ICIAP 2022 Workshops	
IEEE Xplore	30.11.2024	Pescapé		paper	included	Healthcom	
Scopus	06.12.2024	Souza R.; Stanley E.A.M.; Forkert N.D.	Imaging and Global Health Equity	paper	included	LNCS	
IEEE Valana	20.44.2024	R. Lebre; L. Bastião; C. Costa	An Accounting Mechanism for Standard Medical Imaging Services	Conference	Included		
IEEE ADIOR	3U. 1.ZUZ4			1241387	111111111111111111111111111111111111111		
		Chang Y.H.; Foley P.; Azimi V.; Borkar R.; Lefman J.	Primer for image informatics in personalized medicine	Conference	maaaaa	IEEE Access	
Scopus	06.12.2024	Chang Y.H.; Foley P.; Azimi V.; Borkar R.; Lefman J.	Primer for image informatics in personalized medicine	Conference paper	included	Procedia Engineering	
Scopus IEEE Xplore	06.12.2024 05.12.2024	Chang Y.H.; Foley P.; Azimi V.; Borkar R.; Lefman J. J. Ruiz-Alzola; A. Skareb; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Fernandes Rodrigues; L. L. Rivero; R. Kikinis	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa	Conference paper Conference paper	included	Procedia Engineering 2018 IEEE Global Humanitarian Technology Conference (GHTC)	
Scopus IEEE Xplore	06.12.2024 05.12.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akzola; A. Skorety, A. M. Idríss; B. Diao; J. Da Rosa De Pina Ferreira; M.A. Ferrandes Rodrigues; L. Rivero; R. Klinis Sagnaráli ľurkayastka, Surendin Babu Baddi, Siddharda Nuthakki, Blawana Yaday,	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Radiology on Chest X-	Conference paper Conference paper Conference	included included	Procedia Engineering 2018 IEEE Global Humanitarian Technology Conference (GHTC)	
Scopus IEEE Xplore Springer	06.12.2024 05.12.2024 17.12.2024	Chang Y.H.; Foley P.; Azimi V.; Borkar R.; Lefman J. J. Ruiz-Alzola; A. Skareb; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Fernandes Rodrigues; L. L. Rivero, R. Kilshi Saparshi Purkayashla, Surendra Babu Budil, Siddhartha Nuthakki, Blawana Yadav, Jady W. Gichoya Mrudang MathurWilliam D. MeadorMarcin MalinowskiTomasz	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Radiology on Chest X- Rays Texas TirValve 1.0 : a revense-engineered, open model of the human tricuspid	Conference paper Conference paper Conference paper	included included included	Incer Accuss Procedia Engineering 2018 IEEE Global Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing	
Scopus IEEE Xplore Springer Springer	06.12.2024 05.12.2024 17.12.2024 08.12.2024	Chang Y.H.; Foley P.; Azimi V.; Borkar R.; Lefman J. J. Ruiz-Alzola; A. Skareb; A. M. kiriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Fernandes Rodrigues; L. L. Rivero; R. Kikinis Saparshi Purkayasha, Surenda Babu Bukil, Sidhartha Nuthakki, Bhawana Yadav, Jady W. Gichaya Mrudang MathurWilliam D. MeadorMarcin MainowskiTomasz JazwiesTomasz A. TimeManuel K. Rausch	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LitreHealth Radiology on Chest X- Rays Traxas Tivalve 1.0 : a reverse-engineered, open model of the human tricuspid valve	Conference paper Conference paper Conference paper Article	included included included included	Procedia Engineering 2018 IEEE Global Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022)	
Scopus IEEE Xplore Springer Springer	06.12.2024 05.12.2024 17.12.2024 08.12.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akzola; A. Skareto; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Fernandes Rodrigues; L. Ruivero; R. Kilonis Saptanii Purkayasha, Surendra Babu Buddi, Siddhartha Nuthatki, Bhawana Yadav, Jady W. Gichoya Mrudang Mathur/William D. MeadorMarcin MalinowskiTomasz JazwiecTomasz. A. TimeMAnaule K. Rausch Bianchi J.; Paniagua B; De Oliveira Ruellas A.C; Fillion-Robin JC; Prietro D.C; Gonçalves J.R; Hotoci J.; Yataba M; Silyenr K; LIT F; Gurgel M.L;	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Chest X- Rays Texas Trylalve 1.0: a revense-engineered, open model of the human tricuspid valve 3D Silcer Craniomaxilloficial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research	Conference paper Conference paper Conference paper Article Conference paper	included included included included	Proceda Engineering 2018 (EEE Giobal Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022)	
Scopus IEEE Xplore Springer Springer	06.12.2024 05.12.2024 17.12.2024 08.12.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akzola; A. Skarety, A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Fernandes Rodrigues; L. Ruivero, R. Kitelinis Saptanti Putaryasha, Surenda Babu Buddi, Siddhardha Nuthakki, Bhawana Yadav, Judy W. Gichoya Mrudang Mathur/William D. MaadorMarcin Malinowak/Tomasz Jazwec Tomasz A. Timek/Manuel K. Rausch Bianch J.; Paniagua B; De Oliveira Ruellas A.C.; Fillion-Robin JC.; Prietro J.C.; Gongalves J.R.; Hoctor J.; Yatabe MS, Styner M.; U.T.F.; Gurgel ML.; Chares Junic C.M.; Massaro C., Garlo D.G.; Mirono L.; Fenrigues J.F.C.;	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Radiology on Chest X- Rays Texas TriValve 1.0 : a reverse-engineered, open model of the human tricuspid valve 30 Silcer Craniomaxillofacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research	Conference paper Conference paper Conference paper Article Conference paper	included included included included	IFEE Addusse Procedia Engineering 2018 IFEE Giobal Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022)	
Scopus IEEE Xplore Springer Springer	06.12.2024 05.12.2024 17.12.2024 08.12.2024 05.12.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akoda; A. Skarets, A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Fernandes Rodrigues; L. L. Rivero; R. Kilkinis Saparshi Putayashta, Surendra Babu Budui, Sidilarrita Nuthakki, Bhawana Yaday, Jady W. Gichoya Murdang Mathu-William D. Meador/Marcin Malinowsk/Tomasz Jazwiec Tomasz A. TimedManuel K. Rausch Bianch J.; Panaigua B; Do Oliveira Ruellas A.C.; Fillion-Robin JC.; Prietro J.C.; Gonçalves J.R.; Hoctor J.; Yatabe M; Styner M.; U T.F.; Gurgel M.L.; Chaves Junico C.M.; Massaro C.; Garib D. G.; Vilnova L.; Henrigues J.F.C.; Castillo AA-D; Janson G; Iwasaki L.R.; Nickel J.C.; Evangelista K.; Cevidanes L.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Radiology on Chest X- Rays Teaca TriValve 1.0 : a reverse-engineered, open model of the human tricuspid valve 30 Sticer Craniomaxillofacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research	Conference paper Conference paper Conference paper Article Conference paper	included included included included	IFEE Addass Procedia Engineering 2018 IEEE Global Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computer Science	
Scopus IEEE Xplore Springer Springer Scopus	06.12.2024 05.12.2024 17.12.2024 08.12.2024 05.12.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akzola; A. Skarab; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Fernandes Rodrigue; L. I. Revero; R. Kilvinis Saptarali Purkayasha, Surendra Babu Budi, Siddhartha Nuthakki, Bhawana Yadav, Jady W. Gichoya, A. TimeldAnaule K. Rausch Bianchi J; Paniagua B; De Oliveira Ruellas A.C.; Filion-Robin JC.; Prietro J.C.; Gonçalwes J.R; Hotoci J.; Yadabe M; Styner Mi, J. Tef, Gurgel M.L.; Cheres Junior C.M.; Massaro C; Garib D.G.; Vilanova L.; Henriques J.F.C.; Castillo A.A.O., Janson G.; Iwasaki L.R., Nickel J.C.; Evangelista K.; Lethor R.; Pinno E; Silva J.M.; Costa C.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Chest X- Rays Texas Thvalve 1.0: a reverse-engineered, open model of the human tricuspid valve 3D Slicer Craniomaxillofacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research	Conference paper Conference paper Conference paper Article Conference paper	included included included included included	Proceda Engineering 2018 IEEE Global Humanitarian Technology Corference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computer Science IEEE Symposium on Computers and Commonitorium on Computers and	
Scopus IEEE Xplore Springer Springer Scopus Scopus	06.12.2024 05.12.2024 17.12.2024 08.12.2024 05.12.2024 05.12.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akzola; A. Skareix, A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Fernandes Rodrigues; L. Ruivero; R. Kilonis Saptania Purkayasha, Surendra Babu Buddi, Siddhartha Nuthakki, Blawama Yaday, Judy W. Gichoya Mrudang Mathur/William D. MeadorMarcin MalinowskiTomasz Jazviete?omasz. A. TimelManuel K. Ruusch Bianchi J.; Paniagua B; De Oliveira Ruelas A.C; Fillion-Robin JC; Prietro D.C; Gonçalves J.R; Hotor J.; Yatabe M; Shyner W; Li T.F.; Gorgel ML;: Chaves Junior C.M.; Massaro C; Garib D.G; Vilanova L; Henriques J.F.C; Castillo A.A.O., Janson G; Ivasaki L.R; Nickel J.C.; Evangelista K; Lebre R; Pinho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatia M.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Chest X- Rays Texas Trivlave 1.0: a reverse-engineered, open model of the human tricuspid valve 3D Sicer Craniomaxillofacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian	Conference paper Conference paper Conference paper Article Conference paper Conference paper Article	included included included included included included	Proced a Engineering 2019 [EEE Global Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Noles in Computer Science EEE Symposium on Computers and Communications	
Scopus IEEE Xplore Springer Springer Scopus Scopus Scopus	06.12.2024 05.12.2024 17.12.2024 08.12.2024 05.12.2024 05.12.2024 13.11.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akzola; A. Skareky, A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Fernandes Rodrigues; L. Ruivero, R. Kilsnis Saptanih Putayasha, Surenda Babu Baddi, Siddhartha Nuthakki, Bhawana Yadav, Jady W. Gicbaya Mrudang Mathu-William D. MaadorMarcin MalincowskiTomasz Jazviele Tomasz. A. TimeMAnuel K. Rausch Bianch J.; Paniagua B; De Oliveira Ruelas A.G.; Filion-Robin JC.; Prieto D.; Gonçaleva J.R.; Hotor, J.; Vataba M; Shyner Wi, Li T.F.; Gurgel ML; Charese J.R.; Hotor, J.; Vataba M; Shyner Wi, Li T.F.; Gurgel ML; Charese J.R.; Hotor, J.; Vataba M; Shyner Wi, Li T.F.; Gurgel ML; Castillo A.A.O.; Janson G; Iwasaki L.R.; Nickel J.C.; Evangelista K; Cevidenes L. Lebre R; Pinho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatia M. Sinisi S; Alimouzhin V: Mancini T. Tonora E.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Radiology on Chest X- Rays. Texas TrValve 1.0: a reverse-engineered, open model of the human tricuspid valve 3D Sticer Cranicmaxillofacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Modical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telemedicine Sarvice (stanjeevani)	Conference paper Conference paper Conference paper Article Conference paper Conference paper Article	induded induded induded induded induded induded induded	Proceda Engineering 2018 IEEE Global Humanitarian Technology Corderence (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computer Science IEEE Symposium on Computers and Communications Journal of Digital Imaging	
Scopus IEEE Xplore Springer Springer Scopus Scopus Scopus Scopus	06.12.2024 05.12.2024 17.12.2024 08.12.2024 05.12.2024 05.12.2024 13.11.2024 13.11.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akzola; A. Skarab; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Fernandes Rodfigues; L. Ruvero, R. Kilvinis Saptarbi Purkayasha, Surendra Babu Budi, Siddhartha Nuthakki, Bhawana Yudav, Jady W. Gichoya, A. TimeldAnaule K. Rausch Bianchi J.; Paniagua B; De Oliveira Ruellas A.C.; Filion-Robin JC.; Prietro J.C.; Conçaleva J.R.; Hotod J.; Yatabe M; Styner M; Li T.F.; Gurgel M.L.; Chaese Junior C.M.; Massaro C; Gario D.G.; Wianova L; Henriques J.F.C.; Cavidens L.R.; Janson G; Iwasaki L.R.; Nickel J.G.; Evangelista K; Lebre R; Pinho E; Silva J.M.; Costa C. Wadail J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatia M. Sinisi S; Alimguzhin V; Manchi T; Trone E.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Radiology on Chest X- Rays Taxas Tivalve 1.0: a reverse-engineered, open model of the human tricuspid valve 30 Silcer Craniomaxillolacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telemeticine Service (Casingevani) Reconciling interoperability with efficient Verification and Validation within open source simulation environments	Conference paper Conference paper Article Conference paper Article Conference paper Article Article	included included included included included included included included	Procedia Engineering 2018 IEEE Goldal Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computer Science IEEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory	
Scopus IEEE Xplore Springer Springer Scopus Scopus Scopus Scopus Scopus	06.12.2024 05.12.2024 17.12.2024 08.12.2024 05.12.2024 05.12.2024 13.11.2024 13.11.2024 01.07.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akzola; A. Skarejs, A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Fernandes Rodrigues; L. Rivero; R. Kilonis Saptani Purkayasha, Surendra Babu Buddi, Siddhartha Nuthakki, Bhawana Yadav, Jady W. Gichoya Mrudang Mathur/William D. MeadorMarcin MalinowskiTomasz Jazviete?cmasz. A. TimeMAnaule K. Rausch Bianch J.; Paniagua B; De Oliveira Ruellas A.C.; Fillion-Robin JC.; Prietro D.; Gonçalves J.R.; Hotozi J.; Yataba M; Styner W; Li T.F.; Gurgel M.L.; Chaves J.Janson G; Iwasski L.R.; Nickel J.C.; Evangelista K; Cevidanes L. Lebre R; Pinho E; Silva J.M.; Costa C. Wadail J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatlia M. Sinisi S; Alimguzhin V; Mancini T; Tronci E. Valdermana C.E.; Stroux L; Katebi N; Paljug E; Hall-Clifford R; Rohloff P; Marzhonade F: Clifford G.D.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Chest X- Rays Texas Thvalve 1.0 : a reverse-engineered, open model of the human tricuspid valve 30 Silcer Craniomaxillofacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Fines, Open-source, Web-based DICOM Viewers for the Indian Nationalizing Intergrenability withSingerovani) Resonalized Imaging Teaching and Alexander Intergrenability of the Indian Nationaling Intergrenability withSingerovani) Resonaling Intergrenability withSingerovani) An open source audocrinational Docket withResound	Conference paper Conference paper Article Conference paper Article Conference paper Conference paper Article Article Article	included included included included included included included included	Proceda Engineering 2019 [EEE Global Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computer Science IEEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory	
Scopus IEEE Xplore Springer Scopus Scopus Scopus Scopus Scopus	06.12.2024 05.12.2024 17.12.2024 08.12.2024 05.12.2024 05.12.2024 13.11.2024 13.11.2024 01.07.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akzola; A. Skareix, A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Fernandes Rodrigues; L. Ruivero; R. Kilonis Saptania Purkayasha, Surendra Babu Buddi, Siddhartha Nuthakki, Blawama Yaday, Judy W. Gichoya Mrudang Mathur/William D. MeadorMarcin MalinowskiTomasz Jazviete?omasz. A. TimelManuel K. Ruusch Bianchi J.; Paniagua B; De Oliveira Ruelas A.C.; Fillion-Robin JC.; Prietro D.; Gonçalves J.R.; Hotor J.; Yatabe M; Stymer W, Li T.F.; Gorgen ML; Chaves Junior C.M.; Massaro C; Garib D.G.; Vilanova L; Henriques J.F.C.; Castillo A.A.O.; Janson G; Ivasaki L.R; Nickel J.C.; Evangelista K; Lebre R; Pinho E; Silva J.M.; Costa C. Wadali J.S.; Sood S. P; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatia M. Siniai S; Alimguzhin V; Mancini T; Tronci E. Valderrama C.E.; Stroux L; Katebi N; Paljug E; Hall-Olifford R; Rohloff P; Marzbannaf F; Cilfford G.D.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Chest X- Rays Toxas Trivlave 1.0: a reverse-engineered, open model of the human tricuspid valve 3D Sicor Craniomaxillofacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telemedicine Service (Sarjeevani) Reconciling interpretaility with efficient Verification and Validation within open source simulation environments An open source autocorrelation-based method for fetal heart rate estimation from one-dimensional Doppter ultrasound Comparison of Three 3D Segmentation Software Tools for Hip Surgical	Conference paper Conference paper Conference paper Article Conference paper Article Conference paper Article Conference paper Article Conference paper	included included included included included included included included	Proced a Engineering 2018 IEEE Global Humanitarian Technology Cordenance (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Noles in Computer Science IEEE Symposium on Computer sand Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement	
Scopus IEEE Xplore Springer Springer Scopus Scopus Scopus Scopus Scopus Scopus	06.12.2024 05.12.2024 17.12.2024 08.12.2024 05.12.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Alzola; A. Skaraty, A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Fernandes Rodrigue; L. Lervero, R. Kilvinis Saptani Purkayasha, Surendra Babu Budi, Siddhurtha Nuthakki, Blawana Yudav, Jady W. Gichoya Mrudang MathurWilliam D. MeadorMarcin Malinowsk/Tomasz Jazviete Temasz, A. TimeMAnuel K. Rausch Bianch J.; Paniagua B; De Oliveira Ruelias A.C.; Filion-Robin JC.; Prietro D.C.; Conçaleve J.R.; Hotori J.; Yadabe M; Silymer M; Li T.F.; Grugel M.L.; O.C.; Conçaleve J.R.; Hotori J.; Yadabe M; Silymer M; Li T.F.; Grugel M.L.; O.C.; Conçaleve J.R.; Hotori J.; Yadabe M; Silymer M; Li T.F.; Castillo AA. D; Janson G; Iwasaki L.R.; Nickel J.C.; Evangelista K; Cevidanes L. Lebre R; Pinho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatia M. Siniai S.; Alimguzhin V; Mancini T; Tronci E. Valderman G E; Clifford G, Kazzoli A; Forcellese A; Gigante A. J. Tomer, S. Skouras, J. L. Molinuevo; J. D. Gilepert; F, Abiste	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Chest X- Rays Taxas Tivalve 1.0 : a reverse-engineered, open model of the human tricuspid valve Control Craniomaxillofacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telemenvers and tricking the Control Control of the Indian National Telemenvers and the Control of the Indian Advance Simulations Service Singlevant) Reconciling interoperability with efficient Verification and Validation within open source singletion-Bervice Control method for fetal heart rate estimation An open source Subcorrelation-based method for fetal heart rate estimation Comparison of Three 3D Segmentation Software Tools for Hip Surgical Planning Multipurpose Virtua Reality Environment for Disomedical and Health Applications	Conference paper Conference paper Conference paper Article Conference paper Article Conference paper Article Conference paper Article Conference paper Article Conference paper	included included included included included included included included	IFEE Actions Procedia Engineering 2018 IEEE Global Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computer Science IEEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors IEEE Face Name Sust Bababil Engineering	
Scopus IEEE Xplore Springer Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus	06.12.2024 05.12.2024 17.12.2024 08.12.2024 05.12.2024 05.12.2024 13.11.2024 13.11.2024 13.11.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Aizola; A. Skarab; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Fernandes Rodrigue; L. I. Revero; R. Kilvinis Saptarali Purkayasha, Surendra Babu Budi, Siddhartha Nuthakki, Bhawana Yadav, Jady W. Gichoya Mrudang Mathur/William D. Meador/Marcin MalinowskiTomasz Jazviete Tomasz, A. Timel/Anaule K. Rausch Bianchi J.; Paniagua B; De Oliveira Ruellas A.C.; Fillion-Robin JC.; Prietro J.C.; Gonçalwes J.R.; Hotozi J.; Yataba M; Stymer W, Li T.F.; Gurgel M.L.; Cheres Junior C.M.; Measaro C; Garib D.G.; Wilanova L; Henriques J.F.C.; Casillo A.A. D., Janson G.; Iwasaki L.R.; Nickel J.C.; Evangelista K.; Cedero R; Pinho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R.; Syed-Abdul S; Khosla P.K.; Bhatia M. Sinisi S; Alimguzhin V; Mancini T; Tronci E. Valdermana C.E.; Silva J.K.; Facoo G; Mazzol A; Forcellese A; Gigante A. J. Torner; S. Skouras; J. L. Molinuevo; J. D. Gispert; F. Alpiste Hussain MA; Luarger S.G; Kohl M.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Check X- Rays Taxas Thvalve 1.0: a reverse-engineered, open model of the human tricuspid valve 3D Sticer Cranicomaxillofacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Modical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telemeticine Sancie (Casinevani) Reconciling interopenality with efficient Verification and Validation within open source simulational Dopper utilesound Comparison of Three 3D Segmentation Software Tools for Hip Surgical Planning Multipurpose Virtual Reality Environment for Biomedical and Health Applications Multipurpose Virtual Reality Environment for Biomedical and Health Applications Learning HL7 Firli Vising the HAPI FHIR Severe and Its Use in Medical Imaging	Conference paper Conference paper Article Conference paper Conference paper Article Article Article Article Article Article Article Article Article Article Article	included included included included included included included included included	Proceda Engineering 2018 IEEE Global Humanitarian Technology Corference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computer Science IEEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors IEEE Trans Neural Syst Rehabil Eng	
Scopus IEEE Xplore Springer Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus	06.12.2024 05.12.2024 17.12.2024 08.12.2024 05.12.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akzola; A. Skarejs, A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandez Rodrigue; L. Elvero; R. Kilonis Saptani Purkayasha, Surendra Babu Buddi, Siddhartha Nutakki, Bhawana Yadav, Jady W. Gichoya Mrudang Mathur/William D. MeadorMarcin MalinowskiTomasz Jazviete Tomasz. A. TimeMAnaule K. Rausch Bianch J.; Paniagua B; De Oliveira Ruellas A.C.; Fillion-Robin JC.; Prieto D.; Gonçalves J.R.; Hotozi J.; Yataba M; Silymer M; Li T.F.; Gurgel M.L.; Chaves J.Janson G; Iwasski L.R.; Nickel J.C.; Evangel M.L.; Castillo A.A.D.; Janson G; Iwasski L.R.; Nickel J.C.; Evangel M.L.; Castillo A.S., Janson G; Iwasski L.R.; Nickel J.C.; Evangel M.L.; Cavidanes L. Lebre R; Finho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatia M. Sinisi S; Alimguzhin V; Mancini T; Tronci E. Valdermana C.E.; Stroux L; Katebi N: Palyug E; Hall-Clifford R; Rohloff P; Marzbenna B; Cilford G.D. Manddini M; Brunzini A; Facco G, Mazzoli A; Forcellese A; Gigante A. J. Tomer; S. Skouras, J.L. Molinuevo; J.D. Gisperț; F. Alpiste Hussain MA; Langer S.G.; Kohl M.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Chest X- Rays Texas Thvalve 1.0 : a reverse-engineered, open model of the human tricuspid valve 30 Silcer Craniomaxillotacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Fines, Open-source, Web-based DICCM Viewers for the Indian National Teleners-indensitient Verification and Validation within open source simulations and treasourd An open source subcorrelation-based method for fetal heart rate estimation Comparison of Three 3D Segmentation Software Tools for Hip Surgical Plinaning Multipupose Virtual Reality Environment for Biomedical Imaging with the SIMD Dataset Anotory of an Extensible Open Source PACS	Conference paper Conference paper Conference paper Article Conference paper Article Conference paper Article Artide Artide Artide Artide Artide Artide Artide Artide Artide	included included included included included included included included included included	Proceda Engineering 2019 [EEE Global Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computers (2022) Lecture Notes in Computer Science EEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors IEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging Lownal of Digital Imaging	
Scopus IEEE Xplore Springer Scopus Scopus Scopus Scopus Scopus Scopus EEE Xplore Scopus Scopus	06.12.2024 05.12.2024 17.12.2024 08.12.2024 05.12.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akzola; A. Skaraky, A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandes Rodrigues; L. Ruivero; R. Kilonis Saptani Purkayasha, Surendra Babu Buddi, Siddhartha Nuthakki, Blawama Yaday, Judy W. Gickoya Mrudarg Mathur/William D. Meador/Marcin Malinowski/Tormasz Jazviete?omasz. A. Timelk/Anaule K. Ruusch Bianchi J.; Paniagua B; De Oliveira Ruelas A.C; Fillion-Robin JC; Prieto D.; Gonçalves J.R.; Hotor J.; Yatabe M; Stymer W, Li T.F.; Gorgel ML.; Chaves Junior C.M.; Massaro C; Garib D.G.; Vilanova L; Henriques J.F.C.; Castilto A.A.D., Janson G; Iwasaki L.R; Nickel J.C.; Evangelista K; Cevidanes L. Lebre R; Pinho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatia M. Sinisi S; Alimguzhin V; Mancini T; Tronci E. Valderrama C.E.; Stroux L; Katebi N; Paljug E; Hall-Clifford R; Rohloff P; Marzbenrad F; Clifford G.D. Mandolin M.; Buruzini A.; Facco G.; Mazzol A; Forceliese A; Gigante A. J. Torner; S. Skouras; J. L. Molinuevo; J. D. Gispert; F. Alpiste Hussain M.A; Langer S.G.; Kohl M. Valente F; Silva L.A.B; Godinho T.M.; Costa C.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Chest X- Rays Texas Trivlave 1.0 : a reverse-engineered, open model of the human tricuspid valve 3D Sticer Craniomaxillofacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telemedicine Service (Sarijeevani) Reconciling interpretability with efficient Varification and Validation within open source simulation environments An open source simulation-based method for fetal heart rate estimation from one-dimensional Doppter uthasourch Multipurpose Virtual Reality Environment for Biomedical and Health Applications Learning H.7 FHIR Using the HAPI FHIR Server and Its Use in Medical Imaging with the SIM Dataset Anatomy of an Extensible Open Source PACS Texhical Nets: SpekPy v2.0 – a software tookit for modeling x-ray tube spectra	Conference paper Conference paper Article Conference paper Article Conference paper Article Ar	included included included included included included included included included included	IFEE Actions Proced a Engineering 2018 IEEE Global Humanitarian Technology Corference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers Science IEEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors IEEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging	
Scopus IEEE Xplore Springer Scopus Scopus Scopus Scopus Scopus IEEE Xplore Scopus Scopus Scopus Scopus Scopus	06.12.2024 05.12.2024 17.12.2024 08.12.2024 05.12.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Alzola; A. Skarab; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandes Rodfugue; L. Lervero, R. Kilonis Saptanši Purkayasha, Surendra Babu Budi, Siddhartha Nuthakki, Blawana Yudav, Judy W. Gichoya Mrudang MathudWilliam D. MeadorMarcin Malinowsk/Tomasz Jazviete?emas; A. TimeMAnauek K. Ruusch Bianch J.; Paniagua B; De Oliveira Ruelias A.C.; Filion-Robin JC.; Prietro J.C.; Conçaleva J.R.; Hotori J.; Yatabe M; Stymer M; U.T.F.; Gurgel M.L.; O.; Conçaleva J.R.; Hotori J.; Yatabe M; Stymer M; U.T.F.; Gurgel M.L.; O.; Conçaleva J.R.; Hotori J.; Yatabe M; Stymer M; U.T.F.; Gurgel M, Sanson G; Wessel L.R.; Nickel J.C.; Evangelista K; Lebre R; Pinho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatia M. Sinisi S; Alimguzhin V; Mancini T; Tronci E. Valdermana C E; Stroux L; Katebi N; Paljug E; Hall-Clifford R; Rohloff P; Marzbanad F; Cillford G.D. Manddini M.; Brurzin A.; Facco G; Mazzol A; Forcellese A; Gigante A. J. Tomer, S; Skouras, J.L. Mellourevoç J.D. Gispert; F. Alpiste Hussain MA; Langer S,G.; Kohl M. Valenten F; Silva LAB; Godinho TM; Costa C. Poludniowski G; Omar A; Bujia R; Andreo P. Haak D; Page C.=; Desemo TM.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Radiology on Chest X- Rays Taxas Thvalve 1.0 : a reverse-engineered, open model of the human tricuspid valve 30 Siloer Craniomaxillolacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telementations Service (Sargievani) Reconciling interoperability with efficient Verification and Validation within open source simulational Dopper utrasourd An open source autocorrelation-based method for fetal heart rate estimation from one-dimensional Dopper utrasourd Multipupose Virtue Really Environment for Simoradci and Health Applications Learning H.T FHIRU Baing Tenchroment for Simoradci and Health Applications Learning H.T FHIRU Baing the HAPI FHIR Server and Its Use in Medical Imaging with the SIMD Dated.	Conference paper Conference paper Article Conference paper Article Conference paper Article	included included included included included included included included included included included included	Inter Actions Proced a Engineering 2018 IEEE Global Humanitarian Technology Cordinance (GHTC) Advances in Intelligent Systems and Computing Engineering with Computer Science IEEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors IEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging	
Scopus IEEE Xplore Springer Springer Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus	06.12.2024 05.12.2024 17.12.2024 08.12.2024 05.12.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Aizola; A. Skarab; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Fernandes Rodrigue; L. ILevero; R. Kilvinis Saptarali Purkayasha, Surendra Babu Budi, Siddhartha Nuthakki, Bhawana Yadav, Jady W. Gichoya Mrudang Mathur/William D. Meador/Marcin MalinowskiTomsaz Jazviete?tomsaz: J. ArmeldAnauek R. Rausch Bianchi J; Paniagua B; De Oliveira Ruellas A.C.; Fillion-Robin JC.; Prietro J.C.; Gonçalews J.R.; Hotoci J.: Yatabe M; Silymer W; Li T.F.; Gurgel M.L.; Cheres Junior C.M.; Measaro C; Garib D.G.; Vilanova L; Henriques J.F.C.; Casillo A.A. O, Janson G; Iwasaki L.R.; Nickel J.C.; Evangelista K; Cevidanes L. Lebro R; Filmb E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatla M. Sinisi S; Alimguzhin V; Mancini T; Tronci E. Valdermana C.E.; Stroux L; Katebi N; Paljug E; Hall-Clifford R; Rohloff P; Marzbanna G; Clifford G.D. Mancini A; Facco G; Mazcol A; Forcellese A; Gigante A. J. Tomer, S. Skouras; J. L. Melinevor, J. D. Gispert; F. Alpiste Hussain MA; Langer S.G.; Kolwin M.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Check X- Rays. Taxas Thvalve 1.0: a reverse-engineered, open model of the human tricuspid valve 3D Sticer Craniomaxillotacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telementations Socie (Casingevani) Reconciling interoperability with efficient Verification and Validation within open source simulation and Dopper Utasound Comparison of Three 3D Segmentation Software Tools for Hip Surgical Planning Multipupose Virtual Reality Environment for Biomedical and Mealth Applications Learning ML7 FIRI Using the HAPT FIRI Rever and Its Use in Medical Imaging with the SIM Dataset Asurory of D. Extensible Open Source PACS Technical Note: SpekP v2.0—a software toolkit for modeling x-ray tube spectra Survey of PLONU Viewer Software to Integrate Clinical Research and Medical Imaging Comparison Comparison Software Toolkit for modeling x-ray tube spectra Survey of PLONU Viewer Software to Integrate Clinical Research	Conference paper Conference paper Article Conference paper Article Conference paper Article Article Article Article Article Article Article Article Article Article Article Conference paper Article Conference paper Article Article Conference	included included included included included included included included included included included included	IFEE Actions Proceda Engineering 2018 IEEE Giobal Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computers (2022) Lecture Notes in Computers Science IFEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors IEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging Journal of Digital Imaging Medical Physics Journal of Digital Imaging Medical Physics Journal of Digital Imaging Medical Physics Journal of Digital Imaging Journal	
Scopus IEEE Xplore Springer Soringer Scopus Scopus Scopus Scopus IEEE Xplore Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus	06.12.2024 05.12.2024 17.12.2024 08.12.2024 05.12.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akzola; A. Skarejs, A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandez Rodrigues; L. Elvero; R. Kilonis Saptani Purkayasha, Surendra Babu Buddi, Siddhartha Nuthakki, Bhawana Yadav, Jady W. Gichoya Mrudang Mathur/William D. MeadorMarcin MalinowskiTomasz Jazviete Tomasz. A. TimeMAnaule K. Rausch Bianch J.; Paniagua B; De Oliveira Ruellas A.C.; Fillion-Robin JC.; Prietro D.; Gonçalwes J.R.; Hoctor J.; Yataba M; Styner M; Li T.F.; Grupel ML; Chaves Junior C.M.; Massaro C; Garib D.G.; Vilanova L; Henriques J.F.C; Castillo A.A.D.; Janson G; Iwasaki L.R.; Nickel J.C.; Evangelista K; Cevidanes L. Lebre R; Finho E; Silva J.M.; Costa C. Wadail J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatlia M. Sinisi S; Alimguzhin V; Mancini T; Tronci E. Valdermano C.E.; Stroux L; Katebi N; Paljug E; Hall-Clifford R; Rohloff P; Marzbenna G; Clifford O.D. Mandolini M; Brunzini A; Facco G; Mazzoli A; Forcellese A; Gigante A. J. Tomer, S. Skouras, J. L. Molinuevo; J. D. Gispert; F. Alpiste Hussain MA; Langer S.G.; Kohli M. Vielente F; Silva L.A.B; Godinhor TM; Costa C. Poludniowski G; Omar A; Bujita R; Andreo P. Haak D; Page CE; Deesmor TM.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Chest X- Rays. Texas Thylave 1.0: a reverse-engineered, open model of the human tricuspid valve 3D Silcer Craniomaxillotacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Fines-Open-source, Web-based DICCM Viewers for the Indian National Teleners-indian Sensitic (Sargiewani) Reconciling interoperability with efficient Verification and Validation within open source sinulational Depth evalues out the Source of Hip Surgical Planning Multipurpose Virtual Reality Environment for Biomedical Imaging with the SIM Dataset Assurey of DICOM Viewer Software toolskit for modeling x-ray tube spectra Surger Voluce Spectry v2.0—a software toolskit Cm collesing and Medical Imagina FOSS Technolo.colicis in MODELLING SPATILL ACCESSIBILITY of FNIAMY F42.TH CARE In MALWY	Conference paper Conference paper Article Conference paper Article Conference paper Article Artide Artide Artide Artide Artide Artide Artide Conference paper Article Conference	included included included included included included included included included included included included included included	Proceda Engineering 2019 [EEE Global Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computers (2022) Lecture Notes in Computer Science EEEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors EEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging Journal of Digital Imaging Journal of Digital Imaging Journal of Digital Imaging Medical Physics Journal of Digital Imaging 2019 Free and Open Source Software for Geospatia	
Scopus IEEE Xplore Springer Springer Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus	06.12.2024 05.12.2024 17.12.2024 08.12.2024 05.12.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 06.12.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Azola; A. Skarab, A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandes Rodfugue; L. Lervero, R. Kilvinis Saptanii Purkayasha, Surendra Babu Budi, Siddhurtha Nuthakki, Blawana Yudav, Jady W. Gichoya Mrudang Mathu/William D. Meador/Marcin Malinowsk/Tomasz Jazviete Temasz, A. Timed/Anauek K. Rausch Bianch J.; Paniagua B; De Oliveira Ruelias A.C.; Filion-Robin JC.; Prietro J.C.; Conçaleva J.R.; Hotori J. Yatabe M; Sighura M; Li T.F.; Gurgel M.L.; O.; Conçaleva J.R.; Hotori J., Yatabe M; Sighura M; Li T.F.; Gurgel M.L.; O.; Conçaleva J.R.; Hotori J., Yatabe M; Sighura M; Li T.F.; Gurgel M.L.; O.; Conçaleva J.R.; Hotori J., Yatabe M; Sighura M; Li T.F.; Castillo AA. D; Janson G; Iwasaki L.R.; Nickel J.C.; Evangelista K; Cevidanes L. Lebre R; Pinho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatia M. Sinial S; Alinguzhin V; Mancini T; Tronci E. Valderman C E; Stroux L; Katebi N; Paljug E; Hall-Clifford R; Rohloff P; Marzbarnad F; Clifford G.D. Manddini M, Bruzini A.; Faco G, Mazzol A; Forcellese A; Gigante A. J Tomer, S; Skouras, J. L. Molinuevo, J. D. Gispert; F. Alpiste Hussain MA; Langer S,G.; Kohli M. Valenten F; Siva L.A.B; Godiriho T.M.; Costa C. Poludnowald G; Omar A; Buijla R; Andreo P. Haak D; Page C-E; Desemo TM. Phiri Y.D.J; Munthali K.G. P, D. A. Mertadewi; W. Puspitasari; M. Saputra	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Radiology on Chest X- Rays Taxas Thvalve 1.0 : a reverse-engineered, open model of the human tricuspid valve 3D Silone Craniomaxillofacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telementories Service Gragineevani) Reconciling interoperability with efficient Verification and Validation within open source simulations Bervice Gragineevani) An open source autocorrelation-based method for fetal heart rate estimation fromparison of Three 3D Begmentation Software Tools for Hip Surgical Planning of Three 3D Begmentation Software Tools for Hip Surgical Planning of Three 3D Segmentation Software Tools for Hip Surgical Planning of Three Software Software Tools for Hip Surgical Planning of Three Software Software Tools for Hip Surgical Planning of DICOM Viewer Software to Integrate Clinical Research and Medical maning of DICOM Viewer Software to Integrate Clinical Research and Medical Promary of DICOM Viewer Software to Integrate Clinical Research and Medical PrimAry Hearth CARE in MODELLING SPATIAL ACCESSIBILITY of PRIMARY HEALTH CARE in MODELLING SPATIAL ACCESSIBILITY of PRIMARY HEALTH CARE in MODELLING View Public Health Services Using OpenERPs with Quickstant Approach	Conference paper Conference paper Article Conference paper Article Conference paper Article Artide Artide Artide Artide Artide Artide Artide Conference paper Conference paper Conference paper	included included included included included included included included included included included included included included	Inter Actions Proced a Engineering 2018 IEEE Global Humanitarian Technology Corference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computer Science IEEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors IEEE Tarts Neural Syst Rehabil Eng Journal of Digital Imaging 2019 Free and Open Source Software for Geospical 2021 IEEE 1th Annual Computing and 2021 IEEE 1th Annual Computing and 2021 IEEE 1th Annual Computing and	
Scopus IEEE Xplore Springer Sorpinger Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus	06,12,2024 05,12,2024 17,12,2024 08,12,2024 05,12,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 06,12,2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Aizola; A. Skarab; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Fernandes Rodrigue; L. ILevero, R. Kilvinis Saptarali Purkayasha, Surendra Babu Budi, Siddhartha Nuthakki, Bhawana Yadav, Jady W. Gichoya Mrudang Mathur/William D. Meador/Marcin MalinowskiTomasz Jazviete Tomasz, A. TimeldAnauel, K. Rausch Bianchi J.; Paniagua B; De Oliveira Ruellas A.C.; Fillion-Robin JC.; Prietro J.C.; Gonçaleva J.R.; Hotozi J.; Yatabe M; Stymer M; Li T.F.; Gurgel M.L.; Cheres Junior C.M.; Measaro C; Garib D.G.; Wilanova L; Henriques J.F.C.; Casillo A.A. D.; Janson G.; Iwasaki L.R.; Nickel J.C.; Evangelista K.; Celdre R; Pinho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatla M. Sinisi S; Alimguzhin V; Mancini T; Tronci E. Valdermana C.E.; Stroux L, Katebi N; Paliya E; Hall-Clifford R; Rohloff P; Marzbanna E; Clifford G.D. Manddini M. Brunzini A; Facco G; Mazzol A; Forcellese A; Gigante A. J. Torner; S. Skouras; J. L. Molinuevo; J. D. Gispert; F. Alpiste Hussain MA; Langer S.G.; Kohl M. Valente F; Silva L.A.B.; Godinho T.M.; Costa C. Poludriowski G; Omar A; Bujila R; Andreo P. Haak D; Page C-E; Deserno TM. Phir Y.D.J; Munthali K.G. P. D. A. Mertadewi; W. Puspitasari; M. Saputra Narden T, Fusina T, Filmas Mang Babuha S M; Kishabana, B.; Congilo C	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Check X- Rays. Taxas ThValve 1.0: a reverse-engineered, open model of the human tricuspid valve 3D Sticer Cranicomaxillolacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Modical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICCM Viewers for the Indian National Telementione Sancie (Casinevani) Reconciling interopenality with efficient Verification and Validation within open source simulation and Promey Statement of the Indian Advisor and Telemention Borne (edispleximations) An open source subcorrelation-based method for fetal heart rate estimation from on-dimensional Doppet utilescound Comparison of Three 3D Segmentation Software Tools for Hip Surgical Planning Multipupose Virtual Reality Environment for Biomedical and Health Applications Learning HL7 FRIIK Using the HAPI FHIR Server and Its Use in Medical Imaging with the SIM Dataset Anatomy of an Estensible Open Source PACS Tachrical Note: SpekPi v2.0 a offware toolkit for modeling x-ray tube spectra Tachviral Note: SpekPi v2.0 a offware toolkit for modeling x-ray tube spectra Tachrical Note: SpekPi v2.0 a offware toolkit for modeling x-ray tube spectra Tachrical Note: SpekPi v2.0	Conference paper Conference paper Article Conference paper Article Conference paper Article Artide Artide Artide Artide Artide Artide Artide Artide Artide Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference	included included included included included included included included included included included included included included included	IFEE Actions Proceda Engineering 2018 IEEE Giobal Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computer Science IFEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors IEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging JOI Free and Open Source Software for Geospatial Z21 IEEE Tith Annual Computing and Communication Workshop and Conference Information	
Scopus IEEE Xplore Springer Soringer Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus	06,12,2024 05,12,2024 17,12,2024 08,12,2024 05,12,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akzola; A. Skarejs, A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandez Rodrigue; L. Elvero; R. Kilvinis Saptaria Purkayasha, Surendra Babu Buddi, Siddhartha Nutakki, Bhawana Yadav, Jady W. Gichoya Mrudarg Mathur/William D. MeadorMarcin MalinowskiTomasz Jazviete Tomasz. A. TimeMAnaule K. Rausch Bianch J.; Paniagua B; De Oliveira Ruellas A.C.; Fillion-Robin JC.; Prieto D.; Gonçaleve J.R.; Hotor J. Yataba M; Styner W, Li T.F.; Gurgel M.L.; Chaves J.Niero C.M.; Massaro C; Garib D.G.; Vilanova L; Henriques J.F.C.; Castillo A.A. D, Janson G; Iwasski L.R.; Nickel J.C.; Evangel M.L.; Chaves Junior C.M.; Massaro C; Garib D.G.; Vilanova L; Henriques J.F.C.; Castillo A.A. D, Janson G; Iwasski L.R.; Nickel J.C.; Evangel M.L.; Crevidanes L. Lebre R; Finho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatlia M. Sinisi S; Alimguzhin V; Mancini T; Tronci E. Valdermano C.E.; Stroux L; Katebi N: Palyug E; Hall-Clifford R; Rohloff P; Marzbanna G; Cilford G.D. Manddini M; Brunzini A; Facco G.; Mazzoli A; Forcellese A; Gigante A. J. Tomer; S. Skouras, J. L. Molinuevo; J. D. Gisperți F. Alpiste Hussain MA; Langer S.G.; Kohl M. Valente F; Silva L.A.B; Godinho T.M.; Costa C. Poludniowski G; Omar A; Bujila R; Andreo P. Haak D; Page CE; Desemo T.M. Phir Y.D.J; Munthali K.G. P. D. A. Mertadewi; W. Puspitasari; M. Seputra Maclag T; Douglas T; Rizwan Islam Rhythm S.M.; Kyabaggu R; Camillo C. A.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Chest X- Rays Taxas Thvalve 1.0 : a reverse-engineered, open model of the human tricuspid valve 3D Silcer Craniomaxillotacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICCM Viewers for the Indian National Telementatione source (edge)evani) Reconciling interoperability with efficient Verification and Validation within open source simulation and Pomer Adamses and DicCM Viewers for the Indian An open source subcorrelation-based method for fetal heart rate estimation Comparison of Three 3D Segmentation Software Tools for Hip Surgical Planning Multipurpose Virtual Reality Environment for Biomedical and Health Applications Learning HL7 Filth Using the HAPI Filth Server and Its Use in Medical Imaging with the SIM Dataset Assurey of DICCM Viewer Software Tools for Tholeing Aray tube spectra Pasitory of DICCM Viewer Software HAPI Filth Server and Its Use in Medical Imaging With PIM SIM Dataset Assurey of DICCM Viewer Software HAPI Filth Server and the Use in Medical Imaging Posis Technical Note: SpekFy v2.0— a software bookit for modeling x-ray tube spectra Pasitory of DICCM Viewer Software In MatAWI POSIS Technical Research Add Medical PRIMARY HEALTH CARE IMALAWI	Conference paper Conference paper Article Conference paper Article Conference paper Article Artide Artide Artide Artide Artide Artide Artide Conference paper Conference paper Conference paper Conference paper	induded induded induded induded induded induded induded induded induded induded induded induded induded induded induded induded induded induded induded induded	Proceda Engineering 2019 [EEE Global Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computers (2022) Lecture Notes in Computers Science EEEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors IEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging Journal of Digital Imaging JOIP Free and Open Source Software for Geospratic 2019 Free and Open Source Software for Geospratic Communication Workshop and Conference. ICCWC)	
Scopus IEEE Xplore Springer Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus	06,12,2024 05,12,2024 17,12,2024 08,12,2024 05,12,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akzola; A. Skarejs, A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandez Rodrigues; L. Rivero; R. Kilonis Saptani Purkayasha, Surendra Babu Buddi, Siddhartha Nuthakki, Blawana Yadav, Judy W. Gichoya Mrudang Mathur/William D. Meador/Marcin Malinowski/Tomasz Jazviete?omasz. A. TimedManuel K. Rausch Bianch J.; Paniagua B; De Oliveira Ruellas A.C; Fillion-Robin JC; Prieto D.; Gonçaleve J.R.; Hotor J.; Yataba M; Styner M; Li T.F.; Gorgel ML; Chaves Junior C.M.; Massaro C; Garib D.G.; Vilanova L; Henriques J.F.C; Castillo AD., Janson G; Iwasaki L.R.; Nickel J.C.; Evangelista K; Cevidanes L. Lebre R; Filnho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatia M. Sinsis S; Alimguzhin V; Mancini T; Tronci E. Valetarma C.E.; Strouz L; Katebi N; Paljug E; Hall-Clifford R; Rohloff P; Marzbarna F; Cilford G.D. Mandolini M, Brurzini A.; Faco G; Mazzoli A; Forcellese A; Gigante A. J. Tomer, S. Skouras; J. L. Molinuevo; J. D. Gispert; F. Alpiste Hussain MA; Langer S.Q.; Kohl M. Valente F; Silva L.A.B; Godinho T.M.; Costa C. Poludriowaki G; Omar A; Bujia R; Andreo P. Haak D; Page CE; Desemo T.M. Phili Y.D.J; Munthali K.G. P. D. A. Mertadewi; W. Puspitasari; M. Sapulra Maclaga T; Douglas T; Rizwan Islam Rhythm S.M.; Kyabaggu R; Camillo C. A.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Railology on Chest X- Rays Taxas Thvalve 1.0 : a reverse-engineered, open model of the human tricuspid valve 3D Silone Craniomaxillolacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telementione Service (Sarjeevant) Reconciling interoperability with efficient Verification and Validation within open source simulations Service (Sarjeevant) An open source such Dorpher ultrasourd Oromparison of Three 3D Segmentation Software Tools for Hip Surgical Planning Multipurpose Virtual Reality Environment for Biomedical and Health Applications Learning H.T FHIRL Using the HAPI FHIR Server and Its Use in Medical Imaging FOSS TECHNOLOGIES in MODELLING SPATIAL ACCESSIBILITY of PRIMARY HEALT CARE in MALAWII Optimization of Event Management Activity in Public Health Services Using OpenERP with Calcitestar Dynach.	Conference paper Conference paper Article Conference paper Article Conference paper Article Artide Artide Artide Artide Artide Artide Artide Artide Artide Conference paper Conference paper Conference paper Conference paper Conference paper Conference	induded induded induded induded induded induded induded induded induded induded induded induded induded induded	Proced a Engineering 2019 [EEE Global Humanitarian Technology Corforence (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computers Science EEEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors EEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging Medical Physics Journal of Digital Imaging Medical Physics Journal of Digital Imaging Medical Physics Journal of Digital Imaging 2019 Free and Open Source Software for Geospatial 2021 [EEE Tith Annual Computing and Communication Workshop and Conference (ICCMC)	
Scopus IEEE Xplore Springer Sorpus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus ACOD Ojgital Library	06.12.2024 05.12.2024 17.12.2024 08.12.2024 05.12.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 05.12.2024 05.12.2024 05.12.2024 08.11.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Aizola; A. Skarab; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandez Rodrigue; L. ILevero; R. Kilvinis Saptarbi Purkayasha, Surendra Babu Budi, Siddhartha Nuthakki, Bhawana Yadav, Jady W. Gichoya Mrudang MathurWilliam D. MaadorMarcin MalinowskiTomasz Jazvete Tomasz, A. TimelManuel K. Rausch Bianchi J.; Paniagua B; De Oliveira Ruellas A.C.; Filtion-Robin JC.; Prieto J.C.; Gongalews J.R.; Hotod: J. Yatabe M; Stylmer W. Li T.F.; Gurgel M.L.; Chaese Junior C.M.; Messaro C; Gario D.G.; Wianova L; Henriques J.F.C.; Gevidence L., Janson G; Iwasaki L.R.; Nickel J.G.; Evangelista K; Lebre R; Pinho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatia M. Sinisi S; Alimguzhin V; Manchi T.; Tronci E. Valderrama C.E.; Stroux L; Katebi N; Palya E; Hall-Clifford R; Rohloff P; Marzbanna E; Clifford G.D. Manddini M.; Brunzini A.; Facco G; Mazzol A; Forcellese A; Gigante A. J. Tomer; S. Skouras; J. L. Molinuevo; J. D. Gispert; F. Alpiste Hussain MA; Langer S.G.; Kohl M. Valente F; Silva L.A.B.; Godinho T.M.; Costa C. Poludriowski G; Cimar A; Bujlia R; Androe P. Haak D; Page C.=E; Desemo TM. Phiri Y.D.J; Munthali K.G. P. D. A. Mertadewi; W. Puspitasari; M. Saputra Maclag T; Douglas T; Rizwan Islam Rhythm S.M.; Kyabaggu R; Camillo C, A. Anandaya, Riccardo Bruno, and Roberto Barbera Yasaen M.G.; McK SA; Alawan Islam Rhythm S.M.; Kyabaggu R; Camillo C, A.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Radiology on Chest X- Rays. Taxas ThValve 1.0: a reverse-engineered, open model of the human tricuspid valve 30 Silcer Craniomaxillolacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telementione Service (Sarjeevan) Reconciling interoperability with efficient Verification and Validation within open source simulation environments An open source autocorrelation-based method for fetal heart rate estimation Comparison of Three 3D Begimentation Software Tools for Hip Surgical Panatomy of an Extensible Open Source PACS Taching Nub. 2000 Viewers Software Tools for Hip Surgical Panatomy of an Extensible Open Source PACS Taching Nub. 2000 Viewer Software Tools for modeling x-ray tube spectra Source Syndauter to Industra to Integrate Clinical Research and Medical Imaging PEOS TECHNOLOGIES in MODELLING SPATIAL ACCESSIBILITY of PRINARY HEALTH CARE in MALAWI OpenEre With Quickstaf Approach Andeling and Search Approach. Abesign Internon Towards a Muth-Modal Software System for Improving Home Care Experiences in Saskatchewan Building public health: the case of a science gateway for physical activity lifelong modeling and simulation Critical aftors affecting the adoptive proving Home Care Experiences in Saskatchewan	Conference paper Conference paper Article Conference paper Article Conference paper Article Article Article Article Article Article Article Article Article Article Article Article Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Article Article Article Article Article Article Conference paper Conference paper Conference paper Conference paper	induded induded induded induded induded induded induded induded induded induded induded induded induded	Preceda Engineering 2018 IEEE Global Humanitarian Technology Corference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computer Science IEEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors IEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging Journal of Computer Science and	
Scopus IEEE Xplore Springer Soringer Scopus	06,12,2024 05,12,2024 17,12,2024 05,12,2024 05,12,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 12,11,2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akzola; A. Skarab; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Forrandec Rodrigue; L. ILevero; R. Kilvinis Saptarali Purkayasha, Surendra Babu Buddi, Siddhartha Nuthakki, Bhawana Yadav, Jady W. Gichoya Mrudang Mathur/William D. Maador/Marcin MalinowskiTomasz Jazviete Tomasz, A. TimeMAnaule K. Rausch Bianch J.; Paniagua B; De Oliveira Ruellas A.C.; Fillion-Robin JC.; Prieto D.C.; Gonçalews J.R.; Hotoci J.; Yataba M; Silymer W; Li T.F.; Gurgel M.L.; Chaves Junior C.M.; Massaro C; Garib D.G.; Vilanova L; Henriques J.F.C.; Castlito A.A. O, Janson G; Iwasaki L.R.; Nickel J.C.; Evangelista K; Cevidanes L. Lebor R; Finho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatlia M. Sinisi S; Alimguzhin V; Mancini T; Tronci E. Valderama C.E.; Stroux L.; Katebi N; Paljug E; Hall-Clifford R.; Rohioff P; Marzbanna G; Clifford G.D. Manzoli A.; Forcellese A.; Gigante A. J. Tomer, S. Skouras; J. L. Molinuevo; J.D. Gispert; F. Alpiste Hussain M.A.; Langer S.G.; Kohi M. Valente F; Silva L.A.B.; Godinho T.M.; Costa C. Poludniowski G; Omar A; Bujila R; Andreo P. Haak D; Page CE; Desemo T.M. Phiri Y.D.J; Munthali K.G. P. D. A. Mertadewi; W. Puspitasari; M. Saputra Maclag T; Dougtas T; Rizvan Islam Rhythm S.M.; Kyabaggu R; Camillo C. A. Anastalia Anagnostou, Simon J. E. Taylor, Denk Groen, Dana Suleimenova, Nama Anokye, Riccardo Buno, and Roberto Barbera Yaseen M.G; Aldo S.A; Adob I.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Chest X- Rays. Taxas ThValve 1.0 : a reverse-engineered, open model of the human tricuspid valve 3D Sticer Craniomaxillotacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telementations Sovice (Sanjeevan) Reconciling interoperability with efficient Verification and Validation within open source simulation and Dopper util Wasound Comparison of Three 3D Segmentation Software Tools for Hip Surgical Planning Multipupose Virtual Reality Environment for Biomedical and Health Applications Learning HLT Frith Using the HAPI FHIR Server and Its Use in Medical Imaging Whith SIMD Dataset Tachtory I Schenbly 2.0 a software toolkit for modeling x-ray tube spectra Nationa Telement Activity in Public Health Services Using OpenErent Activity and ModelLING SPATIAL ACCESSIBILITY of PRIMARY HEALT CHON CORES in MODELLING SPATIAL ACCESSIBILITY of PRIMARY HEALT CHON MODELLING SPATIAL ACCESSIBILITY of PRIMARY HEALT CHON MARKER in MALAWI OpenErent Management Activity in Public Health Services Using OpenERP with Quickstart Approach A Design Iteration Towards a Mult-Modal Software system for Improving Home Care Experiences in Saskatchewan Building global research capacity in public health: the case of a science gateway for Physical activity Illeving modeling and simulation Critical Rators affecting the adoption of open source software in public organizations	Conference paper Conference paper Article Conference paper Article Conference paper Article Artide Artide Artide Artide Artide Artide Artide Artide Conference paper Conference paper Conference paper Conference paper Conference paper	induded induded induded induded induded induded induded induded induded induded induded induded induded induded induded induded induded induded induded	Proceda Engineering 2018 [EEE Global Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computer Science EEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors IEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging Journal of Digital Imaging Journal of Digital Imaging Journal of Digital Imaging Medical Physics Journal of Digital Imaging Journal of Digital Imaging Medical Physics Journal of Digital Imaging JOI Preve and Open Source Software for Geosprial 2019 Free and Open Source Software for 2019 Free And Open Source Soft	
Scopus IEEE Xplore Springer Soringer Scopus Scopus Scopus Scopus Scopus IEEE Xplore Scopus	06,12,2024           05,12,2024           17,12,2024           08,12,2024           05,12,2024           13,12,0024           13,11,2024           13,12,2024           05,12,2024           06,12,2024           06,12,2024           06,12,2024           12,11,2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Alzola; A. Skarab; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandez Rodrigues; L. Rivero; R. Kilonis Saptani Purkayasha, Surendra Babu Buddi, Siddhartha Nuthakki, Bhawana Yadav, Jady W. Gichoya Mrudang Mathur/William D. MeadorMarcin MalinowskiTomasz. Jazviete Tomasz. A. TimeMAnaule K. Rausch Bianch J.; Paniagua B; De Oliveira Ruellas A.C.; Fillion-Robin JC.; Prieto D.; Gonçalwes J.R.; Hotozi J.; Yataba M; Styner Wi, Li T.F.; Grayen ML; Chaves J.J.; Poniagua B; De Oliveira Ruellas A.C.; Fillion-Robin JC.; Prieto D.; Gonçalwes J.R.; Hotozi J.; Yataba M; Styner W; Li T.F.; Grayen ML; Chaves J.J.; Bonson G; Iwasaki L.R.; Nickel J.C.; Evangelista K; Cevidanes L. Lebre R; Filnho E; Silva J.M.; Costa C. Wadail J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatlia M. Sinisi S; Alimguzhin V; Mancini T; Tronci E. Vialdermano C.E.; Stroux L; Katebi N; Paljug E; Hall-Clifford R; Rohloff P; Marzbenned F; Clifford O. Mandolini M; Brunzini A; Facco G; Mazzoli A; Forcellese A; Gigante A. J. Tomer, S. Skouras, J. L. Molinuevo; J. D. Gispert; F. Alpiste Hussain MA; Langer S.G.; Kohli M. Vielnet F; Silva L.A.B; Godinhor TM; Costa C. Poludniowski G; Omar A; Bujita R; Andrao P. Haak D; Page CE; Deesem T.M. Phiri Y.D.J; Munthali K.G. P. D. A. Mertadewi; W. Puspitasari; M. Saputra Maclaga T; Douglas T; Rizvan Islam Rhythm S.M.; Kyabaggu R; Camillo C. A. anastaia Anagnostou, Simon J. E. Taylor, Deenk Groen, Diana Suleimenova, Anama Anolye, Riccardo Bunov, and Roberto Barbera Yaseen M.G; Abd S.A; Adeb I. Duarte L; Lobo M; Viana J; Freitas A; Teodoro AC.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Chest X- Rays. Texas Thylave 1.0 : a reverse-engineered, open model of the human tricuspid valve 3D Silcer Craniomaxillotacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Fines, Open-source, Web-Jaead DICCM Viewers for the Indian National Telementations Sencie (darginewrai) Reconciling interoperability with efficient Verification and Validation within open source simulations Sencie (darginewrai) Multipurpose Virtual Reality Environment for Biomedical Imaging With Byll Subgementation Software Tools for Hip Surgical Planning Wultipurpose Virtual Reality Environment for Biomedical Imaging with the SIM Dutaset A Survey of DICCM Viewer Software to toolk for modeling x-ray tube spectra Surger Synditics Specify Cop-a software toolk for modeling x-ray tube spectra Surger Synditics Specify Cop-a Software Tools for Hip Surgical Planning POSS TECHNOLOGIES In MODELLING SPATILAL ACCESSIBILITY of PRIMARY HEALTH CARE In MALAWI Optimization Towards a Multi-Modal Software System for Improving Home Carrie Experiments in Stakatchewan Building global research appendix hup holib health: the case of a science gateway for JOCDM Viewer Software	Conference paper Conference paper Article Conference paper Article Conference paper Article Artide Artide Artide Artide Artide Artide Artide Artide Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper	induded induded induded induded induded induded induded induded induded induded induded induded induded induded induded	Proceda Engineering 2019 [EEE Global Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computers (2022) Lecture Notes in Computer Science EEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors EEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging Journal of Digital Imaging Journal of Digital Imaging Journal of Digital Imaging Medical Physics Journal of Digital Imaging 2019 Free and Open Source Software for Geospatian 2021 [EEE 1 thin Annual Controlling and Computer Engineering WSC '19 Ireaj Journal for Computer Science and Mathematics	
Scopus IEEE Xplore Springer Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus Scopus ACCM Digital Library Scopus	06,12,2024 05,12,2024 17,12,2024 08,12,2024 05,12,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 12,11,2024 05,12,2024 05,12,2024 05,12,2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Aizola; A. Skarab, A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandez Rodrigue; L. ILevero, R. Kilvinis Saptarbi Purkayasha, Surendra Babu Budi, Siddharba Nuthakki, Bhavana Yadav, Jady W. Gichoya Mrudang MathudWilliam D. MeadorMarcin MalinowskiTomasz Jazvete Tomasz, A. TimedManuel K. Rausch Bianch J.; Paniagua B; De Oliveira Ruellas A.C.; Filtion-Robin JC.; Pinetro D.; Conçalves J.R.; Hotod J.; Yadabe M; Stylmer W, Li T.F.; Gurgel M.L.; Chaves Junior C.M.; Massaro C; Garb D.G.; Witanova L; Henriques J.F.C.; Cevidense J., Janison G; Iwasaki L.R.; Nickel J.C.; Evangelista K; Lebre R; Pinho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatia M. Siniai S; Alimguzhin V; Manchi T; Tronci E. Valdermana C E; Stroux L; Katebi N; Palyug E; Hall-Clifford R; Rohloff P; Marzbannad F; Clifford G.D. Manddini M., Brunzin A.; Facco G; Mazzol A; Forcellese A; Gigante A. J. Tomar; S. Iskuras; J. L. Molinuevo; J. D. Gispert; F. Alpiste Hussain MA.; Langer S.G.; Kohl M. Valente F; Silva L.A.B.; Godinho T.M.; Costa C. Poludriowski G; Omar A; Bujlia R; Andreo P. Haak D; Page C.=; Desemo TM. Phiri Y.D.J; Munthali K.G. P. D. A. Mertadewi; W. Puspitasari; M. Saputra Maclag T; Douglas T; Rizwan Islam Rhythm S.M.; Kyabaggu R; Camillo C. A. Anastasia Anagonotou, Sirnon J. E. Taylor, Derek Groen, Dana Suleimenova, Nana Anokye, Riccardo Bruno, and Robetro Barbera Yasaen M.G; Md S.A; Addb S.I.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Radiology on Chest X- Rays. Texas ThValve 1.0: a reverse-engineered, open model of the human tricuspid valve 30 Silcer Craniomaxillolacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telementorine Service (Sargievani) Reconciling interoperability with efficient Verification and Validation within open source simulation and Degree Adapteous the Sargiest An open source autocorrelation-based method for fetal heart rate estimation from on-dimensional Dopper universe. Tools for Hip Surgical Multipurpose Virtue Reality Environment for Biomedical and Health Applications. Hultipurpose Virtue Reality Environment for Biomedical and Health Applications Anatomy of an Extensible Open Source PACS Technical Note: SpekPi v2.0—a software toolkit for modeling x-ray tube spectra Assurey of DCOM Viewer Software to Integrate Clinical Research and Medical Imaging Decisity Intertion Toxards a Multi-Modal Software System for Improving Home Care Experiences in Saskatchewan Building upose virtue Realt Environment for Building and simulation Chinal Andre Senet Angenet Andre Virtue States States and Medical Imaging Decisity Intertion Toxards a Multi-Modal Software System for Improving Home Care Experiences in Saskatchewan Building ubota Research capacity in public Health Services Using OpenEmprint Doxards a Multi-Modal Software System for Improving Home Care Experiences in Saskatchewan Building ubota Research expectity in public health Services Using Openizations	Conference paper Conference paper Article Conference paper Article Conference paper Article Article Article Article Article Article Article Article Article Article Article Article Conference paper Conference paper Conference paper Conference paper Article Conference paper Article Conference paper Article Conference paper Article Conference paper Article Conference paper Article Conference paper Article Conference paper Article Conferenc	Included included included included included included included included included included included included included included included included included	IFEE Addass Proceda Engineering Proceda Proceeda Proceda Proceda Proceeda Proceda Proceda Proceeda Proceda Proced	
Scopus IEEE Xplore Springer Soringer Scopus	06,12,2024 05,12,2024 17,12,2024 08,12,2024 05,12,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 13,11,2024 12,11,2024 12,11,2024 12,11,2024 12,11,2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Aizola; A. Skarab; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandez Rodrigue; L. E. Nevero, R. Kilvinis Saptarali Purkayasha, Surendra Babu Budi, Siddhartha Nuthalki, Bhavana Yadav, Jady W. Gichoya Mrudang Mathu/William D. Maador/Marcin MalinowskiTomasz Jazviete Tomasz, A. Timel/Manuel K. Rausch Bianchi J.; Paniagua B; De Oliveira Ruellas A.C.; Fillion-Robin JC.; Prietro J.C.; Gonçalwes J.R.; Hotoci J. Yataba M; Silymer W, Li T.F.; Gurgel M.L.; Chares Junior C.M.; Massaro C; Garib D.G.; Vilanova L; Henriques J.F.C.; Casillo A.A.O.; Janson G.; Iwasaki L.R.; Nickel J.C.; Evangelista K.; Cevidarez L., Janson G, Iwasaki L.R.; Nickel J.C.; Evangelista K.; Ovidarez L., Janson G, J. Wasaki L.R.; Nickel J.C.; Evangelista K.; Cevidarez L.; Barton J.; Janson G, Jiwasaki L.R.; Nickel J.C.; Evangelista K.; Ovidarez L.; Gardin D.A.; Janson G, Jiwasaki L.R.; Nickel J.C.; Evangelista K.; Ovidarez L.; Gardin J., Janson G, Jiwasaki L.R.; Nickel J.C.; Evangelista K.; Ovidarez L.; Gardin J., Janson G, Jiwasaki L.R.; Nickel J.C.; Evangelista K.; Ovidarez L.; Gilford G.D.; Waddali J.S.; Sood S.P.; Kaushish R.; Syed-Abdul S.; Khosla P.K.; Bhatla M. Sinisi S; Alimguzhin V; Mancini T.; Tronci E. Valdermana C.E.; Stroux L.; Katebi N.; Paljug E.; Hall-Clifford R.; Rohloff P.; Marzbanrad F.; Cliford G.D.; Manddini M.; Brunzini A.; Facco G, Mazcol A.; Forcellese A.; Gigante A. J. Torner; S. Skouras; J. L. Molinuevo; J. D. Gispert; F. Alpiste Hussain MA.; Langer S.G.; Kohl M. Valente F.; Silva L.A.B.; Godinho T.M.; Costa C. Poludriowski G.; Omar A.; Bujlia R.; Andreo P. Haak D.; Page CE; Desemor T.M. Phir Y.D.J.; Munthali K.G. P. D. A. Mertadewi; W. Puspitasari; M. Sapulra Maclag T; Douglas T; Rizwan Islam Rhythm S.M.; Kyabaggu R; Camillo C. A. Anastasia Anagnostou, Sirnon J. E. Taylor, Derek Groen, Diana Sudeimenova, Nana Anokye, Riccardo Binno, and Roberto Barbera Yaseem M.G.; Add S.A.; Adob I. Duarte L; Lobo M; Viana J; Freitas A,; Teodoro A.C. Zimmerman R.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Check X- Rays. Taxas Thvalve 1.0: a reverse-engineered, open model of the human tricuspid valve 3D Sticer Cranicomaxillolacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telementations Sovice (Sciapievani) Reconciling interoperability with efficient Verification and Validation within open source simulation environments An open source subcorrelation-based method for fetal heart rate estimation from on-dimensional Dopper util Web. Mathematics and PL FIRS Server and Its Use in Medical Imaging Multipupose Virtual Reality Environment for Biomedical and Health Applications Learning HL7 FIRI Using the HAPI FIRI Server and Its Use in Medical Imaging White SIM Dataset Anatomy of an Extensible Open Source PACS Tachnical Note: SpekP v2.0.—a software toolkit for modeling x-ray tube spectra Anatomy on a Extensible Open Source PACS Tachnical Research Activity in Public Health Services Using OpenERP with Quickstart Approach A Design Iteration Towards a Multi-Modal Software System for Improving Home Care Experiences in Saskatchware Multipupota Virtual Reality in public Health Services Using OpenERP with Quickstart Approach A Design Iteration Towards a Multi-Modal Software System for Improving Home Care Experiences in Saskatchware Model Software System for Improving Home Care Experiences in Saskatchware Model Software System for Improving Home Care Experiences in Saskatchware Model Software System for Improving Home Care Experiences in Saskatchware Robel Software System for Improving Home Care Experiences in Saskatchware model Impaging Building option Experts Approach Care Experiences and Saskation of prevention quality indicators (POIS) Fortholoce Seare application to Defrom the spatial di	Conference paper Conference paper Article Conference paper Article Conference paper Article Article Article Article Article Article Article Article Article Article Article Article Article Article Article Article Article Article Conference paper Conference paper Conference paper Conference paper Conference paper Article Conference paper Article Conference paper Article Conference paper Article Conference paper Article	included inc	IFEE Actorss Proceda Engineering 2018 IEEE Giobal Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computer Science IFEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors IEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging Zotaria Effect Software for Geospatial Zotaria Effect Computer Science and Mathematics GISTAM 2020 BMC Psychiatry	
Scopus IEEE Xplore Springer Soringer Scopus	06,12,2024           05,12,2024           17,12,2024           08,12,2024           05,12,2024           13,12,2024           13,11,2024           05,12,2024           06,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akzola; A. Skarejs, A. M. Kirks; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandez Rodrigue; L. El Nevero; R. Kilvinis Saptani Purkayasha, Surendra Babu Buddi, Siddhartha Nuthakki, Bhawana Yadav, Jady W. Gichoya Mrudarg Mathur/William D. Meador/Marcin MalinowskiTomasz Jazviete Tomasz, A. TimeMAnaule K. Rausch Bianch J.; Paniagua B; De Oliveira Ruellas A.C.; Fillion-Robin JC.; Prieto D.; Gonçaleve J.R.; Hotozi J.; Yataba M; Stymer W, Li T.F.; Gurgel M.L.; Chaves J.R.; Hotozi J.; Yataba M; Stymer W, Li T.F.; Gurgel M.L.; Chaves J.R.; Otozi J., Yataba M; Stymer W, Li T.F.; Gurgel M.L.; Chaves J.Niero C.M.; Massaro C; Garib D.G.; Vilanova L; Henriques J.F.C.; Castillo A.A. D, Janson G; Iwasski L.R.; Nickel J.C.; Evangelista K; Cevidanes L. Lebre R; Finho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatlia M. Sinisi S; Alimguzhin V; Mancini T; Tronci E. Valdermano C.E.; Stroux L; Katebi N: Palyug E; Hall-Clifford R; Rohloff P; Marzbenna G; Cilford G.D. Manddini M; Brunzini A; Facco G.; Mazzoli A; Forcellese A; Gigante A. J. Tomer; S. Skouras; J. L. Molinuevo; J. D. Gispert; F. Apiste Hussain MA; Langer S.G.; Kohl M. Valente F; Silva L.A.B; Godinho T.M.; Costa C. Poludniowski G; Omar A; Bujila R; Andreo P. Haak D; Page CE; Desemo T.M. Phir Y.D.J; Munthali K.G. P. D. A. Mertadewi; W. Puspitasari; M. Saputra Maclag T: Douglas T; Rizwan Islam Rhythm S.M.; Kyabaggu R; Camillo C. A. Anastaia Anagnostou, Jimon J. E. Taylor, Denk Groen, Dana Suleimenova, Nama Anokye, Riccardo Bunno, and Roberto Barbera Yaseen M.G; Abd S.A; Adob I. Duarte L; Lobo M; Viana J; Freitas A; Teodoro AC. Zirmermann R; Konjufca J; Sakejo P; Rilonzo M; Quevedo Y; Blum K; Biba E; Miosha T; Cottin X; Hennández C; Kaway S; Arenilu A; Behn A. Sirnon Fetersen NguyenPetter NielsenJohan Ivar Sabo	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Chest X- Rays. Taxas Thvalve 1.0 : a reverse-engineered, open model of the human tricuspid valve 30 Slicer Craniomaxillotacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICCM Viewers for the Indian National Telementations Source (asgineyavari) Reconciling interoperability with efficient Verification and Validation within open source simulation and Pomey Source Methods for Hip Surgical Planning Multipurpose Virtual Reality Environment for Biomedical Medical Imaging with the SIM Dataset Assurey of DICOM Viewer Software Tools for Hip Surgical Planning Wultipurpose Virtual Reality Environment for Biomedical and Health Applications Eacring HL7* Filt Using the HAPI FiltR Server and Its Use in Medical Imaging with the SIM Dataset Assurey of DICOM Viewer Software Tools for Hip Surgical Planning Congension Colles in MODELLING SPATIAL ACCESSIBILITY of PRIMARY HEALTH CARE in MALAWI Optimization Towards a Multi-Modal Software System for Improving Home Care Experiences in Saskatchewan Building global research capacity in public health: the case of a science gatewy for physical activity lifeting modeling and simulation Critical Raters affecting the adoption of open source software in public Organizations (Cliss) in MODELLING SPATIAL ACCESSIBILITY of PRIMARY HEALTH CARE in MALAWI Optimization Towards a Multi-Modal Software System for Improving Home Care Experiences in Saskatchewan Building global research capacity in public health: the case of a science gatewy for physical activity lifetion modeling and simulation Critical Raters affecting the adoption of open source software in public organizations The Role of Global Standardization Communities in Shaping National	Conference paper Article Conference paper Article Conference paper Article Artide Artide Artide Artide Artide Artide Artide Artide Artide Artide Artide Conference paper Conference paper Artide Artide Art	induded indude	Proceda Engineering 2018 [EEE Global Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computers (2022) Lecture Notes in Computer Science EEEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors EEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging Journal of Digital Imaging Journal of Digital Imaging Journal of Digital Imaging Medical Physics Journal of Digital Imaging 2019 From Journal Syst Rehabil Eng 2019 From Journal Conference on Electrical and Communication Workshop and Conference. ICCWC) EEEE Canadian Conference on Electrical and Computer Engineering WSC '19 Irraj Journal for Computer Science and Mathematics GISTAM 2020 BMC Psychiatry ICT4D 2017	
Scopus IEEE Xplore Springer Springer Scopus	06,12,2024 05,12,2024 17,12,2024 08,12,2024 13,11,2024 14,11,2024 14,11,2024 14,11,2024 14,11,2024 14,11,2024 14,11,2024 14,11,2024 14,11,2024 14,11,2024 14,11,2024 14,11,2024 14,11,2024 14,11,2024 14,11,2024 14,11,2024 14,11,2024 14,11,2024 14,11,2024 14,11,204 14,11,204 14,11,204 14,11,204 14,11,204 14,11,204 14,11,204 14,11,1	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akzola; A. Skarab, A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandez Rodrigue; L. Lervero, R. Kilvinis Saptanił Putsyastła, Surendra Babu Budi, Siddhartha Nuthakki, Blawana Yudav, Judy W. Glebya Mrudang MathudWilliam D. MeadorMarcin MalinowskTomasz. Jazvete Temasz, A. TimeMAnauek K. Rausch Bianch J.; Paniagua B; De Oliveira Ruelias A.C.; Filion-Robin JC.; Prietro D.C.; Conçaleva J.R.; Hotori J.; Yatabe M, Stymer, W. Lit T,F.; Gurgel M.L.; Crastila A. TimeMAnauek K. Rausch Bianch J.; Paniagua B; J. Bo Oliveira Ruelias A.C.; Filion-Robin JC.; Prietro D.C.; Conçaleva J.R.; Hotori J.; Yatabe M, Stymer, W. Lit T,F.; Gurgel M.L.; Crastila Animor C.M.; Messaro C; Garab D.S.; Vitanova L.; Henriques J.F.C.; Cevidenes L. Lebre R; Pinho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatia M. Siniai S; Alimguzhin V; Manchi T; Tronci E. Valdermana C E; Stroux L; Katebi N; Paliyu E; Hall-Clifford R; Rohloff P; Marzbanad F; Clifford G.D. Manddini M.: Bruzini A.; Facco G; Mazzol A; Forcellese A; Gigante A. J. Tomer, S; Skouras, J. L. Melnieuwo, J. D. Gispert; F. Alpiste Hussain MA; Langer S.G.; Kohli M. Valenter F; Silva L.A.B.; Godinho T.M.; Costa C. Poludniowski G; Omar A; Bujita R; Andrao P. Haak D; Page CE; Desemo TM. Phiri Y.D.J; Munthali K.G. P. D. A. Mertadewi; W. Puspitasari; M. Saputra Maclag T; Douglas T; Rizwan Islam Rhythm S.M.; Kyabaggu R; Camillo C. A. Anastasia Anagnostou, Simon J. E. Taylor, Derek Groen, Dana Suleimenova, Nana Anokye, Riccardo Bruno, and Roberto Barbera Yasaen M.G.; Kid S.A; Xdeb I. Duarte L; Lobo M; Viana J; Freitas A; Tacdoro A.C. Zimmermann R; Konju(Ga J; Sakejo P; Klonzo M; Oavendo Y; Blum K; Biba E; Mohad S.A; Addo J.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Radiology on Check X- Rays Taxas Thvalve 1.0: a reverse-engineered, open model of the human tricuspid valve 30 Siloer Craniomaxillolacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telementations Bevice (Stagleevani) Reconciling interoperability with efficient Verification and Validation within open source simulation and Degrad Meditare Stagleevani An open source subcorrelation-Based method for fetal heart rate estimation firms on-dimensional Dopper utwaters Constrained on three SD Begmentation Software Tools for Hip Surgical Multipupose Virtue Reality Environment for Biomedical and Health Applications Learning 4.7 FHIR Using the HAPI FHIR Server and Its Use in Medical Imaging with the SIM Dataset. Anatomy of an Extensible Open Source PACS Technical Note: SpekPy v2.0—a software toolkit for modeling x-ray tube spectra Assurey of DOCM Viewer Software to Integrate Clinical Research and Medical Imaging Ommon Coll Sein MODELLING SPATIAL ACCESSIBILITY of FRMMAY HEATT CARE in MULE-Modal Software System for Improving Home Care Experiences in Saskatchewan Building ubods refers th Approach Allos prose virtue and the Multi-Modal Software System for Improving Home Care Experiences in Saskatchewan Building ubods refers the Approach Care Straination of Open source software toolkit for modeling x-ray tube spectra Care Spreinces in Saskatchewan Building ubods refers the Approach Care Experiences in Saskatchewan Building ubods and Strainstein MULEModal Software System for Improving Home Care Experiences in Saskatchewan Building ubods and Strainstein Multi-Modal Software System for Improving Home Care Experiences in Saskatchewan Building ubods	Conference paper Conference paper Article Conference paper Article Conference paper Article Article Article Article Article Article Article Article Article Article Article Article Conference paper Conference paper Conference paper Article Conference paper Conference paper Conference paper Conference paper Article Conference paper Article Conference paper Article Conference paper Article Conference paper Article Conference paper	Included included included included included included included included included included included included included included included included included included included	IFEE Actorss Proceda Engineering Proceda Proceeda Proceda Proceeda Proceda Proceda Proceeda Proceda Proceda Proceeda Procee	
Scopus IEEE Xplore Springer Sorpus Scopus Sc	06.12.2024 05.12.2024 17.12.2024 05.12.2024 05.12.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 13.11.2024 05.12.2024 05.12.2024 05.12.2024 05.12.2024 05.12.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Aizola; A. Skarab; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandez Rodrigue; L. ILevero, R. Kilvinis Saptarili Putayasha, Surendra Babu Budi, Siddharba Nuthatki, Bhavana Yadav, Jady W. Gichoya Mrudang Mathu-William D. MaadorMarcin MalinowskiTomasz Jazvete Tomasz, A. TimelManuel K. Rausch Bianchi J.; Paniagua B; De Oliveira Ruellas A.C.; Filtion-Robin JC.; Prieto D.; Congalews J.R.; Hotozi J. Yataba M; Silymer W. Li T.F.; Gurgel M.L.; Chaese Junior C.M.; Massaro C; Garb D.G.; Wianova L; Henriques J.F.C.; Gasillo A.A. D, Janson G, Iwasaki L.R.; Nickel J.C.; Evangelista K.; Lebora R; Pinho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatlia M. Sinisi S; Alimguzhin V; Mancini T; Tronci E. Valderma C.E.; Stroux L; Katebi N; Paliya E; Hall-Clifford R; Rohloff P; Marzbanna E; Clifford G.D. Manddini M.; Brunzini A; Facco G; Mazzol A; Forcellese A; Gigante A. J. Tomer, S. Skouras; J. L. Molinuevo; J. D. Gispert; F. Alpiste Hussain MA; Langer S G; Kohl M. Valente F; Silva L.A.B.; Godinho T.M.; Costa C. Poludriowski G; Omar A; Bujlia R; Andreo P. Haak D; Page CE; Desemo TM. Philf Y.D.J; Munthali K G. P. D. A. Mertadewi; W. Puspitasari; M. Saputra Maadagi T; Douglas T; Rizwan Islam Rhythm S.M.; Kyabaggu, R; Camillo C. A. Anastasia Anagnostou, Simon J. E. Taylor, Derek Green, Diana Suleimenova, Nana Androye, Riccardo Buno, and Roberto Barbera Yaseen M, G; Abd S, Adeb I. Duarte L; Lobo M; Viana J; Freitas A; Teodoro A.C. Zimmermann R; Konjufca J; Stekip P; Kilorzo M; Cuevedo Y; Blum K; Biba E; Mosha T; Otiya Ca, Stekip P; Kilorzo M; Cuevedo Y; Blum K; Biba E; Mosha T; Colivita D; Freitas A; Teodoro A.C.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Radiology on Check X- Rays. Taxas ThValve 1.0: a reverse-engineered, open model of the human tricuspid valve 3D Silcer Craniomaxillolacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telementicine Sourcie (Sciencevani) Reconciling interoperability with efficient Verification and Validation within open source simulation environments An open source subcorrelation-based method for fetal heart rate estimation from on-dimensional Doppet utilescound Comparison of Three 3D Begimentation Software Tools for Hip Surgical Planning Multipupose Virtual Reality Environment for Biomedical and Health Applications Learning H.7. Fritl Using the HAPI FHIR Server and Its Use in Medical Imaging with the SIM Dataset Assurey of DICOM Viewer Software Tools for Hip Surgical Plannating Planting COM Viewer Software to Integrate Cinical Research and Medical Imaging POSS TECHNOLOGIES in MODELLING SPATIAL ACCESSIBILITY of PRIMARY HEALTH CARE in MALAWI OpenERP with Quarkstat Approach AD esign Intering Towards a MLAWI OpenERP with Quarkstat Approach AD esign Intering Comporting Assistant (MHRA)—an open-source software facilitating evidence-based assessment for clinical services Using adjota Standerdization Communities In Shapin National Health Information Architectures Community Health Information Reporting Assistant (MHRA)—an open-source software facilitating evidence-based assessment for clinical arkvices (PCI) Simple method basistander/ Community Health Information Reporting Assistant (MHRA)—an open-source software facilitating evidence-based assessment for clinical arkvices (PCI) Simple method based Standerdization C	Conference paper Conference paper Article Conference paper Article Conference paper Article Article Article Article Article Article Article Article Article Article Article Article Article Article Article Article Conference paper Conference paper Conference paper Article	Included Inc	Inter Actors Proceda Engineering 2018 IEEE Ginbal Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computer Science IEEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors IEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging Journal of Digital Imaging 2021 IEEE (Tans Neural Software for Geospatial 2021 IEEE (The Annual Computing and Computer Engineering WSC 19 IEEE Canadan Conference on Electrical and Computer Engineering WSC 19 IEEE Canadan Conference and Mathematics CISTAM 2020 BMC Psychiatry ICT4D 2017 IST-Africa 2022	
Scopus IEEE Xplore Springer Sorpus Scopus Sc	06,12,2024           05,12,2024           17,12,2024           08,12,2024           05,12,2024           13,11,2024           14,11,2024           15,12,2024           12,11,2024           12,11,2024           13,12,2024           13,12,2024           14,11,2024           15,12,2024           12	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Aizola; A. Skarab; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Fernandes Rodrigue; L. Ruevor, R. Kilvinis Saptarali Purkayasha, Surendra Babu Budki, Siddhartha Nuthakki, Bhawana Yadav, Jady W. Gichoya Mrudang Mathur/William D. Maador/Murcin MalinowskiTomasz Jazviete Tomasz, A. TimelManuel K. Rausch Bianchi J; Paniagua B; De Oliveira Ruellas A.C.; Fillion-Robin JC.; Prietro D.C.; Gonçalews J.R.; Hotoci J. Yataba M; Silymer W; Li T.F.; Gurgel M.L.; Craese J.J., Baron G; J. Massaki, L.R.; Nickel J.C.; Evangelista K.; Celvidmes L. Lebro R; Filmb E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatla M. Sinisi S; Alimguzhin V; Mancini T; Tronci E. Valdermana C.E.; Stroux L; Katebi N; Peljug E; Hall-Clifford R; Rohloff P; Marzbenna G; Cilford G.D. Mancini, J. D. Gispert; F. Alpiste Hussain MA; Langer S.G.; Kohl M. Valente F; Silva J.A.B.; Godinho T.M.; Costa C. Poludriowski G; Omar A; Bujlia R; Andreo P. Haak D; Page C-E; Desemor T.M. Phin Y.D.J; Munthali K.G. P. D. A. Mertadewi; W. Puspitasari; M. Saputra Maclag T; Douglas T; Rizwan Islam Rhythm S.M.; Kyabaggu R; Camillo C. A. Anastala Anagnostou, Simon J. E. Taylor, Derek Groen, Diana Suleimenova, Nana Anokye, Riccardo Bruno, and Roberto Barbera Yaseen M.G; Add S.J.; Yadob I. Duarte L; Lobo M; Viana J; Freitas A; Teodoro AC. Zimmerman R, Konjufca J; Skejo P; Kilozo M; Cuevedo Y; Blum K; Biba E; Mosha T; Cottin M; Hernandez C; Kanya S; Arenilu A; Behn A, Simon Petiersen NguyenPetter NielsenJona Iva	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Check X- Rays. Taxas Thvalve 1.0: a reverse-engineered, open model of the human tricuspid valve 3D Sticer Craniomaxillotacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telementations Socie (Sciangevani) Reconciling interoperability with efficient Verification and Validation within open source simulation environments An open source subcorrelation-based method for fetal heart rate estimation form on-dimensional Doppet utilizes out Comparison of Three 3D Segmentation Software Tools for Hip Surgical Planning Multipupose Virtual Reality Environment for Biomedical maging with the SIM Dataset Anatomy of an Extensible Open Source PACS Tachnical Note: SpekP yz 0.— a software toolkit for modeling x-ray tube spectra Anatomy of an Extensible Open Source PACS Tachnical Research addition Surves Software Tools for Hip Surgical Planning Multipupose Virtual Reality Environment for Biomedical and Medical Imaging with the SIM Dataset Anatomy of an Extensible Open Source PACS Tachnical Research ad Medical Imaging CoSS TECHNOLOGIES in MODELLING SPATIAL ACCESSIBILITY of PRIMARY HEALTH CARE in MALAWI OpenERP with Quickstart Approach A Design Iteration Towards a Multi-Modal Software Toxis of a science gateway for physical activity Hiptiong modeling and simulation Critical Research and Model Imaging AG Discopen source apports in public Comparization Information Reporting Assistant (MHRA)—en open source software ficilitating evidence-based assessment for clinical services The Rev of Colds Standerdoppond communities in Shaping National Health Information Apporting Assistant (MHRA)—en open source software ficilitating evidence-based assessment for clinical services The Rev of Colds Standerdoppond coc	Conference paper Conference paper Article Conference paper Article Conference paper Article Artide Artide Artide Artide Artide Artide Artide Artide Artide Artide Conference paper Conference paper Conference paper Artide Conference paper Artide Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper	induded induded induded induded induded induded induded induded induded induded induded induded induded induded induded induded induded induded	IFEE Actorss Proceda Engineering 2018 IEEE Giolal Humanitarian Technology 2018 IEEE Giolal Humanitarian Technology 2018 IEEE Giolal Humanitarian Technology Advances in Intelligent Systems and Computing Engineering with Computer Science IEEE Symposium on Computer Science IEEE Symposium on Computers and Communications Journal of Digital Imaging	
Scopus Springer Springer Springer Scopus Sco	06,12,2024 05,12,2024 17,12,2024 05,12,2024 13,11,2024 14,11,2024 12,11,2024 14,11,11,11,11,11,11,11,11,11,11,11,11,1	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Azola; A. Skarab, A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandez Rodrigue; L. Lervero, R. Kitonis Saptani Purkayasha, Surendra Babu Budi, Siddhurtha Nuthakki, Blawana Yudav, Jady W. Gichoya Mrudang Mathu/William D. Meador/Marcin Malinowsk/Tomasz Jazviete Temasz, A. Timed/Anauet K. Rausch Bianch J.; Panigua B; De Oliveira Ruelias A.C.; Filion-Robin JC.; Prietro D.C.; Conçaleve J.R. Hotor J. Yatabe M; Sylmer M; Li T.F.; Gurgel M.L.; O.C.; Conçaleve J.R.; Hotor J. Yatabe M; Sylmer M; Li T.F.; Gurgel M.L.; O.C.; Conçaleve J.R.; Hotor J. Yatabe M; Sylmer M; Li T.F.; Gurgel M.L.; O.C.; Conçaleve J.R.; Hotor J. Yatabe M; Sylmer M; Li T.F.; Gurgel M.L.; O.C.; Conçaleve J.R.; Hotor J. Yatabe M; Sylmer M; Li T.F.; Gurgel M.L.; Castillo AA. D; Janson G; Iwasaki L.R.; Nickel J.C.; Evangelista K; Cevidanes L. Lebre R; Pinho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatia M. Siniai S; Alimguzhin V; Mancini T; Tronci E. Valderman G E; Clifford G, Kanzoli A; Forcellese A; Gigante A. J. Tomer, S. Skouras, J. L. Molineveç J. D. Gispert; F. Alpiste Hussain MA; Langer S.G.; Kohli M. Valenter F; Siva L.A.B.; Godiriho T.M.; Costa C. Pubudrowald G: Omar A; Buglin R; Andreo P. Haak D; Page CE; Desemo T.M. Phiri Y.D.J; Munthali K.G. P. D. A. Mertadewi; W. Puspitasari; M. Saputra Maclag T; Douglas T; Rizwan Islam Rhythm S.M.; Kyabaggu R; Camillo C. A. Anatataia Anagnostou, Simon J. E. Taylor, Derek Groen, Diana Suleimenova, Nana Anokye, Riccardo Bruno, and Roberto Barbara Yasaen M.S; Ald S.A; Addob I. Duarte L; Lobo M; Viana J; Freitas A; Teodoro A.C. Zimmermann R; Konjifca J; Sakej P; Kilonzo M; Cuevedo Y; Blum K; Ziba E; Moda S A; Aidob I. Duarte L; Lobo M; Viana J; Freitas A; Teodoro A.C. Zimmermann R; Konjifca J; Sakej P; Kilonzo M; Cuevedo Y; Blum K; Siba E; Mosha S A; Addob I.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Rabiology on Chest X- Rays Taxas Thvalve 1.0 : a reverse-engineered, open model of the human tricuspid valve 3D Silone Craniomaxillolacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Dicoogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telementatione Service (Sargievant) Reconciling interopenability with efficient Verification and Validation within open source sinulations Service (Sargievant) Comparison of Three 3D Segmentation Software Tools for Hip Surgical Planning Multipurpose Virtual Reality Environment for Biomedical and Health Applications Learning HL7 FHIR Using the HAPI FHIR Server and Its Use in Medical Imaging FOSS TECHNOLOGIES in MODELLING SPATIAL ACCESSIBILITY of PRIMARY HEALT CARE in MALWH Optimization of Event Maragement Activity in Public Health Services Using CorpeERP with Clearkstore Tools Software System for Improving Home Care Experiences in Saskatchewan Eauling global Exercision of open source software in public orgenation of Levent Maragement Activity in Public Health Services A Design Iteration Towards a Multi-Modal Software System for Improving Home Care Experiences in Saskatchewan Eauling global Calculation Communities in Shaping National Health Information Reporting Assistant (MHIRA)—an open-source gatewary for physical activity Jilebing modeling and simulation orginations A Design Iteration Towards a Multi-Modal Software System for Improving Home Care Experiences in Saskatchewan Eauling global Design of open source software in public orginations Care Experiences in Saskatchewan Eauling global Design proteing Assistant (MHIRA)—an open-source gatewary for physical activity Jilebing modeling and simulation orginations Fire Fire Calculation Source Systems F	Conference paper Conference paper Article Conference paper Article Conference paper Article Artide Artide Artide Artide Artide Artide Artide Artide Artide Artide Conference paper Conference paper Artide Conference paper Artide Conference paper Conference paper Artide Conference C	included inc	Inter Actorss Proceda Engineering 2019 [EEE Global Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computer Science EEEE Symposium on Computers and Communications Journal of Digital Imaging JOH Free and Open Source Software for Geovential QCAL/COMPUTER Science on Electrical and Computer Engineering WSC '19 Fray Journal for Computer Science and Mathematics GISTAM 2020 BMC Psychiatry ICT4D 2017 IST-Africa 2022 Applied Network Science	
Scopus IEEE Xplore Springer Sopus Sco	06,12,2024           05,12,2024           17,12,2024           08,12,2024           05,12,2024           05,12,2024           13,11,2024           13,11,2024           13,11,2024           13,11,2024           13,11,2024           13,11,2024           13,11,2024           13,11,2024           13,11,2024           13,11,2024           13,11,2024           12,11,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           05,12,2024           08,12,2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akola; A. Skarab, A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandez Rodrigue; L. ILevero, R. Kilvinis Saptarbi Purkayasha, Surendra Babu Budi, Siddharba Nuthakki, Bhavana Yadav, Jady W. Gichoya Mrudang Mathu-William D. MeadorMarcin MalinowskiTomasz Jazvete Tomasz, A. TimedManuel K. Rausch Bianchi J.; Paniagua B; De Oliveira Ruellas A.C.; Filtion-Robin JC.; Prietro J.C.; Conçalves J.R.; Hotod J. Yatabe M; Stylmer W. Lit T,F.; Gurgel M.L.; Chares Junior C.M.; Measaro C; Gario D.G.; Wianova L; Henriques J.F.C.; Cevidense L. Cevidense L. J. Janison G; Iwasaki L.R. Nickel J.C.; Evangelista K; Cevidense L. Vadaii J.S.; Stood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatia M. Sinisi S; Alimguzhin V; Manchi T; Tronci E. Valderman C E; Stroux L; Katebi N; Palyug E; Hall-Clifford R; Rohloff P; Marzbanad F; Clifford G.D. Mandbini M.; Brunzini A.; Facco G; Mazzol A; Forceliese A; Gigante A. J. Tomer; S. Ikouras; J. L. Molinuevo; J. D. Gispert; F. Alpiste Hussain MA.; Langer S.G.; Kohl M. Valente F; Silva L.A.B.; Godinho T.M.; Costa C. Poludriowski G; Cimar A; Bujlia R; Androe P. Haak D; Page C.=; Desemo TM. Phiri Y.D.J.; Munthali K.G. P. D. A. Mertadewi; W. Puspitasari; M. Saputra Maclag T; Douglas T; Rizwan Islam Rhythm S.M.; Kyabaggu R; Camillo C. A. Anastasia Anagonotou, Simon J. E. Taylor, Derek Groen, Diana Suleimenova, Nana Anokye, Riccardo Bruno, and Robetto Barbera Yaseen M.G; Mod X.; Adeb I. Duarte L; Lido M; Viana J; Freitas A; Teodoro A.C. Zimmermann R; Konjufca J; Sakeji P; Kilonzo M; Quevedo Y; Blum K; Bia B; Mosha T; Cotlin M; Hernández C; Kaaya S; Aenilu A; Behn A Simon Petfaereen Nguer Pitter M: Sabou C. Kanjo; T. D. Manda; L. Byson; A. Twabi; K. G. Munthali A. E. SchweikertG, F. L'Hert, R. Deinert	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Radiology on Chest X- Rays. Taxas ThValve 1.0: a reverse-engineered, open model of the human tricuspid valve 30 Silcer Craniomaxillolacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telementions Service (Sargievan) Reconciling interoperability with efficient Verification and Validation within open source simulation environments An open source autocorrelation-based method for fetal heart rate estimation from on-dimensional Doppet univesoind Comparison of Three 3D Begmentation Software Tools for Hip Surgical Multipupop Virtual Reality Environment for Biomedical and Health Applications Learning HL, T-FHR Using the HAPI FHIR Server and Its Use in Medical Imaging with the SIM Ditaset Anatomy of an Extensible Open Source PACS Tachrical Note: SpekPy v2.0—a software toolkit for modeling x-ray tube spectra A Survey of DiCON Viewer Software Notiki for Improving Home Care Experiences in Saskatchewan Design Interior Towards a Multi-Modal Software System for Improving Home Care Experiences in Saskatchewan Earling evidence-based assessment for dimoting an simulation circlical factors a Becling the doption of open source Software in public organizations A GIS open source application to perform the spatial distribution of prevention quality indicators (PGIS) methode in Saskatchewan Comparison compared and Software System for Improving Home Care Experiences in Saskatchewan Care Experiences in Saskatchewan Design Interior Towards and Multi-Modal Software System for Improving Home Care Experiences in Saskatchewan Design Interior Towards and Software System for Improving Home Care Experiences in Saskatchewan	Conference paper Conference paper Article Conference paper Article Conference paper Article Article Article Article Article Article Article Article Article Article Article Article Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Article Conference paper Article Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Article	induded induded	IFEE Actorss Proceda Engineering Proceda Proceeda Proceda Proce	
Scopus IEEE Xplore Springer Sorpus Scopus Sc	06,12,2024           05,12,2024           17,12,2024           08,12,2024           05,12,2024           05,12,2024           13,11,2024           13	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Aizola; A. Skarab; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandez Rodrigue; L. Ruevor, R. Kilvinis Saptarali Purkayasha, Surendra Babu Budi, Siddhartha Nuthakki, Bhawana Yadav, Jady W. Gichoya Mrudang Mathu/William D. Maador/Marcin MalinowskiTomasz Jazviete Tomasz. A. TimedManuel K. Rausch Bianch J.; Paniagua B; De Oliveira Ruellas A.C.; Fillion-Robin JC.; Prieto D.C.; Gonçaleva J.R.; Hotozi J. Yataba M; Sighmer M; Li T.F.; Gurgel M.L.; Chares Junior C.M.; Messaro C; Garib D.G.; Vilanova L; Henriques J.F.C.; Casillo A.A. D.; Janson G.; Iwasaki L.R.; Nickel J.C.; Evangelista K.; Cewidarez J., Janson G.; Iwasaki L.R.; Nickel J.C.; Evangelista K.; Owidarez J., Janson G.; Masaki L.R.; Nickel J.C.; Evangelista K.; Cewidarez J., Janson G.; Masaki L.R.; Nickel J.C.; Evangelista K.; Owidarez J., Janson G.; Masaki L.R.; Nickel J.C.; Evangelista K.; Owidarez J., Janson G.; Masaki L.R.; Nickel J.C.; Evangelista K.; Owidarez J., Janson G.; Masaki L.R.; Nickel J.C.; Evangelista K.; Owidarez J., Elford G.D.; Waddai J.S.; Sood S.P.; Kaushish R.; Syed-Abdul S.; Khosla P.K.; Bhatia M. Sinisi S; Alimguzhin V; Mancini T.; Tronci E. Valdermana C.E.; Stroux L.; Katebi N.; Paljug E.; Hall-Clifford R.; Rohloff P.; Marzbenna G.; Cifford G.D.; Mandolini M.; Brunzini A.; Facco G.; Mazzol A.; Forcellese A.; Gigante A. J. Torner; S. Skouras; J. L. Molinuevo; J. D. Gispert; F. Alpiste Hussain MA.; Langer S.G.; Kohl M. Valente F.; Silva L.A.B.; Godinho T.M.; Costa C. Poludriowski G.; Omar A.; Bujlia R.; Andreo P. Haak D.; Page CE; Desemo T.M. Phiri Y.D.J.; Munthali K.G. P. D. A. Mertadewi; W. Puspitasari; M. Saputra Maclag T; Douglas T; Rizwan Islam Rhythm S.M.; Kyabaggu R; Camillo C. A. Anastasia Anagnostou, Sirnon J. E. Taylor, Derek Groen, Diana Sudeimenova, Nana Anokye, Riccardo Binnon, and Roberto Bathera Yaseem M.G.; Add S.A.; Adob J. Duarte L; Lobo M; Viana J; Freitas A.; Teodoro A.C. Zimmerman R.; Konjufca J.; Sakejo P; Kilorzo M; Ouzevedo Y; Blum K.;	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Check X- Rays. Taxas ThValve 1.0: a reverse-engineered, open model of the human tricuspid valve 3D Sticer Crainomaxillotacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Modical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telementione Source (Sarjeevan) Reconciling interopenality with efficient Verification and Validation within open source simulation environments An open source subcorrelation-based method for fetal heart rate estimation from one-dimensional Doppet utilescound Comparison of Three 3D Begmentation Software Tools for Hip Surgical Planning Multipupose Virtual Reality Environment for Biomedical and Health Applications Learning HL7 Firlt Using the HAPT FIRIR Server and Its Use in Medical Imaging with the SIM Dataset Anatomy of an Extensible Open Source PACS Technical Note: SpekPi v2 0.— a software toolkit for modeling x-ray tube spectra Assurey of Dictor Viewer Software to Integrate Clinical Research and Medical Imaging Point Store View Imagement Activity in Public Health Services Using OpenERP with Quickstart Approach A Design Iteration Towards a Multi-Modal Software System for Improving Home Care Experiences in Sastatchewon modeling and simulation Constructions and Sastation force in services of strate solution Building global research capacity in public health Services Using OpenERP with Quickstart Approach A Design Iteration Towards a Multi-Modal Software System for Improving Home Care Experiences in Sastatchewon Communities in Shaping National Health Hormation Architectures Community Of Pacidic in Practice. Successful Implementation of prevention quality indicators (PCI)S Simple method to Information Reporting Ass	Conference paper Conference paper Article Conference paper Article Conference paper Article Article Artide Artide Artide Artide Artide Artide Artide Artide Conference paper Conference paper Conference paper Conference paper Artide Conference paper Conference paper Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide	Included Include Include Include Include Include Include Include Include Include Include Include Inclu	IFEE Actorss Proceda Expineering 2018 IEEE Giolal Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computer Science IFEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors IEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging Journal of Digital Imaging 2019 Free and Open Source Software for Geospital 2021 IEEE Canadian Conference on Electrical and Computer Engineering WSC '19 Iraj Joaranal for Computer Science and Mathematics GISTAM 2020 BMC Psychiatry ICT4D 2017 IST-Africa 2022 Applied Network Science SN Computer Science PLoS Computer Science PLoS Computer Science	
Scopus IEEE Xplore Springer Sorpus Scopus Sc	06,12,2024           05,12,2024           17,12,2024           08,12,2024           05,12,2024           13,12,2024           13,11,2024           13	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Akzola; A. Skarab; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Fernandes Rodrigue; L. El Nevero, R. Kilvinis Saptani Purkayasha, Surendra Babu Buddi, Siddhartha Nuthakki, Bhawana Yadav, Jady W. Gichoya Mrudang Mathur/William D. Meador/Marcin MalinowskiTomasz Jazviete Tomasz. A. TimeMAnaule K. Rausch Bianch J.; Paniagua B; De Oliveira Ruellas A.C.; Fillion-Robin JC.; Prieto D.; Gonçaleve J.R.; Hotozi J.; Yataba M; Stymer W; Li T.F.; Gurgel M.L.; Chaves J.R.; Hotozi J.; Yataba M; Stymer W; Li T.F.; Gurgel M.L.; Chaves J.Niero C.M.; Massaro C; Garib D.G.; Vilanova L; Henriques J.F.C.; Castillo A.A. D; Janson G; Iwasski L.R.; Nickel J.C.; Evangelista K; Cevidanes L. Lebre R; Finho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatlia M. Sinisi S; Alimguzhin V; Mancini T; Tronci E. Valdermano C.E.; Stroux L; Katebi N; Paljug E; Hall-Clifford R; Rohloff P; Marzbanna G; Cifford G.D. Manddini M, Brunzini A; Facco G; Mazzol A; Forcellese A; Gigante A. J. Tomer, S. Skouras; J. L. Molinuevo; J. D. Gispert; F. Apiste Hussain MA; Langer S.G.; Kohl M. Valente F; Shva L.A.B; Godinto T.M.; Costa C. Poludniowski G; Omar A; Bujila R; Andreo P. Haak D; Page CE; Desemor T.M. Phiri Y.D.J; Munthali K.G. P. D. A. Mertadewi; W. Puspitasari; M. Saputra Maclag T; Douglas T; Rizwan Islam Rhythm S.M.; Kyabaggu R; Camillo C. A. Anastalia Anagnostou, Simon J. E. Taylor, Derek Groen, Dana Suleimenova, Nama Anoky, Riccardo Bunno, and Roberto Barbera Yaseen M.G; Abd S.A; Adeb I. Duarte L; Lebo M; Vian J; Fretas A; Teodoro AC. Zimmermann R; Konjufca J; Sakejo P; Kilonzo M; Cuavedo Y; Blum K; Bia E; Misha T; Cottim J: Hennandez C; Kasaya S; Areniu A; Behn A, Smon Petresen NguyenPetter NielsenJonan Ivar Sabo C. Kanjo; T. D. Manda; L. Byson; A. Twab; K. G. Munthali A. E. SchweikertG, F. L'HerM. R. Deinert A. Ravishankar RaoSubrata GaraiSoumyabrata DayHang Peng Ajett-Bullock J; Causta J; Baten A; Beanen A; Hasmik K; Hamar K, B; Husha	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Check X- Rays. Texas ThValve 1.0 : a reverse-engineered, open model of the human tricuspid valve 3D Sticer Craniomaxillotacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Evaluation of Free, Open-source, Web-based DICCM Viewers for the Indian National Telement of Science	Conference paper Article Conference paper Article Conference paper Article Artide Artide Artide Artide Artide Artide Artide Artide Artide Artide Artide Artide Conference paper Conference paper Artide Conference paper Artide Conference paper Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Conference Con	induded indude	Inter Actorss Proceda Engineering 2018 IEEE Global Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computer Science IEEE Symposium on Computer Science IEEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors IEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging Journal of Computer Soltware for Computer Engineering WSC '19 Imaj Journal for Computer Science and Mathematics GISTAM 2020 BMC Psychiatry ICT4D 2017 IST-Africa 2022 Applied Network Science PL oS Computer Science PL oS Computer Science	
Scopus IEEE Xplore Springer Scopus	06.12.2024           05.12.2024           17.12.2024           08.12.2024           05.12.2024           13.11.2024           13.11.2024           13.11.2024           13.11.2024           13.11.2024           13.11.2024           13.11.2024           13.11.2024           13.11.2024           13.11.2024           13.11.2024           13.11.2024           13.11.2024           13.11.2024           13.11.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           06.12.2024           07.12.2024           12.11.2024           12.11.2024           12.11.2024	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Azola; A. Skarab, A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandez Rodrigue; L. Lervero, R. Kitonia Saptanii Purkayasha, Surendra Babu Budi, Siddharba Nuthakki, Blawana Yudav, Judy W. Gickeya Mrudang MathudWilliam D. MeadorMarcin MalinowskTomasz. Jazvete Temasz, A. Trienkfanzule K. Rausch Bianch J.; Paniagua B; De Oliveira Ruellas A.C.; Filton-Robin JC.; Prietro D.C.; Conçaleva J.R.; Hotot J. Y. Yatabe M, Stymer M, Li T.F.; Gurgel M.L.; Chaese Junior C.M.; Messaro C, Garla D.S.; Vitanova L; Henriques J.F.C.; Cevidenes L., Janeson G; Iweaski L.R.; Nickel J.C.; Evangelista K.; Lebre R; Pinho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatia M. Siniai S; Alimguzhin V; Manchi T; Tronci E. Valderman C E; Stroux L; Katebi N; Palyg E; Hall-Clifford R; Rohloff P; Marzbanad F; Clifford G.D. Manddini M.: Brunzin A.; Facco G; Mazzol A; Forcellese A; Gigante A. J. Tomer, S; Skouras; J. L. Molinuevo, J. D. Gispert; F. Alpiste Hussain MA; Langer S.G.; Kohl M. Valenne F; Silva LAB; Godinho T.M.; Costa C. Poludniowski G; Omar A; Bujia R; Andreo P. Haak D; Page CE; Desemo T.M. Phiri Y.D.J; Munthali K.G. P. D. A. Mertadewi; W. Puspitasari; M. Saputra Maclag T: Douglas T; Rizowan Islam Rhythm S.M.; Kyabaggu R; Camillo C. A. Anastasia Anagonotou, Simon J. E. Taylor, Derek Groen, Dana Suleimenova, Nana Anokye, Riccardo Bruno, and Roberto Barbera Yaseen M.; Konjufca J; Sakejo P; Kilonzo M; Cusevedo Y; Blum K; Bha E; Lobo M; Viana J; Freitas A; Teodoro A.C. Zimmermann R; Konjufca J; Sakejo P; Kilonzo M; Cusevedo Y; Blum K; Bha E; Mosha T; Cottin M, Hernandez C; Kaaya S; Arennila A; Behn A Simon Pettanew Nguyen Pitter M: Saebow C. Kanjo; T. D. Manda; L. Byson; A. Twebi; K. G. Munthali A. E. SchweikertG, F. L'HerM. R. Deinert A. Ravishanikar RaoSubrata GaralSournyabrata DeyHang Peng Aylett-Bulock J; Custosh L, Zunosh A; Kange L, Sez J; Ahumada D; Moreira F. Steban Parra, Sonia Haiduo, and Rebecca James	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Radiology on Chest X- Rays. Taxas ThValve 1.0: a reverse-engineered, open model of the human tricuspid valve 30 Siloer Craniomaxillolacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telementions Berrice (Sargievani) Reconciling interoperability with efficient Verification and Validation within open source simulations action and Sargievanian An open source subcorrelation-Based method for fetal heart rate estimation from or-dimensional Doppet utraceurd Comparison of Three 3D Begmentation Software Tools for Hip Surgical Multipurpose Yintu Reality Environment for Biomedical and Health Applications Learning 4.7 FHIR Using the HAPI FHIR Server and Its Use in Medical Imaging with the SIM Delated. Anatomy of an Extensible Open Source PACS Technical Note: SpekFy v2.0—a software toolkit for modeling x-ray tube spectra Assurve of Incidention Approach Automy of an Extensible Open Source PACS Technical Note: SpekFy v2.0—a software toolkit for modeling x-ray tube spectra Assurvey of IDCOM Viewer Software to Integrate Clinical Research and Medical Imaging Comparison in Saskatchevan Edicing todo and Statchevan Delation of Sevent Management Activity in Public Health Services Using OpenERP with Quickstant Approach Care Experiences in Saskatchevan Single for the Surgical Community of Practice (PGIS) Multi-More Stetem Software toolkit for Improving Home Care Experiences in Saskatchevan Edicing todo approving Assistant (MHRA)—an open-source antimative of Information Reporting Assistant (MHRA)—an open-source antimative of Information Reporting Assistant (MHRA)—an open-source antimative of Information Reporting Assistant (MHRA)—an open-source antima	Conference paper Article Conference paper Article Conference paper Article Conference paper Article Article Article Article Article Article Article Article Article Article Article Article Article Article Conference paper Conference paper Conference paper Conference paper Conference paper Article Article Article Article Conference paper Article Conference paper Article Conference paper Article Article Conference Paper Conference Paper Article Conference Paper Conference Paper Conference Paper Conference Paper Article	Included Inc	Inter Actors Proceda Engineering 2018 IEEE Global Humanitarian Technology Corference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computer Science IEEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors IEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging Journal of Digital Imaging 2019 Free and Open Source Software for Geospaial 2021 IEEE Thin Annual Computing and Computer Engineering WSC '19 Imagi Journal of Computer Science and Mathematics GISTAM 2020 BMC Psychiatry ICT4D 2017 IST-Africa 2022 Applied Network Science SN Computer Science on Schware Engineering Companion (ICSE '16) Syringer Science and Biology 38th International Biology 38th International Biology	
Scopus IEEE Xplore Springer Sorpus Scopus Sc	06.12.2024           05.12.2024           17.12.2024           08.12.2024           05.12.2024           05.12.2024           05.12.2024           13.11.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05.12.2024           05	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Aizola; A. Skarab; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandez Rodrigue; L. ILevero, R. Kilvinis Saptarali Purkayasha, Surendra Babu Budi, Siddhartha Nuthalki, Bhavana Yadav, Jady W. Gichoya Mrudang Mathu-William D. MaadorMarcin MalinowskiTomasz Jazvete Tomasz. A. TimedManuel K. Rausch Bianchi J.; Paniagua B; De Oliveira Ruellas A.C.; Filtion-Robin JC.; Pietro D.C.; Gongalews J.R.; Hotod: J. Yataba M; Silymer W. Li T.F.; Gurgel M.L.; Chaese J.R.; Hotod: J. Yataba M; Silymer W. Li T.F.; Gurgel M.L.; Chaese J.Janes C.M.; Massaro C; Garb D.G.; Wianova L; Henriques J.F.C.; Gasillo A.A. D., Janson G, Hwasski L.R.; Nickel J.C.; Evangelista K.; Lebre R; Pinho E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatia M. Sinisi S; Alimguzhin V; Mancini T; Tronci E. Valdermas CE; Stroux L; Katebi N; Palyag E; Hall-Clifford R; Rohloff P; Marzbannad F; Clifford G.D. Manddhini M.; Brunzini A; Facco G; Mazzoli A; Forcellese A; Gigante A. J. Tomer; S. Skouras; J. L. Molinuevo; J. D. Gispert; F. Alpiste Hussain MA; Langer S G; Kohl M. Valente F; Silva L.A.B.; Godinho T.M.; Costa C. Poludriowski G; Omar A; Bujlia R; Andreo P. Haak D; Page C.=k; Desemo TM. Philf Y.D.J; Munthali K G. P. D. A. Mertadewi; W. Puspitasari; M. Saputra Maatag T; Douglas T; Rizwan Islam Rhythm S.M.; Kyabaggu, R; Camillo C, A. Anastasia Anagnonotou, Sirono J. E. Taylor, Demk Groen, Dana Suleimenova, Nana Andrye, Riccardo Buno, and Roberto Barbera Yaseen M; Cabdo J.; Yanaba J; Freitas A; Teodoro A.C. Zimmermann R; Konjufca J; Stekip P; Kilorzo M; Quevedo Y; Blum K; Biba E; Mosha T; Cuthy J; Freitas A; Teodoro A.C. Zimmermann A; Konjufca J; Stekip P; Kilorzo M; Quevedo Y; Blum K; Biba E; Mosha T; Oth M; Hernándoz C; Kaaya S; Arenilu A; Behn A, Simon Pettersen Nyuer/Pettr M: Hernándoz C; Kaaya S; Arenilu A; Behn A, Simon Pettersen Nyuer/Pettr M; Hernándoz C; Kaaya S; Arenilu A; Behn A, Simon Pettersen Nyuer/Nett M; Hernándoz C; Kaaya S; Ak	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Radiology on Chest X- Rays. Taxas ThValve 1.0: a reverse-engineered, open model of the human tricuspid valve 3D Silcer Craniomaxillolacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telementicine Service (Sciapievani) Reconciling interoperability with efficient Verification and Validation within open source simulation environments An open source autocorrelation-based method for fetal heart rate estimation from on-dimensional Doppet utilescound Comparison of Three 3D Begimentation Software Tools for Hip Surgical Phaning Multipupose Virtual Reality Environment for Biomedical and Health Applications Learning H.7 Firlt Using the HAPI FHIR Server and Its Use in Medical Imaging with the SIM Dataset Anatomy of an Extensible Open Source PACS Tachrical Note: SpekiP v2.0—a software toolkit for modeling x-ray tube spectra Source Simulation Colles in MODELLING SPATIAL ACCESSIBILITY of PRIMARY HEALTH CARE in MALAWI OpenERP with Qualcistat Approach AD esign Instant Approach AD esign Instant Approach AD esign Instant Communities in Spatial distribution of prevention gateway for physical activity lifelong modelling and simulation ortical aforts and Sector Share Pacifics. Successful Implementation Grant Experimences in Sastatchewan Building global Standerdization Communities In Sharejin Attion of FreeNouce applementation Approxies. A Surger JO Dictability Teichee States for the Individent AGIS open source application to perform the spatial distribution of prevention gateway for physical activity lifelong modelling and simulation critical aforts aneach capacit on Communities In Sharejin National Health Infor	Conference paper Article Conference paper Article Conference paper Article Conference paper Article Artide Artide Artide Artide Artide Artide Artide Artide Artide Artide Conference paper Conference paper Conference paper Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Artide Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Conference C	Included Inc	Inter Actorss Proceda Engineering 2018 IEEE Ginbal Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers (2022) Lecture Notes in Computer Science IEEE Symposium on Computers and Communications Journal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors IEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging Journal of Digital Imaging Medical Physics Journal of Digital Imaging 2019 Free and Open Source Software for Geospatial 2021 IEEE Tith Annual Computing and Computer Engineering WSC 19 IEEE Canadian Conference on Electrical and Computer Engineering WSC 19 IEEE Canadian Conference on Software STAM 2020 BMC Psychiatry ICT4D 2017 IST-Africa 2022 Applied Network Science SN Computer Science on Software PLoS Computer Science on Software	
Scopus IEEE Xplore Springer Sorpus Scopus Sc	06,12,2024           05,12,2024           17,12,2024           05,12,2024           05,12,2024           13,11,2024           14,11,2024           12,11,2024           12,11,2024           13,12,2024           13,12,2024           13,12,2024           13,12,2024           13,12,2024           13,12,2024           13,12,2024           13,12,2024           13,12,2024           13,12,2024           13,12,2024           13,12,2024           13,12,2024           13,12,2024           14	Chang Y.H.; Foley P; Azimi V; Borkar R; Lefman J. J. Ruiz-Aizola; A. Skarab; A. M. Idriss; B. Diao; J. Da Rosa De Pina Ferreira; M. A. Ferrandez Rodrigue; L. Revero; R. Kilvinis Saptarali Purkayasha, Surendra Babu Budki, Sidflartha Nuthakki, Blawana Yadav, Jady W. Gichoya Mrudang Mathu/William D. Maador/Murcin MalinowskiTomasz JazvieteTomasz, A. TimeMAnauek K. Rausch Bianchi J.; Paniagua B; De Oliveira Ruellas A.C.; Fillion-Robin JC.; Prieto D.C.; Gonçalews J.R.; Hotoci J. Yataba M; Silymer W; Li T.F.; Gurgel M.L.; Craese J.R.; Hotoci J., Yataba M; Silymer W; Li T.F.; Gurgel M.L.; Craese J.J.; Conc, J. Janson G.; Iwasaki L.R.; Nickel J.C.; Evangelista K.; Cevidines L. Lebro R; Filmb E; Silva J.M.; Costa C. Wadali J.S.; Sood S.P.; Kaushish R; Syed-Abdul S; Khosla P.K.; Bhatla M. Sinisi S; Alimguzhin V; Mancini T; Tronci E. Valdermana C.E.; Stroux L.; Katebi N; Paljug E; Hall-Clifford R; Rohloff P; Marzbanna G; Cifford G.D. Manddhin M.; Brunzin A; Facco G; Mazzol A; Forcellese A; Gigante A. J. Tomer, S. Skouras; J. L. Molinuevo; J. D. Gispert; F. Alpiste Hussain MA; Langer S.G.; Kohl M. Valente F; Silva L.A.B; Godinho T.M.; Costa C. Poludriowski G; Omar A; Bujlia R; Andreo P. Haak D; Page C-E; Desenor T.M. Phin Y.D.J; Munthali K.G. P. D. A. Mertadewi; W. Puspitasari; M. Saputra Maclag T; Douglas T; Rizwan Islam Rhythm S.M.; Kyabaggu R; Camillo C. A. Anastala Anagnostou, Simon J. E. Taylor, Derek Groen, Diana Suleimenova, Nana Anokye, Riccardo Bruno, and Roberto Barbera Yasseen M.G; Add S.J.; Yadob J. Duarte L; Lobo M; Viana J; Freitas A; Teodoro AC. Zimmermann R; Kourfuca J; Stekip P; Kilonzo M; Quevedo Y; Blum K; Biba E; Mosha T; Cotim A; Bayon; A. Twabi; K. G. Munthali A. E. SchweikertG, F, L'HerfM. R. Deinert A. Ravishankar RaoSubrata GaraSuoumysbrata DeyHang Peng Aylert-Bulcok J; Loustan Garao; C.; Quera-Boftani A; Katta A, Pham K.H.; Horover B, Strobelt H; Jimenaz R.M.; Sedgewick A; Evene E.S.; Kennedy D; Hartass S; Maina A, G.K.; Hussien A, Luongo-Oroz M.	Primer for image informatics in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Evaluating the Implementation of Deep Learning in LibreHealth Ratiology on Check X- Rays. Taxas ThValve 1.0: a reverse-engineered, open model of the human tricuspid valve 3D Sticer Craniomaxillotacial Modules Support Patient-Specific Decision-Making for Personalized Healthcare in Dental Research Evaluation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telementations Socie (Sciapevani) Reconciling interoperability with efficient Verification and Validation within open source simulation of Dree, Open-source, Web-based DICOM Viewers for the Indian National Telementions Socie (Sciapevani) Reconciling interoperability with efficient Verification and Validation within open source simulation and Dorget with vasound Comparison of Three 3D Segmentation Software Tools for Hip Surgical Planning Multipupose Virtual Reality Environment for Biomedical and Health Applications Learning HL7 Filth Using the HAPT FIHR Server and Its Use in Medical Imaging with the SIM Dataset Assurey of DICOM Viewer Software Tools for Hip Surgical Planning Surget Plant Reality Environment for Biomedical and Health Applications Surger Spruce Software Tools for Hip Surgical Planning Multipupose Virtual Reality Environment for Biomedical and Medical Imaging with the SIM Dataset Assurey of DICOM Viewer Software to Integrate Clinical Research and Medical Imaging COSS TECHNOLOGIES in MODELLING SPATILA.ACCESSIBILITY of PRIMARY HEALTH CARE in MALAWI Optimization of Event Management Activity in Public Health Services Using OpenERP with Quickstart Approach A GB open source application to perform the spatial distribution of prevention Building global research capacity in public health: the case of a science gateway for Plands StandardUstation Communities in Shaping National Health Information Reporting Assistant (MHIRA)—an open-source software ficilitating evidence-based assessement for clinical servi	Conference paper Article Conference paper Article Conference paper Article Conference paper Article Artide Artide Artide Artide Artide Artide Artide Artide Artide Artide Conference paper Conference paper Conference paper Article Conference paper Conference paper Conference paper Conference paper Conference paper Article Conference paper Conference paper Article Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Article Article Article Article Conference paper Conference paper Conference paper Conference paper Conference paper Conference paper Article Article Article Article Article Article Conference paper	induded indude	Inter Actorss Proceda Engineering 2018 IEEE Giolal Humanitarian Technology Conference (GHTC) Advances in Intelligent Systems and Computing Engineering with Computers Science IEEE Symposium on Computers and Communications Iournal of Digital Imaging Simulation Modelling Practice and Theory Physiological Measurement Sensors IEEE Trans Neural Syst Rehabil Eng Journal of Digital Imaging Medical Physics Journal of Digital Imaging Journal of Computer Science and Mathematics IGSTAM 2020 Journal For Computer Science Journal Computer Science and Business Media Deutschland ICEE	

Springer	07.12.2024	Emanuele Lindo SeccoDaniel McHughDavid ReidAtulya Kumar Nagar	Development of an Intuitive EMG Interface for Multi-dexterous Robotic Hand	Conference paper	induded	International Conference on Wireless Mobile Communication and Healthcare	2020
Springer	08 12 2024	Vasiliy LeonenkoAlexander LobachevGeorgiy Bobashev	Spatial Modeling of Influenza Outbreaks in Saint Petersburg Using Synthetic Populations	Conference paper	included	ICCS 2019	2019
IEEE Voloro	05 12 2024	T. Haidegger	Surgical robots of the next decade: New trends and paradigms in the 21th	Conference	included	2017 IEEE 30th Noumann Colloquium (NC)	2017
	12 11 2024	A. B. Clark; V. Mathivannan; N. Rojas	A Continuum Manipulator for Open-Source Surgical Robotics Research and	Article	included	IEEE Transactions on Medical Robotics and	2021
IEEE Xplore	05 12 2024	YH. Su; A. Munawar; A. Deguet; A. Lewis; K. Lindgren; Y. Li; R. H. Taylor; G. S. Fischer, B. Hannaford: P. Kazanzides	Collaborative Robotics Toolkit (CRTK): Open Software Framework for Surgical Robotics Research	Conference	included	IEEE International Conference on Robotic	2020
ACM Digital	00.12.2024		Let your body move: a prototyping toolkit for wearable force feedback with	Conference paper	included	18th International Conference on Human- Computer Interaction with Mobile Devices and	
Library	11.11.2024	Ricardo Cell-PárragaFélix Fernández-PeñaDiana Coello-Fiallos	electrical muscle stimulation Development of a Web-Based Medical Record Management Software for	Conference	induded	Services (MobileHCI '16)	2016
Springer	07.12.2024	Poba-Nzaou P.; Kume N.; Kobavashi S.	Digital Immigrant Users Governance and sustainability of an open source electronic health record: An	paper Conference	included	ICAT 2020 Proceedings	2019
Scopus	06.12.2024	D Davis: X Qu: X Yuan	interpretive case study of opendolphin in Japan Evaluation of WorldVistA and OpenEMR for PCMH care	paper Conference	included	MEDINFO 2019	2017
IEEE Xplore	05.12.2024		A modern loan dealeument of EHD sustame: Two lossons from magningful use	paper	included	SoutheastCon 2017	2016
IEEE Xplore	30.11.2024	A. Houde, D. 2100	in the United States	paper	induded	ICIEV	2010
Scopus	05.12.2024	Wickramasinghe R.; Weerabaddana C.	Sri Lankan Citizens	paper	included	(NITC)	2019
		Vasyl MartsenyukVladyslav PovoroznyukAndriy SemenetsLarysa Martynyuk	On an Approach of the Solution of Machine Learning Problems Integrated with Data from the Open-Source System of Electronic Medical Records: Application	Conference paper			2019
Springer	08.12.2024	Yang CY.: Liu CT.	for Fractures Prediction Performance assessment and tuning for exchange of clinical documents cross	Article	included	ICAISC 2019	2016
Scopus	16.11.2024	Van Biegen W.: Van Cauwenherre D.: Decruivenaere 1.: Leune T.: Stercky S.	healthcare enterprises	Article	included	Computer Standards and Interfaces	2022
Sconus	12 11 2024		implementation of computerized clinical decision support systems using a Qsort annroach	Autor	induded	BMC Medical Informatics and Decision Making	LULL
Casava	00 40 0004	Farhadi M.; Haddad H.; Shahriar H.	Static Analysis of HIPPA Security Requirements in Electronic Health Record	Conference	in aluded	International Computer Software and Applications	2018
Scopus	12.11.2024	Dzwiniel P.; Gola M.; Wójcik-Gryciuk A.; Waleszczyk W.J.	Applications Specvis: Free and open-source software for visual field examination	Article	included	PLoS ONE	2017
Sconus	05 12 2024	Farhadi M.; Haddad H.; Shahriar H.	Compliance checking of open source EHR applications for HIPAA and ONC security and privacy requirements.	Conference	included	IEEE Computer Software and Applications Conference (COMPSAC)	2019
Scopus	01.07.2024	McConnell M.; Schwerin B.; So S.; Richards B.	RR-APET - Heart rate variability analysis software	Article	included	Computer Methods and Programs in Biomedicine	2020
Scopus	14.11.2024	Shaikh M.; Vayani A.H.M.; Akram S.; Qamar N.	Open-source electronic health record systems: A systematic review of most recent advances	Article	included	Health Informatics Journal	2022
Scopus	13 11 2024	Asare E.; Wang L.; Fang X.	Conformance Checking: Workflow of Hospitals and Workflow of Open-Source	Article	included	IEEE Aman	2020
Coopus	15 11 2024	Bloomfield R.A., Jr; Polo-Wood F.; Mandel J.C.; Mandl K.D.	Opening the Duke electronic health record to apps: Implementing SMART on	Article	included	International Journal of Medical Information	2017
Scopus	13.11.2024	Walonoski J.; Kramer M.; Nichols J.; Quina A.; Moesel C.; Hall D.; Duffett C.;	Synthea: An approach, method, and software mechanism for generating	Article	included	Journal of the American Medical Informatics	2018
ACM Digital	17.11.2024	Dube K.; Gallagher I.; McLachlan S.	synthetic patients and the synthetic electronic health care record	Conference paper	included	Association 10th International Conference on Information and Communication Technologies and	
Library ACM Digital	08.11.2024	Sunyoung Kim, Theresa Allen, and Damini Bhatt	Creating an EMR system to support mobile clinics in Haiti	Conference	included	Development (ICTD '19) IEEE/ACM 1st International Workshop on	2019
Library	17.11.2024	Bell Raj Eapen, Norm Archer, Kamran Sartipi, and Yufei Yuan Scagliarini M.; Boccaforno N.; Donati R.; Vandi M.; Ponti E.; Nanni S.	Drishti: a sense-plan-act extension to open mHealth framework using FHIR Process Control Charts in Falls Prevention: The Experience of the Local	paper Article	included	Software Engineering for Healthcare (SEH)	2019 2020
Scopus	17.11.2024	Rivard L.; Lehoux P.; Alami H.	Healthcare Authority of Romagna, Italy "it's not just hacking for the sake of it": A qualitative study of health innovators'	Article	induded	Springer Health and Technology	2021
Scopus	17.11.2024		views on patient-driven open innovations, quality and safety	Conference	induded	BMJ Quality and Safety LOPAL '18: Proceedings of the International	
ACM Digital Library	29.11.2024	Fadratul Hafinaz Hassan, Au Yong Kah Wye, Sharifah Syafiqah Syed Yusof, and Teh Yi Xiang	Predicting Risk of Getting Smoking-Related Cancer: A Comparison of Three Prediction Models	paper	included	Conference on Learning and Optimization Algorithms: Theory and Applications	2018
IEEE Xplore	05.12.2024	V. Chaniotakis; L. Koumakis; H. Kondylakis; G. Notas; D. Plexousakis; M. Tsiknakis	Predictive Analytics Based on Open Source Technologies for Acute Respiratory Distress Syndrome	Conference paper	included	IEEE CBMS	2021
ACM Digital	00.44.0004	Miles I Device Debu D Counts and Dull D Medice	European de la la companya de	Conference		1st International Conference on Advanced	0000
ACM Digital	09.11.2024	Wyrene J. Domingo, Booby D. Gerardo, and Ruji P. Medina	Fuzzy decision tree for breast cancer prediction	Conference	included	2023 ACM Conference on Fairness,	2020
Library	10.11.2024	Yuxin Xiao, Shulammite Lim, Tom Joseph Pollard, and Marzyeh Ghassemi Jagreet KaurKulwinder Singh Mann	In the Name of Fairness: Assessing the Bias in Clinical Record De-identification Al Based HealthCare Platform for Real Time. Predictive and Prescriptive	paper Conference	included	Accountability, and Transparency (FAccT '23)	2023
Springer	06.12.2024	Pagemerite L. Coffithe W.: Allegen L. Europe W.: Distinghouse D.: L.V.:	Analytics	paper	included	Analytics and Networks	2021
Scopus	17.11.2024	Marzo A.; Wood S.; Viceconti M.; Narracott A.	exemplar solution to predict femoral bone strength from CT data	Ande	included	Computer Methods and Programs in Biomedicine	2021
Scopus	05.12.2024	Alnegheimish S.; Alrashed N.; Aleissa F.; Althobaiti S.; Liu D.; Alsaleh M.; Veeramachaneni K.	Cardea: An open automated machine learning framework for electronic health records	paper	included	Science and Advanced Analytics (DSAA)	2020
Scopus	17.11.2024	Reps J.M.; Schuemie M.J.; Suchard M.A.; Ryan P.B.; Rijnbeek P.R.	Design and implementation of a standardized framework to generate and evaluate patient-level prediction models using observational healthcare data	Article	included	Journal of the American Medical Informatics Association	2018
IEEE Xplore	05.12.2024	M. M. Ghassemi; B. E. Moody; LW. H. Lehman; C. Song; Q. Li; H. Sun; R. G. Mark: M. B. Westover: G. D. Clifford	You Snooze, You Win: the PhysioNet/Computing in Cardiology Challenge 2018	Conference	included	2018 Computing in Cardiology Conference (CinC)	2018
ACM Digital	00 11 2024	Juan F. Maestre, Tawanna Dillahunt, Alec Andrew Theisz, Megan Fumess,	Evamining Mobility Among Boople Living with HIV in Rural Areas	Conference	Induded	2021 CHI Conference on Human Factors in	2021
ACM Digital	09.11.2024	valantav Karteswalan, ninany venot, ald Parikk C. Onni	Helping Pregnant Women in the Rural Areas of Pakistan using a Low-cost	Conference	incided	9th International Conference on Information and Communication Technologies and	2021
Library	10.11.2024	Amna Abid and Suleman Shahid D. Thomas: R. Subramanian	Interactive System Value-Mapping IT Platform Options in Global Health: A Multi-year Case of Code	Conference	included	Development (ICTD '17)	2017
IEEE Xplore	05.12.2024	Tuneed V: Keishankumar A : Chitten A L : Kim V: Onnas L	Rot versus the Event-Driven Outbreak Economy	paper	included	System Sciences	2022
Scopus	05.12.2024	D Vesterende O. Teasteller V. Oshina A.L., Kim F., Ogras O.	Al Applications	paper	induded	ISLPED	2022
IEEE Xplore	05.12.2024	r. renonepatit, C. ripamuna, A. Galueron	wearable system for monitoring or numan physical activities	paper	included	eDemocracy & eGovernment (ICEDEG)	2017
IEEE Xplore	30.11.2024	A. Nassem, M. Iamazin, M. H. Aly	A Context-Aware to I-Based Smart Wearable Health Monitoring System	paper	induded	ICCSPA 2020	2021
Scopus	11.11.2024	de Almerda T.F.; Morya E.; Rodrigues A.C.; Dantas A.F.O.A.	Development of a low-cost open-source measurement system for joint angle estimation	Article	induded	Sensors	2021
ACM Digital Library	17.11.2024	Lora Arduser	Impatient patients: a DIY usability approach in diabetes wearable technologies	Article	included	Communication Design Quarterly Review	2018
IEEE Xplore	30.11.2024	V. Choudhari; V. Dandge; N. Choudhary; R. G. Sutar	A portable and low-cost 12-lead ECG device for sustainable remote healthcare	Conference paper	included	2018 International Conference on Communication information and Computing Technology (ICCICT)	2018
Springer	08.12.2024	Ashish SinghAntonio BevilacquaThach Le NguyenFeiyan HuKevin McGuinnessMartin O'ReillyDarragh WhelanBrian CaulfieldGeorgiana Ifrim	Fast and robust video-based exercise classification via body pose tracking and scalable multivariate time series classifiers	Article	included	Data Mining and Knowledge Discovery	2023
Scopus	17.11.2024	Pisha L.; Garudadri H.; Warchall J.; Zubatiy T.; Hamilton S.; Lee CH.; Chockalingam G.; Mercier P.P.; Gupta R.; Rao B.D.	A wearable, Extensible, Open-Source Platform for Hearing Healthcare Research	Article	included	IEEE Access	2019

## Appendix A-2

Title	Topic	Topic keywords	Article URL
Fine-grained mathematical modeling for cost-effectiveness evaluation of public health policies for cervical cancer, with application to a Colombian case study	Open-Source Modeling for Cervical Cancer Public Health Policies	Cancer Care with Open-Source Software	https://www.scopus.com/inward/record.uri?/
Open source hardware, exploring how industry regulation affects knowledge commons governance: An exploratory case study	Open-Source Mobile Application for Cancer Patient Management	Cancer Care with Open-Source Software	https://www.scopus.com/inward/record.uri?/
Design of an ultrasound-navigated prostate cancer biopsy system for nationwide implementation in	Onen-Source Prostate Cencer Bioney System	Cancer Care with Onen-Source Software	https://www.scopus.com/inward/record.uri?
Biomechanical modelling of the pelvic system: improving the accuracy of the location of neoplasms	Open-Source Prostate Cancer Bubysy System		https://www.scopus.com/inward/record.uri?
in MRI-TRUS fusion prostate biopsy Integrating Artificial Intelligence Tools in the Clinical Research Setting: The Ovarian Cancer Use	Biomechanical Modeling for Prostate Biopsy Accuracy	Cancer Care with Open-Source Software	https://www.scopus.com/inward/record.uri?
Case The federated tumor segmentation (FeTS) tool: an open-source solution to further solid tumor	Integration of Al Tools in Clinical Research for Ovarian Cancer	Cancer Care with Open-Source Software	https://www.scopus.com/inward/record.uri?
research SMAPT practicion cancer medicine: A FMIP-based and to provide genomic information at the point	Federated Tumor Segmentation in Healthcare	Cancer Care with Open-Source Software	https://www.econue.com/inward/secord.uri?
of care	Open-Source App for Genomic Information Integration in Healthcare	Cancer Care with Open-Source Software	https://www.scopus.com/inward/record.un//
Integrated Extended SIR-type Epidemiological Modeling Platform for Covid-19 How makers responded to the PPE shortage during the COVID-19 pandemic: an analysis focused	Open-Source Epidemiological Modeling Platform for COVID-19	COVID-19 with Open-Source Software	10.1109/EHB55594.2022.9991549
on the Hauts-de-France region	Collaborative PPE Production in Response to COVID-19	COVID-19 with Open-Source Software	https://doi.org/10.1145/3479986.3479989
The power of open-source hardware in crisis: perspectives from the Wilson Center's open hardware and COVID-19 roundtable	Open-Source Hardware's Role in Crisis Management and Healthcare	COVID-19 with Open-Source Software	https://doi.org/10.1145/3495262
Externalization of software behavior by the mining of norms	Data Dregs and Its Implications for AI Ethics: Lessons from the Pandemic	COVID-19 with Open-Source Software	http://link.springer.com/article/10.1007/s436
Estimating the COVID-19 epidemic trajectory and hospital capacity requirements in South West	Open Source Platforms in Healthcare During COVID-19	COVID-19 with Open-Source Software	https://www.scopus.com/inward/record.uri?
England: A mathematical modelling framework An application to support COVID-19 occupational health and patient tracking at a Veterans Affairs	Open-Source COVID-19 Epidemic Modelling Framework	COVID-19 with Open-Source Software	https://www.scopus.com/inward/record.uri?
medical center	Open-Source Application for COVID-19 Occupational Health and Patient Tracking	COVID-19 with Open-Source Software	https://www.suppus.com/inward/socrat.uri?
of-Care Diagnostics	Radiomics-Based Prediction of COVID-19 Outcomes	COVID-19 with Open-Source Software	
during the first wave of COVID-19	COVID-19 Impact on Primary Care Activity Using Open Source	COVID-19 with Open-Source Software	https://www.scopus.com/inward/record.un//
How Human and Organizational Factors Influence Software Teams Productivity in COVID-19 Pandemic: A Brazilian Survey	Productivity Impact of Remote Work on Software Teams During COVID-19	COVID-19 with Open-Source Software	https://doi.org/10.1145/3422392.3422417
Automated production of research data marts from a canonical fast healthcare interoperability	Open Source Data Transformation for COVID-19 Research	COVID-19 with Open-Source Software	https://www.scopus.com/inward/record.uri?
Inferring actual treatment pathways from patient records	Inferring Treatment Pathways Using Open-Source Tools	Decision Support in Healthcare with Open-Source Software	https://www.scopus.com/inward/record.uri?
Opensource spatial optimization in GIScience for strategic positioning	Open-Source Spatial Optimization in Healthcare	Decision Support in Healthcare with Open-Source Software	https://www.scopus.com/inward/record.uri?/
based clinical decision support: Informatics challenges and issues from three clinical domains	Integration of web-based Clinical Decision Support (CDS) tools with VistA EHR	Decision Support in Healthcare with Open-Source Software	https://ink.springer.com/article/10.1186/s1
An open source decision support system for facility location analysis Open source automated insulin delivery: addressing the challenge	Open-Source Decision Support System for Healthcare Facility Location Analysis	Decision Support in Healthcare with Open-Source Software	https://www.scopus.com/inward/record.uri?/ https://www.scopus.com/inward/record.uri?
D-GDM: A mobile diagnostic decision support system for gestational diabetes	Open-Source Automated Insulin Delivery Systems for Diabetes Management Open-Source Mobile Application for Gestational Diabetes Diagnosis	Diabetes Care with Open-Source Software	https://www.scopus.com/inward/record.uri?
HealthFog: An ensemble deep learning based Smart Healthcare System for Automatic Diagnosis of Heart Diseases in integrated IoT and fog computing environments	Open-Source Framework for Heart Disease Diagnosis Using Fog Computing	Diabetes Care with Open-Source Software	https://www.scopus.com/inward/record.uri?
Health & amp; Play: Addressing together the challenges of creating an open source standard for	Open-Source Framework for Heart Disease Disgroup Carry Fog Computing	Diabetes Care man Open-Cource Contraine	
Assessing Credibility Factors of Short-Form Social Media Posts: A Crowdsourced Online	Open-Source Health Games for Respiratory Health	Digital Health and Open Source Software	https://doi.org/10.1145/2962132.2962138
Experiment	Online Health Information Credibility Assessment	Digital Health and Open Source Software	https://doi.org/10.1145/3605390.3605406
in Cyprus: MYeHealthAppCY: Patient-Centered Teleconsultation	Teleconsultation for Patient-Centered Healthcare	Digital Health and Open Source Software	https://doi.org/10.1145/3606305.3606329
Open Speech Platform: Democratizing Hearing Aid Research	Open-Source Platform for Hearing Aid Research	Digital Health and Open Source Software	https://doi.org/10.1145/3421937.3422017
System for Adolescent Mental Health Screening	Facial Recognition for Mental Health Screening	Digital Health and Open Source Software	https://doi.org/10.1145/3472813.3473183
LibreHealth Cost-of-Care Explorer: Mobile Application for Patient-friendly Access to Hospital Charoemasters	Mobile Application for Patient-Friendly Access to Hospital Chargemasters	Digital Health and Open Source Software	https://ieeexplore.ieee.org/abstract/docume
An Open-Source ACASI Design for Social Collaborative mHealth	Open-Source ACASI System for Social Collaborative mHealth	Digital Health and Open Source Software	https://www.scopus.com/inward/record.uri?
Research Challenges of Emerging Technologies Supporting Life-Long Health and Wellbeing Browser-based Infographic Tailoring Self-service Interface (BITSI)	Emerging Technologies in Health and Wellbeing	Digital Health and Open Source Software	https://doi.org/10.1145/3132635.3132639 10.1109/VAHC57815.2022.10108526
Healthcare analytics and visualization using SEMantic Open Source Software (SEMOSS)	Healthcare Analytics and Visualization Using SEMOSS	Digital Health and Open Source Software	https://www.scopus.com/inward/record.uri?
NHS Number Open Source Software: Implications for Digital Health Regulation and Development	Quality and Regulation of NHS Number Open Source Software	Digital Health and Open Source Software	https://www.scopus.com/inward/record.uri?/
EMICS'21: Eye Movements as an Interface to Cognitive State	Eye Tracking and Cognitive State Analysis Open-Source Machine Learning Frameworks in Healthcare	Digital Health and Open Source Software	https://doi.org/10.1145/3411763.3441357 https://doi.org/10.1145/3208157.3208175
8-year Evaluation of GameBus: Status quo in Aiming for an Open Access Platform to Prototype and	Oran Assass Distance for Disital Markh Ass Brelatinia	Digital Health and Open Source Software	https://doi.org/10.1115/2502022
MobSF: Mobile Health Care Android Applications Through the Lens of Open Source Static Analysis	Mobile Health Care Applications Through Open-Source Static Analysis	Digital Health and Open Source Software	https://www.scopus.com/inward/record.uri?
Open Source Digital Health Software for Resilient, Accessible and Equitable Healthcare Systems	Open Source Software in Healthcare	Digital Health and Open Source Software	https://www.scopus.com/inward/record.uri?/
How Transgender People and Communities Were Involved in Trans Technology Design Processes Design and Implementation WSN Based on Raspberry Pi for Medical Application	Community-Involved Design of Trans Technologies	Digital Health and Open Source Software	https://doi.org/10.1145/3544548.3580972 https://www.scopus.com/inward/record.uri?
AI Explainability 360 Toolkit	Open-Source Toolkit for Al Explainability	Digital Health and Open Source Software	https://doi.org/10.1145/3430984.3430987
Lessons from the deployment of the SPIRIT app to support collaborative care for rural patients with complex psychiatric conditions	Mobile Health App for Collaborative Mental Health Care	Digital Health and Open Source Software	https://doi.org/10.1145/3123024.3125610
Advances in DIY Health and Wellbeing	DIY Open-Source Technologies for Health and Wellbeing	Digital Health and Open Source Software	https://doi.org/10.1145/2851581.2856467
Design and development of a customizable telemedicine platform for improving access to healthcare for underserved populations	Telemedicine Platform for Underserved Populations Using Open Source	Digital Health and Open Source Software	https://ieeexplore.ieee.org/stamp/stamp.jsp
Scoping review and evaluation of SMS/text messaging platforms for mHealth projects or clinical interventions	Open-Source SMS/Text Messaging Platforms for mHealth	Digital Health and Open Source Software	https://www.scopus.com/inward/record.uri?
Low-Cost Open-Source Solution to Optimize Emergency Medical Services in Developing	Onen Source EMS Ontimization for Developing Computition	Emorganey Healthcare Solutions with Open Source Software	10.1109/GHTC46095.2019.9033058
Reasoning on Health Condition Evolution for Enhanced Detection of Vulnerable People in	open oblice and optimization of bereisping communices.	Emergency mannance controls with open course conver	
Emergency Settings Data Sharing and Information Platforms in Crisis Response and Preparedness: Exploring the Role	Condition Evolution Reasoning for Emergency Health Services	Emergency Healthcare Solutions with Open-Source Softwar	/https://doi.org/10.1145/3460210.3493551
of Open Data Sharing Platforms and Collective Intelligence in COVID-19 Response Efforts, and Preparedness for Future Pandemics	Use of Open Source Platforms in Pandemic Response and Healthcare Crisis	Emergency Healthcare Solutions with Open-Source Softwar	https://doi.org/10.1145/3529165
Utilizing Open ERP for Creating Medical Record Management System in Smart Hospital : A Case	One FDD for Medical Departure in the Smart Medical Inc.	Emergency Healthcare Colutions with Open Source Software	10.1109//AICT50021.2020.9172029
Guidelines for the development of a critical software under emergency	Guidelines for Developing Critical Software Under Emergency	Emergency Healthcare Solutions with Open-Source Software	https://www.scopus.com/inward/record.uri?
Crisis, what crisis: does reproducibility in modeling & amp; simulation really matter? In Proceedings of the 2018 Winter Simulation Conference (WSC '18)	Reproducibility in Open-Source Simulation for Healthcare	Emergency Healthcare Solutions with Open-Source Softwar	https://bura.brunel.ac.uk/bitstream/2438/18
Clinical implementation of 3D printed plesiotherapy moulds using free open source software	3D Printing and Open-Source Software for Brachytherapy Moulds	Medical 3D Printing with Open-Source Software	https://www.scopus.com/inward/record.uri?
Three-dimensional printing of orbital computed tomography scan images for use in ophthalmology teaching	3D Printing for Ophthalmology Education	Medical 3D Printing with Open-Source Software	https://www.scopus.com/inward/record.uri?/
An Easy-to-Use Protocol for Segmenting and 3-D Printing Craniofacial CT-Images Using Open- Source Software	Open-Source Protocol for 3D Printing Craniofacial Models	Medical 3D Printing with Open-Source Software	https://www.scopus.com/inward/record.uri?/
Femoral Neck Angle Impacts Hip Disorder and Surgical Intervention: A Patient-Specific 3D Printed	n District March (sell's Queen District		
3D medical objects processing and retrieval using spherical harmonics: a case study with	3D Medical Object Retrieval with Spherical Harmonics for Congestive Heart	Medical SD Printing with Open-Source Soliware	https://doi.org/10.1145/3509199.3509225
congestive heart failure MBI exams NextMed: How to enhance 3D radiological images with Augmented and Virtual Reality	Failure Diagnosis Aurmented and Virtual Reality for 3D Radiological Imaging	Medical 3D Printing with Open-Source Software	https://doi.org/10.1145/3167132.3167168 https://doi.org/10.1145/3284179.3284247
Enhanced Visualisation of Normal Anatomy with Potential Use of Augmented Reality Superimposed	Augmented and virtual Reality of 3D Radiological Imaging	Medical SD Printing with Open-Source Soliware	https://www.scopus.com/inward/record.uri?
3D Virtual Planning for Rhinoplasty Using a Free Add-On for Open-Source Software	Open-Source AR and 3D Printing for Anatomy Education 3D Simulation for Rhinoplasty Using Open-Source Software	Medical 3D Printing with Open-Source Software Medical 3D Printing with Open-Source Software	https://www.scopus.com/inward/record.uri?
Homomorphic Encryption for Machine Learning in Medicine and Bioinformatics	Privacy-Preserving Machine Learning in Medicine Using Homomorphic Encryption	Medical 3D Printing with Open-Source Software	https://doi.org/10.1145/3394658
Architectural Solutions to Mitigate Security Vulnerabilities in Software Systems Exploiting New CPU Extensions for Secure Exchange of eHealth Data at the EU Level	Security Vulnerabilities in Software Systems	Open Source Security Measures in Healthcare Systems	https://doi.org/10.1145/3230833.3233766 https://www.scopus.com/inward/record.uri?
ClinGen Variant Curation Interface: a variant classification platform for the application of evidence	Secure energin Data Excitange with Openiver and intel 30X	open outre occurity measures in realiticale oysients	https://www.scopus.com/inward/record.uri?
A Lot Less Likely Than I Thought: Introducing Evidence-Based Security Risk Assessment for	Open-Source variant Classification for Precision Medicine	Open Source Security Measures in Healthcare Systems	https://ieeexplore.ieee.org/document/10305
Healthcare Software	Security Risk Assessment for Healthcare Software Ecasibility and challenges of archetyping a complete Electronic Health Record dataset in	Open Source Security Measures in Healthcare Systems	
Modeling EHR with the openEHR approach: an exploratory study in China	Chinese hospitals	Open Source Software in Electronic Health Records	https://link.springer.com/article/10.1186/s12
An OpenEHR Adoption in a Portuguese Healthcare Facility ADAM Genomics Schema - Extension for Precision Medicine Research*	An OpenEHR Adoption in a Portuguese Healthcare Facility Extending Genomic Schema for Precision Medicine	Open Source Software in Electronic Health Records Open Source Software in Genomic Medicine	https://www.sciencedirect.com/science/artic https://doi.org/10.1145/3194658.3194669
IGZO-TFT-PDK: Thin-Film Flexible Electronics Design Kit, Standard Cell and Design Methodology	Open-Source Design Kit for Flexible Electronics in Healthcare Sensors	Open Source Software in Genomic Medicine	10.1109/OJCAS.2021.3123396
Resources for healthcare workflow modeling and analysis Open Source Robust Machine Learning Software for Medical Patient Data Analysis and Cloud	Open-Source Tools for Healthcare Workflow Modeling and Redesign Open-Source Machine Learning Software for Medical Patient Data Analysis and	Open Source Software in Health Data Management	https://doi.org/10.1145/3297280.3297637 https://www.scopus.com/inward/racord.uri?
Storage	Cloud Storage	Open Source Software in Health Data Management	
Collaborative Cloud Computing Framework for Health Data with Open Source Technologies	Cloud-based Big Data Framework for Biomedical Literature Mining Cloud-Based Health Data Framework with Open Source Technologies	Open Source Software in Health Data Management Open Source Software in Health Data Management	https://doi.org/10.1145/3342827.3342843 https://doi.org/10.1145/3388440.3412460
A de-identification tool for users in medical operations and public health	Open-Source Tool for Health Data Privacy	Open Source Software in Health Data Management	10.1109/BHI.2016.7455951
Democracy beyond projects: how users may achieve lasting influence on it systems Discovering Potential Effects of Distance Supplements (see Twitter Date	Open Source Telehealth Systems and Democratic Governance in Healthcare	Open Source Software in Health Data Management	https://doi.org/10.1145/2889276
A Versatile and Scalable Platform That Streamlines Data Collection for Patient-Centered Studies:	Sincery Supprement Safety wonitoring Using Social Media Data	Open Source Soliware in Health Data Management	https://www.scopus.com/inward/record.uri?
How to Overcome Lack of Health Record Data and Privacy Obstacles in Initial Phases of Medical	Open-Source Hattorm for Patient-Centered Data Collection	Open Source Software in Health Data Management	https://www.scopus.com/inward/record.uri?
Data Analysis Projects CONSENSUS: a Shiny application of dementia evaluation and reporting for the KU ADC longitudinal	Synthetic EHR Generation Using Open-Source Tools	Open Source Software in Health Data Management	https://www.scopus.com/inward/record uri?
Clinical Cohort database	Open-Source Shiny Application for Dementia Evaluation	Open Source Software in Health Data Management	han block and a second state
usurerum, smallt and secure coordinate-based automated health PDF data batch extraction	Automated Data Extraction Tool for Health PDFs	Open Source Software in Health Data Management	https://doi.org/10.1145/3278576.3281274

COP-E-CAT: cleaning and organization pipeline for EHR computational and analytic tasks A novel technique of motif discovery for medical big data using hadoop EHR Data Processing Pipeline for MIMIC-IV Motif Discovery for ECG Data Analysis Using Hadoop Open Source Software in Health Data Management Open Source Software in Health Data Management https://doi.org/10.1145/3459930.3469536 10.1109/ICEDSS.2017.8073683 n novel technique of motif discovery for medical big data using hadoop Identifying and prioritizing benefits and risks of using privacy-enhancing software through participatory design. A nominal group technique study with patients living with chonic condition Design and implementation of an Open Source Thin SIM System for Collecting Data Bamp Supporting Global Health Case Open-Source Privacy-Enhancing Software for Healthcare Data Open Source Software in Health Data Management tions Open-Source Thin SIM System for Health Data Collection Data Collection and ICT Tools in Malaria Surveillance Leveraging Distributed Data on Big Data Platforms for Healthcare Services Open-Source Framework for Managing Electronic Forms in Healthcare Differential Privacy for Geosphatel Healthcare Data Analysis Open-Source Tool for Health Data De-Identification Open Source Software in Health Data Management https://doi.org/10.1145/3001913.3001923 Data Bites Man: The Production of Malaria by Technology Leveraging Distributed Data over Big Data Analytics Platform for Healthcare Services https://doi.org/10.1145/3274422 https://www.scopus.com/inward/record.uri?e Everaging Unsubute Data Viet Big Data Analysis Flation in the mean de Serivises FHIRForm: An open-source framework for the management of electronic forms in healthcare Leveraging Unsernial Privacy in Geospatial Analyses of Standardized Healthcare Data An Open Source Tool for Game Theoretic Health Data De-Identification 10.1109/BigData50022.2020.9378390 An Open Source Tool for Game Theoretic Health Data De-Mentification Real-time event-driven sensor data analytics at the edge-Internet of Things for smart personal PHIR PTI: An open software application for spatialottemporal integration of clinical data and environmental exposures data. Fred-Bioldet: A General Open-Source Frontend Framework for Federated Learning in Healthcare Using Distributed Data over HBase in Big Data Analytics Platform for Clinical Services A fully integrated open-source toolkit for mining healthcare big-data: Architecture and applications A scalable, secure, and interopretable platform for deep data-driven health management The digital biomarker discovery pipeline: An open-source software platform for the development of digital biomarker discovery pipeline: An open-source software platform for interoperability in type 1 diabetes management. Big data analytics and its application tool for the analysis of peripheral physiological data https://www.scopus.com/inward/record.uri?e https://www.scopus.com/inward/record.uri?e Open-Source Frameworks for Real-Time Sensor Data Analytics in Healthcare Open Source Software in Health Data Management https://www.scopus.com/inward/record.uri?e Open-Source Software for Spatiotemporal Data Integration in Healthcare Fed-BioMed: Open-Source Federated Learning Framework for Healthcare Open-Source Big Data Analytics Platform for Healthcare Open Source Software in Health Data Management Open Source Software in Health Data Management Open Source Software in Health Data Management https://www.scopus.com/inward/record.uri?e https://www.scopus.com/inward/record.uri?e https://www.scopus.com/inward/record.uri?e https://www.scopus.com/inward/record.uri?e https://www.scopus.com/inward/record.uri?e Open-Source Toolkit for Mining Healthcare Big Data Open Source Software in Health Data Managemen Open-Source Platform for Biomedical Data Managemen Open Source Software in Health Data Managemen Open-Source Application for Dementia Evaluation and Reporting Open Source Software in Health Data Management https://www.scopus.com/inward/rec Open-Source Platform for Diabetes Data Management Open Source Software in Health Data Managemen Open-Source Tools for Big Data Analytics in Healthca Open-Source Software for Physiological Data Analysis Open Source Software in Health Data Managemen Open Source Software in Health Data Managemen https://www.scopus.com/inward/record.uri?e Big data analytics and its applications An open-source feature extraction tool for the analysis of peripheral physiological data A FRAMEWORK TO SHARE HEALTHCARE SIMULATIONS ON THE WEB USING FREE AND OPEN SOURCE TOOLS AND PYTHON https://www.scopus.com/inward/record.uri?e https://www.scopus.com/inward/record.uri?e Open Source Software in Health IT Infrastructure Framework for Sharing Healthcare Simulations with Open-Source Tools reanework for sharing Heanthcare Simulations with Open-Source Tools Ontology-Based Systems for Traditional Medicine CARUDY - Care Coach: An Ambient Assisted Living Ecosystem for Supporting Open Data and Open Science Clinhal Standards and Local Development: A Case Study of OpenHIE in Developing Commits: Source Software in Health IT Infrastruct https://doi.org/10.1145/3305160.3305212 http://link.andioaer.com/chapter/10.1007/97/ A Study on Traditional Medicine Ontology CARIOT+ Care Coach – An Ambient Assisted Living Ecosystem for Supporting Open Data and Open S Open Source Software in Health IT Infrastructure Science Projects Standards and Local Developm http://link.springer.com/chapter/10.1007/97 Open Source Software in Health IT Infrastructure Open Source Software in Health IT Infrastructure Multilayer NoC firewall services: case-study on e-health MyCareBudget: Co-creating a Healthcare Digital Commons with and for Disabled Citizens and thei Unpaid Carers Network-on-Chip (NoC) Firewall for Real-Time E-Health Data Protection https://doi.org/10.1145/3479876.3481598 Digital Commons for Self-Directed Healthcare Budgets Standardizing Healthcare IoT with Open-Source Framework IoTivi Quantum Key Distribution for Secure Digital Health Services Open Source Software in Health IT Infrastructure Open Source Software in Health IT Infrastructure Open Source Software in Health IT Infrastructure https://doi.org/10.1145/3544548.3580934 10.23919/ITUK 48006.2019.8996134 althcare IoT Standardization with Open Source: A Case Study on OCF and IoTivity radinating resentance of a semidarization win Open Source. A case Source of the offent A Prototype for GND-secure Servertees Computing with ETSI MEC Developing an Architecture for IoT Interoperability in Healthcare: A Case Study of Real-time SpO2 Signal Monitoring and Analysis Towards Responsible Data Practices in Digital Health: A case study of an open source community's 10.1109/SMARTCOMP58114.2023.00043 https://ieeexplore.ieee.org/document/937820 IoT Architecture for Healthcare Interoperability with Open-Source Platforms Towards Responsible Data Practices in Digital Health: A Case Study of an Open-Source Community's Journey Open Source Software in Health IT Infrastructure https://www.scopus.com/inward/record.uri?e Open Source Software in Health IT Infrastructure Journey Breaking Barriers for Interoperability: A Reference Implementation of CSV-FHIR Transformation Using Open-Source Tools Adaptation of open source software for healthcare in India: Case of CARE2X OCF healthcare proof of concept (PicC) on libelium MySignals Short-term adoption rates for a web-based portal within the intranet of a hospital information system https://www.scopus.com/inward/record.uri?e Open-Source Reference for CSV-to-FHIR Transformation Open Source Software in Health IT Infrastructure https://www.scopus.com/inward/record.uri?e https://www.scopus.com/inward/record.uri?e Adaptation of Open-Source Software for Healthcare Management in India Open Source Software in Health IT Infrastructure Open Connectivity Framework for Healthcare IoT Open-Source CMS Adoption in Hospital Information Systems Open Source Software in Health IT Infrastructure Open Source Software in Health IT Infrastructure https://www.scopus.com/inward/record.uri?e An Empirical Study of Blockchain Repositories in GitHub Building an Open Health Data Analytics Platform: a Case Study Examining Relationships and Trends in Seniority and Performance in Healthcare Providers Authorization solution for full stack FHIR HAPI access Empirical Analysis of Open-Source Blockchain Repositories Open Source Software in Health IT Infrastructure https://doi.org/10.1145/3530019.3530041 //link.springer.com/article/10.1007/s416 Building an Open Health Data Analytics Platform Open-Source Authorization Solution for FHIR HAPI Access Open Source Software in Health IT Infrastruct Open Source Software in Health IT Infrastruct Co-design open-source medical devices: how to minimize the human error using UBORA e-10.1109/EMBC.2019.8856891 Co-Design of Open-Source Medical Devices for Global Health Equity Open Source Software in Health IT Infrastructure infrastructure A 3-Tier Architecture for Network Latency Reduction in Healthcare Internet-of-Things Using Fog A 3-Tier Architecture for Network Latency Reduction in Healthcare Internet-ol-Things Using Fog Computing and Machine Learning Open-Source Based Integration Solution for Hospitals A Systematic Review of Open Source Clinical Software on GiHub for Improving Software Reuse in Smart Healthcare Barriers and facilitators to Electronic Medical Record (EMR) use in an urban slum Fog-Based Architecture for Latency Reduction in Healthcare IoT Open Source Software in Health IT Infrastructure https://doi.org/10.1145/3316615.3318222 https://ieeexplore.ieee.org/document/754600 Fog-Based Architecture for Latency Reduction in Healthcare IoT Open-Source Integration Solution for Hospital Systems Systematic review of open source clinical software on GitHub, focusing on software reuse. Open-Source EMR Deployment in Urban Slume Open-Source Simulator for Healthcare Education Challenges and Opportunities in Student Contributions to Humanitarian FOSS Private Control Con Open Source Software in Health IT Infrastructure https://www.mdpi.com/2076-3417/9/1/150#n Open Source Software in Health IT Infrastructure Open Source Software in Health IT Infrastru Open Source Software in Health IT Infrastru Open Source Software in Medical Education Sustainable Health Education Simulator Using Open-Source Technology https://www.scopus.com/inward/record.uri?e Open Source Software in Medical Education and Traini Student contribution to HFOSS: challenges and opportunities A Virtual Spine Construction Algorithm for a Patient - Specific Pedicle Screw Surgical Simulators https://dl.acm.org/doi/abs/10.5555/32 10.1109/SSCI51031.2022.10022213 Projects Virtual Spine Construction Algorithm for Patient-Specific Surgical Sin Simulation-Based Medical Education Platform Using PVSIo-Web Open-Source Interactive Learning in Pharmacy Education Towards a Simulation-Based Medical Education Platform for PVSic-Web 10.1109/ITCGI.2018.8602845 Reflection on the development of a patient case in twine https://www.scopus.com/inward/record.uri?e Reflection on the development of a patient case in twine Realization of Open Software Chain for 30 Modeling and Printing of Organs in Simulation Centers: Example of Renal Petvis Reconstruction The Role of Freezilizer and Open Source Software in Learning Health Systems Open Simulation Environment for Learning and Practice of Robot-Assisted Surgical Suturing The Case for a Potable Open-Source 30 Ultrasourci Lisseus, Benefits, and Challenges Towards Clinical Translation of Deep Learning-based Classification of DSA Image Sequences for Similar Translation of Deep Learning-based Classification of DSA Image Sequences for Similar Translation of Deep Learning-based Classification of DSA Image Sequences for Similar Translation of Deep Learning-based Classification of DSA Image Sequences for Similar Translation of Deep Learning-based Classification of DSA Image Sequences for Similar Translation of Deep Learning-based Classification of DSA Image Sequences for Similar Translation of Deep Learning-based Classification of DSA Image Sequences for Similar Translation of Deep Learning-based Classification of DSA Image Sequences for Similar Translation of Deep Learning-based Classification of DSA Image Sequences for Similar Translation of Deep Learning-based Classification of DSA Image Sequences for Similar Translation of Deep Learning-based Classification of DSA Image Sequences for Similar Translation of Deep Learning-based Classification of DSA Image Sequences for Similar Translation of Deep Learning-based Classification of DSA Image Sequences for Similar Translation of Deep Learning-based Classification of DSA Image Sequences for Similar Translation of Deep Learning-based Classification of DSA Image Sequences for Similar Translation of Deep Learning-based Classification of DSA Image Sequences for Similar Translation of Deep Learning-based Classification of DSA Image Sequences for Similar Translation of Deep Learning-based Classification of DSA Image Sequences for Similar Based Classification of DSA Image Sequences for Similar Open-Source Noted Instance Dearing in Framma / Source Noted Instance Dearing (InFramma / Source Noted Instance https://www.scopus.com/i Open Source Software in Medical Education and Training Open Source Software in Medical Education and Training Open Source Software in Medical Education and Training Open Source Software in Medical Imaging https://www.scopus.com/inward/n 10 1100/L PA 2022 31/600/ 10.1109/ULTSYM.2019.8926072 ttp://link.springer.com/chapter/10.1007/97 rds Guincai irransiation of Deep Learning-based Glassification of DSA image Sequences for e Treatment enting Biomedical Engineering Education Working Towards the 2030 Agenda for Sustainable Open Source Software in Medical Imaging Reinverting Biomedical Engineering Education Proving Con-Development BrainX3: A NeuroInformatic Tool for Interactive Exploration of Multimodal Brain Datasets http://link.springer.com/chapter/10.1007/97/ Open Source Software in Medical Imaging http://link.springer.com/chapter/10.1007/97 Open-Source Neuroinformatics Tools for Brain Data Exploration Digital Twin of Musculoskieldel Systems Using OpenSim UniToBrain Dataset A Brain Perfusion Dataset Network Performance Evaluation of a Biobanking System for Diagnostic Images oinformatics Tools for Brain Data Exploration Open Source Software in Medical Imaging A Neutronomitatic too ni naractove Exploration of Musimodal Bran Datasets win of a Musculoskeletal System ain Dataset: A Brain Perfusion Dataset rk performance view of a biobanking system for diagnostic images Relationship Between Open Science in Artificial Intelligence for Medical Imaging and Global and Statement of the Statement of Statement of Statement of Statement of Statement Relationship Between Open Science in Artificial Intelligence for Medical Imaging and Global Digital Twin of UniToBrain Da http://link.springer.com/chapter/10.1007/97 10.1109/ISCC50000.2020.9219662 A network periodition periodition periodition and periodic periodition perioditi perioditi Open Science in Al for Medical Imaging and Global Health Equity Open Source Software in Medical Imaging 10.1109/ENBENG.2019.8692545 Accounting Mechanism for Secure Medical Imaging Services Open Source Software in Medical Imaging Primer for image information obtained metadation imaging outries Primer for image information in personalized medicine Train the Trainers: Medical Technology for the Sustainable Development of Africa Open-Source Image Informatics for Personalized Medicine Open-Source Medical Imaging Platform for Education and Sustainable Development in Africa Open Source Software in Medical Imaging https://www.scopus.com/inward/record.ur https://ieeexplore.ieee.org/stamp/stamp.is Open Source Software in Medical Imaging ntation of Deep Learning in LibreHealth Radiology on Chest X-Ray Implementation of Deep Learning in LibreHealth Radiology on Chest X-Rays Open Source Software in Medical In https://link.springer.com/chapter/10.1007/97 Texas TriVa 3D Slicer ( ve 1.0 : a reverse-engineered, open model of the human tricuspid valve raniomaxillofacial Modules Support Patient-Specific Decision-Making for Personalized verse-Engineered, Open Model of the Human Tricuspid Valv Open Source Software in Medical Imagi http://link.springer.com/article/10.1007/s003/ https://www.scopus.com/inward/record.uri?e Texas Invarie 1.0: a reverse-engineered, open model of action 30 Silcer Craniomaxillofacial Modules Support Patient-Specific D Healthcare in Dental Research Diccogle Framework for Medical Imaging Teaching and Research 3D Slicer for Craniomavillofacial Decision-Making Open Source Software in Medical Imaging Open Source Software in Medical Imaging https://www.scopus.com/inward/record.uri?e vork for Open-Source Medical Imaging in Teaching and Research loader - namework for weakan maging reaching and research Ivation of Free, Open-source, Web-based DICOM Viewers for the Indian National Telemedicine vice (@Sanjeevani) .onciling interoperability with efficient Verification and Validation within open source simulation https://www.scopus.com/inward/record.uri?e Evalu Open-Source for Telemedicine Integration Open Source Software in Medical Imaging https://www.scopus.com/inward/record.uri?e Open-Source Simulation for Cyber-Physical Systems Open Source Software in Medical Imaging rce autocorrelation-based method for fetal heart rate estimation from one-dimensional https://www.scopus.com/inward/record.uri?e open source autocorrelation-based meanor no common opler ultrasound mparison of Three 3D Segmentation Software Tools for Hip Surgical Planning Open Source Software in Medical Imaging Open Source Software in Medical Imaging Open Source Software in Medical Imaging Open-Source Method for Fetal Heart Ra https://www.scopus.com/inward/record.uri?e 3D Segmentation Tools for Hip Surgery Planning Open-Source VR Environment for Biomedical and Health Applications comparison or inree 40 Segmentation Software Tools for Hip Surgical Planning Multipurpose Virtual Reality Environment for Biomedical and Health Applications Learning HL7 FHIR Using the HAPI FHIR Server and Its Use in Medical Imaging with the SIIM Dataset 10.1109/TNSRE.2019.2926786 https://www.scopus.com/inward/record.uri?e Open-Source Tools for Learning HL7 FHIR in Medical Imaging Open Source Software in Medical Imaging Dataset Anatomy of an Extensible Open Source PACS Technical Note: SpekPy v2.0—a software tookit for modeling x-ray tube spectra A Survey of DICOM Viewer Software to Integrate Clinical Research and Medical Imaging Open-Source PACS (Picture Archiving and Communication Systems) Framework Open Source Software in Medical Imaging Open-Source Toolkit for Modeling X-Ray Tube Spectra Evaluation of Open-Source DICOM (Digital imaging and communications in medicine) Viewer Software Open Source Software in Medical Imaging https://www.scopus.com/inward/record.uri?e https://www.scopus.com/inward/record.uri?e Open Source Software in Medical Imaging FOSS TECHNOLOGIES in MODELLING SPATIAL ACCESSIBILITY of PRIMARY HEALTH CARE https://www.scopus.com/inward/record.uri?e FOSS Technologies in Modeling Spatial Accessibility to Primary Healthcare Open Source Software in Public Health Sector in MALAWI Optimization of Event Management Activity in Public Health Services Using OpenERP with Quickstart Approach A Design Iteration Towards a Multi-Modal Software System for Improving Home Care Experiences in Saskatchewan Building global research capacity in public health: the case of a science gateway for physical 10.1109/CCWC51732.2021.9375913 OpenERP for Public Health Event Management Open Source Software in Public Health Sector https://www.scopus.com/inward/record.uri?e Open-Source Framework for Improving Home Care Experie Open Source Software in Public Health Sector Building global research capacity in public health: the case of a science gateway for physical activity lifelogn guideling and simulation Critical factors affecting the adoption of open source software in public organizations A GIS open source application to perform the spatial adoption guidely indicators (POIs) Menal Health Information Reporting Assistant (MHIRA)—an open-source software facilitating evidence-based assessment for clinical services in Sradie Standardzaktor communities in Shaping National Health Information The Fall of Global Standardzaktor Communities in Shaping National Health Information Open Science Platform for Public Health Research Capit Open Source Software in Public Health Sector https://dl.acm.org/doi/10.5555/3400397.3400 https://www.scopus.com/inward/record.uri?e Factors Influencing Open Source Software Adoption in Healthcare Open Source Software in Public Health Sector https://www.scopus.com/inward/record.uri?e GIS Open-Source Application for Healthcare Quality Indicators in Portugal Open Source Software in Public Health Sector https://www.scopus.com/inward/record.uri?e Open-Source Software for Evidence-Based Mental Health Assessment Global Standardization Communities in Shaping National Health Information Open Source Software in Public Health Sector http://link.springer.com/chapter/10.1007/97 Open Source Software in Public Health Sector tures nity of Practice in Practice: Successful Implementation of Integrated Community Health 10 23919/IST-Africa56635 2022 9845579 unity of Practice in Practice: Successrul imperimentation ation Systems method for identifying interdependencies in service delivery in critical intrastructu Open Source Software in Public Health Sector Community Health Information Systems with Open Source Platforms Open Source Software in Public Health Sector d for Identifying Interdependencies in Healthcare Service Delivery A Technique to Identify Actionable Trends for Policy-Makers Through Open http://link.springer.com/article/10.1007/s4110 PIKS: A Technique Healthcare Data http://link.springer.com/article/10.1007/s4297 PIKS: A Technique to identify Actionable Trends for Policy-Makers Through Open Healthcare Data Operational response simulation tool for epidemics within refugee and IDP settlements: A scenario-based case study of the Cox's Bazer settlement Making a difference: an overview of humanitarian free open source systems Open Source Software in Public Health Sector Simulation of Epidemic Responses in Refugee Settlements Humanitarian Free and Open Source Software (HFOSS) for Social Good Open Source Software in Public Health Sector Open Source Software in Public Health Sector https://doi.org/10.1145/2889160.2892651

ord uni?e

ord uri?e

rd.uri?e

rd.uri?e

Automatic gait analysis through computer vision: a pilot study	Open-Source Gait Analysis Software	Open Source Software in Robotics for Healthcare	https://www.scopus.com/inward/record.uri?e
A Low-cost Healthcare Bot for Elderly People	Low-Cost Healthcare Bot for Elderly Assistance	Open Source Software in Robotics for Healthcare	10.1109/ICIEV.2019.8858567
Open Robot Hardware: Progress, Benefits, Challenges, and Best Practices	Open-Source Robotic Hardware in Healthcare	Open Source Software in Robotics for Healthcare	10.1109/MRA.2022.3225725
Development of an Intuitive EMG Interface for Multi-dexterous Robotic Hand	Development of an Intuitive EMG Interface for Multi-Dexterous Robotic Hand	Open Source Software in Robotics for Healthcare	http://link.springer.com/chapter/10.1007/97
Spatial Modeling of Influenza Outbreaks in Saint Petersburg Using Synthetic Populations	Spatial Modeling of Influenza Outbreaks in Saint Petersburg Using Synthetic Populations	Open Source Software in Robotics for Healthcare	http://link.springer.com/chapter/10.1007/978
Surgical robots of the next decade: New trends and paradigms in the 21th century	Trends in Surgical Robotics with Open-Source Platforms	Open Source Software in Robotics for Healthcare	10.1109/NC.2017.8263255
A Continuum Manipulator for Open-Source Surgical Robotics Research and Shared Development	Open-Source Continuum Manipulator for Surgical Robotics	Open Source Software in Robotics for Healthcare	10.1109/TMRB.2020.3041391
Collaborative Robotics Toolkit (CRTK): Open Software Framework for Surgical Robotics Research	Collaborative Robotics Toolkit (CRTK) for Surgical Robotics Research	Open Source Software in Robotics for Healthcare	10.1109/IRC.2020.00014
Let your body move: a prototyping toolkit for wearable force feedback with electrical muscle			
stimulation	EMS Prototyping Toolkit for Haptic Feedback in Healthcare	Open Source Software in Robotics for Healthcare	https://doi.org/10.1145/2935334.2935348
Development of a web-Based Medical Record Management Software for Digital Immigrant Users	Immigrant Users	Open-Source Software in Electronic Health Records	http://ink.springer.com/cnapter/10.1007/97
Governance and sustainability of an open source electronic health record: An interpretive case			https://www.scopus.com/inward/record.uri?e
study of opendolphin in Japan	Governance and Sustainability of Open Source EHR: OpenDolphin Case Study	Open-Source Software in Electronic Health Records	han the second sec
Evaluation of wondvistA and OpenEMR for PCMH care	Evaluation of Open-Source EHR Systems for PCMH	Open-Source Software in Electronic Health Records	https://ieeexplore.ieee.org/stamp/stamp.jsp?
A modern, lean deployment of EHR systems: Two lessons from meaningful use in the United States	Lean Deployment of EHR Systems with Open Source Technologies	Open-Source Software in Electronic Health Records	https://ieeexplore.ieee.org/stamp/stamp.jsp?
Cloud-Based Open Source Primary Care Electronic Patient Record System for Sri Lankan Citizens	Cloud-Based Open-Source EHR for Sri Lankan Primary Care	Open-Source Software in Electronic Health Records	https://www.scopus.com/inward/record.uri?e
On an Approach of the Solution of Machine Learning Problems Integrated with Data from the			http://link.springer.com/chapter/10.1007/97
Open-Source System of Electronic Medical Records: Application for Fractures Prediction	Machine Learning Problems Integrated with Data from Open-Source EMR Systems	Open-Source Software in Electronic Health Records	https://www.sconus.com/inward/racord.uri2a
enterprises	Open-Source EHR System for Interoperability	Open-Source Software in Electronic Health Records	https://www.scopus.com/inward/record.on/re
An exploration of expectations and perceptions of practicing physicians on the implementation of	Computerized clinical decision support systems (CDSS) in electronic health		https://www.scopus.com/inward/record.uri?e
computerized clinical decision support systems using a Qsort approach	records (EHR)	Open-Source Software in Electronic Health Records	
Static Analysis of HIPPA Security Requirements in Electronic Health Record Applications	Static Analysis of HIPAA Security Requirements in Open-Source EHR	Open-Source Software in Electronic Health Records	https://www.scopus.com/inward/record.uri?e
Specvis: Free and open-source software for visual field examination	Open-Source Software for Visual Eigld Examination	Open Source Software in Electronic Health Records	https://www.scopus.com/inward/record.uri?e
Compliance checking of open source EHR applications for HIPAA and ONC security and privacy	Compliance Checking of Open-Source EHR Systems for HIPAA and ONC	Open-Source Solaware in Electronic Health Records	https://www.scopus.com/inward/record.uri?e
requirements	Requirements	Open-Source Software in Electronic Health Records	
RR-APET - Heart rate variability analysis software	Open-Source Software for Digital Biomarker Discovery	Open-Source Software in Electronic Health Records	https://www.scopus.com/inward/record.uri?e
Open-source electronic health record systems: A systematic review of most recent advances	Systematic Review of Open-source EHR Systems	Open-Source Software in Electronic Health Records	https://www.scopus.com/inward/record.uri?e
Conformance Checking: Workflow of Hospitals and Workflow of Open-Source EMRs	Conformance Checking of Open-Source EMR Systems with Hospital Workflows	Open-Source Software in Electronic Health Records	https://www.scopus.com/inward/record.uri?e
Opening the Duke electronic health record to apps: Implementing SMART on FHIR	Implementation of Open-Source Standards in EHR (Electronuc Health Record)		https://www.scopus.com/inward/record.uri?e
Outboard a second method and a finance mark sales for an exiting a without and the	Systems	Open-Source Software in Electronic Health Records	https://www.execution.com/figures/figures/
synthesic An approach, method, and soltware mechanism for generating synthesic patients and the synthetic electronic health care record	Open-Source Software for Generating Synthetic Health Records	Open-Source Software in Electronic Health Records	https://www.scopus.com/inward/record.un/e
Creating an EMR system to support mobile clinics in Haiti	EMB System Design for Mobile Clinics in Resource-Limited Settings	Patient-Care with Open-Source Software	https://doi.org/10.1145/3287098.3289282
	Open Source Frameworks in mHealth Using FHIR for Improved Interoperability		
Drishti: a sense-plan-act extension to open mHealth framework using FHIR	and Patient Care	Patient-Care with Open-Source Software	https://doi.org/10.1109/SEH.2019.00016
Process Control Charts in Falls Prevention: The Experience of the Local Healthcare Authority of Remanded Italy	Onen Source Software for Monitoring Patient Falls in Hospitals	Patiant Care with Open Source Software	https://www.scopus.com/inward/record.uri?e
"it's not just hacking for the sake of it": A gualitative study of health innovators' views on patient-	Open-Source Soltware for Monitoring Patient Pails in Hospitals	Faterit-Care with Oper-Source Soliware	https://www.scopus.com/inward/record.uri?e
driven open innovations, quality and safety	Patient-Driven Open Source Health Innovations	Patient-Care with Open-Source Software	
Predicting Risk of Getting Smoking-Related Cancer: A Comparison of Three Prediction Models	Prediction Models for Smoking-Related Cancer Risks	Prediction Models in Healthcare	https://doi.org/10.1145/3230905.3230950
Predictive Analytics Based on Open Source Technologies for Acute Respiratory Distress Syndrome	Open-Source Predictive Analytics for Acute Respiratory Distress Syndrome	Prediction Models in Healthcare	https://ieeexplore.ieee.org/document/947468
Fuzzy decision tree for breast cancer prediction	Predicting Breast Cancer Stages	Prediction Models in Healthcare	https://doi.org/10.1145/3373477.3373489
In the Name of Fairness: Assessing the Bias in Clinical Record De-identification	Bias in Clinical Record De-identification	Prediction Models in Healthcare	https://doi.org/10.1145/3593013.3593982
Al Based HealthCare Platform for Real Time, Predictive and Prescriptive Analytics	Embeddable Prediction System for Hospital Stays Using Open Clinical Data	Prediction Models in Healthcare	http://link.springer.com/chapter/10.1007/97
Delivering computationally-intensive digital patient applications to the clinic: An exemplar solution to			https://www.scopus.com/inward/record.uri?e
predict femoral bone strength from CT data	Predictive Modeling of Bone Strength Using Open-Source Tools	Prediction Models in Healthcare	https://www.execution.com/forumed/secution/2a
Cardea. An open automated machine learning trainework for electronic nearth records	Cardea: Open-Source Automated Machine Learning Framework for EHR	Prediction Models in Healthcare	https://www.scopus.com/inward/record.un/?e
prediction models using observational healthcare data	Open-Source Framework for Healthcare Prediction Models	Prediction Models in Healthcare	https://www.scopus.com/inward/record.un/e
You Snooze, You Win: the PhysioNet/Computing in Cardiology Challenge 2018	Open-Source Tools for Sleen Arousal Detection in PhysioNet Challenge	Prediction Models in Healthcare	10.22489/CinC.2018.049
Examining Mobility Among Beople Living with HIV in Bural Areas	Transportation Barriers and Solutions for Rural HIV Patients	Rural Healthcare Access with Open-Source Solutions	https://doi.org/10.1145/3411764.3445086
Helping Pregnant Women in the Bural Areas of Pakistan using a Low-cost Interactive System	IVR System for Maternal Health in Rural Pakistan	Rural Healthcare Access with Open-Source Solutions	https://doi.org/10.1145/3136560.3136607
Value-Mapping IT Platform Options in Global Health: A Multi-year Case of Code Rot versus the	TVIX Oyatam for Indiana material metalar in Naran Pakialan	Notal realition recess with open-oduloe obtaions	10.1109/HICSS.2016.384
Event-Driven Outbreak Economy	Open-Source IT Platforms for Global Health Response and Surveillance	Wearables in Healthcare with Open-Source Software	
A Domain-Specific System-On-Chip Design for Energy Efficient Wearable Edge AI Applications	Open-Source SoC Design for Wearable Edge AI in Healthcare	Wearables in Healthcare with Open-Source Software	https://www.scopus.com/inward/record.uri?e
Wearable system for monitoring of human physical activities	Wearable Open-Source System for Monitoring Physical Activities	Wearables in Healthcare with Open-Source Software	https://ieeexplore.ieee.org/document/796254
A Context-Aware IoT-Based Smart Wearable Health Monitoring System	IoT-Based Smart Wearable Health Monitoring System	Wearables in Healthcare with Open-Source Software	10.1109/ICCSPA49915.2021.9385761
Development of a low-cost open-source measurement system for joint angle estimation	Open-Source Joint Angle Measurement System	Wearables in Healthcare with Open-Source Software	https://www.scopus.com/inward/record.uri?e
Impatient patients: a DIY usability approach in diabetes wearable technologies	DIY Usability in Diabetes Wearable Technologies	Wearables in Healthcare with Open-Source Software	https://doi.org/10.1145/3188387.3188390
A portable and low-cost 12-lead ECG device for sustainable remote healthcare	Low-Cost 12-Lead ECG Device for Remote Healthcare	Wearables in Healthcare with Open-Source Software	https://ieeexplore.ieee.org/document/832587
East and solvest video, based exercise classification via body page tracking and easiable			
rast and robust video-based exercise classification via body pose tracking and scalable			
multivariate time series classifiers	Fast and robust video-based exercise classification via body pose tracking	Wearables in Healthcare with Open-Source Software	http://link.springer.com/article/10.1007/s1061

## **Appendix B**

Search				_				
venue	Query date	Author Tran A S : Dong VK : Thinh Ngo H O	Title	Type	Status	Reason	Year	Article URL https://www.scopus.com/inward/record.uri?eid=2-
		Han A.S., Dong V.K., Hinningo H.G.	MEDICAL THERAPY FOR PROTOTYPING THE	Anticie			2022	<u>s2.0-</u>
Scopus	30.06.2024		STYROFOAM MASK		excluded	focus not on OS		85132184994&partnerID=40&md5=94e663d5015 8d4239646bf980De425b3
		Torres J.; Malla S.D.; Silveira V.; Mainero L.;	Local clinical informatics investments are required for in silico	Article			2022	https://www.scopus.com/inward/record.uri?eid=2-
		Czeisler C.; Diaz-Rossello J.L.; Maccarrone A.; Medore A.; Sanchez B.; Blasina F.; Otere J. J.	biomarker generation across the globe: Lessons learned from a secondary applying of the PROP trial					37938&partnerID=40&md5=1ab093751f1d3f6781
Scopus	28.06.2024	Dang C : Phuong T : Beddag M : Vega A : Denis C	A data model for clinical legal medicine practice and the	Article	excluded	tocus not on US	2018	0dc90732180cd5 https://www.scopus.com/inward/record.uri?eid=2-
			development of a dedicated software for both practitioners and					s2.0-85007422328&doi=10.1016%26.jflm. 2016 11 002
Commun	02.07.2024		researchers		a valuada d	fearing and an OC		&partneriD=40&md5=617450c952ac2fd11e3b8f5
Scopus	02.07.2024	Alhothali A : Samkari A : Algasemi U S	Enhanced System for Computer-Aided Detection of MRI Brain	Article	excluded	locus not on OS	2023	2d6/coul1 https://www.scopus.com/inward/record.uri?eid=2-
			Tumors					s2.0-85181034788&doi=10.14569%2fiJACSA. 2023.0141028
Sconue	27.06.2024				excluded	focus not on OS		&partnerID=40&md5=8c8aa466657f8fbc82doc90a
ocopus	21.00.2024	Roberts J.; Zhong Q.; Linger R.	Integrating Foundational and Clinical Science Remotely by	Article	excluded	10003 1101 011 03	2023	https://www.scopus.com/inward/record.uri?eid=2-
			Combining Team-Based Learning and Simulation					s2.0-85161861296&doi=10.1007%2fs40670-023- 01817-
Sconus	27 06 2024				excluded	focus not on OS		9&partneriD=40&md5=0f408a7fb93c312f45ca6b8 87b9a66db
		Schneider F.B.A.; de Melo A.S.; Andreão R.V.;	Alternative class III distributed polysomnography system	Article			2018	https://www.scopus.com/inward/record.uri?eid=2-
		Dalvi R.F.						4740.03317
Scopus	02.07.2024				excluded	focus not on OS		<u>&amp;partnerID=40&amp;md5=680df8eb73f2aacd64b944e</u> 2cc1e4e7f
		Monks T.; Harper A.	Computer model and code sharing practices in healthcare	Article			2023	https://www.scopus.com/inward/record.uri?eid=2- s2.0-851734878778doi=10.1080%2(17477778
			discrete-event simulation: a systematic scoping review					2023 2260772
Scopus	28.06.2024				excluded	focus not on OS		6be40e70a8
		Xu Y.; Wu S.; Wang M.; Zou Y.	Design and implementation of distributed RSA algorithm based	Article			2020	https://www.scopus.com/inward/record.uri?eid=2- s2.0-85053371972&doi=10.1007%2fs12652-018-
			on hadoop					1021- v&partnerID=408md5=d9029e0258ab49b6eee1ff
Scopus	01.07.2024				excluded	focus not on OS		11441b0810
		Dimakos I.; Gourzis P.; Jelastopulu E.	for 2021: Nationwide Prevalence in 2–17-Year-Old Children	Arucie			2023	s2.0-85152895448&doi=10.3390%
Scopus	27.06.2024		and Regional Disparities		excluded	focus not on OS		19db60be3da8f14474a46
		Kabakama S.; Ngallaba S.; Musto R.; Montesanti	Assessment of four common underfive children illnesses	Article			2016	https://www.scopus.com/inward/record.uri?eid=2- s2.0-84976467476&doi=10.1016%2fi.limedinf.
		S.; Konje E.; Kishamawe C.	decision making at Ilemela Municipal Council, Northwest					2016.06.003 8partner/D=408md5=c3e3957593d39dc8566059
Scopus	02.07.2024		Tanzania: A case series analysis		excluded	focus not on OS		365aaed469
		Gabrani-Juma H.; Al Bimani Z.; Zuckier L.S.; Klein R	Development and validation of the Lesion Synthesis Toolbox and the Perception Study Tool for quantifying observer limits of	Article			2020	https://www.scopus.com/inward/record.uri/eid=2- s2.0-85084677570&doi=10.1117%2f1.JML
			detection of lesions in positron emission tomography					7.2.022412 &partnerID=40&md5=d018493ccb72b6831480f20
Scopus	01.07.2024	Sameh A	Knowledge based Bayasian natwork construction algorithm for	Article	excluded	focus not on OS	2010	cb98bb8f0 https://www.scopus.com/inward/record.uri?eid=2-
		Sameria	medical data fusion to enhance services and diagnosis	Alucie			2013	s2.0-85067307820&doi=10.3844%2fjcssp.
0	04.07.0004				a control and	( 00		&partneriD=40&md5=96237612b99d8383f6bc76b
Scopus	01.07.2024	Williams R.D.: Reps J.M.: Kors J.A.: Rvan P.B.:	Using Iterative Pairwise External Validation to Contextualize	Article	excluded	tocus not on US	2022	a034e7db7 https://www.scopus.com/inward/record.uri?eid=2-
		Steyerberg E.; Verhamme K.M.; Rijnbeek P.R.	Prediction Model Performance: A Use Case Predicting 1-Year					s2.0-85130135708&doi=10.1007%2fs40264-022- 01161-
Sconus	27 06 2024		Heart Failure Risk in Patients with Diabetes Across Five Data Sources		excluded	focus not on OS		8&partnerID=40&md5=f6f95605cb6eb444314dbfb 58231cfbc
		Harmon D.J.; Klein B.A.; Im C.; Romero D.	Development and implementation of a three-dimensional (3D)	Article			2022	https://www.scopus.com/inward/record.uri?eid=2-
0	07.00.0004		printing elective course for health science students			( 00		s2.0-851232957288001=10.1002%2888, 2133&partnerID=40&md5=b87c906b74078b1153
Scopus	27.06.2024	Velumani R.: Sudalaimuthu H.: Choudhary G.:	Secured Secret Sharing of QR Codes Based on Nonnegative	Article	excluded	tocus not on US	2022	ct9c24dbd88885 https://www.scopus.com/inward/record.uri?eid=2-
		Bama S.; Jose M.V.; Dragoni N.	Matrix Factorization and Regularized Super Resolution					s2.0-85127970448&doi=10.3390% 2fs22082959&partnerID=40&md5=56a5f2c305cd
Scopus	24.06.2024	Post R : Lu R : Kim L Dupp LD	Convolutional Neural Network	Article	excluded	focus not on OS	2024	9560656cbb914d1618ef https://www.scopus.com/inword/incord.url?eid=?-
		Bent B.; Lu B.; Kim J.; Dunn J.P.	Biosignal compression toolbox for digital biomarker discovery	Article			2021	s2.0-85099359027&doi=10.3390%
Scopus	01.07.2024				excluded	focus not on OS		2152102051643partneriD=404md5=4aec163e22be 22437776643c45d4346d
		Murch B.J.; Cooper J.A.; Hodgett T.J.; Gara E.L.;	Modelling the effect of first-wave COVID-19 on mental health	Article			2021	https://www.scopus.com/inward/record.uri?eid=2- s2.0-85114687206&doi=10.1016%2fi.orhc.
		Walker 0.0., Wood R.M.	3011003					2021.100311 &partnerID=40&md5=c6170ce060a6604accfc8e0
Scopus	29.06.2024	Williams B - Brewn B - Kestenantelia E - use Stee	Term acts: A transportant and reproducible representation of	Article	excluded	focus not on OS	2010	bbd71aea5
		T.; Peek N.	clinical code sets	Article			2019	s2.0-85061551921&doi=10.1371%2fournal.pone.
Scopus	01.07.2024				excluded	focus not on OS		e1bb8aa9f7c8a2d0
		VARGHESE B.A.; SHIN H.; DESAI B.; GHOLAMREZANEZHAD A · LELX · PERKINS M ·	Predicting clinical outcomes in COVID-19 using radiomics on chest radiographs	Article			2021	https://www.scopus.com/inward/record.uri?eid=2- s2.0-85115992553&doi=10.1259%2fBJR.
Scopus	27.06.2024	OBERALA ; NANDA N.; CEN S.; DUDDALWAR V.	cheat radiographa		excluded	focus not on OS		20210221&partnerID=40&md5=e82e408ced8f20d b570eaedc65f3baae
		Murri R.; Masciocchi C.; Lenkowicz J.; Fantoni M.;	A real-time integrated framework to support clinical decision	Article			2022	https://www.scopus.com/inward/record.uri?eid=2- s2.0-851244211018.dvi=10.10163/26.cmph
		Cesario A.: Patamello S.: Antonelli M.: Bellantone	making for covid-19 patients					2022.106655
		R.; Bernabei R.; Boccia S.; Calabresi P.; Cambieri						4de0a4c26
		A.; Cauda R.; Colosimo C.; Crea F.; De Maria R.; De Stefano V : Franceschi F : Gasbarrini A :						
		Landolfi R.; Parolini O.; Richeldi L.; Sanguinetti M.;						
Scopus	29.06.2024	Urbani A.; Zega M.; Scambia G.; Valentini V.	Quantificing the indirect impact of COVID 10 condemic on	Article	excluded	focus not on OS	2022	https://www.econus.com/inward/racord.uri?aid=2.
		AP.; Tuti T.; Paton C.; Wambu B.; English M.;	utilisation of outpatient and immunisation services in Kenya: A	Ancie			2022	s2.0-85126424582&doi=10.1136%2fbmjopen-
Seening	20.06.2024	Okiro E.A.	longitudinal study using interrupted time series analysis		oveluded	focus pot on OS		055815&partnerID=40&md5=eed69a4a3b022f3a7
acopus	25.00.2024	Vasquez-Cevallos L.A.; Bobokova J.; González-	Design and Technical Validation of a Telemedicine Service for	Article	excluded	locus not on 0.5	2018	https://www.scopus.com/inward/record.uri?eid=2-
		Granda P.V.; Iniesta J.M.; Gómez E.J.; Hernando	Rural Healthcare in Ecuador					s2.0-85050037675&doi=10.1089%2ftmj. 2017.0130
Scopus	01.07.2024	M.E.			excluded	focus not on OS		<u>&amp;partneriD=40&amp;md5=e0acd20e32aae86afe07d6a</u> 11d1c3be7
		Lesho E.; Clifford R.; Onmus-Leone F.; Appalla L.;	The challenges of implementing next generation sequencing	Article			2016	https://www.scopus.com/inward/record.uri?eid=2-
		Snesrud E.; Kwak Y.; Ong A.; Maybank R.; Waterman P.: Rohrbeck P.: Julius M.: Roth A.	across a large healthcare system, and the molecular enidemiology and antibiotic suscentibilities of carbanenemase-					0155770&pathererD=40&md5=7127bb91d20deb7
		Martinez J.; Nielsen L.; Steele E.; McGann P.;	producing bacteria in the healthcare system of the U.S.					221223/07/00008791
Scopus	02.07.2024	Hinkle M. Binho E : Codinho T : Valente E : Costa C	Department of Defense A Multimodal Search Engine for Medical Imaging Studies	Article	excluded	focus not on OS	2017	https://www.scopus.com/inward/record.uti?eid=2-
		Find E., Oddinio I., Valente F., Odsta O.	A materiodal Search Engine for Medical Imaging Studies	Article			2017	s2.0-84983390819&doi=10.1007%2fs10278-016-
Sconus	02 07 2024				oveluded	focus not on OS		z&partnerID=40&md5=b7609455efd97e761c239d
ocopus	02.07.2024	Bouteraa Y.; Ben Abdallah I.; Elmogy A.	Design and control of an exoskeleton robot with EMG-driven	Article	excluded	10003 1101 011 000	2020	https://www.scopus.com/inward/record.uri?eid=2-
_			electrical stimulation for upper limb rehabilitation					s2.0-85085064374&doi=10.1108%2flR-02-2020- 0041&partnerID=40&md5=0a0a3e96d4b160d3fe7
Scopus	01.07.2024	Wiemken TL : Furmanek S P : Mattingly W A	Googling your hand bygiene data: Lleing Google Forms	Article	excluded	focus not on OS	2018	d231e0c20038a https://www.scopus.com/inward/record.uri?eid=2-
		Haas J.; Ramirez J.A.; Carrico R.M.	Google Sheets, and R to collect and automate analysis of hand	Particito			2010	s2.0-85042443662&doi=10.1016%2fj.ajic.
Sconue	02 07 2024		hygiene compliance monitoring		ovcluded	focus not on OS		ApartneriD=40&md5=41bd6dc8a7a1b8be3fc88eb
ocopus	02.01.2024	Priyadarshini R.; Barik R.K.; Dubey H.	DeepFog: Fog computing-based deep neural architecture for	Article	excluded	10000 1101 011 00	2018	https://www.scopus.com/inward/record.uri?eid=2-
			prediction of stress types, diabetes and hypertension attacks					s2.0-85059228815&doi=10.3390% 2fcomputation6040062&partnerID=40&md5=b40d
Scopus	01.07.2024	Erdoğan G	An open source Spreadsheet Solver for Vehicle Routing	Article	excluded	focus not on OS	2017	dab255b9b683e6841d0771f146d7 https://www.scopus.com/inward/record.uri?eid=2-
		Eldogan 6.	Problems	Anticle			2017	s2.0-85015977384&doi=10.1016%26.cor.
Coonus	02 07 2024				ovaludad	feaux pet en OP		SpartneriD=40&md5=0646dc601a7e04d8f318859
ACM Digital	02.07.2024		A pattern sequence for designing blockchain-based healthcare		excluded	10003 1100 011 03		101337
Library	28.05.2024	Peng Zhang, Douglas C. Schmidt, and Jules White	information technology systems	Article	excluded	focus not on OS	2021	https://dl.acm.org/doi/10.5555/3492252.3492269
		X. Liang; Y. Jiang; Y. Xie; T. Niu	Quantitative Cone-Beam CT Imaging in Radiotherapy: Parallel Computation and Comprehensive Evaluation on the TrueBeam				2019	10.1109/ACCESS.2019.2902168
IEEE Xplore	02.07.2024		System	Article	excluded	focus not on OS		
		S. Kumar; G. Abowd; W. T. Abraham; M. al'Absi; D. H. Chau; F. Ertin; D. Estrin; D. Canacan; T. Hast;	Center of Excellence for Mobile Sensor Data-to-Knowledge (MD2K)				2017	https://eeexplore.ieee.org/document/7891193
		S. M. Hossain; Z. Ives; J. Kerr; B. M. Marlin; S.						
IFFF Yolore	02 07 2024	Murphy; J. M. Rehg; I. Nahum-Shani; V. Shetty; I. Sim: B. Spring: M. Srivastava: D. Watter		Article	evoluted	focus not on OS		
ACM Digital	52.01.2024	Hussain Ahmad, Isuru Dharmadasa, Faheem Ullah.	A Review on C3I Systems' Security: Vulnerabilities. Attacks	ALCOIC.	CACIDUED	iooda not oll Oa		
Library	23.06.2024	and Muhammad Ali Babar	and Countermeasures	Article	excluded	focus not on OS	2023	https://doi.org/10.1145/3558001
ACM Digital Library	21.06.2024	Gopal Mengi	Accessible and Individualized Learning: MIT Computer Science and Artificial Intelligence Laboratory (CSAIL) Cambridge MA	Article	excluded	focus not on OS	2023	https://doi.org/10.1145/3589853
ACM Digital			Computer professionals without borders: what computer				2020	and a second sec
Library	21.06.2024	Norberto Patrignani	protessionals do with their free time? volunteering	Article	excluded	focus not on OS	2021	https://doi.org/10.1145/3447913.3447924

Search								
venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
ACM Digital Library	23.06.2024	Pascal Fechner, Fabian König, Wolfgang Kratsch, Jannik Lockl, and Maximilian Röglinger	Near-Infrared Spectroscopy for Bladder Monitoring: A Machine Learning Approach	Article	excluded	focus not on OS	2023	https://doi.org/10.1145/3563779
ACM Digital	22.06.2024	Samuel Greengard	Finding a healthier approach to managing medical data	Article	excluded	focus not on OS	2018	https://doi.org/10.1145/3193759
ACM Digital	24.06.2024	Zhongyuan He	From ELIZA to ChatGPT: The Evolution of Chatbots in Public	Article	ovoludod	focus not on OS	2022	
ACM Digital	21.00.2024		Powerful Futures: How a Big Tech Company Envisions	Article	excluded	locus not on OS	2023	https://doi.org/10.1145/3589658
Library ACM Digital	24.06.2024	EunJeong Cheon	Humans and Technologies in the Workplace of the Future The Coping with COVID Project: Participatory Public Health	Article	excluded	focus not on OS	2023	https://doi.org/10.1145/3610103
Library ACM Digital	24.06.2024	Kathryn Yankura Swacha	Communication Digital humanities, middleware, and user experience design for	Article	excluded	focus not on OS	2023	https://doi.org/10.1145/3563890.3563891
Library	23.06.2024	J. D. Applen and Sonia H. Stephens Barry Brown Minna Vigren, Asreen Bostami and	public health applications	Article	excluded	focus not on OS	2018	https://doi.org/10.1145/3188173.3188176
Library	23.06.2024	Mareike Glöss	Economy of Software	Article	excluded	focus not on OS	2022	https://doi.org/10.1145/3555774
ACM Digital Library	23.06.2024	Weijia Zhang, Jiuyong Li, and Lin Liu	A Unified Survey of Treatment Effect Heterogeneity Modelling and Uplift Modelling	Article	excluded	focus not on OS	2021	https://doi.org/10.1145/3466818
ACM Digital		Ting Jiang, Yang Zhang, Minhao Zhang, Ting Yu, Yizheng Chen, Chenhao Lu, Ji Zhang, Zhao Li, Jun	A Survey on Contact Tracing: The Latest Advancements and					
Library	24.06.2024	Gao, and Shuigeng Zhou	Challenges	Article	excluded	focus not on OS	2022	https://doi.org/10.1145/3494529
Library	24.06.2024	Isaac Holeman, and Nicola Dell	Feedback Loops in Rural Kenya	Article	excluded	focus not on OS	2019	https://doi.org/10.1145/3359173
ACM Digital Library	23.06.2024	John Jones, Catherine Gouge, and Mariah Crilley	Design principles for health wearables	Article	excluded	focus not on OS	2017	https://doi.org/10.1145/3131201.3131205
ACM Digital Library	24.06.2024	Shachar Siboni, Asaf Shabtai, Nils O. Tippenhauer, Jemin Lee, and Yuval Elovici	Advanced Security Testbed Framework for Wearable IoT Devices	Article	excluded	focus not on OS	2016	https://doi.org/10.1145/2981546
		Zihan Fang, Madeline Endres, Thomas				keyword mismatch:		
Library	23.05.2024	Leach, and Yu Huang	A Four-Year Study of Student Contributions to OSS vs	Article	excluded	to healthcare	2023	https://doi.org/10.1145/3611643.3616250
ACM Digital						keyword mismatch: not directly related		https://www.imir.org/naners/volume23/10.207/10.
Library	23.06.2024	Craig M. Lewis and Francesco Grossetti	A statistical approach for optimal topic model identification	Article	excluded	to healthcare	2022	297.pdf
ACM Digital			Analysis of Organization of the Interactome Using Dominating			not directly related		
Library	23.06.2024	Huiru Zheng, Chaoyang Wang, and Haiying Wang Hamid JahankhaniArshad JamalShaun Lawson	Sets: A Case Study on Cell Cycle Interaction Networks Cybersecurity, Privacy and Freedom Protection in the	Article Conference	excluded	to healthcare	2017	https://doi.org/10.1109/TCBB.2015.2459712 http://link.springer.com/book/10.1007/978-3-
			Connected World	paper		keyword mismatch:		030-68534-8
						not directly related		
Springer	07.12.2024	Andreas TsamadosNikita AggarwalJosh			excluded	to healthcare keyword mismatch:		
Springer	08 12 2024	CowlsJessica MorleyHuw RobertsMariarosaria TaddeoLuciano Floridi	The ethics of algorithms: key problems and solutions	Article	excluded	not directly related to healthcare	2022	http://link.springer.com/article/10.1007/s00146- 021.01154-8
opinigoi	CONTENEDET				011010000	keyword mismatch:		<u>orrentero</u>
Springer	07.12.2024	KirkLuciano Floridi	Auditing large language models: a three-layered approach	Article	excluded	to healthcare	2023	http://link.springer.com/article/10.1007/s43681- 023-00289-2
			Review of the state of the art in autonomous artificial			keyword mismatch: not directly related		
Springer	08.12.2024	Petar RadanlievDavid De Roure	intelligence	Article	excluded	to healthcare	2023	http://iink.springer.com/article/10.1007/s43681- 022-00176-2
		Helena LöfströmTuwe LöfströmUlf	Investigating the impact of calibration on the quality of			not directly related		http://link.springer.com/article/10.1007/s10472-
Springer	08.12.2024	JohanssonCecilia Sönströd	explanations	Article	excluded	to healthcare keyword mismatch:	2023	023-09837-2
Springer	08 12 2024	Rocio GomezMohan SridharanHeather Bilev	What do you really want to do? Towards a Theory of Intentions for Human-Robot Collaboration	Article	evoluded	not directly related	2021	http://link.springer.com/article/10.1007/s10472-
opiniger	00.12.2024	noolo comezhonan orana an ioanor niloy		101010	CACINGED	keyword mismatch:	LOLI	013-03012-4
Springer	08.12.2024	Henry WildeVincent KnightJonathan Gillard	Evolutionary dataset optimisation: learning algorithm quality through evolution	Article	excluded	to healthcare	2020	http://link.springer.com/article/10.1007/s10489- 019-01592-4
		Juri Di RoccoDavide Di RuscioClaudio Di	HybridRec: A recommender system for tagging GitHub			keyword mismatch: not directly related		
Springer	08.12.2024	SipioPhuong T. NguyenRiccardo Rubei	repositories	Article	excluded	to healthcare	2023	http://ink.springer.com/article/10.1007/s10489- 022-03864-y http://www.scopus.com/inwart/record.uri2eid=2-
		ru C.; Qi X.; ran W.; Wu W.; Shen B.	Medium for the Representation and Exchange of NGS Data	Article		keyword mismatch:	2023	s2.0-85124089124&doi=10.1109%2fTCBB. 2022.3144170
Scopus	24.06.2024				excluded	to healthcare		&partneriD=40&md5=75b13857f5cc517b8849525 b261cb001
		Tampouratzis N.; Mousouliotis P.; Papaefstathiou I.	A Novel Integrated Simulation Framework for Cyber-Physical Systems Modelling	Article		keyword mismatch:	2023	https://www.scopus.com/nward/record.un/?eid=z- s2.0-85166744317&doi=10.1109%2fTPDS. 2023.3300081
Scopus	30.06.2024				excluded	to healthcare		&partneriD=40&md5=a343dbd203119fa91a82292 2f8475459
		Choi H.; Moran J.; Matsumoto N.; Hernandez M.E.; Moore J.H.	Aliro: An automated machine learning tool leveraging large language models	Article		keyword mismatch:	2023	https://www.scopus.com/inward/record.uri?eid=2- s2.0-85174751738&doi=10.1093% 20-ininformation%
Scopus	30.06.2024				excluded	not directly related to healthcare		2fbtad606&partnerID=40&md5=a5d225e761820b e6a70e211728f885b2
		Eickelberg G.; Luo Y.; Sanchez-Pinto L.N.	Development and validation of MicrobEx: an open-source package for microbiology culture concept extraction	Article		keyword mismatch:	2022	https://www.scopus.com/inward/record.uri?eid=2- s2.0-85134966657&doi=10.1093%2fjamiaopen%
Scopus	28.06.2024		publicage to minimize bology suitare concept end deter		excluded	to healthcare		2fooac026&partnerID=40&md5=fd98fe7a884ff725 f098799046737825
		Wagner M.; Bothner C.L.; Rau E.; Zeng P.Z.; Waisanen E.H.; Czerwinski M.A.; Fickes B.; Eastin	Gala: An Open-Access Platform for Interactive Learning with Sustainability Case Studies	Article		keyword mismatch:	2021	s2.0+85124753031&doi=10.1109%2fTLT. 2022.3148723
Scopus	27.06.2024	I.; Hardin R.D.			excluded	to healthcare		&partner/D=40&md5=66e1a0536f9fc5042587358 6adec209d
		Raphaël GvoryDavid Restrepo AmarilesGregory	Ant: a process aware annotation software for regulatory	Article		keyword mismatch: not directly related		
Springer	07.12.2024	LewkowiczHugues Bersini	compliance	Article	excluded	to healthcare	2023	023-09372-9
0	00.000	Rahul Kumar SinghManoj Kumar SachanR. B.	360 degree view of cross-domain opinion classification: a			not directly related	00-	http://link.springer.com/article/10.1007/s10462-
Springer	06.12.2024	Pater	survey	Article	excluded	to nealthcare keyword mismatch:	2021	020-09884-9
Springer	06.12 2024	Jamal UddinRozaida GhazaliMustafa Mat DerisRashid NaseemHabib Shah	A survey on bug prioritization		excluded	not directly related to healthcare	2017	http://link.springer.com/article/10.1007/s10462- 016-9478-6
sh		Neville Kenneth KitsonAnthony C.	······································	Article		keyword mismatch:		
Springer	06.12.2024	Chobtham	A survey of Bayesian Network structure learning		excluded	to healthcare	2023	http://link.springer.com/article/10.1007/s10462- 022-10351-w
				Article		keyword mismatch: not directly related		http://ink.springer.com/article/10.4003/-40402
Springer	06.12.2024	Santosh S. RathoreSandeep Kumar Giang NeuwepStefan DlugglingkuMartin Bobák//iet	A study on software fault prediction techniques	Article	excluded	to healthcare	2019	017-9563-5
		TranÁlvaro López Garcíalgnacio HerediaPeter	Machine Learning and Deep Learning frameworks and libraries	1000		not directly related	0040	http://link.springer.com/article/10.1007/s10462-
Springer	08.12.2024	MalikLadislav Hluchy	for large-scale data mining: a survey	Article	excluded	to healthcare keyword mismatch:	2019	<u>018-09679-z</u>
Springer	06.12.2024	Muhammad Azeem AkbarArif Ali KhanSaima Rafi	A systematic decision-making framework for tackling quantum software engineering challenges		excluded	not directly related to healthcare	2023	http://link.springer.com/article/10.1007/s10515- 023-00389-7
			A Nevel Technique for Accelerating Live Migration in Cloud	Article		keyword mismatch:		
Springer	06.12.2024	Ambika GuptaSuyel Namasudra	Computing		excluded	to healthcare	2022	http://link.springer.com/article/10.1007/s10515- 022-00332-2
		Pattaraporn SangaroonsilpMorakot	An empirical study of automated privacy requirements	Article		keyword mismatch: not directly related		http://link.springer.com/article/40.1007/e10545
Springer	07.12.2024	ChoetkiertikulHoa Khanh DamAditya Ghose	classification in issue reports	Article	excluded	to healthcare keyword mismatch	2023	023-00387-9
Code	09 40 000	Amin NikanjamMohammad Mehdi MorovatiFoutse	Faults in deep reinforcement learning programs: a taxonomy		and	not directly related	202.	http://link.springer.com/article/10.1007/s10515-
opringer	08.12.2024	Knominhoussem ben brallek	anu a ustection approach	Article	excluded	keyword mismatch:	2021	<u>021-00313-x</u>
						not directly related		
Sprinaer	06.12.2024	Saad AlqahtanyNathan ClarkeSteven FurnellChristoph Reich	A forensic acquisition and analysis system for IaaS		excluded	to healthcare	2016	http://link.springer.com/article/10.1007/s10586- 015-0509-x
Springer	06.12.2024	Saad AlqahtanyNathan ClarkeSteven FurnellChristoph Reich Anayo Chukwu IkegwuHenry Friday NwekeChioma Virninia Aniwel Izoma Bita AlaObituwalu Banbaci	A forensic acquisition and analysis system for laaS	Article	excluded	to healthcare keyword mismatch:	2016	http://link.springer.com/article/10.1007/s10586- 015-0509-x
Springer Springer	06.12.2024	Saad AlqahtanyNathan ClarkeSteven FurnellChristoph Reich Anayo Chukwu IkegwuHenry Friday NwekeChioma Virginia AnikweUzoma Rita AloObikwelu Raphael Okonkwo	A forensic acquisition and analysis system for laaS Big data analytics for data-driven industry: a review of data sources, tools, challenges, solutions, and research directions	Article	excluded excluded	to healthcare keyword mismatch: not directly related to healthcare	2016	http://link.springer.com/article/10.1007/s10586- 015-0509-x http:/link.springer.com/article/10.1007/s10586- 022-03568-5
Springer Springer	06.12.2024 07.12.2024	Saad AlqahtanyNathan ClarkeSteven FurneliChristoph Reich Anayo Chukwu IkegwuHenry Friday NwekeChioma Virginia AnikweUzoma Rita AloObikwelu Raphael Okonkwo Zahra AmiriArash HeidariNima Jafari	A forensic acquisition and analysis system for laaS Big data analytics for data-driven industry: a review of data sources, tools, challenges, solutions, and research directions Resilient and dependability management in distributed environments: a systematic and comprehensive literature	Article	excluded excluded	to healthcare keyword mismatch: not directly related to healthcare keyword mismatch: not directly related	2016	http://inik.springer.com/article/10.1007/s10586- 015-0509-2 015-0509-2 015-0509-2 http://inik.springer.com/article/10.1007/s10586- 022-03569-5 http://inik.springer.com/article/10.1007/s10586- 022-03569-5

venue	Query date	Author	Title	Type	Status	Reason	Year	Article URL
Tende	autory auto		Are online discussions enough to constitute communities of	Article	oluluo	keyword mismatch:	. our	And the
			practice in professional domain? A case study of ergonomics'			not directly related		http://link.springer.com/article/10.1007/s10111-
Springer	07.12.2024	Flore BarcelliniCatherine DelgouletJulien Nelson	practice in France		excluded	to healthcare	2016	015-0361-z
				Article		keyword mismatch:		
0	00 40 0004	Vieling Charl Jacon VieCon, Chara Zangui Li	A Decade of Sentic Computing: Topic Modeling and		and a dealer of	not directly related	0000	http://link.springer.com/article/10.1007/s12559-
Springer	06.12.2024	Kieling ChenHaoran Kiedary Chengzongxi Li	Bibliometric Analysis	Article	excluded	to nealtricare	2022	021-09861-6
		Tulika SahaDhawal GuptaSriparna SahaPushpak	Emotion Aided Dialogue Act Classification for Task-	Allois		not directly related		
Springer	08.12.2024	Bhattacharyya	Independent Conversations in a Multi-modal Framework		excluded	to healthcare	2021	http://link.springer.com/article/10.1007/s12559- 019-09704-5
		Mukesh Kumar SainiNeeraj GoelHanumant Singh	Agriculture-Centric Computation	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
		ShekhawatJaime Lloret MauriDhananjay Singh	-	paper		not directly related		031-43605-5
Springer	06.12.2024				excluded	to healthcare		
		Christian ZirpinsGuadalupe OrtizZoltan	Advances in Service-Oriented and Cloud Computing	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		NochtaOliver WaldhorstJacopo SoldaniMassimo		paper		not directly related		031-23230-5
Springer	06.12.2024	VillariDamian Tamburri		o /	excluded	to healthcare		http://dist.com/com/com/com/com/com/com/com/com/com/
		Ludovico BorattoStefano FaralliMirko	Advances in Bias and Fairness in Information Retrieval	Conference		keyword mismatch:	2023	031-37249-0
Springer	06 12 2024	Manasolovanni Sulo		paper	ovoludod	to healthcare		
Springer	00.12.2024	Deenak GaroV A NaravanaP N Suganthan Jaume	Advanced Computing	Conference	excluded	keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
		AngueraVijava Kumar KoppulaSuneet Kumar	Advanced computing	paper		not directly related	LOLO	031-35644-5
Springer	06.12.2024	Gupta		P-P	excluded	to healthcare		
		Kothe DougGeist AlSwaroop PophaleHong	Accelerating Science and Engineering Discoveries Through	Conference		keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3-
		LiuSuzanne Parete-Koon	Integrated Research Infrastructure for Experiment, Big Data,	paper		not directly related		031-23606-8
Springer	06.12.2024		Modeling and Simulation		excluded	to healthcare		
		Rabindra Nath ShawMarcin PaprzyckiAnkush	Advanced Communication and Intelligent Systems	Conference		keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-3- 031-45121-8
0	00 40 0004	Ghosh		paper		not directly related		Set Total S
Springer	06.12.2024	Ashish Kumar LukashBraf, Dharm Singh, JatDraf	Advanced Information for Computing Research	Conference	excluded	to nealtricare	2022	http://link.enringer.com/book/10.1007/978-3-
		Kamarul Bin Ghazali HawariProf. Dr. Yiao. Zhi	Advanced informatics for Computing Research	Conterence		not directly related	2022	031-09469-9
Springer	06 12 2024	GaoPawan Lingras		paper	excluded	to healthcare		
opinigoi	CONTRIECT	Xingming SunProf. Xiaorui ZhangZhihua XiaProf.	Advances in Artificial Intelligence and Security	Conference	entitie de d	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		Dr. Elisa Bertino	,	paper		not directly related		031-06761-7
Springer	06.12.2024				excluded	to healthcare		
		Mayank SinghDr. Vipin TyagiDr. P. K. GuptaProf.	Advances in Computing and Data Sciences	Conference		keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3-
		Jan FlusserTuncer Oren		paper		not directly related		031-12030-3
Springer	06.12.2024	Vilacen Russenan Diace Handling Ober Cl	Advenezzate la Interdicielation - Decembra	Ocale	excluded	to healthcare	0000	http://link.ep/inger.com/http://in.inger.com
		vijayan SugumaranDivya UpadhyayShanu Sharma	Advancements in Interdisciplinary Research	Conference		keyword mismatch:	2022	nttp://ink.springer.com/b66k/10.1007/978-3- 031-23724-9
Springer	06 12 2024			haber	excluded	to healthcare		
Springer	00.12.2024	Sridaran BajagonalParvez FarukiKalpesh Popat	Advancements in Smart Computing and Information Security	Conference	excluded	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		Siluaran hajagopararvez rarukikaipesi ropat	Advancements in Smart Computing and mormation Security	paper		not directly related	2022	031-23095-0
Springer	06.12.2024			papor	excluded	to healthcare		
opinigoi	CONTRIBUTI	Ngoc Thanh NguyenJános BotzheimLászló	Advances in Computational Collective Intelligence	Conference	0/10/0000	keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
		GulyásManuel NunezJan TreurGottfried		paper		not directly related		031-41774-0
Springer	06.12.2024	VossenAdrianna Kozierkiewicz			excluded	to healthcare		
		Isaac WoungangSanjay Kumar DhurandherKiran	Advanced Network Technologies and Intelligent Computing	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
		Kumar PattanaikAnshul VermaPradeepika Verma		paper		not directly related		031-28183-9
Springer	06.12.2024				excluded	to healthcare		
		Golnaz Arastoopour IrgensSimon Knight	Advances in Quantitative Ethnography	Conference		keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-3- 031-47014-1
Cosinger	06 10 0004			paper	oveluded	not directly related		
Springer	06.12.2024	Costin Bădică lan TraurDiamal	Advances in Computational Collective Intelligence	Conference	excluded	to nealtricare	2022	http://link.spripger.com/book/10.1007/978-3-
		BenslimaneBogumita HnatkowskaMarek	Advances in computational collective intelligence	naper		not directly related	2022	031-16210-7
Springer	06.12.2024	Krótkiewicz		paper	excluded	to healthcare		
opiniger		Prof. Dr. Xingming SunProf. Xiaorui ZhangZhihua	Advances in Artificial Intelligence and Security	Conference		keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
		XiaProf. Dr. Elisa Bertino		paper		not directly related		030-78618-2
Springer	06.12.2024				excluded	to healthcare		
		Krystian WojtkiewiczProf. Dr. Jan TreurDr. Elias	Advances in Computational Collective Intelligence	Conference		keyword mismatch:	2021	http://ink.springer.com/book/10.1007/978-3- 030-99113-9
		PimenidisMarcin Maleszka		paper		not directly related		000-00110-0
Springer	06.12.2024	Deef Allheim Abstallah De Ostaslauren Manialaren De	Arburnes la Orber Oraurite	0	excluded	to nealthcare	0004	http://link.opringer.com/hook/10.1007/079.091
		Mohammed Anbar	Advances in Cyber Security	Conterence		not directly related	2021	16-8059-5
Springer	06 12 2024	Nonamino Priba		papor	excluded	to healthcare		
opinigoi	CONLECT	Andrew R. RuisSeung B. Lee	Advances in Quantitative Ethnography	Conference	01010000	keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
		0	0 1 7	paper		not directly related		030-67788-6
Springer	06.12.2024				excluded	to healthcare		
		Alvaro David Orjuela-CañónJesus A. LopezJulián	Applications of Computational Intelligence	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		David Arias-LondoñoJuan Carlos Figueroa-García		paper		not directly related		030-51300-3
Springer	07.12.2024	hum Onder Figures OrminOnder Franzel/anid	Applied Computer Oping on in Epsing of	Conference	excluded	to nealthcare	0000	http://ink.opringer.com/hook/10.1007/079-9-
		Disz-GutierrezGermán Hernández-Pérez	Applied Computer Sciences in Engineering	Conterence		not directly related	2022	031-20611-5
Springer	07 12 2024	Diaz-Galienez-German Hernandez-Ferez		paper	excluded	to healthcare		
opinigoi		Hector FlorezHenry Gomez	Applied Informatics	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		,		paper		not directly related		031-19647-8
Springer	07.12.2024				excluded	to healthcare		
		Mufti MahmudCosimo leracitanoM. Shamim	Applied Intelligence and Informatics	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		KaiserNadia MammoneFrancesco Carlo Morabito		paper		not directly related		031-24001-0
Springer	07.12.2024	Man Data Taka Manala Zantana	An a Paul Teacher I a star	0	excluded	to healthcare	0000	http://inicaprimer.com/hook/10.1007/070.2
		VizueteServic Montes LeónPablo Torres	Applied Technologies	Conference		Reyword mismatch:	2023	031-24971-6
Springer	07,12,2024	CarriónBenjamin Durakovic		paper.	excluded	to healthcare		
- F		Ashwani Kumarlztok Fister Jr.P. K. GuptaJohan	Artificial Intelligence and Data Science	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		DebayleZuopeng Justin ZhangMohammed Usman		paper		not directly related		031-21385-4
Springer	07.12.2024				excluded	to healthcare		
		Amita DevProf. Dr. S. S. AgrawalArun Sharma	Artificial Intelligence and Speech Technology	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 030-95711-7
0	07 40 000			paper	and the second	not directly related		sees shift 1.1-1
Springer	07.12.2024	Congress ZhangVanggan- Zhan-	Artificial Intelligence Logic and Applications	Conformer	excluded	to nealthcare	0000	http://link.springer.com/http://10.1007/070.001
		Songmao Znang ronggang Zhang	Artificial Intelligence Logic and Applications	Conference		net directly related	2023	99-7869-4
Springer	07.12.2024			hahai	excluded	to healthcare		
opingo	07.12.2024	A. K. M. Muzahidul IslamJia UddinNafees	Bangabandhu and Digital Bangladesh	Conference	excluded	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		MansoorShahriar RahmanShah Murtaza Rashid Al		paper		not directly related		031-17181-9
Springer	07.12.2024	Masud			excluded	to healthcare		
		Dr. Yuan TianTinghuai MaMuhammad Khurram	Big Data and Security	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-981-
		KhanVictor S. ShengZhaoqing Pan		paper		not directly related		10-202-1
Springer	07.12.2024	Visefare ManeVers Obschlader Occol V. Th	Rie Date and Casial Computing	Orafo	excluded	to healthcare	0000	http://link.opringer.com/http://in.ing7/pro.com
		Alaoleng Mengrang ChenLiming SuoQi XuanZi-Ke	big Data and Social Computing	Conference		keyword mismatch:	2023	99-3925-1
Springer	07 12 2024	znang		paper	excluded	to bealthcare		
opiniger	51.12.2024	D. S. GuruSharath Kumar Y. H. Balakrishna K. B. K	Cognition and Recognition	Conference	avenued	keyword mismatch	2022	http://link.springer.com/book/10.1007/978-3-
		AgrawalManabu Ichino	g	paper		not directly related	2022	031-22405-8
Springer	07.12.2024			·	excluded	to healthcare		
		T.N. NagabhushanAssoc. Prof. V. N. Manjunath	Cognitive Computing and Information Processing	Conference		keyword mismatch:	2018	http://link.springer.com/book/10.1007/978-981-
		AradhyaPrabhudev JagadeeshSeema		paper		not directly related		10-9059-2
Springer	07.12.2024	ShuklaChayadevi M.L.			excluded	to healthcare		
		Fuchun SunAngelo CangelosiJianwei	Cognitive Systems and Information Processing	Conference		keyword mismatch:	2023	nttp://link.springer.com/book/10.1007/978-981- 99-0617-8
Onder	07 40 000 -	ZnangYuanlong YuHuaping LiuBin Fang		paper	evel de l	not directly related		
Springer	07.12.2024	Papiest Singh TomorDr. Shalihar VermaDuli - 1	Communication Networks and Commuting	Conformer	excluded	to nealthcare	2024	http://link.springer.com/hwwk/10.1007/079.091
		Kumar ChaurasiaVrijendra Sinch lemal AbawajuDr	communication, retworks and computing	paper		keyword miematch:	2021	16-8896-6
		Shyam AkasheDr. Pao-Ann HsiungProf. Viiav K.		Prodeco,		not directly related		
Springer	07.12.2024	Bhargava			excluded	to healthcare		
		Wenxing HongYang Weng	Computer Science and Education	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-981-
				paper		not directly related		33-2443-1
Springer	07.12.2024				excluded	to healthcare		

Search	Query date	Author	Title	Type	Status	Reason	Year	Article URI
Vende	query dute	Prof. Yuqing SunTun LuProf. Buqing CaoHongfei	Computer Supported Cooperative Work and Social	Conference	otatas	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-981- 19-4549-6
Springer	07.12.2024	Liping Gao	oonputing	paper	excluded	to healthcare		
		A. Augusto de SousaVlastimil HavranAlexis	Computer Vision, Imaging and Computer Graphics Theory and	Conference		In second and an and a feature of the	2023	http://ink.springer.com/book/10.1007/978-3- 031-25477-2
		Paljic labitha PeckChristophe HurterHelen PurchaseGiovanni Maria FarinellaPetia RadevaKadi	Applications	paper		keyword mismatch: not directly related		
Springer	07.12.2024	Bouatouch			excluded	to healthcare		
		David GerberEvangelos PantazisBiayna BogosianAlicia NahmadConstantinos Miltiadis	Computer-Aided Architectural Design. Design Imperatives: The Future is Now	Conference		keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-981- 19-1280-1
Springer	07.12.2024	bogosianninoia manniadoonstantinos mintadis		paper	excluded	to healthcare		
		Andreas HolzingerHugo Plácido da SilvaJean	Computer-Human Interaction Research and Applications	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-41962-1
Springer	07.12.2024	vanderdoncktLarry Constantine		paper	excluded	to healthcare		
-19		Sanjaya Kumar PandaRashmi Ranjan RoutRavi	Computing, Communication and Learning	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 021-21750-0
Springer	07 12 2024	Chandra SadamBala Venkata Subramaanyam BayanoothalaKuan-Ching LiBaikumar Buyya		paper	evoluded	not directly related to healthcare		031-21/00-0
opinigei	07.12.2024	Giuseppe AgapitoAnna BernasconiCinzia	Current Trends in Web Engineering	Conference	CACINGCO	to notali loaro	2023	http://link.springer.com/book/10.1007/978-3-
		CappielloHasan Ali KhattakInYoung KoGiuseppe		paper		kouword micmatob:		031-25380-5
		PinoliAzzurra RagoneMichele RutaFloriano				not directly related		
Springer	07.12.2024	SciosciaAbhishek Srivastava	Data Apolitics and Management in Data Intensive Domains	Conference	excluded	to healthcare	2022	http://link.springer.com/book/10.1007/978-3-
		ThalheimEva MendezNadezhda Kiselyova	Data Analytics and Management in Data Intensive Domains	paper		not directly related	2022	031-12285-9
Springer	07.12.2024	Prof. Mar. Mar. Mar. Astron. Lond Prof. Dr.	Data Mising	0	excluded	to healthcare	0001	http://link.coriogor.com/book/10.1007/079.091
		Yee Ling BooRichi NayakYanchang ZhaoGraham	Data Mining	paper		not directly related	2021	16-8531-6
Springer	07.12.2024	Williams	Part Million and Pla Parts	0.1	excluded	to healthcare	0010	http://www.com/com/com/com/com/com/com/com/com/com/
		Ying TanProf. Yuhui Shi	Data Mining and Big Data	Conterence paper		not directly related	2019	32-9563-6
Springer	07.12.2024				excluded	to healthcare		
		Pinle QinHongzhi WangGuanglu SunZeguang Lu	Data Science	Conference		keyword mismatch: not directly related	2020	http://link.springer.com/book/10.1007/978-981- 15-7984-4
Springer	07.12.2024			pape.	excluded	to healthcare		
		Chutiporn AnutariyaMarcello M. Bonsangue	Data Science and Artificial Intelligence	Conference		keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-981- 99-7969-1
Springer	07.12.2024			paper	excluded	to healthcare		
		Dr. K. R. VenugopalP. Deepa ShenoyProf. Dr.	Data Science and Computational Intelligence	Conference		keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3- 030-91244-4
Springer	07.12.2024	S. Iyengar		paper	excluded	to healthcare		
		Gabriele KotsisA Min Tjoalsmail KhalilDr. Bernhard	Database and Expert Systems Applications - DEXA 2021	Conference			2021	http://link.springer.com/book/10.1007/978-3- 030-87101-7
		MoserAtif MashkoorJohannes SametingerDr. Anna FenselProf. Jorge Martinez-GilLukas FischerGerald	Workshops	paper		keyword mismatch: not directly related		55555710172
Springer	07.12.2024	CzechFlorian SobieczkySohail Khan			excluded	to healthcare		
		Prof. Dr. Gabriele KotsisA Min Tjoalsmail KhalilDr. Bernhard MoserProf. Dr. Alfred TaudesAtif	Database and Expert Systems Applications - DEXA 2022 Workshops	Conference			2022	http://link.springer.com/book/10.1007/978-3- 031-14343-4
		MashkoorJohannes SametingerJorge Martinez-	the number of th	papor		keyword mismatch:		
0	07 40 0004	GilFlorian SobieczkyLukas FischerRudolf			evel of ed	not directly related		
Springer	07.12.2024	Donatello ConteAna FredOleg GusikhinCarlo	Deep Learning Theory and Applications	Conference	excluded	keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
		Sansone		paper		not directly related		031-39059-3
Springer	07.12.2024	Gang WangArridhana CiptadiAli Ahmadzadeh	Deployable Machine Learning for Security Defense	Conference	excluded	to healthcare keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
				paper		not directly related		030-67639-9
Springer	07.12.2024	Antonio SkarmetaDaniele CanaveseAntonio	Digital Sovereighty in Cyber Security: New Challenges in	Conference	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
		LioySara Matheu	Future Vision	paper		not directly related	2020	031-36096-1
Springer	07.12.2024	Dapiel A AlexandrovAlexander V	Digital Transformation and Global Society	Conformero	excluded	to healthcare	2022	http://link.springer.com/book/10.1007/978-3-
		BoukhanovskyAndrei V. ChugunovYury	Digital transformation and Global Society	paper		keyword mismatch:	2022	030-93715-7
0	07 40 0004	KabanovOlessia Koltsovallya MusabirovSergei				not directly related		
Springer	07.12.2024	Daniel VasićMirela Kundid Vasić	Digital Transformation in Education and Artificial Intelligence	Conference	excluded	keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
<u>.</u>			Application	paper		not directly related		031-36833-2
Springer	07.12.2024	Andrei Chugunovlgor KhodachekYuri	Electronic Governance and Open Society: Challenges in	Conference	excluded	keyword mismatch:	2020	http://link.springer.com/book/10.1007/978-3-
		MisnikovDmitrii Trutnev	Eurasia	paper		not directly related		030-67238-6
Springer	07.12.2024	Lazaros IliadisChrisina JavneAnastasios TefasElias	Engineering Applications of Neural Networks	Conference	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		Pimenidis	5 5 11	paper		not directly related		031-08223-8
Springer	08.12.2024	Davang Norhavati A. Jawawilmran Sarwar	Engineering Software for Modern Challenges	Conference	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		BajwaRafaqut Kazmi		paper		not directly related		031-19968-4
Springer	08.12.2024	Prof. Giancarlo SucciProf. Paolo CiancariniArtem	Frontiers in Software Engineering	Conference	excluded	to healthcare	2021	http://link.springer.com/book/10.1007/978-3-
		Kruglov	Troniters in contware Engineering	paper		not directly related	2021	030-93135-3
Springer	08.12.2024	Farias Parageas Mandias Pasta Zampartia	Competing and Consectial Technologies	Conferences	excluded	to healthcare	0000	http://link.enringer.com/book/10.1007/978-3-
		Enico Bolgogno-Mondino-aola Zampenin	Geomatics and Geospatial rechnologies	paper		not directly related	2022	030-94426-1
Springer	08.12.2024	Farine Demograph Mandia Deale Zemperie	Conmetion for Organ and Digital Transition	Conference	excluded	to healthcare	0000	http://link.opringer.com/book/10.1002/078-2-
		cinco borgogno-monumoPaola Zamperin	Geomatics for Green and Digital Transition	paper		not directly related	2022	031-17439-1
Springer	08.12.2024	Constanting Stankonidial*	UCI Internetional 2020. Late Brouble Brother	Canto	excluded	to healthcare	0000	http://link.environer.com/http://10.1007/070-0
		AntonaStavroula Ntoa	nui international 2020 – Late Breaking Posters	paper		not directly related	2020	030-60703-6
Springer	08.12.2024				excluded	to healthcare		have different and a second provide a second second second
		Prof. Constantine StephanidisDr. Margherita AntonaDr. Stavroula Ntoa	HCI International 2021 - Posters	Conference		keyword mismatch: not directly related	2021	http://ink.springer.com/book/10.1007/978-3- 030-78642-7
Springer	08.12.2024			paper	excluded	to healthcare		
		Constantine StephanidisMargherita	HCI International 2022 – Late Breaking Posters	Conference		keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3- 031-19679-9
Springer	08.12.2024	Antonaotavioua Ntoaoaviel Savendy		paper	excluded	to healthcare		
		Prof. Constantine StephanidisDr. Margherita	HCI International 2022 Posters	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-06394-7
Springer	08.12.2024	Antonabi, Staviotia Nioa		paper	excluded	to healthcare		
		Constantine StephanidisMargherita	HCI International 2023 Posters	Conference		keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-3- 031-35989-7
Springer	08.12.2024	AntonaStavroula NtoaGavriel Salvendy		paper	excluded	to healthcare		
		Philippe NavauxCarlos J. Barrios H.Carla	High Performance Computing	Conference		keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3- 031-23821-5
Springer	08.12.2024	Osmotraines auerrero		paper	excluded	to healthcare		
	/	Isidoro GitlerCarlos Jaime Barrios	High Performance Computing	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-04209-6
Springer	08.12.2024	HernandezEsteban Meneses		paper	excluded	not directly related to healthcare		
		Vladimir Jordanllya TarasovElla ShurinaNikolay	High-Performance Computing Systems and Technologies in	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 021.22744.7
Springer	08 12 2024	FilimonovVladimir Faerman	Scientific Research, Automation of Control and Production	paper	excluded	not directly related to healthcare		55175574941
Shundoi		Dr. Gabriella CasalinoMarta CimitilePietro	Higher Education Learning Methodologies and Technologies	Conference	chloraded	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
Springer	08 12 2024	DucangeNatalia Padilla ZeaRiccardo PecoriProf. Pietro PicernoPaolo Baviolo	Online	paper	excluded	not directly related		<u>uau-a0000-5</u>
opiniger	JU. 12.2024	Dalila DurãesAlfonso González-BrionesMarin	Highlights in Practical Applications of Agents, Multi-Agent	Conference	excluded	keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
Springer	09 12 2024	LujakAlia El BolockJoão Carneiro	Systems, and Cognitive Mimetics. The PAAMS Collection	paper	oveluted	not directly related		<u>u31-37593-4</u>
opninger	JU. 12.2024	Pablo H. RuizVanessa Agredo-DelgadoAndré Luiz	Human-Computer Interaction	Conference	excluded	keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
Corinaan	09 10 0004	Satoshi Kawamoto		paper	oveluder	not directly related		<u>030-92325-9</u>
opringer	00.12.2024				excluded	to rieditricare		

venue	Query date	Author	Title	Type	Status	Reason	Year	Article URL
		Ljupcho AntovskiGoce Armenski	ICT Innovations 2021. Digital Transformation	Conference paper		keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-04206-5
Springer	08.12.2024	Oleksii IgnatenkoVyacheslav KharchenkoVitaliy KobetsHennadiy KravtsovYulia TarasichVadim ErmolayevDavid EstebanVitaliy	ICTERI 2021 Workshops	Conference paper	excluded	keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-14841-5
Springer	08.12.2024	YakovynaAleksander Spivakovsky Maxim BakaevProf. In-Young KoMichael MrissaProf. Cesare PautassoAbhishek Srivastava	ICWE 2021 Workshops	Conference paper	excluded	to healthcare keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 030-92231-3
Springer	08.12.2024	Wang YongtianWu Lifang	Image and Graphics Technologies and Applications	Conference paper	excluded	to healthcare keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-981- 99-7549-5
Springer	06.12.2024	Davide CiucciProf. Dr. Inés CousoJesús MedinaDominik ŚlęzakProf. Davide PetturitiProf. Dr. Bernadette Bouchon-MeunierProf. Ronald R.	Information Processing and Management of Uncertainty in Knowledge-Based Systems	Conference paper	excluded	keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-08974-9
Springer	08.12.2024	Yager Victor TaratukhinMikhail MatveevProf. Dr. Jörg BeckerYury Kupriyanov	Information Systems and Design	Conference paper	excluded	to healthcare keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 030-95494-9
Springer	08.12.2024	Udo R. KriegerGerald EichlerChristian ErfurthGünter Fahrnberger	Innovations for Community Services	Conference paper	excluded	keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-40852-6
Springer	08.12.2024	Dmitrii RodionovTatiana KudryavtsevaAngi SkhvedianiMohammed Ali Berawi	Innovations in Digital Economy	Conference paper	excluded	keyword mismatch: not directly related to healthcare	2021	http://ink.springer.com/book/10.1007/978-3- 030-84845-3
Springer	08.12.2024	Prof. Qinglong HanSean McLooneChen PengProf. Baolin Zhang	Intelligent Equipment, Robots, and Vehicles	Conference paper	excluded	keyword mismatch: not directly related to healthcare	2021	http://link.springer.com/book/10.1007/978-981- 16-7213-2
Springer	08.12.2024	Zhiwen YuXinhong HeiDuanling LiXianhua SongZeguang Lu	Intelligent Robotics	Conference paper	excluded	keyword mismatch: not directly related to healthcare	2023	http://link.springer.com/book/10.1007/978-981- 99-0301-6
Springer	08.12.2024	Akram BennourTolga EnsariYousri KessentiniSean Eom	Intelligent Systems and Pattern Recognition	Conference paper	excluded	keyword mismatch: not directly related to healthcare	2022	http://ink.springer.com/book/10.1007/978-3- 031-08277-1
Springer	08.12.2024	Assoc, Prof. Sule Yildirim YayilganProf. Imran Sarwar BajwaAssoc, Prof. Filippo Sanfilippo	Intelligent Technologies and Applications	Conference paper	excluded	keyword mismatch: not directly related to healthcare	2021	http://ink.springer.com/book/10.1007/978-3- 030-71711-7
Springer	08.12.2024	Revatni venkataramanAnnie Uthravijayan SugumaranR. I. MinuPethuru Raj Chelliah Frans CoenenAna FredDavid Aveiro Ian Dietz Jorce	Internet or Finings	paper	excluded	not directly related to healthcare	2023	http://ink.springer.com/book/10.1007/978-3-
Springer	08.12.2024	BernardinoElio MasciariJoaquim Filipe Michael KampDr. Irena KoprinskaAdrien	Knowledge Management Machine Learning and Principles and Practice of Knowledge	paper	excluded	not directly related to healthcare	2021	031-43471-6 http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	GalarragaJosé OranasLinara AdilovaYamuna KrishnamurtkyBo KangChristine LargeronDr. Jefrey Liffit[Tiphaine ViardPascal WelkeMassimiliano RuccoErlend AuneDr. Claudio GalilochloGregor SchieleFranz PernkopfMichaela BlottHolger FröningGünther SchindlerDr. Riccardo GuidottiAnna MonrealeSalvatore PichenizkiyBodo RosenhainChristopher BuckkyDaniela ClaffiPabio LanillosMaxwell RamsteadTim VerbelenPedro M. FerreiraDr. Giuseppina AndresiniProf. Donato Malerbalbéria MedericoSPhilippe Fournier-VigerM. Sagib NawazSebastian VenturaProf. Meng SumMin ZhouValerio Bitettatlaria BordinoAndrea FerrettiFrancesco GuiloDr. Giovanni PontiLorenzo SeveriniRita Ribeirubabo GamaRicard GavaldAue CooperNaghmeh Chazalehonas RichardiDamian RoqueroDiego Saldana MirandaKonstantinos			excluded	keyword mismatch: not directly related to healthcare		
Springer	08.12.2024	Nilay KhareDeepak Singh TomarMitul Kumar AhirwalVijay Bhaskar SemwalVaibhav Soni	Machine Learning, Image Processing, Network Security and Data Sciences	Conference paper	excluded	keyword mismatch: not directly related to healthcare	2022	http://link.springer.com/book/10.1007/978-3- 031-24352-3
Springer	08.12.2024	Emmanouel GaroufallouAndreas Vlachidis	Metadata and Semantic Research	Conference paper	excluded	keyword mismatch: not directly related to healthcare	2023	http://link.springer.com/book/10.1007/978-3- 031-39141-5
Springer	08.12.2024	Dr. V. ArunachalamK. Sivasankaran	Microelectronic Devices, Circuits and Systems	Conference paper	excluded	keyword mismatch: not directly related to healthcare	2021	http://ink.springer.com/book/10.1007/978-981- 16-5048-2
Springer	08.12.2024	Ilsun YouHwankuk KimPelin Angin	Mobile Internet Security	Conference paper	excluded	keyword mismatch: not directly related to healthcare	2023	http://link.springer.com/book/10.1007/978-981- 99-4430-9
Springer	08.12.2024	Luís Ferreira PiresSlimane HammoudiEdwin Seidewitz	Model-Driven Engineering and Software Development	Conference paper	excluded	keyword mismatch: not directly related to healthcare	2023	http://link.springer.com/book/10.1007/978-3- 031-38821-7
Springer	08.12.2024	Dana SimianLaura Florentina Stoica	Modelling and Development of Intelligent Systems	Conference paper	excluded	keyword mismatch: not directly related to healthcare	2023	http://ink.springer.com/book/10.1007/978-3- 031-27034-5
Springer	08.12.2024	Sun-Yuan HsiehLing-Ju HungRalf KlasingChia-Wei LeeSheng-Lung Peng	New Trends in Computer Technologies and Applications	Conference paper	excluded	keyword mismatch: not directly related to healthcare	2022	http://ink.springer.com/book/10.1007/978-981- 19-9582-8
Springer	08.12.2024	Silvia ChiusanoTania CerquitelliRobert WrembelKjetil NorvågBarbara CataniaGenoveva Vargas-SolarEster Zumpano	New Trends in Database and Information Systems	Conference paper	excluded	keyword mismatch: not directly related to healthcare	2022	http://link.springer.com/book/10.1007/978-3- 031-15743-1 http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	Schwenker	Patters Decembrance of Artificial Intelligence	paper	excluded	not directly related to healthcare	2023	031-31327-1 http://link.springer.com/book/10.1007/078-2
Springer	08.12.2024	BelotserkovskyMarina M. Lukashevich	Pattern Recognition and mormation Processing	paper	excluded	not directly related to healthcare	2022	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	Ventura MartinsRicardo Pérez-Castillo	Quality of Information and Communications Technology	Conference paper	excluded	not directly related to healthcare	2021	030-85347-1 http://link.snringer.com/book/10.1007/978-3-
Springer	08.12.2024	TravassosValentina LenarduzziXiaozhou Li Rajesh Kumar TiwariG, Saboo	Recent Trends in Artificial Intelligence and IoT	paper	excluded	not directly related to healthcare	2023	031-43703-8 http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	Florian NiehlingSander Münsterkleike Messerrer	Research and Education in Lithan History in the Ace of Disite	paper	excluded	not directly related to healthcare	2023	031-37303-9 http://ink.springer.com/book/10.1007/978-9-
Springer	08.12.2024	Péter GalambosErdal Kavacanki meh Madani	Libraries Robotics, Computer Vision and Intalligent Systeme	paper	excluded	not directly related to healthcare keyword mismatch:	2022	030-93186-5 http://ink.springer.com/book/10.1007/978-3-
Springer	08.12.2024	Ram KrishnanH, Raohav RaoSaniav K	Secure Knowledge Management In The Artificial Intelligence	paper	excluded	not directly related to healthcare keyword mismatch	2022	031-19650-8 http://ink.springer.com/book/10.1007/978-3-
Springer	08.12.2024	SahaySagar SamtaniZiming Zhao	Era	paper	excluded	not directly related to healthcare	-966	030-97532-6

venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
		Zhongjie WangShangguang WangHanchuan Xu	Service Science	Conference paper		keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-981- 99-4402-6
Springer	08.12.2024	Johanna Barzen	Service-Oriented Computing	Conference	excluded	to healthcare keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
0	00.40.0004		eerroe energe eerrouwing	paper	and shade of	not directly related	2021	030-87568-8
Springer	08.12.2024	Prof. Dr. Sergio NesmachnowLuis Hernández	Smart Cities	Conference	excluded	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	Callejo		paper	excluded	not directly related to healthcare		030-96753-6
- , <u>.</u>		Fabián R. NarváezJulio ProañoPaulina MorilloDiago VollaioDapiel Conzáloz MontovaCloria	Smart Technologies, Systems and Applications	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 030-99170-8
Springer	08.12.2024	M. Díaz		paper	excluded	to healthcare		
		Marina YusoffTao HaiMurizah KassimAzlinah MohamedEisuke Kita	Soft Computing in Data Science	Conference paper		keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-981- 99-0405-1
Springer	08.12.2024	Henry MucciniParis AvgeriouDr. Barbora	Software Architecture	Conference	excluded	to healthcare	2020	http://link.springer.com/book/10.1007/978-3-
		BuhnovaJavier CamaraMauro CaporuscioMirco		paper		keyword mismatch:		030-59155-7
Springer	08.12.2024	TrubianiDanny WeynsUwe Zdun			excluded	to healthcare		
		Hans-Georg FillMarten van SinderenLeszek A. Maciaszek	Software Technologies	Conference paper		keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-37231-5
Springer	08.12.2024	Aurona GerberMariike Coetzee	South African Institute of Computer Scientists and Information	Conference	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
Coringor	09 12 2024	·	Technologists	paper	oveluded	not directly related		031-39652-6
opinger	00.12.2024	Matthes ElstermannStefanie BetzMatthias Lederer	Subject-Oriented Business Process Management. Dynamic	Conference	excluded	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-19704-8
Springer	08.12.2024		Digital Design of Everything – Designing or being designed?	paper	excluded	to healthcare		301 10101 0
		Pawan WhigNuno SilvaAhmed A. ElngarNagender AnejaPavika Sharma	Sustainable Development through Machine Learning, Al and IoT	Conference paper		keyword mismatch: not directly related	2023	http://ink.springer.com/book/10.1007/978-3- 031-47055-4
Springer	08.12.2024	Murat YilmazPaul ClarkeBichard MessnarzBruno	Systems, Software and Services Process Improvement	Conference	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
Original	00.40.0004	Wöran	cysteria, contrare and controcs riseces improvement	paper	and shade of	not directly related	LULL	031-15559-8
Springer	08.12.2024	Xabier Larrucealzaskun SantamariaProf. Dr. Rory	Systems, Software and Services Process Improvement	Conference	excluded	keyword mismatch:	2018	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	V. O'ConnorRichard Messnarz		paper	excluded	not directly related to healthcare		313-3/323-0
		Prof. Rafael Valencia-GarcíaMartha Bucaram- Leverone-Javier Del Cioppo-MorstadtNéstor Vera-	Technologies and Innovation	Conference paper		keyword mismatch: not directly related	2021	http://ink.springer.com/book/10.1007/978-3- 030-88262-4
Springer	08.12.2024	LucioEmma Jácome-Murillo	Technology and Inneution in Learning, Teaching and	Conforman	excluded	to healthcare	2022	http://link.springer.com/book/10.1007/078-3-
		MartinsAthanassios JimoyiannisRay Yueh-Min	Education	paper		not directly related	2022	031-22918-3
Springer	08.12.2024	HuangRoberto Henriques Miguel Félix Mata-RiveraRoberto Zagal-	Telematics and Computing	Conference	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
Springer	08 12 2024	FloresCristian Barria-Huidobro		paper	excluded	not directly related to healthcare		031-45316-8
opinigoi	CONTRIECT	Rocco FurferiLapo GoverniYary VolpeKate	The Future of Heritage Science and Technologies: ICT and Digital Haritage	Conference	0.00000	keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3- 031-20302-2
Springer	08.12.2024	Seymour Anna Felagour Francesco Grierardini		paper	excluded	to healthcare		
		Henry HanErich Baker	The Recent Advances in Transdisciplinary Data Science	Conference paper		not directly related	2022	031-23387-6
Springer	08.12.2024	Guojun WangKim-Kwang Raymond ChooRyan K.	Ubiquitous Security	Conference	excluded	to healthcare keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-981-
Springer	08 12 2024	L. KoYang XuBruno Crispo		paper	excluded	not directly related		<u>19-0468-4</u>
opinigo		Ana Cristina Eranoo da SilvaDascal Hirmed Iwa	Customization and provisioning of complex quant processing			keyword mismatch:		
Springer	07.12.2024	BreitenbücherOliver KoppBernhard Mitschang	using TOSCA	Article	excluded	to healthcare	2018	http://link.springer.com/article/10.1007/s00450- 017-0386-z
			The Human Infrastructure of Civic Data: A Taxonomy for			keyword mismatch: not directly related		http://link.springer.com/article/10.1007/s10606-
Springer	08.12.2024	Firaz Peer Prakash Das, Shivangi Srivastava, Valentin	Participatory Infrastructuring of Civic Data	Article	excluded	to healthcare keyword mismatch:	2023	023-09470-x
ACM Digital	23.06.2024	Moskovich, Anmol Chaturvedi, Anant Mittal,	CDI-E: an elactic cloud service for data engineering	Article	evoluded	not directly related	2022	
Library	20.00.2024	rengen mue, and mostrata enemanaly		7 0 000	excluded	keyword mismatch:	LULL	100570010101110.1477613504621.3304620
Library	22.06.2024	Blaine Rothrock, Eric Greenlee, and Josiah Hester	Community-Driven Mobile and Ubiquitous Computing	Article	excluded	to healthcare	2023	https://doi.org/10.1145/3631588.3631590
ACM Digital		Geert-Jan Bogaerts, José van Dijck, and Ethan	Creating PublicSpaces: Centering Public Values in Digital			keyword mismatch: not directly related		
Library	23.05.2024	Zuckerman	Infrastructures	Article	excluded	to healthcare keyword mismatch:	2023	https://doi.org/10.1145/3582578
ACM Digital	22.06.2024	Andrew Greasley	Implementing reinforcement learning in simio discrete-event simulation software	Article	evoluded	not directly related	2020	https://dl.com.org/doi/10.5555/2427510.2427520
	LEIGOLEOLY	Line Zhang Union Wass Dishard I Dauburt		144010	excluded	keyword mismatch:	LOLO	negester.acm.org/doi/10.0000/0421010.0421000
Library	23.06.2024	and Rainer Roehe	Bovine Rumen Microbiome	Article	excluded	to healthcare	2020	https://doi.org/10.1109/TCBB.2018.2879342
ACM Digital			Log-related Coding Patterns to Conduct Postmortems of			keyword mismatch: not directly related		
Library	28.05.2024	Farzana Ahamed Bhuiyan and Akond Rahman	Attacks in Supervised Learning-based Projects	Article	excluded	to healthcare keyword mismatch:	2023	https://doi.org/10.1145/3568020
ACM Digital	22.06.2024	Kelly J. Fadel, Matthew L. Jensen, and Thomas O. Meservy	Online Information Filtering: The Role of Contextual Cues in Electronic Networks of Practice	Article	excluded	not directly related to healthcare	2023	https://doi.org/10.1145/3631341.3631347
	LEIGOLOLI	Creaser W Histor Joseph Kukawaka and Lari		744010	excluded	keyword mismatch:	LULU	ritigiastication of the root of the
Library	28.05.2024	Postner	Open source as an extracurricular activity	Article	excluded	to healthcare	2019	https://dl.acm.org/doi/10.5555/3344051.3344075
ACM Digital						keyword mismatch: not directly related		https://dl.acm.org/doi/odf/10.5555/3373669.
Library	22.06.2024	Mary Tedeschi	Patterns for ethical decisions in information systems security	Article	excluded	to healthcare keyword mismatch:	2020	3373683
ACM Digital	23.06.2024	Tirthankar Ghosal, Muskaan Singh, Anja Nedoluzhko, and Ondřej Bojar	Report on the SIGDial 2021 special session on summarization of dialogues and multi-party meetings (SummDial)	Article	evoluted	not directly related	2022	https://doi.org/40.114E/0507E46.2507E61
	20.00.2024		or allogaes and main party meetings (commonal)	74466	CAGIGGGG	keyword mismatch:	LULL	https://doi.org/10.114/J/J221040.3021001
ACM Digital Library	22.06.2024	Jeenisha Shrungare	Robots and You	Article	excluded	to healthcare	2023	https://doi.org/10.1145/3611691
ACM Digital		Sherae Daniel, Shadi Janansefat, E. Ilana Diamant,	Single- and Double-Loop Learning: Linking Free/Libre Open Source Software (FLOSS) Developer Motivation, Contribution,			keyword mismatch: not directly related		
Library	23.05.2024	and Yuqing Ren	and Turnover Intentions	Article	excluded	to healthcare keyword mismatch:	2020	https://doi.org/10.1145/3433148.3433153
ACM Digital	22.06.2024	Jeremiah W. Johnson	Teaching neural networks in the doop learning are	Article	excluded	not directly related	2010	https://dl.acm.org/doi/abs/10.5555/3344051.
Louis	LL.UU.ZUZ4	Secondari IV. Germadil	reasoning neural networks in the deep reaming eta		avoinged	keyword mismatch:	2019	<u>3077988</u>
ACM Digital Library	23.06.2024	Stefan Buchner and Dirk Riehle	The Business Impact of Inner Source and How to Quantify It	Article	excluded	to healthcare	2023	https://doi.org/10.1145/3611648
ACM Digital		Michael Bailey, Christian Conroy, and Vandana	The Data Science Corps: Making a Difference by Connecting			keyword mismatch: not directly related		
Library	23.06.2024	Janeja	Experts to Projects	Article	excluded	to healthcare keyword mismatch:	2017	https://dl.acm.org/doi/book/10.5555/3278159
ACM Digital	23.06.2024	Ninel Sharibolt	The Open Data Revolution	Article	ovolusiasi	not directly related	2022	hiles lidei eseltă și transcena
Lorary	23.00.2024	Zhiyuan Wan, Xin Xia, Yun Zhang, David Lo	What motivates software practitioners to contribute to inner	Arucle	axcluded	keyword mismatch	2023	naps://doi.org/10.1145/3591366.3591381
ACM Digital	20 11 2021	Daibing Zhou, Qiuyuan Chen, and Ahmed E.	Source / In Proceedings of the 30th ACM Joint European Software Engineering Conference and Symposium on the Exundations of Software Engineering (CEEC/CEE 0020)	Article	ovoluted	not directly related	2022	
Lorary	30.11.2024	rissoull	roundations of Software Engineering (ESEC/FSE 2022)	Aruule	excluded	keyword mismatch:	2022	110ps.//001.0rgr10.1145/3540250.3549148
ACM Digital Library	24.06.2024	waryam Amini, Hazvan Stanica, and Catherine Rosenberg	Where Are the (Cellular) Data?	Article	excluded	to healthcare	2023	https://doi.org/10.1145/3610402

Search venue	Query date	Author	Title	Type	Status	Reason	Year	Article URL
· · · · · ·	anory anto			.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	otatao	keyword mismatch:		ALLOW DILL
ACM Digital Library	22.06.2024	Aldo de Moor	Smart cities: heading toward panopticions or smart societies? SIGCAS Comput	Article	excluded	not directly related to healthcare keyword mismatch:	2021	https://doi.org/10.1145/3447913.3447923
ACM Digital Library	24.10.2024	Chris Bopp and Amy Voida	"Showing the Context": A Need for Oligopticonic Information Systems in Homelessness Measurement	Article	excluded	not directly related to healthcare	2023	https://doi.org/10.1145/3579622
ACM Digital Library	23.06.2024	Rui Bordalo, Carlo Bottaini, and António Candeias	A Framework Design for Information Management in Heritage Science Laboratories	Article	excluded	not directly related to healthcare	2021	https://doi.org/10.1145/3417304
ACM Digital Library	24.06.2024	Andrés Domínguez Hernández, Kopo M. Ramokapane, Partha Das Chowdhury, Ola Michalec, Emily Johnstone, Emily Godwin, Alicia G. Cork, and Awais Rashid Galanopoulou R.; Spinellis D.	Co-creating a Transdisciplinary Map of Technology-mediated Harms, Risks and Vulnerabilities: Challenges, Ambivalences and Opportunities A dataset of open-source safety-critical software	Article Conference	excluded	keyword mismatch: not directly related to healthcare keyword mismatch:	2023 2021	https://doi.org/10.1145/3610179 https://www.scopus.com/inward/record.uri?eid=2-
Scopus	05.12.2024	Dubrovenski V.: Karim M.N.: Chen E.: Xu D.	Dynamic Access Control with Administrative Oblications: A	paper Conference	excluded	not directly related to healthcare	2023	s2.0- 85108025545&partnerID=40&md5=01e97189e6e 6005cf63f94c9907ee46 https://www.scopus.com/inward/record.uri?eid=2-
Scopus	06.12.2024		Case Study	paper	excluded	keyword mismatch: not directly related to healthcare		s2.0-85186745146&doi=10.1109%2fQRS- C60940.2023.00071 &partnerID=40&md5=53f0fe6e3339d27b1fff4d5c4 87c064e
0	00 40 0004	Raikar M.M.; Shavi P.; Patil N.; Mudavi S.	IoT prototyping using Block based programming: An use case of smart agriculture	Conference paper		keyword mismatch: not directly related	2023	https://www.scopus.com/inward/record.uri?eid=2- s2.0-85163204432&doi=10.1109%2/AISP57993. 2023.10134851 &partnerID=40&md5=859095d522e5d218413950
Scopus	06.12.2024	Kuryliak Y.; Emmerich M.; Dosyn D.	On the effect of complex network topology in managing epidemic outbreaks	Conference paper	excluded	keyword mismatch: not directly related	2021	120a4actbe https://www.scopus.com/inward/record.uri?eid=2- s2.0-
Scopus	06.12.2024	Shukla S.M.; Mishra G.; Mishra R.	Open source image processing module to improve dental	Conference	excluded	to healthcare	2023	85111897733&partneriD=40&md5=a88aed9108b2 4363d53d405f164463a3 https://www.scopus.com/inward/record.uri?eid=2-
Scopus	06.12.2024		patient care in suburban and rural areas: Review Paper	paper	excluded	keyword mismatch: not directly related to healthcare		s2.0-85179843419&doi=10.1109% 2flCCCNT56998.2023.10307967 &partnerID=40&md5=52a8947cb17281eDe68ad6 d9e4ebb867
		Filaly Y.; Berros N.; Badri H.; mendil F.E.; EL Idrissi Y.E.B.	Security of Hadoop Framework in Big Data	Conference paper		keyword mismatch:	2023	https://www.scopus.com/inward/record.uri?eid=2- s2.0-851511461378doi=10.1007%2f978-3-031- 26254-
Scopus	06.12.2024	Ramos A.R.: Velez F.J.: Gardasevic G.	Source Routing Minimum Cost Forwarding Protocol over	Conference	excluded	to healthcare	2019	8 103&partnerID=40&md5=501b256e26e0f7e6d 30afa7af71aa07 https://www.scopus.com/inward/record.uri?eid=2-
Scopus	06.12.2024		6TISCH Applied to the OpenMote-B Multi-hop Platform	paper	excluded	keyword mismatch: not directly related to healthcare	2010	s2.0-850755989688doi=10.1109%2fPIMRC. 2019.8904356 &partner10=40&md5=115acd0c3a63ec59943d90 d984bda62a
ACM Digital Library	11.11.2024	Oshani Seneviratne and James Hendler (Eds.)	Linking the World's Information: Essays on Tim Berners-Lee's Invention of the World Wide Web (1st	Article	excluded	keyword mismatch: not directly related to healthcare	2023	https://doi.org/10.1145/3591366
ACM Digital Library	23.06.2024	Zoe Kotti, Rafaila Galanopoulou, and Diomidis Spinellis	Machine Learning for Software Engineering: A Tertiary Study	Article	excluded	not directly related to healthcare	2023	https://doi.org/10.1145/3572905
ACM Digital	00.00.0004	Liuyue Jiang, Nguyen Khoi Tran, and Muhammad	Mod2Dash: A Framework for Model-Driven Dashboards			keyword mismatch: not directly related	0000	
ACM Digital	23.06.2024	Ali babar	Providing Detailed Information on National Policies to Cope	Article	excluded	keyword mismatch: not directly related	2022	https://doi.org/10.1145/3534526
Library	23.05.2024	Christophe Gaie Miquel Costa, Diogo Costa, Tiago Gomes, and	with the Covid-19 Pandemic	Article	excluded	to healthcare keyword mismatch: not directly related	2020	https://doi.org/10.1145/3428089
Library	22.06.2024	Sandro Pinto	Shifting Capsule Networks from the Cloud to the Deep Edge	Article	excluded	to healthcare keyword mismatch:	2022	https://doi.org/10.1145/3544562
ACM Digital Library	23.05.2024	Connor Esterwood and Lionel Peter Robert	The Warehouse Robot Interaction Sim: An Open-Source HRI Research Platform	Article	excluded	not directly related to healthcare keyword mismatch:	2023	https://doi.org/10.1145/3568294.3580086
ACM Digital Library	23.05.2024	Richard Halstead-Nussloch and Rebecca Rutherfoord	Tips for Sources of Cost-Free and Open Educational Resources to Reduce Textbook Costs in IT Courses	Article	excluded	not directly related to healthcare	2019	https://doi.org/10.1145/3349266.3351359
ACM Digital Library	24.06.2024	Sharifah Roziah Binti Mohd Kassim, Shujun Li, and Budi Arief	Understanding How National CSIRTs Evaluate Cyber Incident Response Tools and Data: Findings from Focus Group Discussions	Article	excluded	keyword mismatch: not directly related to healthcare	2023	https://doi.org/10.1145/3609230
ACM Digital		Agrim Sachdeva, Ben Lazarine, Hongyi Zhu, and	User Profiling and Vulnerability Introduction Prediction in Social Coding Repositories: A Dynamic Graph Embedding Approach: Vulnerability Introduction Prediction in Social Coding			keyword mismatch: not directly related		
Library	23.05.2024	Sagar Samtani	Repositories	Article	excluded	to healthcare keyword mismatch: not directly related	2023	https://doi.org/10.1145/3607505.3607512
Library	23.06.2024	Advait Deshpande	Assessing the quantum-computing landscape	Article Article	excluded	to healthcare keyword mismatch:	2022	https://doi.org/10.1145/3524109
ACM Digital Library	24.06.2024	Justin Middleton, Emerson Murphy-Hill, and Kathryn T. Stolee	Data Analysts and Their Software Practices: A Profile of the Sabermetrics Community and Beyond		excluded	not directly related to healthcare	2020	https://doi.org/10.1145/3392859
ACM Digital	21.06.2024	Farah E. Shamout and Dana Abu Ali	The strategic pursuit of artificial intelligence in the United Arab	Article	oveluded	keyword mismatch: not directly related	2024	
Library	21.00.2024	Deepthi S.A.; Raol E.S.; farhan M.D.S.	Handwritten to Text Document Converter	Conference	excluded	keyword mismatch:	2021	https://doi.org/10.1145/3447723 https://www.scopus.com/inward/record.uri?eid=2- s2.0-85144201782&doi=10.1007%2f978-981-19-
Scopus	06.12.2024				excluded	not directly related to healthcare		0000- 1_18&partnerID=40&md5=1a0db7c707acdbee77 ce5c3c7738ac0f
		Phalaprom S.; Jitngernmadan P.	Ireedingbot: A vision-based feeding robotic arm prototype based on open source solution	paper		keyword mismatch: not directly related	2020	s2.0-85091530891&doi=10.1007%2f978-3-030- 58805- 2_53&partnerID=40&md5=a01fe7c69b35de95f66
Scopus	06.12.2024	Maikantis T.; Tsintzira AA.; Ampatzoglou A.; Arvanitou EM.; Chatzigeorgiou A.; Stamelos I.;	Software Architecture Reconstruction via a Genetic Algorithm: Applying the Move Class Refactoring	Conference paper	excluded	keyword mismatch:	2020	1c53fab86eab2 https://www.scopus.com/inward/record.uri?eid=2- s2.0-85102359191&doi=10.1145%2f3437120.
Scopus	06.12.2024	Bibi S.; Deligiannis I. Patrick Keller, Abdoul Kader Kaboré, Laura Plein,		F - F	excluded	to healthcare keyword mismatch:		3437292&partnertD=40&md5=836766510b99549 b842921843a79bc57
ACM Digital Library	24.06.2024	Jacques Klein, Yves Le Traon, and Tegawendé F. Bissyandé	What You See is What it Means! Semantic Representation Learning of Code based on Visualization and Transfer Learning	Article	excluded	not directly related to healthcare keyword mismatch:	2021	https://doi.org/10.1145/3485135
ACM Digital Library	22.06.2024	Linda Northrop, Ipek Ozkaya, George Fairbanks, and Michael Keeling	Designing the Software Systems of the Future	Article	excluded	not directly related to healthcare	2019	https://doi.org/10.1145/3282517.3302397
ACM Digital Library	22.06.2024	Misael C. Junior, Domenico Amalititano, Lina Garcés, Anna Rita Fasolino, Stevão A. Andrade, and Márcio Delamaro	Dynamic Testing Techniques of Non-functional Requirements in Mobile Apps: A Systematic Mapping Study	Article	excluded	not directly related to healthcare	2022	https://doi.org/10.1145/3507903
ACM Digital Library	24.06.2024	Vishal Sharma, Neha Kumar, and Bonnie Nardi	Post-growth Human–Computer Interaction	Article	excluded	keyword mismatch: not directly related to healthcare	2023	https://doi.org/10.1145/3624981
ACM Digital	22.06.2024	Badie klammi Sharali Zaadaliy, and Jamal Nabhan	Security Threats, Countermeasures, and Challenges of Digital	Articlo	oveluded	keyword mismatch: not directly related	2022	
ACM Digital	20.00.2024	Cite Mannin, Gristian Zeadaily, dila Jamei Nebhen	Toward Responsible Al: An Overview of Federated Learning	- uuule	excluded	keyword mismatch: not directly related	2023	1846-7001.010110.1140/3000999
Library ACM Digital	24.06.2024	Qiang Yang	for User-centered Privacy-preserving Computing	Article	excluded	to healthcare keyword mismatch: not directly related	2021	https://doi.org/10.1145/3485875
Library	24.06.2024	Riemer van Rozen	Languages of Games and Play: A Systematic Mapping Study	Article	excluded	to healthcare keyword mismatch:	2021	https://doi.org/10.1145/3412843
ACM Digital Library	23.06.2024	Boyuan Chen and Zhen Ming (Jack) Jiang	A Survey of Software Log Instrumentation	Article	excluded	not directly related to healthcare keyword mismatch:	2021	https://doi.org/10.1145/3448976
ACM Digital Library	23.06.2024	Jorge Goncalves, Simo Hosio, and Vassilis Kostakos	Eliciting Structured Knowledge from Situated Crowd Markets	Article	excluded	not directly related to healthcare	2017	https://doi.org/10.1145/3007900
ACM Digital Library	24.06.2024	Denys Flores, Jonathan Burton, and Matthew Butler	Threat-modeling-guided Trust-based Task Offloading for Resource-constrained Internet of Things	Article	excluded	not directly related to healthcare	2022	https://doi.org/10.1145/3510424
,								

Search	Query date	Author	Title	Type	Status	Reason	Vear	Article UPI
venue	Query date	Autor	Title	Type	Status	keyword mismatch:	Teal	ANICIO ORL
ACM Digital Library	22.06.2024	Nischal Shrestha, Titus Barik, and Chris Parnin	Remote, but Connected: How #TidyTuesday Provides an Online Community of Practice for Data Scientists	Article	excluded	not directly related to healthcare	2021	https://doi.org/10.1145/3449126
ACM Digital	22.06.2024	Pantelis Koutroumpis, Aija Leiponen, and Llewellyn	Disital instruments as invention machines	Articlo	oveluded	not directly related	2020	
Library	20.00.2024	Rivero M.H.	An open-source application built with R programming language for clinical laboratories to innovate process of excellence and	Conference	excluded	keyword mismatch:	2020	https://www.scopus.com/inward/record.uri?eid=2- s2.0-85113400941&doi=10.1109%2fCSCI51800.
Scopus	05.12.2024		overcome the uncertain outlook during the global healthcare crisis		excluded	not directly related to healthcare		2020-00163 &partnerID=40&md5=1543c00254cc48c00b2dfa3 75ec5bb2a
Pagaug	05 12 2024	Schuler A.	Application of search-based software engineering methodologies for test suite optimization and evolution in mission critical mobile application development	Conference paper	ovaludad	keyword mismatch: not directly related	2017	https://www.scopus.com/inward/record.uri?eid=2- s2.0-85030786031&doi=10.1145%2f3106237, 3119876&partnerID=40&md5=59c32bbbba94806
Scopus	05.12.2024	Zervakis G.; Xydis S.; Soudris D.	Performance-power exploration of software-defined big data analytics: The AEGLE cloud backend	Conference paper	excluded	keyword mismatch:	2017	https://www.scopus.com/inward/record.uri?eid=2- s2.0-85013807853&doi=10.1109%2/SAMOS.
Scopus	06.12.2024				excluded	not directly related to healthcare		2016.7818364 &patherID=40&md5=51bee9d2149c766368d15b d85ac925a7
		Kaur A.; Verma D.; Kaur N.	Utilizing Quantitative Data Science Salary Analysis to Predict Job Salaries	paper		keyword mismatch: not directly related	2022	ntps://www.scopus.com/inwarb/record.un/e86=2- s2.0-85149444156&doi=10.1109%2fCISCT55310, 2022.10046491 &nartner/D=40&md5=eab/7927be40893e5625f07b
Scopus	06.12.2024				excluded	to healthcare keyword mismatch:		a37890092
ACM Digital Library	24.06.2024	Arash Heidari, Nima Jafari Navimipour, Mehmet Unal, and Guodao Zhang	Machine Learning Applications in Internet-of-Drones: Systematic Review, Recent Deployments, and Open Issues	Article	excluded	not directly related to healthcare	2023	https://doi.org/10.1145/3571728
ACM Digital	24 06 2024	Chen Song, Zhengxiong Li, Wenyao Xu, Chi Zhou, Zhanpeng Jin, and Kui Ren	My Smartphone Recognizes Genuine QR Codes! Practical Unclonable QR Code via 3D Printing	Article	excluded	not directly related to healthcare	2018	https://doi.org/10.1145/3214286
ACM Digital						keyword mismatch: not directly related	20.0	
Library	22.06.2024	Shaukat Ali, Tao Yue, and Rui Abreu	When software engineering meets quantum computing	Article	excluded	to healthcare keyword mismatch:	2022	https://doi.org/10.1145/3512340
ACM Digital Library	22.06.2024	Logan Kugler	Why virtual reality will transform a workplace near you	Article	excluded	not directly related to healthcare	2017	https://doi.org/10.1145/3105444
ACM Digital Library	28.05.2024	Yu Huang, Denae Ford, and Thomas Zimmermann	Leaving My Fingerprints: Motivations and Challenges of Contributing to OSS for Social Good	Article	excluded	not directly related to healthcare	2021	https://dl.acm.org/doi/10.1109/ICSE43902. 2021.00096
ACM Digital		Rajiv Krishnamurthy, Varghese Jacob, Suresh	Peripheral Developer Participation in Open Source Projects: An			keyword mismatch: not directly related		
Library	23.05.2024	Radhakrishnan, and Kutsal Dogan	Empirical Analysis	Article	excluded	to healthcare keyword mismatch:	2016	https://doi.org/10.1145/2820618
ACM Digital Library	23.06.2024	Varsha S. Lalapura, J. Amudha, and Hariramn Selvamuruga Satheesh	Recurrent Neural Networks for Edge Intelligence: A Survey	Article	excluded	to healthcare	2021	https://doi.org/10.1145/3448974
ACM Digital Library	23.06.2024	Brian McFee, Justin Salamon, and Juan Pablo Bello	Adaptive Pooling Operators for Weakly Labeled Sound Event Detection	Article	excluded	not directly related to healthcare	2018	https://doi.org/10.1109/TASLP.2018.2858559
ACM Digital		Yanjiao Chen, Xu Wang, Baochun Li, and Qian	An Incentive Mechanism for Crowdsourcing Systems with			keyword mismatch: not directly related		
Library	24.06.2024	Zhang Thom J.; Das T.; Shrestha B.; Sengupta S.; Arslan	Network Effects Casting a Wide Net: An Internet of Things Testbed for	Article Conference	excluded	to healthcare keyword mismatch:	2019 2021	https://doi.org/10.1145/3347514 https://www.scopus.com/inward/record.uri?eid=2- s2.0-85124043600&doi=10.23919%
Scopus	05.12.2024	les t	Cybersediny Edicator and research	paper	excluded	not directly related to healthcare		2fSPECTS52716.2021.9639278 &partnerID=40&md5=a2206a8756e53e271d1a73 8b993afd38
		Eltenahy S.; Fayez N.; Obayya M.; Khalifa F.	Conversion of Videoconference Speech into Text based on WebRTC and Web Speech APIs	Conference paper		keyword mismatch: not directly related	2021	https://www.scopus.com/inward/record.uri?eid=2- s2.0-85113963302&doi=10.1109%28TC- Egypt52936.2021.9513968
Scopus	05.12.2024	Cheng B.; Wang O.; Shao D.; Arora C.; Hoang T.;	Edge4Emotion: An edge computing based multi-source	Conference	excluded	to healthcare	2021	apartneriD=40&md5=500605baar336abbeee0682 81f6c6031 https://www.scopus.com/inward/record.uri?eid=2- 0.051148749738.dei=10.4000/
Scopus	06 12 2024	Liu X.	emotion recognition platform for human-centric software engineering	paper	excluded	not directly related to healthcare		2/C-031146746738004-10,110236 2fCCGrid51090.2021.00071 &partnerID=40&md5=882c08b14bfdeb3e3271580 9974291651
000000	00.12.2024	Popa DL.; Popa RT.; Barbulescu LF.; Ivanescu RC.; Mocanu ML.	Segmentation of Different Human Organs on 3D Computer Tomography and Magnetic Resonance Imaging using an Open	Conference paper	excitated	keyword mismatch:	2022	https://www.scopus.com/inward/record.uri?eid=2- s2.0-851341611228doi=10.1109%2flCCC54292. 2022.9805911
Scopus	06.12.2024		Source 3D U-Net Framework		excluded	to healthcare		&partneriD=40&md5=e71301659c8b315c6beede 1693495b7e
ACM Digital Library	23.05.2024	Matt Germonprez, Georg J.P. Link, Kevin Lumbard, and Sean Goggins	Eight Observations and 24 Research Questions About Open Source Projects: Illuminating New Realities	Article	excluded	not directly related to healthcare	2018	https://doi.org/10.1145/3274326
ACM Digital		Daniel Abadi, Anastasia Ailamaki, David Andersen, Peter Bailis, Magdalena Balzinska, Philip Bernstein, Peter Boncz, Surajit Chaudhuri, Alvin Chaung, Arthai Doan, Lunca Dong, Michael J Franklin, Juliana Freire, Alon Halevy, Joseph M. Heilerstein, Stratos Idroso, Donald Kossmann, Tim Kraska, Sailesh Krishnamurthy, Volker Markl, Sergey Melnik, Tova Milo, C. Mohan, Thomas Neumann, Beng Chin Coi, Fatma Ozcan, Jignesh Patel, Andrew Pavlo, Raluca Popa, Raghu Ramakrishnan, Christopher Re, Michael				keyword mismatch: not directly related		
Library	23.06.2024	Stonebraker, and Dan Suciu	The Seattle Report on Database Research Toward a Progress Indicator for Machine Learning Model	Article	excluded	to healthcare keyword mismatch:	2020	https://doi.org/10.1145/3385658.3385668
ACM Digital Library	23.06.2024	Gang Luo	Building and Data Mining Algorithm Execution: A Position Paper	Article	excluded	not directly related to healthcare	2017	https://doi.org/10.1145/3166054.3166057
ACM Digital Library	23.06.2024	Cheol-Ho Hong, Eyal De Lara, Weisong Shi, and Christopher Stewart	A Survey on Edge Performance Benchmarking	Article	excluded	not directly related to healthcare	2021	https://doi.org/10.1145/3444692
ACM Digital		Guoliang Zhao, Safwat Hassan, Ying Zou, Derek	Predicting Performance Anomalies in Software Systems at			keyword mismatch: not directly related		
Library	24.06.2024	Truong, and Toby Corbin Horst Treiblmaier, Melanie Swan, Primavera de Filippi Mary Lecity, Thomas Harringo, and Henry	Run-time	Article	excluded	to healthcare keyword mismatch: not directly related	2021	https://doi.org/10.1145/3440757
Library	23.06.2024	Kim Blandino S.; Ropitault T.; Caromi R.; Chakareski J.;	Topics Using a Multidisciplinary Perspective Head Rotation Model for Virtual Reality System Level	Article Conference	excluded	to healthcare	2021 2021	https://doi.org/10.1145/3447934.3447938 https://www.scopus.com/inward/record.uri?eid=2-
Scopus	06.12.2024	Khan M.; Golmie N.	Simulations	paper	excluded	not directly related to healthcare		52.0-651250184298d01=10.1109%2058752913. 2021.00016 &partnerID=40&md5=911e2518fbc6e8c77763d66 711b24d47
0	00 40 0004	Shrestha S.L.; Chowdhury S.A.; Csallner C.	SLNET: A Redistributable Corpus of 3rd-party Simulink Models	Conference paper		keyword mismatch: not directly related	2022	https://www.scopus.com/inward/record.uri?eid=2- s2.0-85134001801&doi=10.1145%2/3524842. 3528001&partner/D=40&md5=89d4856398d614d
ACM Digital	06.12.2024	Celestine Iwendi, Saif Ur Rehman, Abdul Rehman	Sustainable Security for the Internet of Thinns Using Artificial		excluded	keyword mismatch: not directly related		04071451408087458
Library	23.06.2024	Javed, Suleman Khan, and Gautam Srivastava	Intelligence Architectures	Article	excluded	to healthcare keyword mismatch:	2021	https://doi.org/10.1145/3448614
ACM Digital Library	23.06.2024	Daniel Russo and Klaas-Jan Stol	PLS-SEM for Software Engineering Research: An Introduction and Survey	Article	excluded	not directly related to healthcare	2021	https://doi.org/10.1145/3447580
ACM Digital Library	24.06.2024	Pratima Sharma, Rajni Jindal, and Malaya Dutta Borah	Blockchain Technology for Cloud Storage: A Systematic Literature Review	Article	excluded	not directly related to healthcare	2020	https://doi.org/10.1145/3403954
		Ruiker A.; Mohammed S.; Kumar P.; Chavan R.; Jadhav S.	CFD Modelling of Contaminant Control in Designing Ventilation System for an Airborne Infection Isolation Room (AIIR)	Conference paper		keyword mismatch:	2022	https://www.scopus.com/inward/record.uri?eid=2- s2.0-85147797601&doi=10.26868%2f25222708, 2021.30969
Scopus	05.12.2024	Zhang J.; Blandino S.; Varshney N.; Wang J.;	Multi-User MIMO Enabled Virtual Reality in IEEE 802.11ay	Conference	excluded	to healthcare	2022	&partnerID=40&md5=745ce0f2b2bd2b6a644c772 b9b427e2b https://www.scopus.com/inward/record.uri?eid=2-
Scopus	06.12.2024	Gentile C.; Golmie N.	WLAN	paper	excluded	keyword mismatch: not directly related to healthcare		ac.u-62130/4520/Xd0i=10.1109% 2fWCNC51071.2022.9771778 &partnerID=40&md5=d8622d58e9179f35d7166de 2f8aeab31
ACM Digital	00.1212.024	Ilya Verenich, Marlon Dumas, Marcello La Rosa,	Survey and Cross-benchmark Comparison of Remaining Time	Article		keyword mismatch: not directly related		
Library	24.06.2024	Fabrizio Maria Maggi, and Irene Teinemaa	Prediction Methods in Business Process Monitoring		excluded	to healthcare	2019	https://doi.org/10.1145/3331449

Search	Query date	Author	Title	Type	Status	Reason	Year	Article URL
Vende	autry auto	Igbal Alam, Kashif Sharif, Fan Li, Zohaib Latif, M.	140	Article	otatas	keyword mismatch:	Tour	ANDIO ORE
ACM Digital		M. Karim, Sujit Biswas, Boubakr Nour, and Yu	A Survey of Network Virtualization Techniques for Internet of			not directly related		
Library	23.06.2024	Wang	Things Using SDN and NFV		excluded	to healthcare	2020	https://doi.org/10.1145/3379444
		Chavan S.; Yu X.; Sanile J.	High Precision Analog Gauge Reader Using Optical Flow and	Conference		keyword mismatch:	2022	https://www.scopus.com/inward/record.uri?eid=2- s2.0-85134332968&doi=10.1109%2felT53891.
			Computer vision	paper		not directly related		2022 9813954 8 pathod D=408 md5=c2190b22d7f=06cbc0ccbd1
Scopus	06.12.2024				excluded	to healthcare		a70522040
		Cankar M.; Petrovic N.; Pita Costa J.; Cernivec A.;	Security in DevSecOps: Applying Tools and Machine Learning	Conference		keyword mismatch:	2023	https://www.scopus.com/inward/record.uri?eid=2- s2.0.851589124618doi=10.1145%2f3578245
		Antic J.; Martincic T.; Stepec D.	to Verification and Monitoring Steps	paper	and a dealer of	not directly related		3584943&partnerID=40&md5=a4ddc68414b1f959
Scopus	06.12.2024			Articlo	excluded	to healthcare		a832a1cf1ec90de8
ACM Digital		Ana Juan Ferrer Joan Manuel Marquès and Josen	Towards the Decentralized Cloud: Survey on Approaches and	Article		not directly related		
Library	24.06.2024	Jorba	Challenges for Mobile. Ad hoc. and Edge Computing		excluded	to healthcare	2019	https://doi.org/10.1145/3243929
,		Abadeer M.; Sabetzadeh M.	Machine Learning-based Estimation of Story Points in Agile	Conference			2021	https://www.scopus.com/inward/record.uri?eid=2-
			Development: Industrial Experience and Lessons Learned	paper		keyword mismatch:		s2.0-85118461307&doi=10.1109%2tREW53955. 2021.00022
Sconus	06 12 2024				excluded	to healthcare		&partnerID=40&md5=3c8f05991b322e5b9f7f6c68
000000	00.12.2024			Article	excluded	keyword mismatch:		30204117
ACM Digital		Farah Aït Salaht, Frédéric Desprez, and Adrien	An Overview of Service Placement Problem in Fog and Edge			not directly related		
Library	24.06.2024	Lebre	Computing		excluded	to healthcare	2020	https://doi.org/10.1145/3391196
				Article		keyword mismatch:		
ACM Digital	24.06.2024	Carlo Puliafito, Enzo Mingozzi, Francesco Longo, Antonio Puliafito, and Omer Pana	For Computing for the Internet of Things: A Super-		ovoludod	not directly related	2010	
Library	24.00.2024	Charith Perera, Yongrui Oin, Julio C, Estrella	Fog Computing for the Internet of Things: A Survey		excluded	keyword mismatch:	2019	https://doi.org/10.1145/3301443
ACM Digital		Stephan Reiff-Marganiec, and Athanasios V.				not directly related		
Library	24.06.2024	Vasilakos	Fog Computing for Sustainable Smart Cities: A Survey	Article	excluded	to healthcare	2017	https://doi.org/10.1145/3057266
				Article		keyword mismatch:		
ACM Digital		Miltiadis Allamanis, Earl T. Barr, Premkumar				not directly related		
Library	23.06.2024	Sved Wasif Abbas Hamdani, Haider Abbas, Abdul	A Survey of Machine Learning for Big Code and Naturalness		excluded	to healthcare	2018	https://doi.org/10.1145/3212695
		Behman Janiua, Waleed Bin Shahid, Muhammad						
		Faisal Amjad, Jahanzaib Malik, Malik Hamza				keyword mismatch:		
ACM Digital		Murtaza, Mohammed Atiquzzaman, and Abdul	Cybersecurity Standards in the Context of Operating System:			not directly related		
Library	24.06.2024	Waheed Khan	Practical Aspects, Analysis, and Comparisons	Article	excluded	to healthcare	2021	https://doi.org/10.1145/3442480
						keyword mismatch:		
ACM Digital	22.05.2024	Peter Norvig, Alfred Spector, Chris Wiggins, Jeappette M Wing, Rep Eried, and Michael Tinglov.	More Than Just Algorithms	Article	excluded	to OS	2022	https://doi.org/10.4145/0000000
Library	23.03.2024	Seamene w. wing, ben neu, and wichaer nigley	More man sust Algonitims	Anioe	excluded	keyword mismatch:	2023	https://doi.org/10.1145/3600099
ACM Digital						not directly related		
Library	22.06.2024	Peter G. Neumann	Risks to the Public	Article	excluded	to OS	2023	https://doi.org/10.1145/3573074.3573077
						keyword mismatch:		
ACM Digital						not directly related		
Library	28.05.2024	Kory Kraft	Hobots Against Infectious Diseases	Article	excluded	to OS	2016	https://dl.acm.org/doi/10.5555/2906831.2907014
ACM Digital		Paolo Manohi Leonardo Candela, Emma Lazzeri				not directly related		
Library	21.06.2024	and Gianmaria Silvello	Digital Libraries: Supporting Open Science	Article	excluded	to OS	2020	https://doi.org/10.1145/3385658.3385669
2.010.7			Harmonizing Sensitive Data Exchange and Double-spending			keyword mismatch:		The contract of the contract o
ACM Digital		Vincent Schlatt, Johannes Sedlmeir, Janina Traue,	Prevention Through Blockchain and Digital Wallets: The Case			not directly related		
Library	21.06.2024	and Fabiane Völter	of E-prescription Management	Article	excluded	to OS	2023	https://doi.org/10.1145/3571509
						keyword mismatch:		
ACM Digital	23.05.2024	Rashmeet Toor and Inderveer Chana	Application of IT in healthcare: a systematic review	Article	evoluded	to OS	2016	https://doi.org/10.1145/2092212.2092215
clorary	20.00.2024	Rashineet foor and inderveer onand	Application of thin inclaimedic, a systematic review	Autore	CACIDACA	keyword mismatch:	2010	https://doi.org/10.1145/2505515.2505515
ACM Digital			Piracy and the Impaired Cyborg: Assistive Technologies,			not directly related		
Library	21.06.2024	Priyank Chandra	Accessibility, and Access	Article	excluded	to OS	2021	https://doi.org/10.1145/3432941
						keyword mismatch:		
ACM Digital	21.06.2024	Eman Badr	Images in Space and Time: Peal Pig Data in Healthcare	Articlo	ovoludod	to OS	2021	http://doi.org/10.4445/0450057
Library	21.00.2024	Eman badi	English Space and Time. Real big Data in Healthcare	Anicie	excluded	keyword mismatch:	2021	https://doi.org/10.1145/3453657
ACM Digital		Gautam Srivastava, Jerry Chun-Wei Lin, Calton Pu,	Architectures Challenges & amp: Future Research Directions"			not directly related		
Library	22.06.2024	and Yudong Zhang	ACM Trans	Article	excluded	to OS	2023	https://doi.org/10.1145/3627990
		Udaya Lakshmi, Megan Hofmann, Stephanie				keyword mismatch:		
ACM Digital		Valencia, Lauren Wilcox, Jennifer Mankoff, and	"Point-of-Care Manufacturing": Maker Perspectives on Digital			not directly related	0040	
Library	24.10.2024	Rosa I. Arriaga Gong S.: Ochiai H.: Esaki H	Fabrication in Medical Practice Scan-Based Self Anomaly Detection: Client-Side Mitigation of	Conference	excluded	10 US	2019	https://doi.org/10.1145/3359193 https://www.scopus.com/inward/record.uri?eid=2-
		Song S., Schlarn, Esakin.	Channel-Based Man-in-the-Middle Attacks against Wi-Fi	paper		keyword mismatch:	2020	s2.0-85094168544&doi=10.1109%
						not directly related		2RCOMPSAC48688.2020.00- 43&partneriD=40&md5=51e8901638b0e5da051d
Scopus	06.12.2024	Longia D. Wester, J. Oblidence A. Educid F.	O de sent Audio (estas estas directos de sentes de	0	excluded	to healthcare	0000	a9c40737dd5d
		Lenain R.; Weston J.; Snivkumar A.; Fristed E.	Surboard: Audio feature extraction for modern machine	Conterence		keyword mismatch:	2020	s2.0-85098196470&doi=10.21437%2finterspeech.
			loanning	paper		not directly related		2020- 2879&partner/D=40&md5=26111db989a9942102
Scopus	06.12.2024				excluded	to healthcare		45db5012ada238
						keyword mismatch:		
ACM Digital	21.06.2024	Supanno Badkar and Marton Kuna	Participaton, Design that Mattern - Easing the Big Jacuas	Article	oveluded	not directly related	2019	
Library	21.00.2024	Chepkoech M.: Mombeshora N.: Malila B.:	Evaluation of Open-Source Mobile Network Software Stacks: A	Conference	excluded	10 03	2018	https://doi.org/10.1145/3152421 https://www.scopus.com/inward/record.uri?eid=2-
		Mwangama J.	Guide to Low-cost Deployment of 5G Testbeds	paper		keyword mismatch:		s2.0-85150682482&doi=10.23919%
-						not directly related		&partnerID=40&md5=290f258d15f967461db9a81
Scopus	06.12.2024	Teampoulatidie I : Kompinge N : Surmos E :	Universality and Interonerability Across Smart City Ecosystems	Conference	excluded	to nealthcare	2022	98521c25f https://www.scopus.com/inward/record.uri?eid=2-
		Bechtsis D.	oniversancy and interoperability Across offair only Ecosystems	paper		keyword mismatch:	2022	s2.0-85133287174&doi=10.1007%2/978-3-031-
						not directly related		1_16&partnerID=40&md5=7b439a0afa27fdbd51a
Scopus	06.12.2024				excluded	to healthcare		4f7b816134cec
ACM Digital		Teung Ting Kup, Hugo Zavalata Boigo, Lucilo	Comparison of blockchain platforma: a custometic review			not directly related		
Library	13.05.2024	Ohno-Machado	healthcare examples	Article	excluded	to OS	2019	https://academic.oup. com/jamia/article/26/5/462/54193217/poin=true
ACM Digital		Junaid Qadir, Arjuna Sathiaseelan, Liang Wang.	"Resource Pooling" for Wireless Networks: Solutions for the		and and	type: not peer		A REAL PROPERTY OF THE REAL PROPERTY OF
Library	24.10.2024	and Jon Crowcroft	Developing World	Article	excluded	reviewed	2016	https://doi.org/10.1145/3027947.3027953
		Gao R.; Jiang Z.M.	An Exploratory Study on Assessing the Impact of Environment	Conference		keyword miematob:	2017	https://www.scopus.com/inward/record.uri?eid=2- =2.0.850265319328doi=10.1100%20MSR
			variations on the Results of Load Tests	paper		not directly related		2017.22
Scopus	05.12.2024				excluded	to healthcare		<u>ApartneriD=40&amp;md5=03a200e6abfb877aa227471</u> d93e2091b
ACM Digital		Luiz Alexandre Costa, Awdren Fontão, and	Investigating Proprietary Software Ecosystem Governance and	Conference				
Library	11.11.2024	Rodrigo Santos	Health: An Updated and Refined Perspective	paper	excluded	focus not on OS	2021	https://doi.org/10.1145/3466933.3466955
		B. Salami, K. Parasyris, A. Cristal, O. Unsal, X.						
		Martorell, P. Carpenter, R. De La Cruz, L. Bautista,						
		Pericas, P. Trancoso, M. Abduliabbar, J. Chen, P.						
		N. Soomro, M Manivannan, M. Berge, S. Krupop,						
		F. Klawonn, Al Mekhlafi, S. May, T. Becker, G.						
		Gaydadjiev, H. Salomonsson, D. Dubhashi, O.						
		Kaiser N. Kucza J. Hagemover P. Griegel J.						
ACM Digital		Tigges, K. Mika, A. Hüffmeier, M. Pasin V	LEGaTO: low-energy, secure, and resilient toolset for	Conference				
Library	22.06.2024	Schiavoni, I. Rocha, C. Göttel, and P. Felber	heterogeneous computing	paper	excluded	focus not on OS	2020	https://ieeexplore.ieee.org/document/9116362
		Bombarda A.; Bonfanti S.; Gargantini A.; Radavelli	Combining Model Refinement and Test Generation for	Conference		Instantian and and and and and and and and and a	2019	https://www.scopus.com/inward/record.uri?eid=2-
		M.; Duan F.; Lei Y.	Conformance Testing of the IEEE PHD Protocol Using Abstract	paper		not directly related		31280-
Scopus	05.12.2024		State MdGIIIIes		excluded	to healthcare		0_5&partnerID=40&md5=7213798343659fe8b695 a7378240b46d
		Esther Han Beol Jang, Nicola Bidwell, Jen Liu.						
		Phoebe Sengers, Naveen Bagalkot, Nervo				keyword mismatch:		
ACM Digital		Verdezoto, Melissa Densmore, Morgan Vigil-	Situating Network Infrastructure with People, Practices, and	Conference		not directly related		
Library	30.11.2024	Hayes, and Shaddi Hasan	Beyond: A Community Building Workshop	paper	excluded	to healthcare	2022	https://doi.org/10.1145/3500868.3560716
ACM Diates			An MPD Player with Expert Knowledge-basedSingle Lloss	paper		not directly related		
Library	04.11.2024	Hen-Hsen Huang	Music Recommendation	· · · · · ·	excluded	to healthcare	2019	https://doi.org/10.1145/3358695.3360919
,		-		Conference		keyword mismatch:		
ACM Digital				paper		not directly related		
Library	30.11.2024	Albert Bifet and Jesse Read	Ubiquitous Artificial Intelligence and Dynamic Data Streams		excluded	to healthcare	2018	https://doi.org/10.1145/3210284.3214345

Search venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
		Huang H.; Dominguez-Kuhne M.; Satish V.; Dapielozuk M.; Sanders K.; Johngwski, J.; Leo A.;	Mechanical Search on Shelves using Lateral Access X-RAY	Conference		keyword mismatch:	2021	https://www.scopus.com/inward/record.uri?eid=2- s2.0-85123704338&doi=10.1109%2fIROS51168.
_		Angelova A.; Vanhoucke V.; Goldberg K.		paper		not directly related		2021.9636629 &partnerID=40&md5=f3b3d611ae2450481395c76
Scopus	06.12.2024			Conference	excluded	to healthcare keyword mismatch:		9e2c60807
ACM Digital		Johanna Plattner, Elena Oberrauner, Daniela	Using IoT middleware solutions in interdisciplinary research	paper		not directly related		
Library	30.11.2024	Elisabeth Ströckl, and Johannes Oberzaucher	projects in the context of AAL	Conference	excluded	to healthcare keyword mismatch:	2020	https://doi.org/10.1145/3389189.3397986
ACM Digital		Stephen McCarthy, Paidi O'Raghallaigh, Ciara	Impacts of Openness on the Success of Information System	paper		not directly related		
Library	10.11.2024	Fitzgerald, and Frédéric Adam Ge S : Isah H : Zulkernine F : Khan S	Development Research Projects A scalable framework for multilevel streaming data analytics	Conference	excluded	to healthcare	2017	https://doi.org/10.1145/3126673.3126675 https://www.scopus.com/inward/record.uri?eid=2-
		,	using deep learning	paper		keyword mismatch:		s2.0-85072677284&doi=10.1109%2fCOMPSAC. 2019.10205
Scopus	05.12.2024				excluded	to healthcare		&partneriD=40&md5=3c7b971f5300f84747d34cc a43e9f1c3
		Carrott Elvan, Jackus Provistar, Dana Sana, and		Conference		keyword mismatch:		
Library	08.11.2024	Marientina Gotsis	Developing Brain-Computer Interfaces with Everyone	paper	excluded	to healthcare	2023	https://doi.org/10.1145/3594806.3594810
		Christen Mikelana Nikelana Bakalan and Dimitrian	Towards Association the Adaption of Federated Learning for	Conference		keyword mismatch:		
ACM Digital Library	30.11.2024	Kalogeras	Heterogeneous Data	hahei	excluded	to healthcare	2023	https://doi.org/10.1145/3594806.3596568
		Won Kim, Ahyoung Choi, Jaehyuk Choi, Ok-Ran	-	Conference				
		Jeong, Yong Ju Jung, Sangwoo Kang, Joohyung Lee, Sang-Woong Lee, Woong-Kee Loh, Joon		paper				
		Yoo, Seongcheol Chung, Hyungchul Lee, Sungmi				to a second and second allow		
ACM Digital		Kim, Jung-Hun Lee, Youna Min, Geun Sil Song,				not directly related		
Library	11.11.2024	and Sun Ok Yang	Korea's software education initiative	0	excluded	to healthcare	2017	https://doi.org/10.1145/3151759.3151800
		Saini S.; Punnani R.; Bathia R.; Shukia V.K.	Sentiment Analysis on Twitter Data using R	paper		keyword mismatch:	2019	s2.0-85070656093&doi=10.1109%2fiCACTM. 2019.8776685
Sconus	06 12 2024				evoluded	not directly related to healthcare		&partnerID=40&md5=dfbbaae9a6c426c984e3491 22faadcba
ocopus	00.12.2024		Engaging with Immersive Technologies in Medical Library: An	Conference	excluded	keyword mismatch:		2210033300
ACM Digital	09 11 2024	Yuxiang Chris Zhao, Jingwen Lian, Shijie Song, and Jun Ying	Investigation of Students' Perceived Affordances and Constraints	paper	evoluded	not directly related	2022	https://doi.org/10.1145/0409368.3505937
Library	05.11.2024	and our ring	onatrainta	Conference	excluded	keyword mismatch:	2022	ntips://doi.org/10.1145/3496306.3505627
ACM Digital	20 11 2024	Winawaa Navam and Yachai Limpiyakom	XGBoost for Classifying Ethereum Short-term Return Based	paper	ovoluded	not directly related	2022	
Library	30.11.2024	Park D.; Lee J.; Lee J.; Lee K.	Deep learning based food instance segmentation using	Conference	excluded	to nearricare	2023	https://doi.org/10.1145/3605423.3605462 https://www.scopus.com/inward/record.uri?eid=2-
			synthetic data	paper		keyword mismatch: not directly related		s2.0-851124204198doi=10.1109%2/UR52253. 2021.9494704
Scopus	05.12.2024				excluded	to healthcare		ApartneriU=40Amd5=ccca182d66ba8ae780eff4d 76fa5e56a
		Moore S.; Armstrong P.; McDonald T.; Yampolskiy M.	Vulnerability analysis of desktop 3D printer software	Conference paper		keyword mismatch:	2016	https://www.scopus.com/inward/record.uri?eid=2- s2.0-84991800699&doi=10.1109%2fRWEEK.
Coopus	06 12 2024				ovoludod	not directly related		2016.7573305 &partneriD=40&md5=7cd02871c8b84ff162aa21a
Scopus	00.12.2024			Conference	excluded	keyword mismatch:		090084801
ACM Digital	00.44.0004	Conner Bradley and David Barrara	Escaping Vendor Mortality: A New Paradigm for Extending IoT	paper	a such solar al	not directly related	0000	
Library	09.11.2024	Panwar A.; Bhatnagar V.	Distributed ledger technology (DLT): The beginning of a	Conference	excluded	to nearthcare	2023	https://doi.org/10.1145/3633500.3633501 https://www.scopus.com/inward/record.uri?eid=2-
			technological revolution for blockchain	paper		keyword mismatch: not directly related		s2.0-85091429352&doi=10.1109%2HDEA49133. 2020.9170699
Scopus	06.12.2024				excluded	to healthcare		ApartneriU=40&md5=3578113588000a74bed7508 b99c68f5e
ACM Digital		Juho Jung, Chaewon Kang, Jeewoo Yoon, Simon	SAFE: Sequential Attentive Face Embedding with Contrastive	Conference paper		keyword mismatch: not directly related		
Library	30.11.2024	S. Woo, and Jinyoung Han	Learning for Deepfake Video Detection	p - p	excluded	to healthcare	2023	https://doi.org/10.1145/3583780.3615279
ACM Digital		Dimitris Stripelis, Chrysovalantis Anastasiou, Patrick Toral, Armaghan Asghar, and José Luis	MetisEL: An Embarrassingly Parallelized Controller for	Conference paper		keyword mismatch: not directly related		
Library	11.11.2024	Ambite	Scalable & amp; Efficient Federated Learning Workflows	P-P	excluded	to healthcare	2023	https://dl.acm.org/doi/10.1145/3630048.3630186
		Zook Butler, lyong Rezékové, and Angeling	Putting a Contact in Contact: Investigating the Contact of	Conference		keyword mismatch:		
Library	29.11.2024	Brilliantova	Pencil Puzzles in Multiple Academic Environments	pape.	excluded	to healthcare	2023	https://doi.org/10.1145/3545945.3569848
		Razvan Raducu, Camino Fernández, Miguel Ángel		Conference		keyword mismatch:		
Library	10.11.2024	Francisco Javier Castaño	Innovation in financial education: ULe-Bank	paper	excluded	to healthcare	2017	https://doi.org/10.1145/3144826.3145432
		Ohen Liv, William W. Ohung Weng, and Online	Disited Dublic Marith Councillance with Online Marith	Conference		keyword mismatch:		
ACM Digital Library	08.11.2024	Khan	Consultation Data: An Example of HIV Monitoring	paper	excluded	to healthcare	2022	https://doi.org/10.1145/3545729.3545748
				Conference		keyword mismatch:		
ACM Digital Library	09.11.2024	Victor Asanza	Evaluation of ML-based Positioning Systems with HSSI Measured on the User's Device or APs	paper	excluded	to healthcare	2023	https://doi.org/10.1145/3616394.3618261
		Martin Kilanda Annala Ormania Martinia		Conference		keyword mismatch:		
ACM Digital Library	30.11.2024	Victor Kristof, Aswin Suresh, Matthias Grossglauser, and Patrick Thiran	War of Words II: Enriched Models of Law-Making Processes	paper	excluded	to healthcare	2021	https://doi.org/10.1145/3442381.3450131
		Fu J.; Norouzi M.; Nachum O.; Tucker G.; Wang Z.;	BENCHMARKS FOR DEEP OFF-POLICY EVALUATION	Conference		keyword mismatch:	2021	https://www.scopus.com/inward/record.uri?eid=2- e2.0-
Scopus	05.12.2024	A.; Paduraru C.; Levine S.; Le Paine T.		paper	excluded	not directly related to healthcare		85105982739&partnerID=40&md5=1377c17869a 8abf24b499218aa475ee9
						keyword mismatch:		
Springer	07 12 2024	Mattis Jacobs Judith Simon	Assigning Obligations in Al Regulation: A Discussion of Two Frameworks Proposed By the European Commission	Article	excluded	not directly related to healthcare	2022	http://link.springer.com/article/10.1007/s44206-
opinigoi	OTTELEDE.				CACINGCO	keyword mismatch:	Long	<u>522-000572</u>
Springer	07 12 2024	Matti Minkkinen loakim LaineMatti Mäntymäki	Continuous Auditing of Artificial Intelligence: a Concentualization and Assessment of Tools and Frameworks	Article	evoluded	not directly related	2022	http://link.springer.com/article/10.1007/s44206-
opinger	07.12.2024	Water Ministerior Docater in California to Marky mark	Conceptualization and Accessment of Yools and Hameworks	741010	GACIDUEU	keyword mismatch:	LOLL	022-00022-2
Springer	06 12 2024	Rustam ShadieuChuanwan Dang	A systematic review study on integrating technology-assisted	Article	ovoludod	not directly related	2022	http://link.springer.com/article/10.1007/s10639-
Springer	00.12.2024	Hustam Shadievondariwen Dang	Intercultural learning in various learning context	Article	excluded	keyword mismatch:	2022	021-10877-6
Cariana	06 40 0004	Shah I Miahlan Salamanidaa Jahn C. Cammaak	A design-based research approach for developing data-	Article	a valuate d	not directly related	2020	http://link.springer.com/article/10.1007/s10639-
Springer	00.12.2024	Shari J. Mianian SolomonidesJohn G. Ganmack	locussed business curricula	Article	excluded	keyword mismatch:	2020	019-09981-5
0	07 10 0001	Elias Melchor-FerrerMaria Angeles Davia-	Computer games and the study of terminology: An application	Autola		not directly related	0000	http://link.springer.com/article/10.1007/s10639-
Springer	07.12.2024	Rodriguez	to national accounts	Article	excluded	keyword mismatch:	2023	022-11138-w
			Design requirements of a modern business Master's degree			not directly related		http://link.springer.com/article/10.1007/s10639-
Springer	07.12.2024	Shah J Miahlan Solomonides	course: perspectives of industry practitioners	Article	excluded	to healthcare keyword mismatch:	2021	020-10285-2
			HELA: A novel hybrid ensemble learning algorithm for			not directly related		http://link.springer.com/article/10.1007/s10639-
Springer	08.12.2024	Sinem Bozkurt KeserSevda Aghalarova	predicting academic performance of students	Article	excluded	to healthcare keyword mismatch:	2022	021-10780-0
			Interactive notebooks for achieving learning outcomes in a			not directly related		http://link.springer.com/article/10.1007/s10639-
Springer	08.12.2024	Ghada AmoudiDina Tbaishat	graduate course: a pedagogical approach	Article	excluded	to healthcare keyword mismatch:	2023	023-11854-x
			Investigating the structural relationship for the determinants of			not directly related		http://link.springer.com/article/10.1007/s10639-
Springer	08.12.2024	Veera BhatiaseviMichael Naglis	cloud computing adoption in education	Article	excluded	to healthcare	2016	015-9376-6
						not directly related		http://ink.springer.com/article/10 1007/s10630.
Springer	08.12.2024	Sadhna Shukla	M-learning adoption of management students': A case of India	Article	excluded	to healthcare	2021	020-10271-8
		Juan Manuel FlorezLaura MorenoZenong				not directly related		http://ink.springer.com/article/10-1007/e10864-
Springer	07.12.2024	ZhangShiyi WeiAndrian Marcus	An empirical study of data constraint implementations in Java	Article	excluded	to healthcare	2022	022-10175-w
		N. C. ShrikanthWilliam NicholsFahmid Morshed				not directly related		http://link.springer.com/article/10 1007/+10664
Springer	07.12.2024	FahidTim Menzies	Assessing practitioner beliefs about software engineering	Article	excluded	to healthcare	2021	021-09957-5
		Waleed ZogaanIbrahim MujhidJoanna C. S.	Automated training-set creation for software architecture			not directly related		http://link.springer.com/article/10-4007/+4088+
Springer	07.12.2024	SantosDanielle GonzalezMehdi Mirakhorli	traceability problem	Article	excluded	to healthcare	2017	016-9476-y

venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
		Suppawong TuarobNoppadol AssavakamhaenghanWaralee TanaphantarukPonlakit SuwanworaboonSaeed-Ul	Automatic team recommendation for collaborative software			keyword mismatch: not directly related		http://link.springer.com/article/10.1007/s10664-
Springer	07.12.2024	HassanMorakot Choetkiertikul	development	Article	excluded	to healthcare keyword mismatch:	2021	021-09966-4
Springer	07.12.2024	KhomhZhen Ming (Jack) Jiang Edson OliveiraEduardo FernandesIgor	bugs in machine learning-based systems, a radiidad benchmark	Article	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/article/10.1007/s10664- 023-10291-1
Springer	07.12.2024	SteinmacherMarco Cristo Iayana ConteAlessandro Garcia	code and commit metrics of developer productivity: a study on team leaders perceptions	Article	excluded	to healthcare keyword mismatch:	2020	http://link.springer.com/article/10.1007/s10664- 020-09820-z
Springer	07.12.2024	Yisi HanZhendong WangYang FengZhihong ZhaoYi Wang	Cross-status communication and project outcomes in OSS development	Article	excluded	not directly related to healthcare keyword mismatch:	2023	http://link.springer.com/article/10.1007/s10664- 023-10298-8
Springer	07.12.2024	Damiano TorreFrantzy MesadieuAnitha Chennamaneni	Deep learning techniques to detect cybersecurity attacks: a systematic mapping study	Article	excluded	not directly related to healthcare	2023	http://link.springer.com/article/10.1007/s10664- 023-10302-1
Springer	07.12.2024	Sarah ElderNusrat ZahanRui ShuMonica MetroValeri KozarevTim MenziesLaurie Williams	Do I really need all this work to find vulnerabilities?	Article	excluded	not directly related to healthcare keyword mismatch:	2022	http://link.springer.com/article/10.1007/s10664- 022-10179-6
Springer	08.12.2024	Nicolas E. GoldJens Krinke	Ethics in the mining of software repositories	Article	excluded	not directly related to healthcare	2021	http://link.springer.com/article/10.1007/s10664- 021-10057-7
Springer	08.12.2024	Mika V. MäntyläMagne JørgensenPaul RalphHakan Erdogmus	Guest editorial for special section on success and failure in software engineering	Article	excluded	not directly related to healthcare	2017	http://link.springer.com/article/10.1007/s10664- 017-9505-5
Springer	08.12.2024	Zainab MasoodRashina HodaKelly Blincoe	How agile teams make self-assignment work: a grounded theory study	Article	excluded	keyword mismatch: not directly related to healthcare	2020	http://link.springer.com/article/10.1007/s10664- 020-09676-x
Springer	08.12.2024	Önder BaburEleni ConstantinouAlexander Serebrenik	Language usage analysis for EMF metamodels on GitHub	Article	excluded	keyword mismatch: not directly related to healthcare	2023	http://link.springer.com/article/10.1007/s10664- 023-10368-x
opinigoi		Christian BirchlerSajad KhatiriBill BosshardAlessio	Machine learning-based test selection for simulation-based			keyword mismatch: not directly related		http://link.springer.com/article/10.1007/s10664-
Springer	08.12.2024	GambiSebastiano Panichella Jiachi ChenXin XiaDavid LoJohn GrundyXiaohu	testing of self-driving cars software Maintenance-related concerns for post-deployed Ethereum smart contract development: issues, techniques, and future	Article	excluded	to healthcare keyword mismatch: not directly related	2023	023-10286-y
Springer	08.12.2024	Yang	challenges	Article	excluded	to healthcare keyword mismatch:	2021	021-10018-0
Springer	08.12.2024	Ying ShiMing LiSteven ArndtCarol Smidts	application	Article	excluded	to healthcare keyword mismatch:	2017	http://link.springer.com/article/10.1007/s10664- 016-9425-9
Springer	08.12.2024	Jacob KrügerChristian Lausbergerlvonne von Nostitz-WallwitzGunter SaakeThomas Leich	Search. Review. Repeat? An empirical study of threats to replicating SLR searches	Article	excluded	not directly related to healthcare keyword mismatch:	2020	http://link.springer.com/article/10.1007/s10664- 019-09763-0
Springer	08.12.2024	Mazen MohamadJan-Philipp SteghöferRiccardo Scandariato	Security assurance cases-state of the art of an emerging approach	Article	excluded	not directly related to healthcare	2021	http://link.springer.com/article/10.1007/s10664- 021-09971-7
Springer	08.12.2024	Zoe KottiKonstantinos KravvaritisKonstantina DritsaDiomidis Spinellis	Standing on shoulders or feet? An extended study on the usage of the MSR data papers	Article	excluded	not directly related to healthcare	2020	http://link.springer.com/article/10.1007/s10664- 020-09834-7
Springer	08.12.2024	Anh Nguyen-DucKai-Kristian KemellPekka Abrahamsson	The entrepreneurial logic of startup software development: A study of 40 software startups	Article	excluded	keyword mismatch: not directly related to healthcare keyword mismatch:	2021	http://link.springer.com/article/10.1007/s10664- 021-09987-z
Springer	07.12.2024	Marcello lenca	Democratizing cognitive technology: a proactive approach	Article	excluded	not directly related to healthcare	2019	http://link.springer.com/article/10.1007/s10676- 018-9453-9
Springer	07.12.2024	Chengliang WangYayun PengDebraj DeWen-Zhan Song	on activity pattern knowledge mined from trajectories in smart environments	Article	excluded	not directly related to healthcare	2016	http://link.springer.com/article/10.1007/s11704- 015-4571-6
Springer	08.12.2024	Edje E. AbelMuhammad Shafie Abd Latiff	The utilization of algorithms for cloud internet of things application domains: a review	Article	excluded	keyword mismatch: not directly related to healthcare	2021	http://link.springer.com/article/10.1007/s11704- 019-9056-6
Springer	06.12.2024	Shan-zhi ChenShao-li Kang	A tutorial on 5G and the progress in China	Article	excluded	not directly related to healthcare	2018	http://link.springer.com/article/10.1631/FITEE. 1800070
Springer	08 12 2024	Fabio ValdésBalf Hartmut Güting	Index-supported pattern matching on tuples of time- dependent values	Article	excluded	keyword mismatch: not directly related to healthcare	2017	http://link.springer.com/article/10.1007/s10707- 016-028-6
		S. Ansari; F. Ali; M. Chaudhary; K. Varshney	A Survey on Smart City Surveillance System	Conference		keyword mismatch: not directly related	2023	10.1109/ICNTE56631.2023.10146671
IEEE Xplore	30.11.2024	F. Amrollahi; S. P. Shashikumar; P. Kathiravelu; A. Sharma; S. Nemati	AIDEx - An Open-source Platform for Real-Time Forecasting Sepsis and A Case Study on Taking ML Algorithms to	Conference	excluded	keyword mismatch: not directly related	2020	10.1109/EMBC44109.2020.9175947
IEEE Xplore	30.11.2024	DT. Le; LS. Lê; N. Thoai; TV. Nguyen	An Old Problem with a New Therapy: Coupling Topic Modeling and Mining Sequential Patterns in Recommending Source	paper Conference	excluded	to healthcare keyword mismatch: not directly related	2018	10.1109/ACOMP.2018.00026
IEEE Xplore	30.11.2024	T. Okumura; S. Takeuchi; K. Chubachi	Code An Open-Source FaxOCR for Emergency Communication	paper	excluded	to healthcare keyword mismatch:	2018	10.1109/ICEIEC.2018.8473570
IEEE Xplore	30.11.2024	R. W. K. Boon; Y. Z. Chong; Y. Q. Tan; V. Sundar;	Effect of Running Shoe Cushioning on Muscle Activation using	paper	excluded	to healthcare keyword mismatch:	2022	10.1109/IECBES54088.2022.10079302
IEEE Xplore	05.12.2024	Y. Selva; S. C. Chan M. Xie; M. F. Awan; J. F. Pajo; A. Cherifi; E. Lajoie;	Opensim	Conference paper	excluded	not directly related to healthcare keyword mismatch:	2023	10.1109/GCWkshps58843.2023.10464493
IEEE Xplore	05.12.2024	J. Gill	End 5G Services	Conference paper	excluded	not directly related to healthcare	2024	10 1100/NSS/MIC44867 2021 9875880
IEEE Xplore	05.12.2024	H. Su; C. W. Stearns; R. Twyman; E. Ovtchinnikov; K. Thielemans	Recent Flogress III 3 HR 3.0	Conference paper	excluded	not directly related to healthcare	2021	
IEEE Xplore	05.12.2024	B. Gao; B. Yang; K. Guo; T. Chen; M. Zheng	Research on Vehice Motion Control Based on Electromyography Signal Collected and Processed by MCU and Computer Platform	Conference paper	excluded	keyword mismatch: not directly related to healthcare	2021	10.1109/ICMA52036.2021.9512774
IEEE Xplore	05 12 2024	E. Abu-Shanab; F. Estatiya	Utilizing Cloud Computing in public sector cases from the world	Conference	evoluded	keyword mismatch: not directly related	2017	10.1109/ICASI.2017.7988265
ILLE ADOIS	03.12.2024	S. Iwami	A Methodology to Categorize Rapidly Changing Projects of Open Source Software	Conference	excluded	keyword mismatch: not directly related	2022	10.23919/PICMET53225.2022.9882581
IEEE Xplore	30.11.2024	D. B. Adbullah; M. R. Mohammed	A Survey on Real Time Cloud-based Robotics Systems	paper Conference	excluded	to healthcare keyword mismatch: not directly related	2022	10.1109/ICCITM56309.2022.10031975
IEEE Xplore	30.11.2024	P. P. Talan; K. U. Sharma	An overview and an Approach for Graph Data Processing	paper	excluded	to healthcare keyword mismatch:	2018	10.1109/ICCMC.2018.8487681
IEEE Xplore	30.11.2024	O. Aziz; M. S. Farooq; A. Khelifi	Domain and Challenges of Big Data and Archaeological	paper	excluded	to healthcare keyword mismatch:	2022	10.1109/ACCESS.2022.3206805
IEEE Xplore	05.12.2024	B. S. Manoj; Y. Puneeth; S. Yuvaraj	Automated Vehicle Braking System and Driver Health	conference paper	excluded	to healthcare keyword mismatch:	2023	10.1109/OTCON56053.2023.10113976
IEEE Xplore	30.11.2024	P. Rangel: K. Brotherton: E. Anderson: A. Hennad	Monitoring System Using IOT Technology	Conference paper	excluded	not directly related to healthcare keyword mismatch:	2019	10.1109/ISMCR47492.2019.8955724
IEEE Xplore	05.12.2024	A. Vega; M. Plotkin	Open-Source Based Humanoid Socially Assistive Robot	Conference paper	excluded	not directly related to healthcare	0010	10 110081928410 2010 0024544
IEEE Xplore	05.12.2024	n. o. kang; n. iasnima; S. J. Hong; I. Yamaya	Opurnization of a high resolution Small Animal SPECT System using GATE and STIR Software	Conference paper	excluded	not directly related to healthcare	2018	10.110//ff33//f6.2016.0524511

venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
		M. Y. Jung; P. Kazanzides	An architectural approach to safety of component-based	0		keyword mismatch:	2016	10.1109/ICRA.2016.7487511
IEEE Xplore	30.11.2024		TODOLIC SYSTEMS	paper	excluded	to healthcare		
		S. A. C. Yohanandan; C. Perera; M. Jones; R. F.	Objective video-based tremor assessment for movement			keyword mismatch:	2017	10.1109/HIC.2017.8227617
IEEE Valore	05 12 2024	Peppard; T. Perera	disorders using open-source software	Conference	ovoludod	not directly related		
ILLE Aploie	03.12.2024	S. Laato; E. Sutinen	Integrating Theological Aspects to Engineering Education	paper	excluded	keyword mismatch:	2020	10.1109/FIE44824.2020.9274089
	05 40 0004			Conference	a such solar at	not directly related		
IEEE Apiore	05.12.2024	Y. Zhu; C. Detig; S. Kane; G. Lourie	Kinematic Motion Analysis with Volumetric Motion Capture	paper	excluded	keyword mismatch:	2022	10.1109/IV56949.2022.00019
				Conference		not directly related		
IEEE Xplore	05.12.2024	M. Guedev: A. Pfeiffer: D. Uckelmann	Transferring Research on IoT Applications for Smart Buildings	paper	excluded	to healthcare keyword mismatch:	2020	10.1109/EDUCON45650.2020.9125150
		in outday, set follow, b. outdaniant	into Engineering Education	Conference		not directly related	LOLO	
IEEE Xplore	05.12.2024	M Nahian A Noubi V Vin S Octadabbas	A biograph specific processing tool for machine learning and	paper	excluded	to healthcare	2017	10 1109/HIC 2017 8227588
		W. Nabian, A. Nouni, T. Tin, S. Ostadabbas	pattern recognition	Conference		not directly related	2017	
IEEE Xplore	30.11.2024	D. D. Harder, O. Dalahar	hand the Development is a Delegan second second in the Original	paper	excluded	to healthcare	0040	10 1100/01=D-1=47000 2010 0002224
		D. R. Harris; C. Delcher	Big Data Big Data	Conference		keyword mismatch: not directly related	2019	10.1109/BigData47090.2019.9006234
IEEE Xplore	30.11.2024			paper	excluded	to healthcare		
		M. M. Volkova; P. V. Antonova; A. R. Shameeva	High-Performance Complex Event Processing	Conference		keyword mismatch:	2020	10.1109/FarEastCon50210.2020.9271338
IEEE Xplore	05.12.2024			paper	excluded	to healthcare		
		L. T. Mohammed; A. A. AlHabshy; K. A. ElDahshan	Big Data Visualization: A Survey	Conformação		keyword mismatch:	2022	10.1109/HORA55278.2022.9799819
IEEE Xplore	05.12.2024			paper	excluded	to healthcare		
		G. Jiao; L. Li; H. Deng; G. Zheng; Y. Zou; J. Zhao	Exploration on Cultivation of Practical Ability of Artificial			keyword mismatch:	2020	10.1109/CSEI50228.2020.9142488
IEEE Xplore	05.12.2024		and Entrepreneurship Education	paper	excluded	to healthcare		
		O. Toutsop; K. Kornegay; E. Smith	A Comparative Analyses of Current IoT Middleware Platforms			keyword mismatch:	2021	10.1109/FiCloud49777.2021.00067
IEEE Xolore	30 11 2024			Conference	evoluded	not directly related to healthcare		
ILLE Aplois	00.11.2024	M. Kanwal; S. Thakur; R. Lashkari	An app based on static analysis for android ransomware	paper	GAGIGGEG	keyword mismatch:	2017	10.1109/ICCCNT.2017.8204124
IEEE Valora	20 11 2024			Conference	ovoludod	not directly related		
IEEE Apiore	30.11.2024	D. Ottolini; I. Zyrianoff; C. Kamienski	Interoperability and Scalability Trade-offs in Open IoT	paper	excluded	keyword mismatch:	2022	10.1109/CCNC49033.2022.9700622
			Platforms	Conference		not directly related		
IEEE Xplore	05.12.2024	P. P. Shinde: K. S. Oza: R. K. Kamat	Big data predictive analysis: Using R analytical tool	paper	excluded	to healthcare keyword mismatch:	2017	10.1109/I-SMAC.2017.8058297
				Conference		not directly related		
IEEE Xplore	05.12.2024	P. Ani: E. Maria: L. L. Javea: V. Sakkaravarthy: M.	Smart Space: Voice sesisted text reading system for visually	paper	excluded	to healthcare	2017	10 11098GEHT 2017 8094103
		A. Raja	impaired persons using TTS method	Conference		not directly related	2017	
IEEE Xplore	05.12.2024	M. P. Ashabarik M. M. Janania	Development of a Oland David Maritarian Outland Union	paper	excluded	to healthcare	0040	10 1100004000 2018 8495418
		K. E. Adetunji; M. K. Joseph	4Duino: Applications in Agriculture	Conference		keyword mismatch: not directly related	2018	10.1109/ICABCD.2018.8465418
IEEE Xplore	05.12.2024			paper	excluded	to healthcare		
		A. Grafberger; M. Chadha; A. Jindal; J. Gu; M. Gerndt	FedLess: Secure and Scalable Federated Learning Using Serverless Computing	Conference		keyword mismatch: not directly related	2021	10.1109/BigData52589.2021.9672067
IEEE Xplore	05.12.2024			paper	excluded	to healthcare		
		S. Y. Song; Y. Pei; E. T. Hsiao-Wecksler	Estimating Relative Angles Using Two Inertial Measurement			keyword mismatch:	2022	https://ieeexplore.ieee.org/document/9888780
IEEE Xplore	02.07.2024		Units Without Magnetometers	Article	excluded	to healthcare		
		G. G. M. Dalveren; N. E. Cagiltay	Evaluation of Ten Open-Source Eye-Movement Classification			keyword mismatch:	2019	https://ieeexplore.ieee.org/document/8890843
IEEE Xplore	02.07.2024		Algorithms in Simulated Surgical Scenarios	Article	excluded	to healthcare		
		Matthew R. JonesArunima S. MukherjeeDevinder	After Latour: Globalisation, Inequity and Climate Change			keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-3-
Springer	06 12 2024	ThapaYingqin Zheng		Conference	excluded	not directly related to healthcare		01-00104-0
opingoi	00.12.2024	Prof. Gilbert PetersonSujeet Shenoi	Advances in Digital Forensics XVII	behoi	excluded	keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
Coringer	06 12 2024			Conference	ovoludod	not directly related		030-88381-2
Springer	00.12.2024	Alexandre DolguiProf. Alain BernardDavid	Advances in Production Management Systems. Artificial	paper	excluded	keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
0	00 40 0004	LemoineGregor von CieminskiDavid Romero	Intelligence for Sustainable and Resilient Production Systems	Conference	a valuate d	not directly related		030-85914-5
Springer	00.12.2024	Duck Young KimGregor von CieminskiDavid	Advances in Production Management Systems. Smart	paper	excluded	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
. ·		Romero	Manufacturing and Logistics Systems: Turning Ideas into	Conference		not directly related		031-16411-8
Springer	06.12.2024	Ilias MaglogiannisLazaros IliadisAntonios	Action Artificial Intelligence Applications and Innovations, AIAI 2023	paper	excluded	keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
		Papaleonidasloannis Chochliouros	IFIP WG 12.5 International Workshops	Conference		not directly related		031-34171-7
Springer	07.12.2024	Ilias MaglogiannisProf. Lazaros IliadisProf. John	Artificial Intelligence Applications and Innovations	paper	excluded	to healthcare keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3-
		MacintyrePaulo Cortez		Conference		not directly related	LOLL	031-08333-4
Springer	07.12.2024	Iliae MaglagiannieProf, John MacintureProf	Artificial Intelligence Applications and Innovations, AIAI 2021	paper	excluded	to healthcare	2021	http://link.spripger.com/book/10.1007/978-3-
		Lazaros Iliadis	IFIP WG 12.5 International Workshops	Conference		not directly related	2021	030-79157-5
Springer	07.12.2024	Prof. Dr. Evolite Mercies LoursetDrof M. Önslig	Astificial Istallizance for Knowledge Menorement	paper	excluded	to healthcare	0001	http://link.epringer.com/book/10.1007/078-9-
		KayalicaProf. Dr. Mieczyslaw Lech Owoc	Artificial Intelligence for Knowledge Management	Conference		not directly related	2021	030-80847-1
Springer	07.12.2024	Amony Elbonno Mol auchie Verach K	Co. creating for Context in the Transfer and Diffusion of T	paper	excluded	to healthcare	0000	http://link.spripper.com/bask/10.1007/076.0
		DwivediBrian DonnellanDavid Wastell	oo-oreating for context in the transfer and Diffusion of IT	Conference		not directly related	2022	031-17968-6
Springer	07.12.2024			paper	excluded	to healthcare		
		Luis M. Camarinha-MatosXavier BoucherAngel Ortiz	Collaborative Networks in Digitalization and Society 5.0	Conference		keyword mismatch: not directly related	2023	031-42622-3
Springer	07.12.2024			paper	excluded	to healthcare		
		Aleksander ByrskiProf. Tadeusz CzachórskiErol GelenbeDr. Krzysztof GrochlaProf. Dr. Yuko	Computer Science Protecting Human Society Against	Conference		keyword mismatch:	2021	http://ink.springer.com/book/10.1007/978-3- 030-86582-5
Springer	07.12.2024	Murayama	Lpidemica	paper	excluded	to healthcare		
		Eunika Mercier-LaurentXavier FernandoAravindan	Computer, Communication, and Signal Processing. Al,	Conformação		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-39811-7
Springer	07.12.2024	Chandrabose	Knowledge Engineering and lot for Smart Systems	paper	excluded	to healthcare		
		Jason StaggsDr. Sujeet Shenoi	Critical Infrastructure Protection XV	Contract		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 030-93511-5
Springer	07.12.2024			paper	excluded	to healthcare		and the second s
,		Yingqin ZhengPamela AbbottJose Antonio Robles-	Freedom and Social Inclusion in a Connected World			keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-19429-0
Springer	08.12 2024	FIORES		Conference paper	excluded	not directly related to healthcare		ALL LUTION M
- F		Prof. Audun JøsangLynn FutcherJanne Hagen	ICT Systems Security and Privacy Protection			keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
Springer	08 12 2024			Conference	excluded	not directly related		<u>500-70120-0</u>
opiniger	50.12.2024	Prof. Jun SasakiProf. Dr. Yuko MurayamaDimiter	Information Technology in Disaster Risk Reduction	Property	evended	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
Pagin	00 10 000 1	VelevPlamena Zlateva		Conference	auchust	not directly related		031-04170-9
opringër	06.12.2024	Luis M. Camarinha-MatosLuis RibeiroLeon Strous	Internet of Things. IoT through a Multi-disciplinary Perspective	hahei	excluded	keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3-
0	00.40.000			Conference		not directly related	-	031-18872-5
Springer	08.12.2024	Prof. Dr. Luis M. Camarinha-MatosGeert	Internet of Things. Technology and Applications	paper	excluded	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		HeijenkProf. Srinivas KatkooriLeon Strous	-	Conference		not directly related		030-96466-5
Springer	U8.12.2024	Kevin CrowstonImed HammoudaBiörn	Open Source Systems: Integrating Communities	paper	excluded	to nearthcare keyword mismatch:	2016	http://ink.springer.com/book/10.1007/978-3-
		LundellGregorio RoblesJonas GamalielssonJuho	· · · · · · · · · · · · · · · · · · ·	Conference		not directly related		319-39225-7
Springer	08.12.2024	Lingman		paper	excluded	to healthcare		

venue	Query date	Author	Title	Type	Status	Reason	Year	Article URL
		Felix BiekerJoachim MeyerSebastian Papelna SchieringAndreas Weich	Privacy and Identity Management	Conference		keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-31971-6
Springer	08.12.2024	Prof. Dr. Felix NyffeneggerJosé RíosLouis	Product Lifecycle Management Enabling Smart X	paper	excluded	to healthcare keyword mismatch:	2020	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	RivestProf. Dr. Abdelaziz Bouras		Conference paper	excluded	not directly related to healthcare		030-62807-9
		Prof. Clement FortinLouis RivestProf. Alain BernardProf. Dr. Abdelaziz Bouras	Product Lifecycle Management in the Digital Twin Era	Conference		keyword mismatch: not directly related	2019	http://link.springer.com/book/10.1007/978-3- 030-42250-9
Springer	08.12.2024	Paolo ChiabertProf. Abdelaziz BourasProf. Frédéric	Product Lifecycle Management to Support Industry 4.0	Casterance	excluded	keyword mismatch:	2018	http://link.springer.com/book/10.1007/978-3- 030-01614-2
Springer	08.12.2024	Rof Osiris Cancialiari JuniorProf. Erédéria	Product Lifecuela Management, Green and Dive Technologies	paper	excluded	to healthcare	2022	http://link.springer.com/book/10.1007/978-3-
Springer	08 12 2024	NoëlLouis RivestProf. Dr. Abdelaziz Bouras	to Support Smart and Sustainable Organizations	Conference	excluded	not directly related to healthcare	2022	030-94335-6
opinigoi	OUT LEVE T	Prof. Dr. Luis M. Camarinha-MatosProf. Dr. Xavier BoucherProf. Dr. Hamideh Afsarmanesh	Smart and Sustainable Collaborative Networks 4.0	Conference	01010000	keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-85969-5
Springer	08.12.2024	Prof. Svetan Ratchev	Smart Technologies for Precision Assembly	paper	excluded	to healthcare keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024			Conference paper	excluded	not directly related to healthcare		030-72632-4
		Prof. Dr. Luis M. Camarinha-MatosPedro FerreiraGuilherme Brito	Technological Innovation for Applied AI Systems	Conference		keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-78288-7
Springer	08.12.2024	Prof. Luis M. Camarinha-Matos	Technological Innovation for Digitalization and Virtualization	Conference	excluded	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-07520-9
Springer	08.12.2024	Therese KeaneCathy LewinTorsten BrindaRosa	Towards a Collaborative Society Through Creative Learning	paper	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	Bottino	· · · · · · · · · · · · · · · · · · ·	Conference paper	excluded	not directly related to healthcare		031-43393-1
						keyword mismatch: not directly related		http://link.soringer.com/article/10.1007/s11633-
Springer	08.12.2024	Tobias TiemerdingSergej Fatikow	Software for Small-scale Robotics: A Review	Article	excluded	to healthcare keyword mismatch:	2018	018-1130-2
Springer	07.12.2024	Sdenka Zobeida Salas-PilcoYuqin Yang	Artificial intelligence applications in Latin American higher education: a systematic review	Article	excluded	to healthcare	2022	http://ink.springer.com/article/10.1186/s41239- 022-00326-w
Springer	08 12 2024	Shuo-Chang TsaiCheng-Huan ChenYi-Tzone Shiao lin-Shuei CiouTrong-Neng Wu	Precision education with statistical learning and deep learning:	Article	evoluded	not directly related	2020	http://link.springer.com/article/10.1186/s41239-
Springer	00.12.2024	Emre SürenFredrik HeidingJohannes		74 1010	excluded	keyword mismatch: not directly related	2020	<u>020-00186-2</u>
Springer	08.12.2024	OlegårdRobert Lagerström	PatrloT: practical and agile threat research for IoT	Article	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/article/10.1007/s10207- 022-00633-3
Springer	08.12.2024	Hira AhmadMuhammad Ahtazaz AhsanAdnan Noor Mian	Trends in publishing blockchain surveys: a bibliometric perspective	Article	excluded	not directly related to healthcare	2023	http://link.springer.com/article/10.1007/s10207- 022-00653-z
		Serin V. SimpsonY. Ravi RajuK. Bhanu Rajesh	SECURE TRUST: a blockchain-enabled trust and reputation			keyword mismatch: not directly related		http://link.springer.com/article/10.1007/s41870-
Springer	08.12.2024	NaiduGopika Venu	system for malicious node detection in IOT networks	Article	excluded	to healthcare keyword mismatch:	2023	023-01467-5
Springer	08.12.2024	Belkhir Scott C EdmundePoter LiChristopher LikunterSi	WebRTC-based MOSR remote control of mobile manipulators	Article	excluded	to healthcare	2023	http://link.springer.com/article/10.1007/s41315- 023-00281-3
Springer	08.12.2024	Zhe XiaoRobert L DavidsonNicole NogoyLaurie Goodman	Experiences in integrated data and research object publishing using GigaDB	Article	excluded	not directly related to healthcare	2017	http://link.springer.com/article/10.1007/s00799- 016-0174-6
opinigoi	CONTRIE OF T					keyword mismatch: not directly related		http://ink.epringer.com/article/10.1196/c/0527
Springer	06.12.2024	Feras A. BatarsehLaura FreemanChih-Hao Huang	A survey on artificial intelligence assurance	Article	excluded	to healthcare keyword mismatch:	2021	021-00445-7
Springer	06.12.2024	Srikanth ThudumuPhilip BranchJiong JinJugdutt (Jack) Singh	A comprehensive survey of anomaly detection techniques for high dimensional big data	Article	excluded	not directly related to healthcare	2020	http://link.springer.com/article/10.1186/s40537- 020-00320-x
Springer	07 12 2024	Peter Olucavi Obitarla	Big data analytics: a link between knowledge management	Article	ovaludad	keyword mismatch: not directly related	2019	http://link.springer.com/article/10.1186/s40537-
Springer	07.12.2024	E. A. HuertaAsad KhanEdward DavisColleen BushellWilliam D. GroppDaniel S. KatzVolodymyr		741010	excluded	to neartheare	2013	013-0223-3
		KindratenkoSeid KoricWilliam T. C. KramerBrendan McGintyKenton McHenryAaron	Convergence of artificial intelligence and high performance			keyword mismatch: not directly related		http://ink.springer.com/article/10.1188/s40537-
Springer	07.12.2024	Saxton	computing on NSF-supported cyberinfrastructure	Article	excluded	to healthcare keyword mismatch:	2020	020-00361-2
Springer	07.12.2024	Athanassios M. KintsakisFotis E. PsomopoulosPericles A. Mitkas	Data-aware optimization of bioinformatics workflows in hybrid clouds	Article	excluded	not directly related to healthcare	2016	http://ink.springer.com/article/10.1186/s40537- 016-0055-2
Springer	08 12 2024	Altan CakirÖznün AkınHalil Fanık DenizAli Yılmaz	Enabling real time big data solutions for manufacturing at scale	Article	evoluded	not directly related	2022	http://ink.springer.com/article/10.1186/s40537-
opiniger	00.12.2024			1000	excluded	keyword mismatch: not directly related	LULL	
Springer	08.12.2024	Gousiya BegumS. Zahoor Ul HuqA. P. Siva Kumar	Sandbox security model for Hadoop file system The anatomy of the data-driven smart sustainable city:	Article	excluded	to healthcare keyword mismatch:	2020	020-00356-z
Springer	08.12.2024	Simon Elias Bibri	instrumentation, datafication, computerization and related applications	Article	excluded	not directly related to healthcare	2019	http://link.springer.com/article/10.1186/s40537- 019-0221-4
Springer	08 12 2024	Simon Elias Bibri John Krogetia	The core enabling technologies of big data analytics and context-aware computing for smart sustainable cities: a review and surthesis	Article	oveluded	not directly related	2017	http://link.springer.com/article/10.1186/s40537-
opiniger	00.12.2024	onnon zina biblionn raogata	A systematic review of the purposes of Blockchain and foo	7 8 4010	excluded	keyword mismatch: not directly related	Lon	017-0091-0
Springer	06.12.2024	Yehia Ibrahim AlzoubiAsif GillAlok Mishra	computing integration: classification and open issues	Article	excluded	to healthcare keyword mismatch:	2022	022-00353-y
Springer	07.12.2024	Justice Opara-MartinsReza SahandiFeng Tian	Critical analysis of vendor lock-in and its impact on cloud computing migration: a business perspective	Article	excluded	not directly related to healthcare	2016	http://link.springer.com/article/10.1186/s13677- 016-0054-z
0	07.40.0004	Mahdi Azasuzil eti Ban Domelaana	An Efficient Two-Phase Model for Computing Influential Nodes	Article		keyword mismatch: not directly related	2019	http://link.springer.com/article/10.1007/s11390-
Springer	07.12.2024	Li Zhang lia-Hao Tian ling liangYi- lun LiuMeng-	In Social Networks Using Social Actions	Article	excluded	keyword mismatch:	2018	018-1820-9
Springer	08.12.2024	Yuan PuTao Yue	Survey	Article	excluded	to healthcare keyword mismatch:	2018	http://link.springer.com/article/10.1007/s11390- 018-1864-x
Springer	08.12.2024	Jian-Qiu XuRalf Hartmut GütingYu ZhengOuri Wolfson	Moving Objects with Transportation Modes: A Survey	Article	excluded	not directly related to healthcare	2019	http://ink.springer.com/article/10.1007/s11390- 019-1938-4
			Topic Modeling Based Warning Prioritization from Change			keyword mismatch: not directly related		http://link.springer.com/article/10.1007/s11390-
Springer	08.12.2024	Jung-Been LeeTaek LeeHoh Peter In	Sets of Software Repository	Article	excluded	to healthcare keyword mismatch:	2020	020-0047-8
Springer	07.12.2024	Mina EbrahimSeyed Alireza Hashemi Golpayegani	network analysis ManBeduce and Its Applications, Challenges, and	Article	excluded	to healthcare	2022	http://link.springer.com/article/10.1007/s11416- 021-00398-8
Springer	08.12.2024	Seyed Nima KhezrNima Jafari Navimipour	Architecture: a Comprehensive Review and Directions for Future Research	Article	excluded	not directly related to healthcare	2017	http://link.springer.com/article/10.1007/s10723- 017-9408-0
		Eduardo HuedoRubén S. MonteroRafael Moreno- VozmedianoConstantino VázquezVlastimil	Opportunistic Deployment of Distributed Edge Clouds for			keyword mismatch: not directly related		http://ink.springer.com/article/10.1007/s10703-
Springer	08.12.2024	HolerIgnacio M. Llorente	Latency-Critical Applications	Article	excluded	to healthcare keyword mismatch:	2021	021-09545-3
Springer	08.12.2024	Daniel Woodsloannis AgrafiotisJason R. C. NurseSadie Creese	mapping the coverage of security controls in cyber insurance proposal forms	Article	excluded	not directly related to healthcare	2017	http://link.springer.com/article/10.1186/s13174- 017-0059-y
Springer	08.12.2024	Carlos Eduardo Da SilvaThomás DinizNelio CachoRogério de Lemos	Self-adaptive authorisation in OpenStack cloud platform	Article	excluded	not directly related to healthcare	2018	http://link.springer.com/article/10.1186/s13174- 018-0090-7

Search venue	Querv date	Author	Title	Type	Status	Reason	Year	Article URL
Terrae	aucry unto	Muhammad Umar FaroogMuhammad		1300	otatas	keyword mismatch:	rear	And the one
		ZeeshanMuhammad Talha JahangirMuhammad	A Novel Cooperative Micro-Caching Algorithm Based on			not directly related		http://link.springer.com/article/10.1007/s10922-
Springer	06.12.2024	Asif	Fuzzy Inference Through NFV in Ultra-Dense IoT Networks	Article	excluded	to healthcare	2021	021-09632-6
						keyword mismatch:		
Springer	07 12 2024	Samaneh MahdavifarDima AlbadidiAli A Ghorbani	Lising Pseudo-Label Stacked Auto-Encoder	Article	excluded	not directly related	2021	http://link.springer.com/article/10.1007/s10922-
Springer	07.12.2024	Nouven Huu ThanhNguyen Ngoc TuanDang Anh	Using Pseudo-Laber Stacked Auto-Lincoder	Allicie	excluded	keyword mismatch:	2021	021-09634-4
		KhoaLe Cong TuanNguyen Trung KienNguyen	On Profiling, Benchmarking and Behavioral Analysis of SDN			not directly related		http://link.epringer.com/article/10.1007/s10022-
Springer	08.12.2024	Xuan DungNgo Quynh ThuFlorian Wamser	Architecture Under DDoS Attacks	Article	excluded	to healthcare	2023	023-09732-5
						keyword mismatch:		
			Big data in cybersecurity: a survey of applications and future			not directly related		http://link.springer.com/article/10.1007/s40860-
Springer	07.12.2024	Mohammed M. Alani	trends	Article	excluded	to healthcare	2021	020-00120-3
		Zahra Boozbahani Jalal RezaeenourHanif	A systematic survey on collaborator finding systems in			not directly related		
Springer	06.12.2024	EmamgholizadehAmir Jalaly Bidgoly	scientific social networks	Article	excluded	to healthcare	2020	http://link.springer.com/article/10.1007/s10115- 020-01483-v
		Dr. Maria PaasivaaraDr. Philippe Kruchten	Agile Processes in Software Engineering and Extreme	Conference		keyword mismatch:	2020	http://link.springer.com/book/10.1007/978-3-
			Programming – Workshops	paper		not directly related		030-58858-8
Springer	06.12.2024	De Janeifes HaderffDref, Estafania Garrel Jalana	Advanced Information Origination Facility and Mediatrans	0	excluded	to healthcare	0000	http://ink.apripage.com/hook/10.1007/079-9-
		Zdravkovic	Advanced information systems Engineering workshops	Daper		not directly related	2022	031-07478-3
Springer	06.12.2024	Landinovio		paper	excluded	to healthcare		
		Dr. Viktoria StrayDr. Klaas-Jan StolDr. Maria	Agile Processes in Software Engineering and Extreme	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		PaasivaaraDr. Philippe Kruchten	Programming	paper		not directly related		031-08169-9
Springer	06.12.2024	Prof Witeld Abromowiez Con Klain	Publicase Information Sustama	Conforman	excluded	to nealthcare	2020	http://link.springer.com/book/10.1007/978-3-
		FIOL WILDIG ADIantowiczGary Rielin	Business information systems	Daper		not directly related	2020	030-53337-3
Springer	07.12.2024			p-p	excluded	to healthcare		
		Rachid El AyachiMohamed FakirMohamed Baslam	Business Intelligence	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
				paper		not directly related		031-37872-0
Springer	07.12.2024	Dr. Baria Shiahkay	Pusiness Medaling and Software Design	Conforman	excluded	to healthcare	2021	http://link.spripger.com/book/10.1007/978-3-
		Dr. Bons Shishkov	Business modeling and Software Design	conterence		not directly related	2021	030-79976-2
Springer	07.12.2024			paper	excluded	to healthcare		
		Dr. Dirk FahlandChiara GhidiniProf. Dr. Jörg	Business Process Management Forum	Conference		keyword mismatch:	2020	http://link.springer.com/book/10.1007/978-3-
		BeckerProf. Marlon Dumas		paper		not directly related		<u>uau-38638-6</u>
Springer	07.12.2024	Dr. Chiara Di EmponenamerineDomos Diller et hus	Pusiness Process Management Markehone	Conformer	excluded	to healthcare	2010	http://link.springer.com/book/10.1007/079.2
		Zdun	pusitiess Process Management Workshops	paper		not directly related	2019	030-37453-2
Springer	07.12.2024			horbon	excluded	to healthcare		
		Prof. Ernest TenienteProf. Dr. Matthias Weidlich	Business Process Management Workshops	Conference		keyword mismatch:	2018	http://link.springer.com/book/10.1007/978-3-
				paper		not directly related		<u>319-74030-0</u>
Springer	07.12.2024				excluded	to healthcare		
		Dr. José González EnriquezProf. Søren DeboisProf.	Business Process Management: Blockchain and Robotic	Conference		keyword mismatch:	2021	http://ink.springer.com/book/10.1007/978-3- 030-85867-4
Springer	07 12 2024	WeerdProf. Dr. Ingo Weber	Process Automation Forum	paper	excluded	to healthcare		
Springer	07.12.2024	Yiliu TuMaomao Chi	E-Business, Digital Empowerment for an Intelligent Future	Conference	excluded	keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
				paper		not directly related		031-32299-0
Springer	07.12.2024				excluded	to healthcare		
		Adriano AugustoAsif GillSelmin NurcanDr. Iris	Enterprise, Business-Process and Information Systems	Conference		keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3- 030-79186-5
Springer	08 12 2024	Zdravkovic	Modeling	paper	excluded	to healthcare		
Springer	00.12.2024	Han van der AaDominik BorkHenderik A.	Enterprise, Business-Process and Information Systems	Conference	excluded	keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
		ProperRainer Schmidt	Modeling	paper		not directly related		031-34241-7
Springer	08.12.2024				excluded	to healthcare		
		Maria PapadakiPaulo Rupino da CunhaMarinos	Information Systems	Conference		keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-3- 031-30694-5
Springer	08 12 2024	I nemistocieouskiitos Christodoulou		paper	oveluded	not directly related		
Springer	00.12.2024	Dr. Jochen De WeerdtDr. Artem Polyvyanyy	Intelligent Information Systems	Conference	excluded	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
				paper		not directly related		031-07481-3
Springer	08.12.2024				excluded	to healthcare		
		Fabiano DalpiazJelena ZdravkovicPericles	Research Challenges in Information Science	Conference		keyword mismatch:	2020	http://ink.springer.com/book/10.1007/978-3- 030-50316-1
Springer	08 12 2024	Loucopoulos		paper	evoluded	to healthcare		
opinigoi	CO.TE.LOL4	Renata GuizzardiDr. Jolita RalytéProf. Dr. Xavier	Research Challenges in Information Science	Conference	excluded	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		Franch		paper		not directly related		031-05760-1
Springer	08.12.2024				excluded	to healthcare		
		Daniel MendezDietmar WinklerJohannes	Software Quality: Higher Software Quality through Zero Waste	Conference		keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-3- 031-31488-9
Springer	08 12 2024	KrossStefan BiffiJonannes Bergsmann	Development	paper	evoluded	not directly related		
opiniger	00.12.2024	Balbir S. BarnKurt Sandkuhl	The Practice of Enterprise Modeling	Conference	CACIDOCO	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
				paper		not directly related		031-21488-2
Springer	08.12.2024			o (	excluded	to healthcare	0000	http://itel.environment/fort/0707070
		Massimo MarchioriFrancisco José Dominguez	web information Systems and Technologies	Conterence		keyword mismatch:	2023	031-24197-0
Springer	08 12 2024	Mayoooaquim Filipe		paper	excluded	to healthcare		
opinigoi	00.12.2024	Joemon M. JoseProf. Emine YilmazJoão	Advances in Information Retrieval	Conference	Choradoa	keyword mismatch:	2020	http://link.springer.com/book/10.1007/978-3-
		MagalhãesDr. Pablo CastellsNicola FerroMário J.		paper		not directly related		030-45442-5
Springer	06.12.2024	SilvaFlávio Martins	Advanced Date Mising and Acceleration	Continu	excluded	to healthcare	0001	http://link.enringer.com/back/00.4007/070.7
		Wang Jiang Bing Li Huaijia Zhu Ninghing Chi	Advanced Data Mining and Applications	conterence paper		net directly related	2023	031-46671-7
Springer	06.12.2024	construction of the second sec		logino.	excluded	to healthcare		
, , , , , , , , , , , , , , , , , , , ,		Tiago Prince SalesJoão AraújoJosé	Advances in Conceptual Modeling	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
		BorbinhaGiancarlo Guizzardi		paper		not directly related		<u>usi1-4/112-4</u>
Springer	06.12.2024	Reports QuizzardiReport Neurope	Advances in Concentral Medeline	Conference	excluded	to healthcare	2000	http://link.spripper.com/beak/10.1007/076.0
		nenata duizzardiberno Neumayr	nuvances in conceptual Modeling	Daper		not directly related	2022	031-22036-4
Springer	06.12.2024			here at	excluded	to healthcare		
,		Haoran XieChiu-Lin LaiWei ChenGuandong	Advances in Web-Based Learning - ICWL 2023	Conference		keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-981-
		XuElvira Popescu		paper		not directly related		99-8385-8
Springer	06.12.2024	Visashua Vasalilaru Cubartasta Ousean WasaBia	Advanced Data Minima and Applications	Conference	excluded	to healthcare	0000	http://link.apripage.com/book/10.1007/078-9-
		Xiaochun YangHeru SunartantoGuoren WangBin Wang ling JiangBing LiHusijie ZhuNingning Cui	Advanced Data Mining and Applications	Conterence		keyword mismatch:	2023	031-46664-9
Springer	06.12.2024			habo	excluded	to healthcare		
		Ana Cristina Bicharra GarciaMariza FerroJulio	Advances in Artificial Intelligence – IBERAMIA 2022	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		Cesar Rodríguez Ribón		paper		not directly related		<u>ua1-22419-5</u>
Springer	06.12.2024	Innesia ReissContrala Investada - C. L.	Advances in Computational Intelligence	Conferr	excluded	to healthcare	0005	http://link.springer.com/book/40.4007/020.0
		Ignacio RojasGonzalo JoyaAndreu Gatala	Advances in Computational Intelligence	conterence		keyword mismatch:	2023	031-43085-5
Springer	06.12.2024			Paper	excluded	to healthcare		
		Philippe MathieuFrank DignumPaulo	Advances in Practical Applications of Agents, Multi-Agent	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
		NovaisFernando De la Prieta	Systems, and Cognitive Mimetics. The PAAMS Collection	paper		not directly related		031-37616-0
Springer	06.12.2024				excluded	to healthcare	04-	halo diffici - and an an a final distance of the
		Dr. IIdar BatyrshinAlexander GelbukhProf. Dr.	Advances in Soft Computing	Conference		keyword mismatch:	2021	nttp://link.springer.com/book/10.1007/978-3- 030-89820-5
Springer	06.12.2024	ungen oldolov		hahai	excluded	to healthcare		
-H	JULLEULT	Weitong ChenLina YaoTaotao CaiShirui PanTao	Advanced Data Mining and Applications	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		ShenXue Li		paper		not directly related		031-22064-7
Springer	06.12.2024				excluded	to healthcare		han direction of the second
		Haris AzizDébora CorréaTim French	AI 2022: Advances in Artificial Intelligence	Conference		keyword mismatch:	2022	nttp://ink.springer.com/book/10.1007/978-3- 031-22695-3
Springer	06.12.2024			hahei	excluded	to healthcare		
.F		Tassadit BouadiElisa FromontProf. Dr. Eyke	Advances in Intelligent Data Analysis XX	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		Hüllermeier	-	paper		not directly related		031-01333-1
Springer	06.12.2024				excluded	to healthcare		

venue	Query date	Author	Title	Type	Status	Reason	Year	Article URL
Vende	duci y dute	Bohan LiLin YueJing JiangWeitong ChenXue LiGuodong LongFei FangHan Yu	Advanced Data Mining and Applications	Conference paper	otatus	keyword mismatch: not directly related	2022	http://ink.springer.com/book/10.1007/978-3- 030-95405-5
Springer	06.12.2024	Halimah Badioze ZamanAlan F. SmeatonTimothy K. ShihProf. Sergio VelastinTada TerutoshiProf. Dr. Bo Nørregaard JørgensenDr. Hazleen ArisNazrita	Advances in Visual Informatics	Conference paper	excluded	to healthcare keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-90235-3
Springer	06.12.2024	Ibrahim Prof. Nadia Magnenat-ThalmannVictoria InterranteDaniel ThalmannProf. Dr. George PapagiannakisAssoc. Prof. Bin ShengAssoc. Prof.	Advances in Computer Graphics	Conference paper	excluded	to healthcare keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-89029-2
Springer	06.12.2024	Jinman KimProf. Marina Gavrilova Marta Indulskalris Reinhartz-BergerCarlos CetinaOscar Pastor	Advanced Information Systems Engineering	Conference paper	excluded	to healthcare keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-34560-9
Springer	06.12.2024	George BebisVassilis AthitsosTong YanManfred LauFrederick LiConglei ShiXiaoru YuanChristos	Advances in Visual Computing	Conference paper	excluded	to healthcare keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-90436-4
Springer	06.12.2024	MousasGerd Bruder Guodong LongProf. Xinghuo YuSen Wang	Al 2021: Advances in Artificial Intelligence	Conference paper	excluded	to healthcare keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 030-97546-3
Springer	06.12.2024	Prof. Dr. Xavier FranchGeert PoelsDr. Frederik GaillyMonique Snoeck	Advanced Information Systems Engineering	Conference paper	excluded	to healthcare keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-07472-1
Springer	06.12.2024	Hamido FujitaPhilippe Fournier-VigerMoonis AliYinglin Wang	Advances and Trends in Artificial Intelligence. Theory and Practices in Artificial Intelligence	Conference paper	excluded	to healthcare keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-08530-7.
Springer	06.12.2024	Jaap KampsLorraine GoeuriotFabio CrestaniMaria MaistroHideo JohoBrian DavisCathal GurrinUdo	Advances in Information Retrieval	Conference paper	excluded	to healthcare keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-28238-6
Springer	06.12.2024	KruschwitzAnnalina Caputo Marcello La RosaShazia SadiqProf. Ernest Teniente	Advanced Information Systems Engineering	Conference paper	excluded	to healthcare keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-79382-1
Springer	06.12.2024	Djoerd HiemstraProf. Marie-Francine MoensJosiane MotheRaffaele PeregoMartin	Advances in Information Retrieval	Conference paper	excluded	to healthcare keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-72113-8
Springer	06.12.2024	PotthastFabrizio Sebastiani Prof. Kamal KarlapalemHong ChengNaren RamakrishnanR. K. AgrawalP. Krishna ReddyDr. Jardeon Scingtown Amorit. Prof. Tampoy.	Advances in Knowledge Discovery and Data Mining	Conference paper	excluded	to healthcare keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3- 030-75762-5
Springer	06.12.2024	Chakraborty Prof. Dr. Schahram DustdarDr. Eric YuCamille SalinasiDominique Biau/dk Bant	Advanced Information Systems Engineering	Conference	excluded	to healthcare keyword mismatch:	2020	http://link.springer.com/book/10.1007/978-3- 030-49435-3
Springer	06.12.2024	Hady W. LauwRaymond Chi-Wing	Advances in Knowledge Discovery and Data Mining	Conference	excluded	to healthcare keyword mismatch:	2020	http://link.springer.com/book/10.1007/978-3- 030-47426-3
Springer	06.12.2024	NgSinno Jialin Pan Paolo GiorginiBarbara Weber	Advanced Information Systems Engineering	Conference	excluded	to healthcare keyword mismatch:	2019	http://link.springer.com/book/10.1007/978-3- 030-21290-2
Springer	06.12.2024	Eric DuboisKlaus Pohl	Advanced Information Systems Engineering	Conference	excluded	to healthcare keyword mismatch:	2017	http://ink.springer.com/book/10.1007/978-3- 319-59536-8
Springer	06.12.2024	Stefania BandiniFrancesca GaspariniViviana	AlxIA 2021 – Advances in Artificial Intelligence	Conference	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-08421-8
Springer	07.12.2024	Agostino DovierAngelo MontanariAndrea Orlandini	AlxIA 2022 - Advances in Artificial Intelligence	Conference	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-27181-6
Springer	07.12.2024	Panagiotis KanellopoulosMaria	Algorithmic Game Theory	Conference	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-15714-1
Springer	07.12.2024	João CorreiaStephen SmithRaneem Qaddoura	Applications of Evolutionary Computation	Conference	excluded	to healthcare keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-3- 031-30229-9
Springer	07.12.2024	Giuseppe AtenieseDaniele Venturi	Applied Cryptography and Network Security	Conference	excluded	to healthcare keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3- 031-09234-3
Springer	07.12.2024	Mehdi TibouchiXiaoFeng Wang	Applied Cryptography and Network Security	Conference	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-33488-7
Springer	07.12.2024	Kazue SakoNils Ole Tippenhauer	Applied Cryptography and Network Security	Conference	excluded	to healthcare keyword mismatch:	2021	http://ink.springer.com/book/10.1007/978-3- 030-78372-3
Springer	07.12.2024	Georgios GoumasSven TomfordeJürgen BrehmStafan WildermannThilo Pionteck	Architecture of Computing Systems	Conference	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-42785-5
Springer	07.12.2024	Patrick HammerMarjan AlirezaieClaes Strannegård	Artificial General Intelligence	Conference	excluded	to healthcare keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-33469-6
Springer	07.12.2024	Sergei M. KovalevProf. Sergei O. KuznetsovDr. Aleksandr I. Panov	Artificial Intelligence	Conference	excluded	to healthcare keyword mismatch:	2021	http://ink.springer.com/book/10.1007/978-3- 030-86855-0
Springer	07.12.2024	Lu FangDaniel PoveyGuangtao ZhaiTao MeiBuining Wang	Artificial Intelligence	Conference	excluded	to healthcare keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-20503-3
Springer	07.12.2024	Dr. Maria Mercedes RodrigoProf. Noburu MatsudaProf. Alexandra I. CristeaDr. Vania	Artificial Intelligence in Education. Posters and Late Breaking Results. Workshops and Tutorials. Industry and Innovation	Conference	excluded	to healthcare keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-11647-6
Springer	07.12.2024	Dimitrova Xiuqin PanTing JinLiang-Jie Zhang	Tracks, Practitioners' and Doctoral Consortium Artificial Intelligence and Mobile Services – AIMS 2022	Conference	excluded	to healthcare keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-23504-7
Springer	07.12.2024	Xingming SunProf. Xiaorui ZhangZhihua XiaProf. Dr. Elisa Bertino	Artificial Intelligence and Security	Conference	excluded	to healthcare keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-06788-4
Springer	07.12.2024	Leszek RutkowskiRafał SchererMarcin KorytkowskiWitold PedryczRyszard	Artificial Intelligence and Soft Computing	Conference	excluded	to healthcare keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-23480-4
Springer	07.12.2024	TadeusiewiczJacek M. Zurada Helmut DegenLauren Reinerman-Jones	Artificial Intelligence in HCI	Conference	excluded	to healthcare keyword mismatch: not directly related	2020	http://link.springer.com/book/10.1007/978-3- 030-50334-5
Springer	07.12.2024	Elias PimenidisPlamen AngelovChrisina JayneAntonios PapaleonidasMehmet Aydin	Artificial Neural Networks and Machine Learning - ICANN 2022	Conference	excluded	to healthcare keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-15919-0
Springer	07.12.2024	Neamat El GayarEdmondo TrentinMirco RavanelliHazem Abbas	Artificial Neural Networks in Pattern Recognition	Conference paper	excluded	to healthcare keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-20650-4
Springer	07.12.2024	Shunli ZhangBo HuLiang-Jie Zhang	Big Data – BigData 2023	Conference paper	excluded	to healthcare keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-44725-9
Springer	07.12.2024	Matteo GolfarelliRobert WrembelGabriele KotsisA Min Tjoalsmail Khalil	Big Data Analytics and Knowledge Discovery	Conference paper	excluded	to healthcare keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-86534-4
Springer	07.12.2024	Shelly SachdevaYutaka WatanobeSubhash Bhalla	Big Data Analytics in Astronomy, Science, and Engineering	Conference paper	excluded	to healthcare keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-28350-5
Springer	07.12.2024				excluded	to healthcare		

Search	Query date	Author	Title	Type	Status	Reason	Year	Article URI
	anory anto	Ching-Hsien HsuMengwei XuHung CaoHojjat	Big Data Intelligence and Computing	Conference		keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-981-
C	07 40 0004	BaghbanA. B. M. Shawkat Ali		paper	evel of edge	not directly related		99-2233-8
Springer	07.12.2024	Asbjørn FølstadTheo AraujoSymeon	Chatbot Research and Design	Conference	excluded	keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
		PapadopoulosEffie LC. LawEwa LugerMorten		paper		not directly related		030-68288-0
Springer	07.12.2024	GoodwinPetter Bae Brandtzaeg Subevla Cetin-KaravumakDaan ChristiaensMatteo	Computational Diffusion MBI	Conference	excluded	to healthcare	2021	http://link.springer.com/book/10.1007/978-3-
		FiginiPamela GuevaraNoemi GyoriVishwesh		paper		not directly related	LOLI	030-87615-9
Springer	07.12.2024	NathTomasz Pieciak	Ocean taking I Diffusion MDI	0	excluded	to healthcare	0000	http://ink.opringer.com/book/10.1007/079.2
		PowellFrancois RheaultStefan Winzeck	Computational Diffusion MHI	Daper		not directly related	2023	031-47292-3
Springer	07.12.2024			p-4	excluded	to healthcare		
		Dr. João M. F. RodriguesDr. Pedro J. S. CardosoDr. Jônio MontairoProf. Pabarto LamDr.	Computational Science – ICCS 2019	Conference		kousuord micmotoby	2019	http://ink.springer.com/book/10.1007/978-3- 030-22744-9
		Valeria V. KrzhizhanovskayaMichael H. LeesProf.		paper		not directly related		
Springer	07.12.2024	Jack J. DongarraProf. Dr. Peter M.A. Sloot			excluded	to healthcare		
		Dr. Valeria V. KrzhizhanovskayaDr. Gábor ZávodszkyMichael H. LeesProf. Jack J.	Computational Science – ICCS 2020	Conference		keyword mismatch:	2020	http://link.springer.com/book/10.1007/978-3- 030-50423-6
		DongarraProf. Dr. Peter M. A. SlootSérgio		paper		not directly related		
Springer	07.12.2024	BrissosJoão Teixeira			excluded	to healthcare		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
		Prot. Maciej PaszynskiProt. Dr. Dieter KranzlmüllerDr. Valeria V. KrzhizbanovskavaProf.	Computational Science – ICCS 2021	Conference paper		keyword mismatch: not directly related	2021	030-77967-2
Springer	07.12.2024	Jack J. DongarraProf. Dr. Peter M.A. Sloot		helter.	excluded	to healthcare		
		Derek GroenClélia de MulatierProf. Maciej Paszunski/Dr. Valoria V. KrzhizbanowskawaProf.	Computational Science – ICCS 2022	Conference		keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3- 031-08760-8
Springer	07.12.2024	Jack J. DongarraProf. Dr. Peter M. A. Sloot		paper	excluded	to healthcare		
		Jiří MikyškaClélia de MulatierMaciej	Computational Science – ICCS 2023	Conference		keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-3- 031-36027-5
Springer	07 12 2024	PaszynskiValeria V. KrzhizhanovskayaJack J. DongarraPeter M.A. Sloot		paper	excluded	not directly related to healthcare		MAL-SOUGLESS
opinger	01.12.2024	Prof. Dr. Osvaldo GervasiBeniamino MurganteDr.	Computational Science and Its Applications - ICCSA 2020	Conference	CACINGOU	to notatilibaro	2020	http://link.springer.com/book/10.1007/978-3-
		Sanjay MisraDr. Chiara Garaulvan BlečićDavid		paper		keysuerd mismetals		030-58820-5
		RochaProf. Eufemia TarantinoProf. Carmelo Maria				not directly related		
Springer	07.12.2024	TorreProf. Yeliz Karaca			excluded	to healthcare		
		Prof. Dr. Osvaldo GervasiBeniamino MurganteDr. Sapiav MieraDr. Chiara Garaulyan BlečićDavid	Computational Science and Its Applications – ICCSA 2021	Conference		kowword micmatch:	2021	http://ink.springer.com/book/10.1007/978-3- 030-86960-1
		TaniarBernady O. ApduhanAna Maria A.C.		paper		not directly related		
Springer	07.12.2024	RochaEufernia TarantinoCarmelo Maria Torre			excluded	to healthcare		
		Prof. Dr. Osvaldo GervasiBeniamino MurganteEligius M. T. HendrixDavid TaniarProf	Computational Science and Its Applications – ICCSA 2022	Conference		keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3- 031-10522-7
Springer	07.12.2024	Bernady O. Apduhan		paper	excluded	to healthcare		
		Osvaldo GervasiBeniamino MurganteSanjay	Computational Science and Its Applications - ICCSA 2022	Conference		keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3- 031-10536-4
Springer	07 12 2024	MisraAna Maria A. C. HochaChiara Garau	worksnops	paper	excluded	to healthcare		
opinigoi	OTTILLOL T	Osvaldo GervasiBeniamino MurganteDavid	Computational Science and Its Applications - ICCSA 2023	Conference	onoradoa	keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-3-
		TaniarBernady O. ApduhanAna Cristina		paper		not directly related		031-36805-9
Springer	07.12.2024	Osvaldo GervasiBeniamino MurganteAna Maria A.	Computational Science and Its Applications - ICCSA 2023	Conference	excluded	keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
		C. RochaChiara GarauFrancesco ScorzaYeliz	Workshops	paper		not directly related		031-37123-3
Springer	07.12.2024	KaracaCarmelo M. Torre	Computer Information Systems and Industrial Management	Conforance	excluded	to healthcare	2022	http://ink.spripper.com/book/10.1007/978-3-
		Khalid SaeedSin Dvolsky	computer mormation systems and industrial wanagement	paper		not directly related	2022	031-10539-5
Springer	07.12.2024				excluded	to healthcare		
		Roberto Moreno-DiazFranz PichlerAlexis Quesada- Arencibia	Computer Aided Systems Theory – EUROCAST 2022	Conference		keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3- 031-25312-6
Springer	07.12.2024	, a choicht		paper	excluded	to healthcare		
		Prof. Alexandra SilvaK. Rustan M. Leino	Computer Aided Verification	Conference		keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3- 030-81685-8
Springer	07.12.2024			paper	excluded	not directly related to healthcare		300 S1302 S
opinigoi		Constantin EneaAkash Lal	Computer Aided Verification	Conference		keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-3-
0	07.40.0004			paper	and the stand	not directly related		031-37709-9
Springer	07.12.2024	Dr. Nicolas TsapatsoulisDr. Andreas PanavidesDr.	Computer Analysis of Images and Patterns	Conference	excluded	keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
		Theo TheocharidesDr. Andreas LanitisProf. Dr.		paper		not directly related		030-89128-2
Springer	07.12.2024	Constantinos PattichisMario Vento	Computer Safety, Baliability, and Security	Conformero	excluded	to healthcare	2022	http://ink.springer.com/book/10.1007/978-3-
		SpisländerFriedemann Bitsch	computer sarety, neirability, and security	paper		not directly related	2022	031-14835-4
Springer	07.12.2024			0. /	excluded	to healthcare	0000	http://init.api/page.com/back/40.4007/078-0
		Jeremie GuiochetStefano TonettaFriedemann Bitsch	Computer Safety, Reliability, and Security	Conference		keyword mismatch: not directly related	2023	031-40923-3
Springer	07.12.2024			hoho.	excluded	to healthcare		
		Prof. Dr. Elisa BertinoProf. Haya ShulmanProf. Dr.	Computer Security – ESORICS 2021	Conference		keyword mismatch:	2021	http://ink.springer.com/book/10.1007/978-3- 030-88428-4
Springer	07.12.2024			paper	excluded	to healthcare		
		Sokratis KatsikasFrédéric CuppensChristos	Computer Security. ESORICS 2022 International Workshops	Conference			2023	http://ink.springer.com/book/10.1007/978-3- 031-25460-4
		KalloniatisJohn MylopoulosFrank PallasJörg PohleM Angela SasseHabtamu AbieSilvio		paper				031-20-00-4
		RaniseLuca VerderameEnrico CambiasoJorge						
		Maestre VidalMarco Antonio Sotelo				keyword mismatch:		
Springer	07.12.2024	PirbhulalAnkur Shukla			excluded	to healthcare		
		Lei WangJuergen GallTat-Jun ChinImari SatoRama	Computer Vision – ACCV 2022	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-26316-3
Springer	07.12 2024	Cnellappa		paper	excluded	not directly related to healthcare		
opinigo	01.12.2024	Andrea VedaldiHorst BischofProf. Dr. Thomas	Computer Vision – ECCV 2020	Conference	CACINGCO	keyword mismatch:	2020	http://link.springer.com/book/10.1007/978-3-
Code	07 40 000 4	BroxJan-Michael Frahm		paper	outer	not directly related		<u>uau-08030-0</u>
Springer	07.12.2024	Prof. Adrien BartoliAndrea Fusiello	Computer Vision - ECCV 2020 Workshops	Conference	excluded	keyword mismatch:	2020	http://link.springer.com/book/10.1007/978-3-
				paper		not directly related		030-66415-2
Springer	07.12.2024	Shai AvidanQabrial ProstowMoustanha	Computer Vision - ECCV 2022	Conformen	excluded	to healthcare	2022	http://ink.springer.com/book/10.1007/978-3-
		CisséGiovanni Maria FarinellaTal Hassner	Computer vision = ECCV 2022	paper		not directly related	2022	031-20059-5
Springer	07.12.2024				excluded	to healthcare		
		Leonid KarlinskyTomer MichaeliKo Nishino	Computer Vision – ECCV 2022 Workshops	Conference paper		keyword mismatch: not directly related	2023	http://ink.springer.com/book/10.1007/978-3- 031-25069-9
Springer	07.12.2024			paper	excluded	to healthcare		
		Henrik I. ChristensenPeter CorkeRenaud	Computer Vision Systems	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-44137-0
Springer	07,12,2024	DetryJean-Baptiste WeidelMarkus Vincze		paper	excluded	to healthcare		
F		Dr. Gillian DobbieProf. Dr. Ulrich FrankProf. Dr.	Conceptual Modeling	Conference		keyword mismatch:	2020	http://ink.springer.com/book/10.1007/978-3-
Poring	07 10 0004	Gerti KappelStephen W. LiddleProf. Dr. Heinrich C.		paper	ovoluted	not directly related		<u>uau-62522-1</u>
opringer	07.12.2024	Dr. Aditya GhoseJennifer HorkoffVítor E. Silva	Conceptual Modeling	Conference	excluded	keyword mismatch	2021	http://link.springer.com/book/10.1007/978-3-
		SouzaProf. Jeffrey ParsonsJoerg Evermann		paper		not directly related		030-89022-3
Springer	07.12.2024	Viibua Luo	Cooperative Design Visualization and Engineering	Conference	excluded	to healthcare	2024	http://link.springer.com/book/10.1007/978-9-
		Tonica EUO	ooperative Design, visualization, and Engineering	paper		not directly related	2021	030-88207-5
Springer	07.12.2024				excluded	to healthcare	0.000	hite Web enderer ere fra de reenen -
		Bernnard HammeriiUdo HeimbrechtWolfgang HommelLeonhard KunczikStefan Pickl	Untical Information Intrastructures Security	Conference paper		keyword mismatch: not directly related	2023	031-35190-7
Springer	07.12.2024			· -+ **	excluded	to healthcare		
		Pei-Luen Patrick Rau	Cross-Cultural Design	Conference		keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-3- 031-35936-1
Springer	07.12.2024			haher	excluded	to healthcare		

venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
		Prof. Dr. Matthias Rauterberg	Culture and Computing	Conference paper		keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-05434-1
Springer	07.12.2024	Shlomi DolevJonathan KatzAmnon Meisels	Cyber Security, Cryptology, and Machine Learning	Conference	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-07689-3
Springer	07.12.2024	Jieren ChengXiangvan TangXiaozhang Liu	Cyberspace Safety and Security	Conference	excluded	to healthcare keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
Springer	07.12.2024			paper	excluded	not directly related to healthcare		030-73671-2
Continuer	07 10 0004	Prof. Dr. Ken BarkerKambiz Ghazinour	Data and Applications Security and Privacy XXXV	Conference paper	avaluate at	keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-81242-3
Springer	07.12.2024	Prof. Shamik SuralHaibing Lu	Data and Applications Security and Privacy XXXVI	Conference	excluded	keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-10684-2
Springer	07.12.2024	Vijayalakshmi AtluriAnna Lisa Ferrara	Data and Applications Security and Privacy XXXVII	Conference	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
Springer	07.12.2024			paper	excluded	not directly related to healthcare		031-37586-6
Springer	07 12 2024	Hien V. NguyenSharon X. HuangYuan Xue	Data Augmentation, Labelling, and Imperfections	Conference paper	excluded	not directly related	2022	nttp://iink.springer.com/book/10.1007/978-3- 031-17027-0
opinigoi	OTTELEORY	Christine StraussToshiyuki AmagasaGabriele KotsisA Min Tjoalsmail Khalii	Database and Expert Systems Applications	Conference paper	CAGGGGGG	keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-39847-6
Springer	07.12.2024	Wen HuaHua WangLei Li	Databases Theory and Applications	Conference	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-15512-3
Springer	07.12.2024	Prof. Shengchao QinProf. Dr. Jim	Dependable Software Engineering, Theories, Tools, and	Conference	excluded	to healthcare keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-030-
Springer	07.12.2024	WoodcockWenhui Zhang	Applications	paper	excluded	not directly related to healthcare	LUET	91265-9
		Gavriel SalvendyDr. June Wei	Design, Operation and Evaluation of Mobile Communications	Conference paper		keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-77025-9
Springer	07.12.2024	Aaron MarcusElizabeth RosenzweigMarcelo M. Soares	Design, User Experience, and Usability	Conference	excluded	to healthcare keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-35708-4
Springer	07.12.2024	Assoc. Prof. Emi IshitaNatalie Lee San PangProf.	Digital Libraries at Times of Massive Societal Transition	Conference	excluded	to healthcare keyword mismatch:	2020	http://link.springer.com/book/10.1007/978-3-
Springer	07.12.2024	Lihong Zhou	-	paper	excluded	not directly related to healthcare		030-64452-9
Coringor	07 12 2024	Dr. Ilias O. PappasProf. Dr. Patrick MikalefProf. Yogesh K. DwivediLetizia JaccheriProf. John KronstiaMatti Mäntumäki	Digital Transformation for a Sustainable Society in the 21st Century	Conference paper	oveluded	keyword mismatch: not directly related	2019	http://ink.springer.com/book/10.1007/978-3- 030-29374-1
Springer	07.12.2024	Carlos SoaresLuis Torgo	Discovery Science	Conference paper	excluded	keyword mismatch: not directly related	2021	http://ink.springer.com/book/10.1007/978-3- 030-88942-5
Springer	07.12.2024	Davide CeolinTommaso CaselliMarina Tulin	Disinformation in Open Online Media	Conference	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
Springer	07.12.2024	Miguel Maten Fabiala Oraya	Distributed Applications and Intercognitude Sustains	paper Conference	excluded	not directly related to healthcare	0001	http://ink.epripag.com/book/10.1007/078-3-
Springer	07.12.2024	Miguel Matosrabiola Greve	Distributed Applications and Interoperable Systems	paper	excluded	not directly related to healthcare	2021	030-78198-9
		Dr. Norbert A. StreitzDr. Shin'ichi Konomi	Distributed, Ambient and Pervasive Interactions. Smart Environments, Ecosystems, and Cities	Conference paper		keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-05463-1
Springer	07.12.2024	Gernot A. FinkRajiv JainKoichi KiseRichard Zanibbi	Document Analysis and Recognition - ICDAR 2023	Conference	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-41682-8
Springer	07.12.2024	Dr. Konstantinos Tearnas lörn Altmann losá Ángal	Fronomics of Gride Clouds Systems and Services	Conference	excluded	to healthcare	2021	http://ink.springer.com/book/10.1007/978-3-
Springer	07.12.2024	BañaresOrna Agmon Ben-YehudaProf. Karim DjemameProf. Vlado StankovskiDr. Bruno Tuffin		paper	excluded	not directly related to healthcare	LULI	030-92916-9
		Isabel HilligerPedro J. Muñoz-MerinoTinne De LaetAlejandro Ortega-ArranzTracie Farrell	Educating for a New Future: Making Sense of Technology- Enhanced Learning Adoption	Conference paper		keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-16290-9
Springer	07.12.2024	lda LindgrenCsaba CsákiEvangelos KalampokisMarijn JanssenGabriela Viale	Electronic Government	Conference	excluded	to nearthcare	2023	http://ink.springer.com/book/10.1007/978-3- 031-41138-0
Springer	07.12.2024	PereiraShefali VirkarEfthimios TambourisAnneke Zuiderwijk		hehe.	excluded	not directly related to healthcare		
0		Hans Jochen SchollJ. Ramon Gil-GarciaProf. Dr. Marijn JanssenEvangelos KalampokisIda	Electronic Government	Conference paper		keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-84789-0
Springer	07.12.2024	LindgrenManuel Pedro Hodriguez Bolivar Marijn JanssenCsaba Csákilda LindgrenEuripidis Loukisl IIf MelinGabriela Viale PereiraManuel Pedro	Electronic Government	Conference	excluded	to healthcare keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-15086-9
Springer	07.12.2024	Rodríguez BolívarEfthimios Tambouris Noella EdelmannCsaba CsákiSara	Electronic Participation	Conference	excluded	to healthcare	2021	http://link.springer.com/book/10.1007/978-3-
		HofmannThomas J. LampoltshammerProf. Laura Alcaide MuñozPeter ParycekGerhard		paper		keyword mismatch: not directly related		030-82824-0
Springer	07.12.2024	SchwabeEthimios Tambouris Alex OrailogluMatthias JungMarc Reichenbach	Embedded Computer Systems: Architectures, Modeling, and	Conference	excluded	to healthcare keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-04580-6
Springer	07.12.2024	Cristina SilvanoChristian PilatoMarc Reichenbach	Embedded Computer Systems: Architectures, Modeling, and	Conference	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
Springer	07.12.2024		Simulation	paper	excluded	not directly related to healthcare		031-46077-7
Carianaa	09 40 0004	Andrei CiorteaMehdi DastaniJieting Luo	Engineering Multi-Agent Systems	Conference paper	and a date	keyword mismatch: not directly related	2023	http://ink.springer.com/book/10.1007/978-3- 031-48539-8
Springer	00.12.2024	Natasha AlechinaMatteo BaldoniBrian Logan	Engineering Multi-Agent Systems	Conference paper	excluded	keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 030-97457-2
Springer	08.12.2024	Don HarrisWen-Chin Li	Engineering Psychology and Cognitive Ergonomics	Conference	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	Deale Oissessieidanale Di Javie Jaleuri		paper	excluded	not directly related to healthcare	0000	131-35392-5
Springer	08.12.2024	HavacsFrancesco Poggi	Entertainment Computing – ICEC 2023	paper	excluded	not directly related to healthcare	2023	99-8248-6
opingoi	00.12.2024	José CanoMarios D. DikaiakosGeorge A. PapadopoulosMiquel PericàsRizos Sakellariou	Euro-Par 2023: Parallel Processing	Conference paper	CAGIGGEG	keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-39698-4
Springer	08.12.2024	Amaud LiefoogheDr. Luis Paquete	Evolutionary Computation in Combinatorial Optimization	Conference	excluded	to healthcare keyword mismatch:	2019	http://ink.springer.com/book/10.1007/978-3- 030-16711-0
Springer	08.12.2024	Dr. Davide CalvaresiAmro NaijarProf Michael	Explainable and Transparent AI and Multi-Agent Systems	Conference	excluded	to healthcare keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	WinikoffKary Främling		paper	excluded	not directly related to healthcare	-961	030-82017-6
		Aggelos Kiayias	Financial Cryptography and Data Security	Conference paper		keyword mismatch: not directly related	2017	http://link.springer.com/book/10.1007/978-3- 319-70972-7.
Springer	U8.12.2024	Prof. Nikita BorisovClaudia Diaz	Financial Cryptography and Data Security	Conference	excluded	to nealthcare keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 662-64322-8
Springer	08.12.2024	Yuen-Hsien TsengMarie KatsuraiHoa N. Nguyen	From Born-Physical to Born-Virtual: Augmenting Intelligence in	Conference	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024		Digital Libraries	paper	excluded	not directly related to healthcare		031-21756-2
Springer	08 12 2024	Juliana BowlesGiovanna BrocciaRoberto Pellungrini	From Data to Models and Back	Conference paper	evolution	keyword mismatch: not directly related to healthcare	2022	nttp://link.springer.com/book/10.1007/978-3- 031-16011-0
opinger	JU. 12.2024				avoinged	- main follo		

venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
		Alfredo CapozuccaSophie EbersoldJean-Michel BruelBertrand Meyer	Frontiers in Software Engineering Education	Conference paper		keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-48639-5
Springer	08.12.2024	Sebastia MassanetSusana MontesDaniel Ruiz-	Fuzzy Logic and Technology, and Aggregation Operators	Conference	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-39965-7
Springer	08.12.2024	Gisele PannaMarin GiacobiniZdenek Vasicek	Genetic Programming	Conference	excluded	to healthcare	2023	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024		echeric regiumming	paper	excluded	not directly related to healthcare	LOLO	031-29573-7
		Rodrigo MianiLasaro CamargosBruno ZarpelãoErika RosasRafael Pasquini	Green, Pervasive, and Cloud Computing	Conference paper		keyword mismatch: not directly related	2019	http://link.springer.com/book/10.1007/978-3- 030-19223-5
Springer	08.12.2024	Charalampos SaitisProf. Ildar FarkhatdinovStefano	Haptic and Audio Interaction Design	Conference	excluded	to healthcare keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3-
Springer	08.12.2024	Papetti		paper	excluded	not directly related to healthcare		<u>131-13013-7</u>
Springer	08 12 2024	Hasti SelfiAstrid M. L. KappersUliver Schneiderbr. Knut DrewingClaudio PacchierottiAlireza AbbasimoshaeiGiis HuismanThorsten A. Kern	Haptics: Science, Technology, Applications	paper	evoluded	not directly related	2022	031-06249-0
Springer	00.12.2024	Abbas Moallem	HCI for Cybersecurity, Privacy and Trust	Conference paper	excluded	keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-77392-2
Springer	08.12.2024	Dr. Fiona Fui-Hoon NahDr. Keng Siau	HCl in Business, Government and Organizations	Conference	excluded	to healthcare keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024			paper	excluded	not directly related to healthcare		030-77750-0
Caringer	00 10 0004	Xiaowen Fang	HCl in Games: Serious and Immersive Games	Conference paper	evoluted	keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-77414-1
Springer	08.12.2024	Heidi Krömker	HCI in Mobility, Transport, and Automotive Systems	Conference	excluded	keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-78358-7
Springer	08.12.2024	Masaaki KurosuSakae YamamotoHirohiko	HCI International 2022 - Late Breaking Papers. Multimodality	Conference	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	MoriDylan D. SchmorrowCali M. FidopiastisNorbert A. StreitzShin'ichi Konomi	in Advanced Interaction Environments	paper	excluded	not directly related to healthcare		<u>031-17618-0</u>
		Hirohiko MoriYumi AsahiAdela ComanSimona VasilacheMatthias Rauterberg	HCI International 2023 – Late Breaking Papers	Conference paper		keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-48044-7
Springer	08.12.2024	Amanda BienzMichèle WeilandMarc BaboulinCarola Kruse	High Performance Computing	Conference	excluded	keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-40843-4
Springer	08.12.2024	Heike JagodeHartwig AnztHatem LtaiefPiotr	High Performance Computing	Conference	excluded	to healthcare keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	Luszczek	·····	paper	excluded	not directly related to healthcare		030-90539-2
		Bradford L. ChamberlainAna-Lucia VarbanescuHatem LtaiefPiotr Luszczek	High Performance Computing	Conference paper		keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-78713-4
Springer	08.12.2024	Hartwig AnztAmanda BienzPiotr LuszczekMarc	High Performance Computing. ISC High Performance 2022	Conference	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-23220-6
Springer	08.12.2024	Sakae YamamotoHirobiko Mori	International workshops	Conference	excluded	to healthcare	2022	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024		Applications in Complex Technological Environments	paper	excluded	not directly related to healthcare	2022	031-06509-5
		Sakae YamamotoHirohiko Mori	Human Interface and the Management of Information. Information-Rich and Intelligent Environments	Conference paper		keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-78361-7
Springer	08.12.2024	Masaaki KurosuAyako Hashizume	Human-Computer Interaction	Conference	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-35602-5
Springer	08.12.2024	Carmelo ArditoBosa I anzilottiAlessio MaliziaHelen	Human-Computer Interaction - INTERACT 2021	Conference	excluded	to healthcare	2021	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	PetrieAntonio PiccinnoDr. Giuseppe DesoldaKori Inkpen		paper	excluded	not directly related to healthcare	2021	030-85613-7
		José Abdelnour NoceraMarta Kristín LárusdóttirHelen PetrieAntonio PiccinnoMarco	Human-Computer Interaction – INTERACT 2023	Conference paper		keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-42293-5
Springer	08.12.2024	Winckler Prof. Stan SclaroffCosimo DistanteMarco LeoDr.	Image Analysis and Processing - ICIAP 2022	Conference	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-06433-3
Springer	08.12.2024	Gian Luca ForestiAndrea FusielloEdwin Hancock	Image Analysis and Processing - ICIAP 2023	Conference	excluded	to healthcare	2023	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024		······	paper	excluded	not directly related to healthcare		031-43153-1
		Pier Luigi MazzeoEmanuele FrontoniStan SclaroffCosimo Distante	Image Analysis and Processing. ICIAP 2022 Workshops	Conference paper		keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-13324-4
Springer	08.12.2024	Fakhri KarrayAurélio CampilhoFarida Cheriet	Image Analysis and Recognition	Conference	excluded	to healthcare keyword mismatch:	2017	http://link.springer.com/book/10.1007/978-3- 319-59876-5
Springer	08.12.2024	Huchuan LuWanli QuyanoHui Huang liwan	Image and Graphics	Conference	excluded	to healthcare	2023	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	LuRisheng LiuJing DongMin Xu	inage and enaprice	paper	excluded	not directly related to healthcare	LOLO	031-46314-3
		Sihan QingEiji OkamotoKwangjo KimDongmei Liu	Information and Communications Security	Conference paper		keyword mismatch: not directly related	2016	http://ink.springer.com/book/10.1007/978-3- 319-29814-6
Springer	08.12.2024	Ding WangMoti YungZheli LiuXiaofeng Chen	Information and Communications Security	Conference	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-981- 99-7356-9
Springer	08.12.2024	Natalia Graene TaulorCaitlin Obviction-	Information in Contemporary Society	Conference	excluded	to healthcare	2010	http://ink.spripper.com/book/10.1007/978-3-
Springer	08.12.2024	LambMichelle H. MartinBonnie Nardi	mormation in contemporary doubty	paper	excluded	not directly related to healthcare	2015	030-15742-5
opinigoi		Alejandro FrangiMarleen de BruijneDemian WassermannNassir Navab	Information Processing in Medical Imaging	Conference paper		keyword mismatch: not directly related	2023	http://ink.springer.com/book/10.1007/978-3- 031-34048-2
Springer	08.12.2024	Joonsang BaekDr. Sushmita Ruj	Information Security and Privacy	Conference	excluded	to healthcare keyword mismatch:	2021	http://ink.springer.com/book/10.1007/978-3-
Springer	08.12.2024	These weblets Mar	Information Consults And Particular	paper	excluded	not directly related to healthcare	0004	http://link.anrianes.com/back/10.1007/078-2
Springer	08 12 2024	Hyoungsnick Kim	Information Security Applications	paper	excluded	not directly related	2021	030-89432-0
e buu Bai	00.12.2024	Ilsun YouTaek-Young Youn	Information Security Applications	Conference paper	choldred	keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-25659-2
Springer	08.12.2024	Weizhi MengZheng YanVincenzo Piuri	Information Security Practice and Experience	Conference	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-981-
Springer	08.12.2024	Operate The share of the state of the		paper	excluded	not directly related to healthcare	0.00	http://iiok.opdogog.com/to-st/d/
Springer	09 12 2024	somanath TripathyRudrapatna K. ShyamasundarRajiv Ranjan	Information Systems Security	Conference paper	evolution	keyword mismatch: not directly related	2021	nttp://inik.springer.com/book/10.1007/978-3- 030-92571-0
opiniger	vo. rz.z024	Maurice H. ter BeekDr. Rosemary Monahan	Integrated Formal Methods	Conference paper	excluded	keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-07727-2
Springer	08.12.2024	Katsuhiro HondaBac LeVan-Nam HuynhMasahiro	Integrated Uncertainty in Knowledge Modelling and Decision	Conference	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	InuiguchiYouji Kohda	Making	paper	excluded	not directly related to healthcare	0.00	<u>U31-46/81-3</u>
Springer	08.12.2024	Peter J. STUCKey	integration of Constraint Programming, Artificial Intelligence, and Operations Research	paper	excluded	keyword mismatch: not directly related to healthcare	2021	030-78230-6

Search	Query date	Author	Title	Type	Status	Reason	Vear	Article URI
venue	Query unte	De-Shuang HuangProf. Kang-Hyun JoJianqiang	Intelligent Computing Theories and Application	Conference	Status	keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3- 030-84532-2
Springer	08.12.2024	Livalenya Ghbovarrashan Fremarathe		paper	excluded	to healthcare		
		Dr. Hujun YinDavid CamachoPeter TinoRichard	Intelligent Data Engineering and Automated Learning - IDEAL	Conference			2021	http://ink.springer.com/book/10.1007/978-3-
		AllmendingerDr. Antonio J. Tallon-BallesterosKe TangProf, Sung-Bae ChoPaulo NovaisDr. Susana	2021	paper		keyword mismatch: not directly related		<u></u>
Springer	08.12.2024	Nascimento			excluded	to healthcare		
		Jong-Hoon KimMadhusudan SinghJaved	Intelligent Human Computer Interaction	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 030-98404-5
Springer	08.12.2024	KhanProf. Uma Shanker TiwaryProf. Marigankar SurDhananiay Singh		paper	excluded	to healthcare		<u></u>
opiniger	CONTRACTO	Huayong YangHonghai LiuJun ZouZhouping	Intelligent Robotics and Applications	Conference	0/10/0000	keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-981-
0	00.40.0004	YinLianqing LiuGeng YangXiaoping		paper	avaluate d	not directly related		99-6483-3
Springer	00.12.2024	Murilo C. NaldiReinaldo A. C. Bianchi	Intelligent Systems	Conference	excluded	keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
				paper		not directly related		031-45392-2
Springer	08.12.2024	André BrittoKarina Valdivia Delgado	Intalligent Systems	Conference	excluded	to healthcare	2021	http://link.springer.com/book/10.1007/978-3-
				paper		not directly related		030-91702-9
Springer	08.12.2024	João Carlos Xavier- Junior Picardo Araújo Pice	Intelligent Systems	Conference	excluded	to healthcare	2022	http://link.springer.com/book/10.1007/978-3-
		blab banbs yavier banbrinbards readjo mos	intelligent Oysterno	paper		not directly related	LOLL	031-21686-2
Springer	08.12.2024	Andrew Danabia Danasa Manhahara shawi7hara	Internation Online prototo	0	excluded	to healthcare	0000	http://ink.opringer.com/hook/10.1007/079.2
		Xiantong	Interactive Collaborative Robotics	paper		not directly related	2022	031-23609-9
Springer	08.12.2024	In case in 16 and the law of the company to a	Internet Online of	0	excluded	to healthcare	0047	http://link.optioner.com/book/10.1007/070.2
		SatsiouGeorg CarleDr. Antonella PassaniEfstratios	Internet Science	paper		not directly related	2017	<u>319-70284-1</u>
Springer	08.12.2024	KontopoulosSotiris DiplarisDonald McMillan			excluded	to healthcare		
		Fernando Ortiz-RodriguezBoris Villazon- TerrazasSaniu TiwariCarlos Bobed	Knowledge Graphs and Semantic Web	Conference		keyword mismatch: not directly related	2023	031-47745-4
Springer	08.12.2024			pape.	excluded	to healthcare		
		Gerard MemmiBaijian YangProf. Linghe	Knowledge Science, Engineering and Management	Conference		keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3- 031-10989-8
Springer	08.12.2024	Kong hanwei zhang roit. Dr. weikang die		paper	excluded	to healthcare		
		Dr. Ivan LirkovProf. Svetozar Margenov	Large-Scale Scientific Computing	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 030-97549-4
Springer	08.12.2024			paper	excluded	to healthcare		
		Panayiotis ZaphirisAndri Ioannou	Learning and Collaboration Technologies. Novel Technological	Conference		keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3-
Springer	08 12 2024		Environments	paper	excluded	not directly related to healthcare		531-53073-0
opinger	00.12.2024	Tiziana MargariaProf. Dr. Bernhard Steffen	Leveraging Applications of Formal Methods, Verification and	Conference	excluded	keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
0	00 40 0004		Validation	paper	a valued and	not directly related		030-89159-6
Springer	06.12.2024	Omar AlonsoHelena CousijnGianmaria	Linking Theory and Practice of Digital Libraries	Conference	excluded	keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
		SilvelloMónica MarreroCarla Teixeira LopesStefano		paper		not directly related		031-43849-3
Springer	08.12.2024	Marchesin Georg GottlobDaniela InclezanMarco Maratea	Logic Programming and Nonmonotonic Beasoning	Conference	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		debig dottobbuncia molezarina do maraca	Logic Frogramming and Hormonotome Reasoning	paper		not directly related	2022	031-15707-3
Springer	08.12.2024	Massib Pera AminiStánbana CanuAsia EischerTian	Machine Learning and Knowledge Discovery in Databases	Conference	excluded	to healthcare	2022	http://link.springer.com/book/10.1007/978-3-
		GunsPetra Kralj NovakGrigorios Tsoumakas	Wachine Leaning and Knowledge Discovery in Databases	paper		not directly related	2020	031-26422-1
Springer	08.12.2024	Andreas HelsingerProf. Dr. Dates Kingshove Drof	Machine Learning and Kanudadas Estration	Conference	excluded	to healthcare	0000	http://link.corionar.com/back/10.1007/079-2-
		Dr. A Min TjoaEdgar Weippl	Machine Learning and Knowledge Extraction	paper		not directly related	2020	030-57321-8
Springer	08.12.2024				excluded	to healthcare		
		Giuseppe NicosiaVarun OjhaEmanuele La MalfaGabriele La MalfaPanos PardalosGiuseppe Di	Machine Learning, Optimization, and Data Science	Conference		keyword mismatch: not directly related	2023	http://ink.springer.com/book/10.1007/978-3- 031-25599-1
Springer	08.12.2024	FattaGiovanni GiuffridaRenato Umeton			excluded	to healthcare		
		Liang-Jie Zhang	Metaverse – METAVERSE 2022	Conference		keyword mismatch: not directly related	2022	http://ink.springer.com/book/10.1007/978-3- 031-23518-4
Springer	08.12.2024			papor	excluded	to healthcare		
		Irfan AwanMuhammad YounasProf. Aneta	Mobile Web and Intelligent Information Systems	Conference		keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3- 031-14391-5
Springer	08.12.2024	Politszewska-tviararida		paper	excluded	to healthcare		
		Philippe Fournier-VigerAhmed HassanLadjel	Model and Data Engineering	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-21595-7
Springer	08.12.2024	Bellatreche		paper	excluded	to healthcare		
		Jan MazalAdriano FagioliniPetr VašíkAgostino	Modelling and Simulation for Autonomous Systems	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-31268-7
Springer	08.12.2024	StodolaStefano Lo Storto		paper	excluded	to healthcare		
		Vadim MalvoneAniello Murano	Multi-Agent Systems	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-43264-4
Springer	08 12 2024			paper	excluded	not directly related to healthcare		
opinigei		Raghava MorusupalliTeja Santosh DandibhotlaVani	Multi-disciplinary Trends in Artificial Intelligence	Conference		keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-3-
Springer	09 12 2024	Vathsala AtluriDavid WindridgePawan		paper	oveluded	not directly related		031-30402-0
Springer	00.12.2024	Xin HePh.D. En ShaoGuangming Tan	Network and Parallel Computing	Conference	excluded	keyword mismatch:	2021	http://ink.springer.com/book/10.1007/978-3-
Springer	08 12 2024			paper	ovcluded	not directly related		<u>uau-rs478-1</u>
Springer	00.12.2024	Xingliang YuanGuangdong BaiCristina	Network and System Security	Conference	excluded	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
Springer	09 12 2024	AlcarazSuryadipta Majumdar		paper	oveluded	not directly related		<u>ua1-23020-2</u>
opniger	JU. 12.2024	Marijn JanssenLuiz PinheiroRicardo	New Sustainable Horizons in Artificial Intelligence and Digital	Conference	excluded	keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-3-
Corinana	09 10 0004	MatheusFernanda FrankenbergerYogesh K.	Solutions	paper	oveluster	not directly related		<u>ua1-50040-4</u>
opiniger.	JO.12.2024	Hong ShenDr. Yingpeng SangYong ZhangNong	Parallel and Distributed Computing, Applications and	Conference	excluded	keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3-
0-rin	00.40.000	XiaoDr. Hamid R. ArabniaGeoffrey FoxProf. Ajay	Technologies	paper		not directly related		030-96772-7
Springer	08.12.2024	Victor Malvshkin	Parallel Computing Technologies	Conference	excluded	to nealthcare keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
				paper		not directly related		030-86359-3
Springer	08.12.2024	Günter BudolohAnna V. KononovaHernán	Parallel Problem Solving from Nature - PPSN XVII	Conference	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		AguirrePascal KerschkeGabriela OchoaTea Tušar	Futures Frobient coloring non-nature - FF on XVI	paper		not directly related	LOLL	031-14714-2
Springer	08.12.2024	Roman Wyrzykowski Jack Dongorga Ewo	Parallal Processing and Applied Mathematics	Conference	excluded	to healthcare	2022	http://link.springer.com/book/10.1007/079-9-
		DeelmanKonrad Karczewski	a anon nocessing and Applied Mathematics	paper		not directly related	2023	031-30445-3
Springer	08.12.2024	Annal Vean Badringer Operation to the Dr	Pottora Possanition	Contant	excluded	to healthcare	0000	http://ink.springer.com/back/10.1007/070.0
		Ansei toan Kodriguez-GonzálezHumberto Pérez- EspinosaJosé Francisco Martínez-TrinidadJesús	Pattern Necognition	paper		keyword mismatch: not directly related	2023	031-33783-3
Springer	08.12.2024	Ariel Carrasco-OchoaJosé Arturo Olvera-López			excluded	to healthcare		hate (Mal) and an end of the state of the
		Mounim El YacoubiEric GrangerPong Chi YuenUmapada PalNicole Vincent	Pattern Recognition and Artificial Intelligence	Conference paper		keyword mismatch: not directly related	2022	mtp://ink.springer.com/book/10.1007/978-3- 031-09037-0
Springer	08.12.2024			holes.	excluded	to healthcare		
		Shiqi YuZhaoxiang ZhangPong C. YuenJunwei HanTieniu TanYike Guo lianhuang Lai liangun	Pattern Recognition and Computer Vision	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-18907-4
Springer	08.12.2024	Zhang		hahai	excluded	to healthcare		
		Jean-Jacques RousseauBill Kapralos	Pattern Recognition, Computer Vision, and Image Processing.	Conference		keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-3- 031-37742-6
Springer	08.12.2024		CFT 2022 International Workshops and Challenges	paper	excluded	to healthcare		
		Roxane LicandroAndrew MelbourneEsra Abaci	Perinatal, Preterm and Paediatric Image Analysis	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-17117-8
Springer	08.12.2024	ruikonnstopher macgowanjana Hutter		paper	excluded	to healthcare		
,		Sankalp KhannaJian CaoQuan BaiGuandong Xu	PRICAI 2022: Trends in Artificial Intelligence	Conference		keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3- 031-20868-3
Springer	08.12.2024			paper	excluded	to healthcare		And

venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
		Tim MillerNir OrenYuko Sakurailtsuki NodaBastin Tony Roy SavarimuthuTran Cao Son	PRIMA 2018: Principles and Practice of Multi-Agent Systems	Conference paper		keyword mismatch: not directly related	2018	http://link.springer.com/book/10.1007/978-3- 030-03098-8
Springer	08.12.2024	Agnicentes Counterprésie Proprietour Poloécké Nile	Privacy Technologica, and Policy	Conforance	excluded	to healthcare	2022	http://link.springer.com/book/10.1007/978-3-
Springer	08 12 2024	GruschkaProf. Dr. Kai RannenbergMonika Adamczyk	Privacy recimologies and Policy	paper	excluded	not directly related	2022	031-07315-1
opinigoi		Pekka AbrahamssonAndreas JedlitschkaAnh Nguyen DucMichael FeldererSousuke	Product-Focused Software Process Improvement	Conference paper		keyword mismatch: not directly related	2016	http://link.springer.com/book/10.1007/978-3- 319-49094-6
Springer	08.12.2024	AmasakiTommi Mikkonen Prof. Dr. Xavier FranchTomi MännistöSilverio	Product-Focused Software Process Improvement	Conference	excluded	to healthcare keyword mismatch:	2019	http://ink.springer.com/book/10.1007/978-3-
Springer	08.12.2024	Martínez-Fernández		paper	excluded	not directly related to healthcare		
Springer	08 12 2024	KlünderPekka Abrahamsson	Product-Pocused Software Process Improvement	paper	excluded	not directly related	2022	031-21388-5
opiniger	00.12.2024	Dr. Maurizio MorisioProf. Marco TorchianoAndreas Jedlitschka	Product-Focused Software Process Improvement	Conference paper	excluded	keyword mismatch: not directly related	2020	http://link.springer.com/book/10.1007/978-3- 030-64148-1
Springer	08.12.2024	Dr. Marco KuhrmannProf. Dr. Kurt	Product-Focused Software Process Improvement	Conference	excluded	to healthcare	2018	http://link.springer.com/book/10.1007/978-3-
		SchneiderDietmar PfahlSousuke AmasakiMarcus CiolkowskiProf. Dr. Regina HebigDr. Paolo TellJil		paper		keyword mismatch: not directly related		030-03673-7
Springer	08.12.2024	Riunderstetten Rupper Goreti MarreirosBruno MartinsAna PaivaBernardete BibeiroAlberto Sardinha	Progress in Artificial Intelligence	Conference	excluded	to neartncare keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-16474-3
Springer	08.12.2024	Nuno MonizZita ValeJosé CascalhoCatarina	Progress in Artificial Intelligence	Conference	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	SilvaRaquel Sebastião		paper	excluded	not directly related to healthcare		031-49008-8
Springer	08 12 2024	Prot. João Manuel R. S. TavaresJoão Paulo PapaManuel González Hidalgo	Progress in Pattern Hecognition, image Analysis, Computer Vision, and Applications	paper	excluded	not directly related to healthcare	2021	030-93420-0
g		Boris GlavicVanessa BraganholoDr. David Koop	Provenance and Annotation of Data and Processes	Conference paper		keyword mismatch: not directly related	2021	http://ink.springer.com/book/10.1007/978-3- 030-80960-7
Springer	08.12.2024	Erika ÁbrahámMarco Paolieri	Quantitative Evaluation of Systems	Conference	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-16336-4
Springer	08.12.2024	Alessio FerrariBirnit Penzenstadler	Requirements Engineering: Foundation for Software Quality	Conference	excluded	to healthcare	2023	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	Alessio Ferraribligit Perizenstadier	requirements Engineering, roundation or Gortware quality	paper	excluded	not directly related to healthcare	2020	031-29786-1
		Haixu Tang	Research in Computational Molecular Biology	Conference paper		keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-29119-7
Springer	08.12.2024	Dr. Denis DennehyAnastasia GrivaDr. Nancy	Responsible AI and Analytics for an Ethical and Inclusive	Conference	excluded	to healthcare keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3- 030-85447-8
Springer	08.12.2024	PouloudiProt. Yogesh K. DwivediDr. Illas PappasMatti Mäntymäki Mariá HattinghMachdel MattheeHanlie SmutsDr	Digitized Society Responsible Design Implementation and Lise of Information	Conference	excluded	to healthcare	2020	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	Ilias PappasPh.D. Yogesh K. DwivediMatti Mäntymäki	and Communication Technology	paper	excluded	not directly related to healthcare	2020	030-44999-5
0	00.40.0004	Slim KallelMohamed JmaielMohammad ZulkernineAhmed Hadj KacemFrédéric	Risks and Security of Internet and Systems	Conference paper		keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-31108-6
Springer	08.12.2024	Stephan ChalupTim NiemuellerJackrit SuthakornMary-Anne Williams	RoboCup 2019: Robot World Cup XXIII	Conference paper	excluded	keyword mismatch: not directly related	2019	http://link.springer.com/book/10.1007/978-3- 030-35699-6
Springer	08.12.2024	Sotiris MoschoyiannisProf. Rafael PeñalozaJan	Rules and Reasoning	Conference	excluded	to healthcare keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3- 030-91167-6
Springer	08.12.2024	Hans P. ReiserMarcel Kvas	Secure IT Systems	Conference	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024			paper	excluded	not directly related to healthcare		031-22295-5
Springer	08 12 2024	Budi AriefAnna MonrealeMichael SirivianosShujun Li	Security and Privacy in Social Networks and Big Data	Conference paper	excluded	keyword mismatch: not directly related to healthcare	2023	http://link.springer.com/book/10.1007/978-981- 99-5177-2
opinger	00.12.2024	Carmelo ArditoRosa LanzilottiAlessio MaliziaProf. Marta LarusdottirLucio Davide SpanoDr. José	Sense, Feel, Design	Conference paper	excluded	to nearricare	2022	http://link.springer.com/book/10.1007/978-3- 030-98388-8
-		CamposMorten HertzumTilo MentlerJosé Abdelnour NoceraDr. Lara PiccoloStefan				keyword mismatch: not directly related		
Springer	08.12.2024	SauerGemt van der veer George A. PapadopoulosFlorian Bademacher Jacopo Soldani	Service-Oriented and Cloud Computing	Conference	excluded	to nealthcare keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-46235-1
Springer	08.12.2024	Javier TroyaBrahim MedjahedMario PiattiniLina	Service-Oriented Computing	Conference	excluded	to healthcare keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3-
Springer	08.12.2024	YaoPablo FernándezAntonio Ruiz-Cortés		paper	excluded	not directly related to healthcare	0001	031-20984-0
Springer	08.12.2024	MohaHye-young Paik	Service-Oriented Computing	paper	excluded	not directly related to healthcare	2021	030-91431-8
		Hakim HacidFatma OutayHye-young PaikAmira AlloumMarinella PetrocchiMohamed Reda	Service-Oriented Computing – ICSOC 2020 Workshops	Conference paper		keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3- 030-76352-7
Springer	08.12.2024	LiuAbderrahmane Maaradji Prof. Dr. Meikang QiuKeke GaiHan Qiu	Smart Computing and Communication	Conference	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	-		paper	excluded	not directly related to healthcare		030-97774-0
Springer	08 10 2024	Gabriele Meiselwitz	Social Computing and Social Media: Applications in Education and Commerce	Conference paper	excluded	keyword mismatch: not directly related to healthcare	2022	nttp://link.springer.com/book/10.1007/978-3- 031-05064-0
opinger	00.12.2024	Frank HopfgartnerKokil JaidkaPhilipp MayrJoemon JoseJan Breitsohl	Social Informatics	Conference paper	excluded	keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-19097-1
Springer	08.12.2024	Alan R. WagnerPh.D. David Feil-SeiferKerstin S.	Social Robotics	Conference	excluded	to healthcare keyword mismatch:	2020	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	HaringProf. Silvia RossiThomas WilliamsHongsheng HeProf. Dr. Shuzhi Sam Ge Robert ThomsonSamer Al-khateehAnnetta	Social Cultural and Behavioral Modeling	Conference	excluded	to healthcare	2023	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	BurgerPatrick ParkAryn A. Pyke	coona, ourara, and penavioral modelinly	paper	excluded	not directly related to healthcare	2023	031-43129-6
		Simon ParkinLuca Viganò	Socio-Technical Aspects in Security	Conference paper		keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-10183-0
Springer	U8.12.2024	Bedir TekinerdoganCatia TrubianiChouki TibermacinePatrizia ScandurraCarlos E. Cuesta	Software Architecture	Conference	excluded	to healthcare keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-42592-9
Springer	08.12.2024	UnivProf. Dr. Stefan BifflElena NavarroWelf	Software Architecture	Conference	excluded	to healthcare keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	LöweProf. Marjan SirjaniProf. Raffaela MirandolaProf. Dr. Danny Weyns	Polituara Ambitantura	paper	excluded	not directly related to healthcare	0.000	USU-55044-8
Springer	08.12.2024	mas GerostathopoulosGrace LewisThais BatistaTomáš Bureš	Sontware Architecture	conference paper	excluded	keyword mismatch: not directly related to healthcare	2022	031-16697-6
		Thais BatistaTomáš BurešClaudia RaibuletHenry Muccini	Software Architecture. ECSA 2022 Tracks and Workshops	Conference paper	manarad	keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-36889-9
Springer	08.12.2024	Carla FerreiraTim A. C. Willemse	Software Engineering and Formal Methods	Conference	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-47115-5
Springer	08.12.2024			haher	excluded	to healthcare		

venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
rende	autry auto	Antonio CeroneMarco AutiliAlessio	Software Engineering and Formal Methods. SEFM 2021	Conference	otatao	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		BucaioniCláudio GomesPierluigi GrazianiMaurizio	Collocated Workshops	paper		not directly related		031-12429-7
Springer	08.12.2024	PalmieriMarco TemperiniGentiane Venture			excluded	to healthcare		
		Alexey KarpovK. SamudravijayaK. T. DeepakBajash M. HeadeSbyam S. AgrawalS. R	Speech and Computer	Conference		keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-3- 031-48309-7
Springer	08 12 2024	Deepakkajesh M. Hegdeshyam S. Agrawais. H. Mabadeva Prasanna		paper	evoluded	to healthcare		
opiligei	00.12.2024	Silvia BonfantiAngelo GargantiniPaolo Salvaneschi	Testing Software and Systems	Conference	excluded	keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
				paper		not directly related		031-43240-8
Springer	08.12.2024			<u> </u>	excluded	to healthcare		http://finit.com/sect/cite/1007/070-0
		Andreas HarthSabrina KirraneAxei-Cyrille Ngonga NgomoHeiko PaulheimAnisa BulaAnna Lisa	The Semantic Web	Conterence		keyword mismatch:	2020	030-49461-2
Springer	08.12.2024	GentileDr. Peter HaaseMichael Cochez		paper	excluded	to healthcare		
		Dr. Aldo GangemiProf. Roberto NavigliMaria-	The Semantic Web	Conference		keyword mismatch:	2018	http://link.springer.com/book/10.1007/978-3-
		Esther VidalProf. Dr. Pascal HitzlerRaphaël		paper		not directly related		319-93417-4
Springer	08.12.2024	IroncyLaura HollinkAnna TordaiMenwish Alam	The Semantic Web	Conference	excluded	to healthcare	2021	http://link.springer.com/book/10.1007/978-3-
		Antoine ChampinMaria MaleshkovaOscar	The Gentantic Web	paper		not directly related	2021	030-77385-4
Springer	08.12.2024	CorchoPetar RistoskiDr. Mehwish Alam			excluded	to healthcare		
		Paul GrothMaria-Esther VidalFabian	The Semantic Web	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-06981-9
Springer	08 12 2024	SuchanekPedro SzekleyPavan KapanipathiCatia PesquitaHala Skaf-MolliMinna Tamper		paper	evoluded	to healthcare		
opinger	00.12.2024	Denny VrandečićKalina BontchevaMari Carmen	The Semantic Web – ISWC 2018	Conference	CACINGGO	to notatilotaro	2018	http://link.springer.com/book/10.1007/978-3-
		Suárez-FigueroaValentina Presuttilrene		paper		keyword mismatch:		030-00668-6
0	00 40 0004	CelinoMarta SabouLucie-Aimée KaffeeElena			and a dealer of	not directly related		
Springer	08.12.2024	Chiara GhidiniOlaf HartigMaria MaleshkovaVoitěch	The Semantic Web - ISWC 2019	Conference	excluded	keyword mismatch:	2019	http://link.springer.com/book/10.1007/978-3-
		Sváteklsabel CruzAidan HoganJie SongMaxime		paper		not directly related	2010	030-30796-7
Springer	08.12.2024	LefrançoisFabien Gandon			excluded	to healthcare		
		Jeff Z. PanValentina TammaClaudia d'	The Semantic Web – ISWC 2020	Conference		keyword mismatch:	2020	http://link.springer.com/book/10.1007/978-3- 030-62466-8
Springer	08 12 2024	Amatokrzysztor Janowiczbo FuAxel PolleresOshani Seneviratnel alana Kagal		paper	excluded	not directly related		
opinigei	00.12.2024	Prof. Dr. Andreas HothoAssoc. Prof. Eva	The Semantic Web - ISWC 2021	Conference	CACINGGO	to notation of	2021	http://link.springer.com/book/10.1007/978-3-
		BlomqvistStefan DietzeAchille FokoueYing		paper		keyword mismatch:		030-88361-4
0	00.40.0004	DingPayam BarnaghiArmin HallerDr. Mauro			and a start	not directly related		
Springer	08.12.2024	Ulrike SattlerAidan HoganMaria KeetValentina	The Semantic Web - ISWC 2022	Conference	excluded	to nealtricare	2022	http://link.springer.com/book/10.1007/978-3-
		PresuttiJoão Paulo A. AlmeidaHideaki		paper		keyword mismatch:	2022	031-19433-7
		TakedaPierre MonninGiuseppe PirròClaudia d'				not directly related		
Springer	08.12.2024	Amato	The Competie Web ICWC 0000	Conference	excluded	to healthcare	0005	http://link.spripper.com/back/10.1002/076-0
		rerry n. Paynevalentina PresuttiGuilin QiMaria Poveda-VillalónGiorgos Stoilost aura HollinkZoi	The Semantic Web - ISWG 2023	Conterence paper		keyword mismatch: not directly related	2023	031-47243-5
Springer	08.12.2024	KaoudiGong ChengJuanzi Li		paper	excluded	to healthcare		
		Aldo GangemiAnna Lisa GentileAndrea Giovanni	The Semantic Web: ESWC 2018 Satellite Events	Conference			2018	http://link.springer.com/book/10.1007/978-3-
		NuzzoleseSebastian RudolphMaria		paper		keyword mismatch:		718-80182-2
Springer	08 12 2024	Alam			excluded	to healthcare		
opinger	CO.TE.LOL4	Andreas DrechslerAurona GerberAlan Hevner	The Transdisciplinary Reach of Design Science Research	Conference	Choradoa	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
				paper		not directly related		031-06516-3
Springer	08.12.2024	Versies All AsseurFlacis Origins	Theoretical Associate of Collinson Engineering	Castanana	excluded	to healthcare	0000	http://link.opringer.com/back/10.1007/078-2-
		Tamine Alt-Aneumonn Gracium	Theoretical Aspects of Software Engineering	paper		not directly related	2022	031-10363-6
Springer	08.12.2024				excluded	to healthcare		
		Charles FoxJunfeng GaoAmir Ghalamzan	Towards Autonomous Robotic Systems	Conference		keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3- 030-89177-0
Springer	08 12 2024	Estananimini Saajmarc HanneideSimon Parsons		paper	excluded	to healthcare		
		Fumiya lidaPerla MaiolinoArsen AbdulaliMingfeng	Towards Autonomous Robotic Systems	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
0	00 40 0004	Wang		paper		not directly related		031-43300-3
Springer	08.12.2024	Prof Hao-Ben KeChei Sian LeeKazunari Sugiyama	Towards Open and Trustworthy Digital Societies	Conference	excluded	to nealthcare	2021	http://link.springer.com/book/10.1007/978-3-
		non nao non noonor olan coonacanan ougiyana	torial do opon and indonionity bigital obolotios	paper		not directly related	LOLI	030-91669-5
Springer	08.12.2024				excluded	to healthcare		
		Simone Fischer-HübnerProf. Costas	Trust, Privacy and Security in Digital Business	Conference		keyword mismatch:	2021	http://ink.springer.com/book/10.1007/978-3- 030-86586-3
Springer	08.12.2024	Min Tjoalsmail Khalil		paper	excluded	to healthcare		
		Margherita AntonaConstantine Stephanidis	Universal Access in Human-Computer Interaction	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-35897-5
Springer	08 12 2024			paper	excluded	to healthcare		
		Bohan LiLin YueChuanqi TaoXuming HanDiego	Web and Big Data	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
		CalvaneseToshiyuki Amagasa		paper		not directly related		031-25158-0
Springer	08.12.2024	Irene Garrigós Juan Manuel Murillo	Web Engineering	Conference	excluded	to neartricare keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
		RodríguezManuel Wimmer		paper		not directly related		031-34444-2
Springer	08.12.2024				excluded	to healthcare		
		Prof. Tommaso Di NoiaDr. In-Young KoDr. Markus SchediCarmelo Ardito	Web Engineering	Conference		keyword mismatch:	2022	http://ink.springer.com/bdok/10.1007/978-3- 031-09917-5
Springer	08.12.2024	Schedioanneio Aldiio		paper	excluded	to healthcare		
		Ph.D. Marco BrambillaProf. Richard ChbeirFlavius	Web Engineering	Conference		keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
		FrasincarDr. Ioana Manolescu		paper		not directly related		030-74296-6
Springer	08.12.2024	Chunyiao XingProf Xiaoming EuVong	Web Information Systems and Applications	Conference	excluded	to neartricare	2021	http://link.springer.com/book/10.1007/978-3-
		ZhangGuigang ZhangChaolemen Borjigin		paper		not directly related	2321	030-87571-8
Springer	08.12.2024				excluded	to healthcare		
		wenjie ZhangLei ZouProf. Dr. Zakaria MaamarDr. Lu Chen	web information Systems Engineering – WISE 2021	Conference		keyword mismatch:	2021	030-90888-1
Springer	08.12.2024	100 00 001		hahai	excluded	to healthcare		
opinigei		Richard ChbeirHelen HuangFabrizio SilvestriYannis	Web Information Systems Engineering – WISE 2022	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		ManolopoulosYanchun Zhang		paper		not directly related		031-20891-1
Springer	08.12.2024	Zhe LiuEan WuProf Saial K Das	Wireless Algorithms Systems and Applications	Conference	excluded	to nearthcare keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
			The second generation of second and high second	paper		not directly related	LULI	030-85928-2
Springer	08.12.2024			o (	excluded	to healthcare		http://www.com/www.com/www.com/www.com/
		Thangaprakash SengodanM. MurugappanSanjay Misra	Advances in Electrical and Computer Technologies	conterence		not directly related	2022	19-1111-8
Springer	06.12.2024			bedra:	excluded	to healthcare		
		Ankur ChoudharyArun Prakash AgrawalRajasvaran	Applications of Artificial Intelligence and Machine Learning	Conference		keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-981- 16-3067-5
Springer	07.12 2024	LogeswaranDr. Bhuvan Unhelkar		paper	excluded	not directly related to healthcare		
opiniger	07.12.2024	Prof. Rajeev R. RajeDr. Farookh HussainDr. R.	Artificial Intelligence and Technologies	Conference	excluded	keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-981-
		Jagadeesh Kannan		paper		not directly related		10-0448-9
Springer	07.12.2024	Prof. Oilian LiangWei Wang Jiasong MuDr. Yin	Artificial Intelligence in China	Conference	excluded	to nealthcare keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-981-
		LiuProf. Zhenyu Na		paper		not directly related	LULL	16-9423-3
Springer	07.12.2024				excluded	to healthcare		had a second
		Prof. Ch. SatyanarayanaProf. Debasis	High Performance Computing and Networking	Conference		keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-981- 16-9885-9
Springer	08.12.2024	Kapoor		paper	excluded	to healthcare		
		Yan PeiJia-Wei ChangJason C. Hung	Innovative Computing	Conference		keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-981- 19-4132-0
Springer	09 10 0004			paper	ovolustast	not directly related		- 1.0 1.0 - M
opiniger	50.12.2024	Jason C. HungJia-Wei ChangYan Pei	Innovative Computing Vol 1 - Emerging Topics in Artificial	Conference	excidued	keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-981-
			Intelligence	paper		not directly related		99-2092-1
Springer	08.12.2024	Jason C. Hung Jia-Wei Chang Van Bai	Innovative Computing Vol 2 - Emorping Topics in Future	Conference	excluded	to healthcare	2022	http://link.springer.com/book/10.1007/978-981-
		oncon on nungoia-wei Onang tall Pël	Internet	paper		not directly related	2023	99-2287-1
Springer	08.12.2024				excluded	to healthcare		

venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
		Siddhartha BhattacharyyaMario KoeppenDebashis DeVincenzo Piuri	Intelligent Systems and Human Machine Collaboration	Conference		keyword mismatch: not directly related	2023	http://ink.springer.com/book/10.1007/978-981- 19-8477-8
Springer	08.12.2024	Devincenzo Fidir		paper	excluded	to healthcare		
		Koushlendra Kumar SinghManish Kumar BainaiAkbar Sheikh Akbari	Machine Vision and Augmented Intelligence	Conference		keyword mismatch: not directly related	2023	http://ink.springer.com/book/10.1007/978-981- 99-0189-0
Springer	08.12.2024	bajparkoar oneknirkoar		ραροι	excluded	to healthcare		
		Sarika JainSven GroppeBharat K. Bhargava	Semantic Intelligence	Conference		keyword mismatch: not directly related	2023	http://ink.springer.com/book/10.1007/978-981- 19-7126-6
Springer	08.12.2024			paper	excluded	to healthcare		
		Md. Sazzad HossainSatya Prasad MajumderNazmul SiddigueMd, Shahadat Hossain	The Fourth Industrial Revolution and Beyond	Conference		keyword mismatch: not directly related	2023	http://ink.springer.com/book/10.1007/978-981- 19-8032-9
Springer	08.12.2024	Wajumderwazmur oldulquewid, onanadat nossain		рары	excluded	to healthcare		
		Ph.D. Rodrigo Tapia-McClungProf. Oscar Sánchez-SiordiaMag, Karime González-	Advances in Geospatial Data Science	Conference		keyword mismatch: not directly related	2022	http://ink.springer.com/book/10.1007/978-3- 030-98096-2
Springer	06.12.2024	ZuccolottoDr. Hugo Carlos-Martínez		paper	excluded	to healthcare		
		Danilo TardioliVicente MatellánGuillermo HerediaManuel E. Silval ino Marques	ROBOT2022: Fifth Iberian Robotics Conference	Conference		keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-21062-4
Springer	08.12.2024	The column and the onvalue of warques		paper	excluded	to healthcare		
		Yu-Dong ZhangTomonobu SenjyuChakchai So- InAmit Joshi	Smart Trends in Computing and Communications	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-981- 16-9967-2
Springer	08.12.2024			ραροι	excluded	to healthcare		
		Weina FuLin Yun	Advanced Hybrid Information Processing	Conference		keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-3- 031-28867-8
Springer	06.12.2024			paper	excluded	to healthcare		
		Ao LiYao ShiLiang Xi	6GN for Future Wireless Networks	Conference		keyword mismatch: not directly related	2023	http://ink.springer.com/book/10.1007/978-3- 031-36014-5
Springer	06.12.2024			рары	excluded	to healthcare		
		Shuai LiuDr. Xuefei Ma	Advanced Hybrid Information Processing	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 030-94551-0
Springer	06.12.2024			paper	excluded	to healthcare		
		Mian Ahmad JanFazlullah Khan	Application of Big Data, Blockchain, and Internet of Things for	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-23950-2
Springer	07.12.2024		Education mormatization	paper	excluded	to healthcare		
		Jingqiang LinQiang Tang	Applied Cryptography in Computer and Communications	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-17081-2
Springer	07.12.2024			paper	excluded	to healthcare		
		Rui HouHuan HuangDeze ZengGuisong XiaKareem	Big Data Technologies and Applications	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-33614-0
Springer	07.12.2024	Ramar A. GhanyHossam W. Zawbaa		paper	excluded	to healthcare		
		Honghao GaoXinheng WangWei WeiTasos	Collaborative Computing: Networking, Applications and	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-24383-7
Springer	07.12.2024	Dagiunias	workshanng	paper	excluded	to healthcare		
		Tanya ZlatevaGeorgi Tuparov	Computer Science and Education in Computer Science	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-44668-9
Springer	07.12.2024			paper	excluded	to healthcare		
		Yahya H. Sheikhldris A. RaiAbubakar D. Bakar	e-Infrastructure and e-Services for Developing Countries	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-06374-9
Springer	07.12.2024			paper	excluded	to healthcare		
		Weina FuShuai LiuJianhua Dai	e-Learning, e-Education, and Online Training	Conference		keyword mismatch:	2021	http://ink.springer.com/book/10.1007/978-3- 030-84383-0
Springer	07.12.2024			paper	excluded	to healthcare		
		Sachi Nandan MohantyVicente Garcia DiazG. A. E.	Intelligent Systems and Machine Learning	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-35081-8
Springer	08.12.2024	Satish Kumar		paper	excluded	to healthcare		
		Md. Shahriare SatuMohammad Ali MoniM.	Machine Intelligence and Emerging Technologies	Conference		keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-3- 031-34619-4
Springer	08.12.2024	Shamim Kaiserwohanimad Shamsul Arelin		paper	excluded	to healthcare		
		Shangguan LongfeiPriyantha Bodhi	Mobile and Ubiquitous Systems: Computing, Networking and	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-34776-4
Springer	08.12.2024		Services	paper	excluded	to healthcare		<u></u>
		Yan ChenggangWang HonggangLin Yun	Mobile Multimedia Communications	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-23902-1
Springer	08.12.2024			paper	excluded	to healthcare		
		Shui-Hua WangYu-Dong Zhang	Multimedia Technology and Enhanced Learning	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-18123-8
Springer	08.12.2024			paper	excluded	to healthcare		
		Fengjun LiKaitai LiangZhiqiang LinSokratis K.	Security and Privacy in Communication Networks	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-25538-0
Springer	08.12.2024	Katsikas		paper	excluded	to healthcare		
		Prof. Joaquin Garcia-AlfaroProf. Shujun LiProf.	Security and Privacy in Communication Networks	Conference		keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3- 030-90022-9
Springer	08.12.2024	Radna PoovendranHerve DebarProt. Moti Yung		paper	excluded	to healthcare		
		Wenbo ShiProf. Xiaofeng ChenProf. Kim-Kwang	Security and Privacy in New Computing Environments	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 030-96791-8
Springer	08.12.2024	Raymond Choo		paper	excluded	to healthcare		
		Prof. Dingde JiangHoubing Song	Simulation Tools and Techniques	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 030-97124-3
Springer	08.12.2024			paper	excluded	to healthcare		
		Eric C. K. ChengRekha B. KoulTianchong	Artificial Intelligence in Education: Emerging Technologies,	Conference		keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-981- 16-7527-0
Springer	07.12.2024	wangxinguo tu	models and Applications	paper	excluded	to healthcare		
			Pabuat Lagal Light Field Suptagia via Opalusian awara			keyword mismatch:		
Springer	08.12.2024	Wenpeng XingJie ChenYike Guo	Sampling and Deep Visual Feature Fusion	Article	excluded	to healthcare	2023	http://link.springer.com/article/10.1007/s11633- 022-1381-9
		Ping ZhouYixue Hao lun YangWei Litu				keyword mismatch:		
Springer	07.12.2024	WangYiming MiaoJeungeun Song	Cloud-assisted hugtive robot for affective interaction	Article	excluded	to healthcare	2017	http://ink.springer.com/article/10.1007/s11042- 016-3849-5
		Anja KeskinarkausSami HuttunenAntti SiipoJukka				keyword mismatch-		
		VäyrynenJanne HeikkiläMatti LehtihalmesTapio	MORE - a multimodal observation and analysis system for			not directly related		http://ink.springer.com/article/10.1007/s11042.
Springer	08.12.2024	SeppänenSeppo Laukka	social interaction research	Article	excluded	to healthcare	2016	015-2574-9
		Federico CabitzaDaniela FogliRosa	Rule-based tools for the configuration of ambient intelligence			not directly related		http://ink.epringer.com/article/10.1007/e11042-
Springer	08.12.2024	LanzilottiAntonio Piccinno	systems: a comparative user study	Article	excluded	to healthcare	2017	016-3511-2
		ChiromaHabib ShahAbdulsalam Y. Gitallbrahim	Deep learning application in smart cities: recent development,			not directly related		http://ink.opringer.com/article/10.1007/c00521
Springer	07.12.2024	Abaker Targio Hashem	taxonomy, challenges and research prospects	Article	excluded	to healthcare	2021	020-05151-8
		Zaru Abdi Aikareem AiyasseriOsama Ahmad AlomariMohammed Azmi Al-BetarSharif Naser				keyword mismatch:		
Coningent	09 10 000 1	Makhadmehiyad Abu DoushMohammed A.	Recent advances of bat-inspired algorithm, its versions and	Article	and the second	not directly related	2000	http://link.springer.com/article/10.1007/s00521-
opringer	08.12.2024	AwayananAmmar Kamar AbasiAShrat Einagar	Some new hybrid hesitant fuzzy weighted aggregation	Arucie	excluded	keyword mismatch:	2022	<u>U22-07662-y</u>
Contractor	00.40.0001	Abbilit SabaDabilt Dutte Samadit Van	operators based on Archimedean and Dombi operations for	Article	auchus .	not directly related	0001	http://link.springer.com/article/10.1007/s00521-
Springer	08.12.2024	Abnijit SanaDebjit DuttaSamarjit Kar	multi-attribute decision making	Article	excluded	keyword mismatch:	2021	<u>020-05623-x</u>
Corina	09 10 000 /	R Pando A Dávila	Software Testing in the DevOps Context: A Systematic	Article	aughorie i	not directly related	2000	http://link.springer.com/article/10.
opringer	08.12.2024	D. Fantooa, Davila	wapping Study	Article	excluded	keyword mismatch:	2022	1134/S0361768822080175
Coderan	09 10 0001	Diego Güemes-PeñaCarlos López-NozalRaúl	Emerging topics is mining activers reported	Article	auchus .	not directly related	2010	http://ink.springer.com/article/10.1007/s13748-
opringer	08.12.2024	manuourena-bancriezJesus maudes-Haedo	Linerging topics in mining software repositories	Arucle	excluded	keyword mismatch:	2018	<u>u18-0147-7</u>
Coringer	07 10 0004	Waleed AbdeenXingru ChenMichael	An approach for performance requirements verification and	Article	ovoluted	not directly related	2022	http://link.springer.com/article/10.1007/s00766-
opninger.	07.12.2024	Gitter Administeriner	sos oraionnens genelation	ALCIO	excluded	keyword mismatch:	2023	<u>982-00319-3</u>
Coring	00.10.0007	Savaaaba MaadalRabia Coosthillor oo Sir	Modular norm models: practical representation and analysis of	Article	evel at a	not directly related	2020	http://link.springer.com/article/10.1007/s00766-
opinger	00.12.2024	Gayorinna manuainooni Ganuniffarvey Siy	consactual rights and obligations	mucle	excluded	to nearchCare	2020	<u>018-00323-y</u>

venue	Query date	Author	Title	Type	Status	Reason	Year	Article URL
			Organization of Information Works, Information, Analytical,	.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		keyword mismatch:		
			Hardware, and Technological Support for Decision Making of			not directly related		http://link.springer.com/article/10,
Springer	08.12.2024	O. A. Melnikov	the State Authorities	Article	excluded	to healthcare	2021	3103/S0147688221030102
			Mapping of topics in DESIDOC, Journal of Library and			keyword mismatch:		
Springer	08 12 2024	Manika LambaMargam Madhusudhan	Information Technology, India: a study	Article	excluded	to healthcare	2019	http://link.springer.com/article/10.1007/s11192- 019-03137-5
opiniger	00.12.2024		internation reenhology, mata a classy	70000	CADIDUCU	keyword mismatch:	2010	010-0010/-0
		Bachar KachouhKhalil HarissLayth SlimanAbed	Privacy preservation of genome data analysis using			not directly related		http://link.springer.com/article/10.1007/s11761-
Springer	08.12.2024	Ellatif SamhatTamim Alsuliman	homomorphic encryption	Article	excluded	to healthcare	2021	021-00326-0
			Two layers service middleware for non-smart IoT services			keyword mismatch:		
Springer	08.12.2024	Yuran DongZuoving ZengCheng Xie	case studies on industrial applications	Article	excluded	to healthcare	2022	http://link.springer.com/article/10.1007/s11761- 022.00339-3
opinigoi	CONTRIECT				enterer enterer	keyword mismatch:		MAR NUMBER OF
		Sebastian HerzogDaniel Schiepellsabella	A Probabilistic Particle Tracking Framework for Guided and			not directly related		http://link.springer.com/article/10.1007/s42979-
Springer	06.12.2024	GuidoRobin BartaClaus Wagner	Brownian Motion Systems with High Particle Densities	Article	excluded	to healthcare	2021	021-00879-z
		Mohammed Imran Basheer AhmedShamsah	A Device on Machine Languise Assessments in Island Washing of			keyword mismatch:		
Springer	06 12 2024	NabilAbdullah Omar AlTurki	Pediatric Enilepsy	Article	evoluded	to healthcare	2022	http://link.springer.com/article/10.1007/s42979- 022.01258-0
opinigoi	CONTENEDET	Said DaoudaghEda MarchettiAntonello			Choladoa	keyword mismatch:		022-01000-0
		CalabròFilipa FerradaAna Inês OliveiraJosé	DAEMON: A Domain-Based Monitoring Ontology for IoT			not directly related		http://link.springer.com/article/10.1007/s42979-
Springer	07.12.2024	BarataRicardo PeresFrancisco Marques	Systems	Article	excluded	to healthcare	2023	023-01975-y
			Data-Driven Requirements Elicitation: A Systematic Literature			reyword mismatch:		
Springer	07.12.2024	Sachiko LimAron HenrikssonJelena Zdravkovic	Review	Article	excluded	to healthcare	2021	http://link.springer.com/article/10.1007/s42979- 020-00416-4
						keyword mismatch:		
0		Shalini DhimanGanesh Kumar MahatoSwarnendu	Homomorphic Encryption Library, Framework, Toolkit and	Antinia		not directly related	0000	http://link.springer.com/article/10.1007/s42979-
Springer	08.12.2024	Kumar Chakraborty	Accelerator: A Review How is People's Awareness of "Biodiversity" Measured? Lising	Article	excluded	to nealthcare	2023	023-02316-9
			Sentiment Analysis and LDA Topic Modeling in the Twitter			not directly related		
Springer	08.12.2024	Shimon Ohtani	Discourse Space from 2010 to 2020	Article	excluded	to healthcare	2022	http://ink.springer.com/article/10.1007/s42979- 022-01276-w
						keyword mismatch:		
0	07 40 0004	Md Rafiqul IslamShaowu LiuXianzhi	Deep learning for misinformation detection on online social	Article	and solved	not directly related	0000	http://link.springer.com/article/10.1007/s13278-
Springer	07.12.2024	Martin HaeuslerThomas Trojer Johannes	networks: a survey and new perspectives	Article	excluded	to nealtricare	2020	020-00696-x
		KesslerMatthias FarwickEmmanuel	ChronoSphere: a graph-based EMF model repository for IT			not directly related		http://link.springer.com/article/I/D-40071-40070-
Springer	07.12.2024	NowakowskiRuth Breu	landscape models	Article	excluded	to healthcare	2019	019-00725-0
		Antonia BertolinoAntonello Calabro'Felicita Di						
		GiandomenicoGiuseppe LamiFrancesca	A tour of secure software engineering colutions for connected			keyword mismatch:		
Springer	06 12 2024	MatteucciPaolo Mori	vehicles	Article	evoluded	to healthcare	2018	http://link.springer.com/article/10.1007/s11219- 017.0202.2
opiniger	00.12.2024		1011010	70000	CACINGCO	keyword mismatch:	2010	017-000-00
			Special issue on "software quality in software-intensive			not directly related		http://link.springer.com/article/10.1007/s11219-
Springer	08.12.2024	Emilia MendesDietmar Winkler	systems"	Article	excluded	to healthcare	2018	018-9406-x
		Abd Alla GadDalia ElfikyAbdelazim NegmSalwa	Applications of Remote Sensing and GIS Based on an	Conference		keyword mismatch:	2023	http://ink.springer.com/book/10.1007/978-3- 031-40447-4
Springer	07 12 2024	Elbein	Innovative vision	paper	evoluded	to healthcare		
opiniger	07.12.2024				CACIDOCO	keyword mismatch:		
		Antonio M. Fernández-GómezDavid Gutiérrez-	A new Apache Spark-based framework for big data streaming			not directly related		http://link.springer.com/article/10.1007/s11227.
Springer	06.12.2024	AvilésAlicia TroncosoFrancisco Martínez-Álvarez	forecasting in IoT networks	Article	excluded	to healthcare	2023	023-05100-x
		Lisbeth Rodriguez-MazahuaCristian-Aaron Redriguez Epriguez José Luis Sénchez				kowword micmatch-		
		Cervantes Jair Cervantes Jorge Luis García-	A general perspective of Big Data: applications, tools,			not directly related		
Springer	06.12.2024	AlcarazGiner Alor-Hernández	challenges and trends	Article	excluded	to healthcare	2016	http://ink.springer.com/article/10.1007/s11227- 015-1501-1
						keyword mismatch:		
0	00 40 0004	Manar Abu TalibSohaib MajzoubQassim NasirDina	A systematic literature review on hardware implementation of	Article		not directly related	0001	http://link.springer.com/article/10.1007/s11227-
Springer	06.12.2024	Jamai	artificial intelligence algoritrinis	Article	excluded	to nealtricare keyword mismatch:	2021	020-03325-8
		Chao-Tung YangTzu-Yang ChenEndah	The implementation of data storage and analytics platform for			not directly related		http:///init.and/and/and/and/and/and/and/and/and/and/
Springer	08.12.2024	KristianiShyhtsun Felix Wu	big data lake of electricity usage with spark	Article	excluded	to healthcare	2021	020-03505-6
						keyword mismatch:		
Caringer	06 40 0004	Fabio ValdésBalf Hartmut Güting	A framework for efficient multi-attribute movement data	Article	a valuate d	not directly related	2010	http://link.springer.com/article/10.1007/s00778-
Springer	00.12.2024	Robert StablbockGary M. WeissMahmourd Abour	Advances in Data Science and Information Engineering	Conference	excluded	keyword mismatch:	2019	018-0525-6 http://link.springer.com/book/10.1007/978-3-
		NasrCheng-Ying YangDr. Hamid R.		paper		not directly related		030-71704-9
Springer	06.12.2024	ArabniaLeonidas Deligiannidis			excluded	to healthcare		
		Manuam Calababaux DadMabubabbab Milaabilialina	Information technology adaptions a various of the literature and			keyword mismatch:		
Springer	08 12 2024	Mohamed Dahlan	classification	Article	excluded	to healthcare	2018	http://link.springer.com/article/10.1007/s10209-
opinger	00.12.2024	Jason SherbelMarilia YatabeLucia	and an	70000	excluded	keyword mismatch:	2010	017-02042
		CevidanesAntonio RuellasSharon AronovichLauren	A method of comparing virtual reality orthognathic surgical			not directly related		http://link.soringer.com/article/10.1007/s10055-
Springer	06.12.2024	EhardtMatthew AmesHera Kim-Berman	predictions and postsurgical treatment outcomes	Article	excluded	to healthcare	2023	023-00855-3
		Oile WangOingi ZhangWeitong SunChadwick	A scoping review of the use of lab streaming laver framework			keyword mismatch:		
Springer	06.12.2024	BoulayKangsoo KimRoghayeh Leila Barmaki	in virtual and augmented reality research	Article	excluded	to healthcare	2023	http://link.springer.com/article/10.1007/s10055- 023-00799-8
		, , , , ,				keyword mismatch:		
						not directly related		http://link.springer.com/article/10.1007/s11280-
Springer	06.12.2024	Georgia M. Kapitsaki I neodoros Charalambous	Adapting HTML5 Web applications to user privacy preferences	Article	excluded	to nealthcare	2019	018-0628-4
		Wiem TounsiNora Cuppens-BoulahiaFrédéric	systems: Proposal and implementation on the fosstrak			not directly related		
Springer	06.12.2024	CuppensGuy Pujolle	platform	Article	excluded	to healthcare	2016	015-0325-5
				Conference		keyword mismatch:		
ACM Digital	20 11 2024	Ihonatan Boarim and Ana Begins C. do Books	Quality of CRM Systems: a systematic manning study	paper	excluded	not directly related	2040	https://doi.org/10.1145/0004044.0004055
Library	29.11.2024	enonatan boarim and And Regina C. da NoCha	agoancy or only systems, a systematic mapping study	Conference	excinded	keyword mismatch	2019	naps://doi.org/10.1145/3364641.3364655
ACM Digital				paper		not directly related		
Library	23.06.2024	Quang Vinh Nguyen and Ulrich Engelke	Big data visual analytics: fundamentals, techniques, and tools		excluded	to healthcare	2017	https://doi.org/10.1145/3134472.3134516
		Steve Hodges, Per Ola Kristensson, Josiah Hester,	Hardware is Hard-is it Worth it? In Extended Abstracts of the	Conference		keyword mismatch:		
ACM Digital	09 11 2024	and Yvonne Bogers	2023 CHI Conference on Human Factors in Computing Systems (CHI EA 23)	paper	evoluded	to healthcare	2023	https://doi.org/10.114E/0544540.0500751
Library	03.11.2024	and Home Hogers	How Did They Build the Free Encyclonedia? A Literature	Conference	excluded	keyword mismatch:	2020	https://doi.org/10.1145/3544548.3563751
ACM Digital			Review of Collaboration and Coordination among Wikipedia	paper		not directly related		
Library	24.06.2024	Yuqing Ren, Haifeng Zhang, and Robert E. Kraut	Editors		excluded	to healthcare	2023	https://doi.org/10.1145/3617369
			Quantified Lie: A group in the loop approach to team network	Conference		keyword mismatch:		
Library	29.11.2024	Raphael Tackx, Leo Blondel, and Marc Santolini	reconstruction	paper	excluded	to healthcare	2021	https://doi.org/10.1145/3460418-3479363
2.0.0.7	LOTTILOET	Eduardo F. Felix, Fernando A. A. Lins, Obionor O.		Conference	011010000	keyword mismatch:	LOLI	
ACM Digital		Nóbrega, Diego R. Gomes, Bruno A. Jesus, and	Compliance Evaluation of Cryptographic Security	paper		not directly related		
Library	08.11.2024	Marco Vieira	Hequirements on IOT Gateways	Conformerer	excluded	to healthcare	2023	https://doi.org/10.1145/3569902.3569915
ACM Digital		Henry Edison, Noel Carroll, Kieran Conboy, and	An Investigation into Inner Source Software Development:	paper		not directly related		
Library	04.11.2024	Lorraine Morgan	Preliminary Findings from a Systematic Literature Review		excluded	to healthcare	2018	https://doi.org/10.1145/3233391.3233529
		Sandesh Dhawaskar Sathyanarayana, Jihoon Lee,		Conference				
		Jinsung Lee, Youngbin Im, Parisa Rahimzadeh,	CARTLE averation Alson Distributed Onto 111 Control	paper		keyword mismatch:		
ACM Digital	08 11 2024	Alaoxi Zhang, Max Hollingsworth, Carlee Joe- Wong, Dirk Grunwald, and Sangtae Ha	Data Transmissions (demo)		excluded	not directly related	2010	https://doi.org/10.1145/3207224.3356576
Library	00.11.2024		and the second section	Conference	sholuded	keyword mismatch:	2019	
ACM Digital		Samantha Luu, Arun Ravindran, Armin Danesh		paper		not directly related		
Library	30.11.2024	Pazho, and Hamed Tabkhi	VEI: a multicloud edge gateway for computer vision in IoT		excluded	to healthcare	2022	https://doi.org/10.1145/3565385.3565877
ACM DI			Keynote: Recearch Challenges and Opportunities in IsT	Conference		keyword mismatch:		
AGM Digital Library	11.11.2024	Elisa Bertino	Security	Pahoi	excluded	to healthcare	2017	https://doi.org/10.1145/3139531.3139535
,				Conference		keyword mismatch:		
ACM Digital			A review of how whistleblowing is studied in software	paper		not directly related		
Library	28.10.2024	Lucy Hunt and Maria Angela Ferrario	engineering, and the implications for research and practice		excluded	to healthcare	2022	https://doi.org/10.1145/3510458.3513013

venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
	just j unit			Conference		keyword mismatch:		
ACM Digital	09.11.2024	Andrew McCrabb, Hellina Nigatu, Absalat Getachew, and Valeria Bertacco	DyGraph; a dynamic graph generator and benchmark suite	paper	excluded	not directly related to healthcare	2022	https://doi.org/10.1145/3534540.3534603
and y	00.11.2024	Zhongwei ZhangJianxiong WangXiaohua Hu	A Framework of Developing Health Care Application Systems		andidded	keyword mismatch:	2022	http://ink.springer.com/chapter/10.1007/978-3-
			Using 6LoWPAN Based Wireless Sensor Networks	Conference		not directly related		030-69431-9_12
Springer	06.12.2024	Sanket Kumar GuptaBahar SchwichtenberoGregor	A Reference Architecture for Enhanced Design of Software	paper	excluded	to nearthcare keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978-3-
		Engels	Ecosystems	Conference		not directly related		030-79976-2_4
Springer	06.12.2024	K S Lliiwal BeddyS Sibi ChakkaravarthyM	A Study on Android Malware Detection Lieing Machine	paper	excluded	to healthcare	2023	http://link.spripper.com/chapter/10.1007/978-3-
		GopinathAditya Mitra	Learning Algorithms	Conference		not directly related	2023	031-35078-8_20
Springer	06.12.2024			paper	excluded	to healthcare		
ACM Digital			A Comprehensive Overview of BIG DATA Technologies: A	Conterence		keyword mismatch: not directly related		
Library	25.10.2024	Muhammad Umair Raza and Zhao XuJian	Survey	F-F	excluded	to healthcare	2020	https://doi.org/10.1145/3404687.3404694
		Carlos Gavidia-Calderon, Amel Bennaceur, Tamara	Mastury Maker A Casial Identity Applying of Debaties	Conference		keyword mismatch:		
ACM Digital Library	11.11.2024	Bashar Nuseibeh	Software Engineering	paper	excluded	to healthcare	2023	https://doi.org/10.1145/3597512.3600206
				Conference		keyword mismatch:		
ACM Digital	20 11 2024	Sriram Birudavolu, Biswajit Nag, and Omprakash	Open Innovation in ICT Services for Quadruple Helix Model: The Cloud Proposition	paper	excluded	not directly related	2016	https://doi.acc/10.444E/0002007.0002448
Library	20.11.2024	Edna Dias Canedo, Larissa Soares, Geovana		Conference	excluded	keyword mismatch:	2010	https://doi.org/10.1145/2090307.2090410
ACM Digital		Ramos Sousa Silva, Verônica Souza Dos Santos,	Do you see what happens around you? Men's Perceptions of	paper		not directly related		
Library	09.11.2024	and Fabiana Freitas Mendes	Gender inequality in Software Engineering	Conference	excluded	keyword mismatch:	2023	https://doi.org/10.1145/3613372.3613394
ACM Digital			Search-based optimization for the testing resource allocation	paper		not directly related		
Library	30.11.2024	Roberto Pietrantuono and Stefano Russo	problem: research trends and opportunities	Conference	excluded	to healthcare	2018	https://doi.org/10.1145/3194718.3194721
ACM Digital		Ákos Milánkovich, Gergely Eberhardt, and Dávid		paper		not directly related		
Library	30.11.2024	Lukács	The AssureMOSS security certification scheme		excluded	to healthcare	2022	https://doi.org/10.1145/3538969.3543804
ACM Digital			"Out of Luck": Socio-Economic Differences in Student Coping	Conference paper		keyword mismatch: not directly related		
Library	24.10.2024	Gwen Petro, Amy Gonzales, and Jessica Calarco	Responses to Technology Problems	P - P	excluded	to healthcare	2020	https://doi.org/10.1145/3313831.3376156
			Communication Chills Training Intervention Deced on	Conference		keyword mismatch:		
ACM Digital Library	08.11.2024	Monica Pereira and Kate Hone	Automated Recognition of Nonverbal Signals	paper	excluded	to healthcare	2021	https://doi.org/10.1145/3411764.3445324
				Conference		keyword mismatch:		
ACM Digital	09 11 2024	Petr Skoda, David Bernhauer, Martin Nečaský, Jakub Klímek, and Tomáš Skopal	Evaluation Framework for Search Methods Focused on Dataset Findability in Open Data Catalogs	paper	excluded	not directly related	2024	https://doi.org/10.1145/9409757.0400070
LIUIdIY	09.11.2024	Andrei Zlotchevski, Dawn Drain, Alexey	Saturder in open Data Oatalogs	Conference	excluded	keyword mismatch:	2021	neps.//doi.org/10.1145/3428757.3429973
ACM Digital		Svyatkovskiy, Colin B. Clement, Neel Sundaresan,	Exploring and evaluating personalized models for code	paper		not directly related		
Library	09.11.2024	and Michele Tufano	generation	Conference	excluded	to healthcare	2022	https://doi.org/10.1145/3540250.3558959
ACM Digital			Exploring the under-explored terrain of non-open source data	paper		not directly related		
Library	09.11.2024	Shriram Shanbhag and Sridhar Chimalakonda	for software engineering through the lens of federated learning	o (	excluded	to healthcare	2022	https://doi.org/10.1145/3540250.3560883
ACM Digital			Harnessing Large Language Models for Simulink Toolchain Testing and Developing Diverse Open-Source Corpora of	Conference paper		keyword mismatch: not directly related		
Library	09.11.2024	Sohil Lal Shrestha	Simulink Models for Metric and Evolution Analysis	P - P	excluded	to healthcare	2023	https://doi.org/10.1145/3597926.3605233
		Lin Venn, Zahul Zhan, Yuannuan Zau, Li Chan	Applying a Blended Board Game System with Robotic Arm for	Conference		keyword mismatch:		
ACM Digital Library	04.11.2024	Zhihang Lin, and Weiyu Zhan	Machine Competition	paper	excluded	to healthcare	2023	https://doi.org/10.1145/3625704.3625766
		Chian-Hsueng Chao, I-Hsien Ting, Yi-Jun Tseng,		Conference		keyword mismatch:		
ACM Digital	20 11 2024	Bing-Wen Wang, Shin-Hua Wang, and Yu-Qing	The Study of DeFi in Virtual Community and Financial	paper	ovaludad	not directly related	2022	
Library	30.11.2024	wang	Supervision	Conference	excluded	keyword mismatch:	2022	https://doi.org/10.1145/3561278.3561297
ACM Digital		Elizabett Hillery, Mark Daniel Ward, Jenna Rickus,	Undergraduate Data Science and Diversity at Purdue	paper		not directly related		
Library	30.11.2024	Alex Younts, Preston Smith, and Eric Adams	University	Conference	excluded	to healthcare keyword mismatch:	2019	https://doi.org/10.1145/3332186.3332202
ACM Digital		Samar Sabie, Maha Salman, and Steve	Situating shelter design and provision in ICT discourse for	paper		not directly related		
Library	30.11.2024	Easterbrook	scarce-resource contexts	Conference	excluded	to healthcare	2016	https://doi.org/10.1145/2926676.2926686
ACM Digital		Samuel Yigzaw, Ilkka Jormanainen, and Markku	Information Systems in Developing Countries: Opportunities	paper		not directly related		
Library	10.11.2024	Tukiainen	and Challenges		excluded	to healthcare	2021	https://doi.org/10.1145/3486011.3486563
			Service Interruption: Managing Commitment to Community	Conference		keyword mismatch:		
Library	30.11.2024	Lecia J. Barker, Amy Voida, and Vaughan Nagy	Partners During a Crisis	paper	excluded	to healthcare	2021	https://doi.org/10.1145/3446871.3469756
		Occurrence Darks House Man Management Darks and	And the BBO Barlin Attack and Oceaning Barry Okting with	Conference		keyword mismatch:		
ACM Digital Library	04.11.2024	Dowon Kim	Practical Method	paper	excluded	to healthcare	2022	https://doi.org/10.1145/3538969.3544448
clotary	01.11.LOL I			Conference	Choladad	keyword mismatch:	LULL	110000000000000000000000000000000000000
ACM Digital	20.44.2024	Lan Anh Tran, Benedikt Hensen, Ralf Klamma, and	Privacy and Security in Mixed Reality Learning Environments	paper	a valudad	not directly related	0000	
Library	29.11.2024	Soamsin Ghamarasku	by input and osen bot interaction Protection	Conference	excluded	keyword mismatch:	2022	https://doi.org/10.1145/3512353.3512363
ACM Digital		Maor Yankovitch, Ilya Kolchinsky, and Assaf	HYPERSONIC: A Hybrid Parallelization Approach for Scalable	paper		not directly related		
Library	10.11.2024	Schuster	Complex Event Processing	Conference	excluded	to healthcare keyword mismatch:	2022	3
ACM Digital				paper		not directly related		
Library	09.11.2024	Ann Barcomb	Episodic volunteering in open source communities	0(-	excluded	to healthcare	2016	https://doi.org/10.1145/2915970.2915972
		Valente, Cole Makuch, Owen Correc, Long Ve		Conference paper				
		Dharmashankar Subramanian, Werner Geyer,						
ACM Dictor		Rahul Nair, Inge Vejsbjerg, Radu Marinescu, Paulito Palmes, Elizabeth M. Dalu, Loraino Franko	AutoDOViz: Human-Centered Automation for Decision			keyword mismatch: not directly related		
Library	04.11.2024	and Daniel Haehn	Optimization		excluded	to healthcare	2023	https://doi.org/10.1145/3581641.3584094
		Marco EspositoLeonardo Picchiami	A Comparative Study of Al Search Methods for Personalised	Casta		keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978-3- 031-08421-8 44
Springer	06.12.2024		Cancer Therapy Synthesis In COPASI	paper	excluded	to healthcare		
-buildoi		Nyalleng MoorosiMamello ThinyaneVukosi	A Critical and Systemic Consideration of Data for Sustainable	F	Shortond	keyword mismatch:	2017	http://ink.springer.com/chapter/10.1007/978-3-
Springer	06 12 2024	Marivate	Development in Africa	Conference	ovoluted	not directly related		010-00111-7_20
opringer.	00.12.2024	Alex KismulHaider Al-KhateebHamid Jahankhani	A Critical Review of Digital Twin Confidentiality in a Smart City	paper	excluded	keyword mismatch:	2023	http://link.springer.com/chapter/10.1007/978-3-
				Conference		not directly related		031-20160-8_25
Springer	06.12.2024	Abu SaveedAzmain Yakin SrizonMd, Mahadi	A Hybrid Campus Security System Combined of Face	paper	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978-9-
		HasanJungpil ShinMd. Al Mehedi HasanM. Rasel	Number-Plate, and Voice Recognition	Conference		not directly related	2023	031-23599-3_27
Springer	06.12.2024	Mahmud	A Learning distance with the Linkings of the West State of the State	paper	excluded	to healthcare	0000	http://link.op/oper.ope/otttip.topper.o
		Hans-Jorg SchulzMartin RöhligLars NonnemannMarius HogräferMario AehneltBodo	A Layered Approach to Lightweight Toolchaining in Visual Analytics	Conference		keyword mismatch: not directly related	2020	030-41590-7_13
Springer	06.12.2024	UrbanHeidrun Schumann		paper	excluded	to healthcare		
		Jaroslav KrálPetr NovákMichal Žemlička	A System Based on Intelligent Documents	Conformer		keyword mismatch:	2017	http://link.springer.com/chapter/10.1007/978-3- 319-62407-5_12
Springer	06.12.2024			paper	excluded	to healthcare		
- shirildoi				Conference	and a deal	keyword mismatch:		
ACM Digital	30 11 2024	Abdellah Maatallaoui Hicham Touil and Larbi Cotti	The Impact of Radio Frequency (RF) Attacks on Security and Privacy: A Comprehensive Review	paper	eveluded	not directly related	2022	
LIDIALLÀ	30.11.2024	rsocolari maatallaoui, nicharri louli, and Larbi Setti	r macy. A comprehensive neview	Conference	excluded	keyword mismatch:	2023	naps://doi.org/10.1145/3607720.3607771
ACM Digital		Rachita Misra, Bijayalaxmi Panda, and Mayank		paper		not directly related		
Library	04.11.2024	Tiwary Arthi Paiu, Muhammad Owais, Joyod, Juba Mumi	Big data and ICT applications: A study	Conference	excluded	to healthcare	2016	https://doi.org/10.1145/2905055.2905099
ACM Digital		Juha Savimäki, Brian McGillion, and Billy Bob	SoK: A Systematic Review of TEE Usage for Developing	paper		not directly related		
Library	30.11.2024	Brumley	Trusted Applications	Conformer	excluded	to healthcare	2023	https://doi.org/10.1145/3600160.3600169
ACM Digital			ICT, Economic Effects and Business Patterns: A Text-Mining	paper		not directly related		
Library	10.11.2024	Kirill Sarachuk and Missler-Behr Magdalena	of Existing Literature		excluded	to healthcare	2020	https://doi.org/10.1145/3383845.3383851

Search venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
		Argyro Mavrogiorgou, Vasileios Koukos, Eleftheria		Conference		keyword mismatch:		
ACM Digital	27 10 2024	Kouremenou, Athanasios Kiourtis, Alexandros Baikos, George Manias, and Dimosthenis Kyriazis	A Cross-domain Data Marketolace for Data Sharing	paper	excluded	not directly related to healthcare	2023	https://doi.org/10.1145/3571607.3571707
Library	27.10.2024	hanos, deorge manus, and prinostions tynazis	A cross domain bata manophate for bata ona ing	Conference	GACIDUCU	keyword mismatch:	2025	https://doi.org/10.1145/3571097.3571707
ACM Digital		Leena Ventä-Olkkonen, Marianne Kinnula, Heidi	Embedded assumptions in design and Making projects with	paper		not directly related		
Library	09.11.2024	Hartikainen, and Netta livari Aikaterini Reufee, Riebard Einkorn, Martiin van	children	Conference	excluded	to healthcare	2021	https://doi.org/10.1145/3441000.3441077
ACM Digital		Kaauwen, Mark Kramer, and Ioannis N.	Managing Variant Calling Files the Big Data Way: Using HDFS	paper		not directly related		
Library	11.11.2024	Athanasiadis	and Apache Parquet	o . /	excluded	to healthcare	2017	https://doi.org/10.1145/3148055.3148060
		Markus I Illrich Thomas Weise Abbiebek Awasthi	A generic problem instance generator for discrete optimization	Conference		keyword mismatch:		
Library	27.10.2024	and Jörg Lässig	problems	paper	excluded	to healthcare	2018	https://doi.org/10.1145/3205651.3208284
						keyword mismatch:		
ACM Digital	40 44 0004	Jelena Marjanović, Nikola Dalčeković, and Goran	Improving Critical Infrastructure Protection by Enhancing	Conference	a secolar al	not directly related	2024	
Library	10.11.2024	Sladic	Software Acquisition Process Through Biockchain	Conference	excluded	to neartricare keyword mismatch:	2021	https://doi.org/10.1145/3459960.3459973
ACM Digital			Two Decades of Empirical Research on Developers'	paper		not directly related		
Library	30.11.2024	Abir Bouraffa and Walid Maalej	Information Needs: A Preliminary Analysis	Conforman	excluded	to healthcare	2020	https://doi.org/10.1145/3387940.3391485
			GitHub? In Proceedings of the 2022 ACM/IEEE 44th	paper		keyword mismatch:		
ACM Digital	40.44.0004	Hourieh Khalajzadeh, Mojtaba Shahin, Humphrey	International Conference on Software Engineering: Software		a contrada al	not directly related	2022	
Library	10.11.2024	Jing Lu	A Data-Driven Framework for Business Analytics in the		excluded	keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978-3-
			Context of Big Data	Conference		not directly related		030-00063-9_32
Springer	06.12.2024	Dable Rusteel vie 1 Manae Iven D. Banders Adrián	A Lipitian Internal Depresentation of the Outer World for Social	paper	excluded	to healthcare	2016	http://link.springer.com/chapter/10.1007/978-3-
		Romero-GarcésLuis V. CalderitaRebeca	Robotics	Conference		not directly related	2010	319-27149-1_57
Springer	06.12.2024	MarfilAntonio Bandera		paper	excluded	to healthcare		
ACM Disital		Daniel Avery Hos Khanh Dam Bastin Tony Boy		Conference		keyword mismatch:		
Library	09.11.2024	Savarimuthu, and Aditya Ghose	Externalization of software behavior by the mining of norms	papa.	excluded	to healthcare	2016	https://doi.org/10.1145/2901739.2901744
		Made Kallana Made Only Onder Tailors and		Conference		keyword mismatch:		
ACM Digital	08 11 2024	Martin Kellogg, Martin Schat, Serdar Tasiran, and Michael D. Ernst	Continuous compliance	paper	excluded	to healthcare	2021	https://doi.org/10.1145/3324884.3416593
Library	OU.THEORY		Contraction Contraction	Conference	Choladoa	keyword mismatch:	LOLI	11022100701910.11197022-1001.0110202
ACM Digital	10 11 000 -	Paul Duckworth, Logan Graham, and Michael	Inferring Work Task Automatability from ALE+ E-i-d-	paper	ovelut	not directly related	2015	
Library	10.11.2024	Capolite	interning work task Automatability from AI Expert Evidence		excluded	keyword mismatch	∠019	ntips://doi.org/10.1145/3306618.3314247
ACM Digital			Motivations for Participating in Online Initiatives: Exploring	Conference		not directly related		
Library	11.11.2024	Sarah Gilbert	Contributory Behaviour Across Initiative Types	paper	excluded	to healthcare	2017	https://doi.org/10.1145/3022198.3024941
ACM Digital			MetaV: A Meta-Verifier Approach to Task-Appostic Model	paper		not directly related		
Library	11.11.2024	Xudong Pan, Yifan Yan, Mi Zhang, and Min Yang	Fingerprinting		excluded	to healthcare	2022	https://doi.org/10.1145/3534678.3539257
		View Orshidenende, Orber Dheimer and Musel	OVER events dies with each life data allow for lat through an	Conference		keyword mismatch:		
ACM Digital Library	29.11.2024	Elovici	adaptable and automated static analysis framework	paper	excluded	to healthcare	2020	https://doi.org/10.1145/3341105.3373930
Library	EO: TTEOET	David MeloFilipe SoaresSimão FelgueirasJoão	A New Compact Optical System Proposal and Image Quality		CACINGUUG	keyword mismatch:	2019	http://link.springer.com/chapter/10.1007/978-3-
		GonçalvesPedro Vieira	Comparison Against Other Affordable Non-mydriatic Fundus	Conference		not directly related		030-29196-9_2
Springer	06.12.2024		Cameras	Conference	excluded	to nealthcare keyword mismatch:		
ACM Digital		Flavio Oquendo, Jérémy Buisson, Elena Leroux,	The sosADL studio: an architecture development environment	paper		not directly related		
Library	30.11.2024	Gersan Moguérou, and Jean Quilbeuf	for software-intensive systems-of-systems	Conforman	excluded	to healthcare	2016	https://doi.org/10.1145/3175731.3176180
ACM Digital		Mohamed M. Kassem, Mahesh K. Marina, and	DIY Model for Mobile Network Deployment: A Step Towards	paper		not directly related		
Library	09.11.2024	Bozidar Radunovic	5G for All		excluded	to healthcare	2018	https://doi.org/10.1145/3209811.3212703
			Supporting systematic literature reviews in computer science:	Conference		keyword mismatch:		
Library	30.11.2024	Sebastian Götz	the systematic literature review toolkit	paper	excluded	to healthcare	2018	https://doi.org/10.1145/3270112.3270117
				Conference		keyword mismatch:		
ACM Digital	29 11 2024	Dharun Anandayuyarai and James C. Davis	Reflecting on Recurring Failures in IoT Development	paper	evoluded	not directly related	2023	https://doi.org/10.1116/0551240.2550545
Library	29.11.2024	Dhardin Anandayuvaraj and barries C. Davis	Helecting of Hecuming Failures in of Development	Conference	excluded	keyword mismatch:	2023	https://doi.org/10.1145/3551349.3559545
ACM Digital				paper		not directly related		
Library	04.11.2024	Shuli Jia, Liyong Ma, and Shuting Zhang	Big data prototype practice for unmanned surface vehicle	Conference	excluded	to healthcare keyword mismatch:	2018	https://doi.org/10.1145/3290420.3290466
ACM Digital			Smart city knowledge management: Holistic review and the	paper		not directly related		
Library	30.11.2024	Vasja Roblek and Maja Meško	analysis of the urban knowledge management	0	excluded	to healthcare	2020	https://doi.org/10.1145/3396956.3398263
ACM Digital			Long-Term Evaluation of Technical Debt in Open-Source	paper		not directly related		
Library	11.11.2024	Arthur-Jozsef Molnar and Simona Motogna	Software		excluded	to healthcare	2020	https://doi.org/10.1145/3382494.3410673
10110-11-1				Conference		keyword mismatch:		
Library	09.11.2024	John M. Pierre	End-to-End Deep Learning for Robotic Following	paper	excluded	to healthcare	2018	https://doi.org/10.1145/3185066.3185084
				Conference		keyword mismatch:		
ACM Digital	11 11 2024	Chinchu Thomas	and Online Educational Settings	paper	excluded	not directly related to healthcare	2018	https://doi.org/10.1145/3242060.3264060
Library	11.11.2024		and online Educational octaingo	Conference	excluded	keyword mismatch:	2010	https://doi.org/10.1145/3242909/3204909
ACM Digital		Beer Weer Mercet Here and Andrew Oracle	Mesa-Geo: A GIS Extension for the Mesa Agent-Based	paper		not directly related		
Library	11.11.2024	Thanasis G. PapaioannouVlado StankovskiPetar	A New Blockchain Ecosystem for Trusted. Traceable and		excluded	to nearthcare	2022	http://doi.org/10.1145/3557989.3566157 http://link.springer.com/chapter/10.1007/978-3-
		KochovskiAnthony Simonet-BoulogneCaroline	Transparent Ontological Knowledge Management	1.00		keyword mismatch:		030-92916-9_8
Springer	06 10 0004	BarelleAlberto CiaramellaMarco CiaramellaGeorge		Conference	oveluded	not directly related		
opinger	50.12.2024	er etamouno		Conference	avenued	keyword mismatch:		
ACM Digital				paper		not directly related		
Library	10.11.2024	Janoon Koo and Young-Gab Kim Bajendra K. Baj, Carol J. Romanowski, John	interoperability requirements for a smart city	Conference	excluded	to healthcare	2021	https://doi.org/10.1145/3412841.3441948
		Impagliazzo, Sherif G. Aly, Brett A. Becker, Juan		paper				
		Chen, Sheikh Ghafoor, Nasser Giacaman, Steven I.	Link Dedemone Computing Education Computer of			keyword mismatch:		
ACM Digital Library	10.11.2024	Robson, Gruz izu, Snanram Rahimi, Michael P. Robson, and Neena Thota	and Future Directions		excluded	to healthcare	2020	https://doi.org/10.1145/3437800.3439203
				Conference		keyword mismatch:		
ACM Digital	11 11 0004	Robert Philipp, Andreas Mladenow, Christine Strauss, and Alexander Välz	Machine Learning as a Service: Challenges in Research and Applications	paper	excluded	not directly related	2024	biles//dei.ess/10.4145/0400757.000005
Library	11.11.2024	Strauss, and Alexander Volz	Applications	Conference	excluded	keyword mismatch:	2021	https://doi.org/10.1145/3428757.3429152
ACM Digital		Klervie Toczé, Johan Lindqvist, and Simin Nadjm-		paper		not directly related		
Library	29.11.2024	Tehrani Nikhil Krishna Gonalakrishna, Dhasua	Performance Study of Mixed Reality for Edge Computing	Conference	excluded	to healthcare	2019	https://doi.org/10.1145/3344341.3368816
ACM Digital		Anandayuvaraj, Annan Detti, Forrest Lee Bland,	"If security is required": engineering and security practices for	paper		not directly related		
Library	24.10.2024	Sazzadur Rahaman, and James C. Davis	machine learning-based IoT devices	Canto	excluded	to healthcare	2023	https://doi.org/10.1145/3528227.3528565
		viiliam Fornaciari, Giovanni Agosta, David Atienza, Carlo Brandolese, Leila Cammoun, Luca Cremona		paper				
		Alessandro Cilardo, Albert Farres, José Flich,						
		Carles Hernandez, Michal Kulchewski, Simone						
		Ariel Oleksiak, Anna Pupykina, Federico				keyword mismatch:		
ACM Digital		Reghenzani, Rafael Tornero, Michele Zanella,	Reliable power and time-constraints-aware predictive			not directly related		
Library	29.11.2024	Marina Zapater, and Davide Zoni	management of heterogeneous exascale systems	Conference	excluded	to nealthcare keyword mismatch	2018	https://doi.org/10.1145/3229631.3239368
ACM Digital				paper		not directly related		
Library	09.11.2024	Emily First and Yuriy Brun	Diversity-driven automated formal verification	01-	excluded	to healthcare	2022	https://doi.org/10.1145/3510003.3510138
ACM Digital		Andreas M, Klein, Andreas Hinderks, Martin	Construction of UEQ+ scales for voice quality: measuring user	conterence paper		keyword mismatch: not directly related		
Library	08.11.2024	Schrepp, and Jörg Thomaschewski	experience quality of voice interaction		excluded	to healthcare	2020	https://doi.org/10.1145/3404983.3410003
ACHE		Thomas W. Malone, Jeffrey V. Nickerson, Robert J.	Putting the Diaces Back Together Agein: Centest Water for	Conference		keyword mismatch:		
ACM Digital Library	29.11.2024	Yue Han, and W. Ben Towne	Large-Scale Problem Solving	Paper	excluded	to healthcare	2017	https://doi.org/10.1145/2998181.2998343

venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
	,			Conference		keyword mismatch:		
ACM Digital Library	28.10.2024	Joshua Introne. Brvan Semaan, and Sean Googins	A Sociotechnical Mechanism for Online Support Provision	paper	excluded	not directly related to healthcare	2016	https://doi.org/10.1145/2858036.2858582
Library	LOTTOLEGET			Conference	0/10/0000	keyword mismatch:	2010	1193-110-019 10-1140-00000-2000002
ACM Digital	20 11 2024	Arash Fard, Anh Le, George Larionov, Waqas	Vartice MI - Distributed Machine Learning in Vartice Database	paper	oveluded	not directly related	2020	
Library	30.11.2024	Dinion, and Chuck Bear	Ventica-IVIC. Distributed Iviacinine Learning in Ventica Database	Conference	excluded	keyword mismatch:	2020	https://doi.org/10.1145/3318464.3386137
ACM Digital	44.44.0004	Kundi Yao, Guilherme B. de Pádua, Weiyi Shang,	Log4Perf: Suggesting Logging Locations for Web-based	paper	and other	not directly related	0040	
Library	11.11.2024	Steve Sporea, Andrei Torna, and Salari Sajedi	Systems Performance Monitoring	Conference	excluded	keyword mismatch:	2010	https://doi.org/10.1145/318440/.3184416
ACM Digital		Astroia M. Dissidiana Ocisticas Duras	A compatible base of second states and second s	paper		not directly related		
Library	28.10.2024	Antonio M. Rinaldi and Cristiano Russo	A semantic-based model to represent multimedia big data	Conference	excluded	to healthcare keyword mismatch:	2018	https://doi.org/10.1145/3281375.3281386
ACM Digital		Alan Freihof Tygel, Judie Attard, Fabrizio Orlandi,	"How Much?" is not Enough: an Analysis of Open Budget	paper		not directly related		
Library	24.10.2024	Maria Luiza Machado Campos, and Sören Auer	Initiatives	Conference	excluded	to healthcare keyword mismatch:	2016	https://doi.org/10.1145/2910019.2910054
ACM Digital		Ivano Malavolta, Grace Lewis, Bradley Schmerl,	How do you architect your robots? state of the practice and	paper		not directly related		
Library	10.11.2024	Patricia Lago, and David Garlan	guidelines for ROS-based systems	Conference	excluded	to healthcare	2020	https://doi.org/10.1145/3377813.3381358
			have, and what do they (feel they can) do about them? In	paper		keyword mismatch:		
ACM Digital	11 11 2024	David Gray Widder, Derrick Zhen, Laura Dabbish, and James Herbsleb	Proceedings of the 2023 ACM Conference on Fairness, Accountability, and Transparency (EAccT '23)		excluded	not directly related to healthcare	2023	https://doi.org/10.1145/3593013.3594012
Library				Conference	0/10/0000	keyword mismatch:	LOLO	11,237,001,01,277,174,230,300 10,000 10,
ACM Digital	20 11 2024	Daniele Di Mitri, Jan Schneider, Roland Klemke, Marcus Specht, and Hendrik Drachsler	Read Between the Lines: An Annotation Tool for Multimodal	paper	oveluded	not directly related	2010	
clorary	20.11.2024		But to Examing	Conference	excluded	keyword mismatch:	2010	https://doi.org/10.114/03303112.3303170
ACM Digital	10 11 2024	Arindam Giri, Subrata Dutta, Sarmistha Neogy, Keshay Dahal, and Zeeshan Pervez	Internet of things (IoT): a survey on architecture, enabling technologies, applications and challenges	paper	excluded	not directly related	2017	https://doi.org/10.0145/0400764.0400769
Library	10.11.2024	Resnav Danai, and Zeesnan Pervez	How is video game development different from software	Conference	excluded	to rieatricare	2017	https://doi.org/10.1145/3109761.3109768
ACM Digital		Luca Pascarella, Fabio Palomba, Massimiliano Di	development in open source? In Proceedings of the 15th International Conference on Mining Software Repositories	paper		keyword mismatch: not directly related		
Library	10.11.2024	Penta, and Alberto Bacchelli	(MSR '18)		excluded	to healthcare	2018	https://doi.org/10.1145/3196398.3196418
ACM Digital		Xin Xia, Emad Shihah, Yasutaka Kamei, David Lo		Conference paper		keyword mismatch: not directly related		
Library	29.11.2024	and Xinyu Wang	Predicting Crashing Releases of Mobile Applications	hoho.	excluded	to healthcare	2016	https://doi.org/10.1145/2961111.2962606
		Doris Via Eva Visuai Wu Doris Jung Lin Lee	Whither AutoMI 2 Understanding the Dole of Automation in	Conference		keyword mismatch:		
Library	30.11.2024	Niloufar Salehi, and Aditya Parameswaran	Machine Learning Workflows	paper	excluded	to healthcare	2021	https://doi.org/10.1145/3411764.3445306
			The Deliver of Directory Marshare from Name to	Conference		keyword mismatch:		
ACM Digital Librarv	30.11.2024	Nora McDonald and Andrea Forte	Vulnerabilities	paper	excluded	to healthcare	2020	https://doi.org/10.1145/3313831.3376167
		Sanjay Misra	A Step by Step Guide for Choosing Project Topics and Writing			keyword mismatch:	2021	http://ink.springer.com/chapter/10.1007/978-3-
Springer	06 12 2024		Research Papers in ICT Related Disciplines	Conference	excluded	not directly related to healthcare		030-09143-1_33
Springer	00.12.2024			Conference	excluded	keyword mismatch:		
ACM Digital	04 11 2024	Charles E. Catlett, Peter H. Beckman, Rajesh	Array of things: a scientific research instrument in the public	paper	oveluded	not directly related	2017	
Library	04.11.2024	Sankaran, and Kate Kusiak Galvin	way, platform design and early lessons learned	Conference	excluded	keyword mismatch:	2017	https://doi.org/10.1145/3063386.3063771
ACM Digital		Hanna Schneider, Malin Eiband, Daniel Ullrich, and		paper		not directly related		
Library	09.11.2024	Andreas Butz	Empowerment in HCI - A Survey and Framework	Conference	excluded	to nealthcare keyword mismatch:	2018	https://doi.org/10.1145/3173574.3173818
ACM Digital				paper		not directly related		
Library	30.11.2024	Bo Han, Yu Liu, and Feng Qian	ViVo: visibility-aware mobile volumetric video streaming		excluded	to healthcare keyword mismatch:	2020	https://doi.org/10.1145/3372224.3380888
ACM Digital		Christopher W. Weimer, J. O. Miller, and Raymond		Conference		not directly related		https://dl.acm.org/doi/pdf/10.5555/3042094.
Library	04.11.2024	R. Hill Varen Bilge Kava, Soobia Mantell, Kavse Lee	Agent-based modeling: an introduction and primer	paper	excluded	to healthcare	2016	3042112
ACM Digital		Maass, Renata Konrad, Andrew C. Trapp, Geri L.	Discrete Event Simulation to Evaluate Shelter Capacity	Conference		not directly related		
Library	22.06.2024	Dimas, and Meredith Dank	Expansion Options for LGBTQ+ Homeless Youth	paper	excluded	to healthcare	2023	https://ieeexplore.ieee.org/document/10015303
ACM Digital		Daniel M. German, Marco A. Gerosa, and Igor	Do I Belong? Modeling Sense of Virtual Community Among	Conference		not directly related		
Library	28.05.2024	Steinmacher	Linux Kernel Contributors	paper	excluded	to healthcare	2023	https://doi.org/10.1109/ICSE48619.2023.00038
ACM Digital		Rodrigo De la Fuente, Ignacio Erazo, and Raymond	Enabling intelligent processes in simulation utilizing the	Conference		not directly related		
Library	21.06.2024	L. Smith	TensorFlow deep learning resources	paper	excluded	to healthcare	2018	https://ieeexplore.ieee.org/document/8632539
ACM Digital		Agnieszka Chodorek, Robert R. Chodorek, and	terrestrial station and a drone: the case of a flying IoT system	Conference		not directly related		
Library	22.06.2024	Krzysztof Wajda	to monitor parking	paper	excluded	to healthcare	2020	https://ieeexplore.ieee.org/document/8958706
ACM Digital		Divvashikha Sethia. Shalini Sheoran, and Huzur		Conference		keyword mismatch: not directly related		
Library	28.05.2024	Saran	Optimized MapFile based Storage of Small files in Hadoop	paper	excluded	to healthcare	2017	https://doi.org/10.1109/CCGRID.2017.83
		G. W. Hislop; H. J. C. Ellis	Humanitarian Open Source Software in Computing Education	Article		keyword mismatch: not directly related	2017	https://eeexplore.ieee.org/document/8057313
IEEE Xplore	02.07.2024				excluded	to healthcare		
		Thoralf ReisMarco X. BornschleglMatthias L. Hemmie	Al2VIS4BigData: Qualitative Evaluation of an Al-Based Big Data Analysis and Visualization Reference Model	Conference		keyword mismatch: not directly related	2021	http://link.springer.com/chapter/10.1007/978-3- 030-68007-7_9
Springer	07.12.2024	. isoninge		paper	excluded	to healthcare		
		Areti KaramanouEvdokia MangouKonstantinos Tarabanis	An Ecosystem for Deploying Artificial Intelligence in Public Administration	Conference		keyword mismatch:	2023	http://link.springer.com/chapter/10.1007/978-3- 031-41138-0_13
Springer	07.12.2024			paper	excluded	to healthcare		
		Selene Caro-ViaAlbert Garangou-CulebrasAlex Falcó-OlmoCarlota Parés-MorlansBaquel Ros	An Educational, Semi-autonomous Telepresence Robot Called Sally	Conference		keyword mismatch: not directly related	2022	http://ink.springer.com/chapter/10.1007/978-3- 031-05675-8_12
Springer	07.12.2024			paper	excluded	to healthcare		
		Anand KhandareNipun AgarwalAmruta BodhankarAnkur Kulkarnilebaan Mana	Analysis of Python Libraries for Artificial Intelligence	Conference		keyword mismatch: not directly related	2023	http://ink.springer.com/chapter/10.1007/978- 981-99-0071-8_13
Springer	07.12.2024	eren an andre man transar mendadi i Marie		paper	excluded	to healthcare		
		Nicolas KourtellisGianmarco de Francisci Moralos Albort Bifot	Analyzing Big Data Streams with Apache SAMOA	Conformer		keyword mismatch:	2019	http://link.springer.com/chapter/10.1007/978-3- 030-34407-8_3
Springer	07.12.2024	WordlesAlbert Bilet		paper	excluded	to healthcare		
		Bianca TrovòNazzareno Massari	Ants-Review: A Privacy-Oriented Protocol for Incentivized	Contract		keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978-3- 030-71593-9_2
Springer	07.12.2024		Open reef Heviews on Ethereum	paper	excluded	to healthcare		
		Chih-Che ChangChia-Mei ChenHan-Wei Hsiao	Applying an IoT Analytics Framework in East Asia Area	Oracla		keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978- 981-19-9582-8.37
Springer	07.12.2024			Conference paper	excluded	not directly related to healthcare		
		Babu Veeresh Thummadi	Artificial Intelligence (AI) Capabilities, Trust and Open Source			keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978-3-
Springer	07.12.2024		Software Team Performance	Conference paper	excluded	not directly related to healthcare		and define the
- Pringor	57.7E.2027	Nour El MawasJean-Marie GilliotSerge	As One Size Doesn't Fit All, Personalized Massive Open Online		Chordood	keyword mismatch:	2019	http://link.springer.com/chapter/10.1007/978-3-
Springer	07 12 2024	GarlattiReinhardt EulerSylvain Pascual	Courses Are Required	Conference paper	excluded	not directly related		<u>uau-21131-6_22</u>
opninger	07.12.2024	Ioannis StelliosKostas MokosPanayiotis	Assessing Vulnerabilities and IoT-Enabled Attacks on Smart	hahei	excluded	keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978-3-
Corio	07 10 000 1	Kotzanikolaou	Lighting Systems	Conference	ovel	not directly related		030-95484-0_13
Springer	07.12.2024	Mst Shapna AkterHossain ShahriarZakirul Alam	Automated Vulnerability Detection in Source Code Using	paper	excluded	keyword mismatch:	2023	http://link.springer.com/chapter/10.1007/978-
	07.40	Bhuiya	Quantum Natural Language Processing	Conference		not directly related		981-99-0272-9_6
Springer	07.12.2024	Agung WahyudiRicardo MatheusMariin Janssen	Benefits and Challenges of a Reference Architecture for	paper	excluded	to nealthcare keyword mismatch:	2017	http://link.springer.com/chapter/10.1007/978-3-
		c c	Processing Statistical Data	Conference		not directly related		319-68557-1_41
Springer	07.12.2024	Zhenzhen BaoPeng LuoDongdai Lin	Bitsliced Implementations of the PRINCE LED and	paper	excluded	to healthcare keyword mismatch:	2016	http://link.springer.com/chapter/10.1007/978-3-
		Last and Last and Last	RECTANGLE Block Ciphers on AVR 8-Bit Microcontrollers	Conference		not directly related	0	319-29814-6_3
Springer	07.12.2024	Firini Kalliamvakou lens WeberAlessia Knouss	Certification of Open Source Software - A Scoping Review	paper	excluded	to healthcare keyword mismatch:	2016	http://link.springer.com/chapter/10.1007/978-9-
		LINE MERCENERS WEDER/ISSIE MILUSS	Socialization of Open Source Socialize – A Scoping Review	Conference		not directly related	2010	319-39225-7_9
Springer	07.12.2024			paper	excluded	to healthcare		

Search								
venue	Query date	Author Kumar Dron ShrivastavAnkan Mukherjee	Title Classification of Colposcopic Cervigrams Using EMD in R	Туре	Status	Reason keyword mismatch:	Year 2019	Article URL http://ink.springer.com/chapter/10.1007/978-
Springer	07.12.2024	DasHarpreet SinghPriya RanjanRajiv Janardhanan		Conference paper	excluded	not directly related to healthcare		961-13-5758-9_25
opinigoi	OTTILLEDE T	David CuttingAndrew McDowellEsha Barlaskar	Co-design of Modern Technology Modules with Industry and		Cholodod	keyword mismatch:	2023	http://link.springer.com/chapter/10.1007/978-3- 031_48639-5_2
Springer	07.12.2024		Students as Partners	paper	excluded	to healthcare		
		Yingying ChenXiaozhe Yao	CVTron Web: A Versatile Framework for Online Computer Vision Services	Conference		keyword mismatch: not directly related	2018	http://ink.springer.com/chapter/10.1007/978-3- 319-94472-2_5
Springer	07.12.2024			paper	excluded	to healthcare		
		Harald FoidIMichael Felderer	Data Science Challenges to Improve Quality Assurance of Internet of Things Applications	Conference		keyword mismatch: not directly related	2016	http://link.springer.com/chapter/10.1007/978-3- 319-47169-3_54
Springer	07.12.2024	Xingwei LiangYou ZouTian XiaQi Zhou	DCRNNX: Dual-Channel Recurrent Neural Network with	paper	excluded	to healthcare	2022	http://link.springer.com/chapter/10.1007/978-3-
		Anglier Elang for 200 har Alogi 2100	Xgboost for Emotion Identification Using Nonspeech	Conference		not directly related	2022	031-23504-7_2
Springer	07.12.2024	Artur KlepaczkoEli EikefjordArvid Lundervold	Vocalizations Deep Convolutional Neural Networks in Application to Kidney	paper	excluded	to healthcare keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978-3-
Springer	07 12 2024		Segmentation in the DCE-MR Images	Conference	evoluded	not directly related		030-77967-2_50
opingoi	OTTELEDE!	Brian NicholsonPetter NielsenJohan Ivar SæbøAna	Digital Public Goods for Development: A Conspectus and	p	CACIDADA	keyword mismatch:	2022	http://ink.springer.com/chapter/10.1007/978-3- 031-19429-0_27
Springer	07.12.2024	Paula lavares	Hesearch Agenda	paper	excluded	to healthcare		
		Oliver SmithSamuel WhiteMartin Stoelen	Eduardo: A Low Cost Assistive Robot Development Platform, Featuring a Compliant End Effector	Conference		keyword mismatch: not directly related	2019	http://link.springer.com/chapter/10.1007/978-3- 030-25332-5_4
Springer	07.12.2024	Heavie They logadeeses layerder	EMDO CI AM: Beel Time High Recelution Reconstruction of	paper	excluded	to healthcare	2021	http://ink.springer.com/chapter/10.1007/978-3-
		naoyin zhousagadeesan sayender	Soft Tissue Surface from Stereo Laparoscopy Videos	Conference		not directly related	2021	030-87202-1_32
Springer	07.12.2024	Jenny Guberlris Reinhartz-BergerMarina Litvak	Empirical Exploration of Open-Source Issues for Predicting	paper	excluded	to healthcare keyword mismatch:	2023	http://link.springer.com/chapter/10.1007/978-3-
Springer	08 12 2024	, , , , , , , , , , , , , , , , , , , ,	Privacy Compliance	Conference	ovcluded	not directly related		031-47112-4_6
ophinger	00.12.2024	Stefano BerlatoUmberto MorelliRoberto	End-to-End Protection of IoT Communications Through	paper	excluded	keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978-3-
Springer	08.12.2024	CarboneSilvio Ranise	Cryptographic Enforcement of Access Control Policies	Conference paper	excluded	not directly related to healthcare		
		Szilvia ZörgöGjalt-Jorn Ygram Peters	Epistemic Network Analysis for Semi-structured Interviews and Other Continuous Narratives: Challenges and Insights	Conference		keyword mismatch:	2019	http://ink.springer.com/chapter/10.1007/978-3- 030-33232-7_23
Springer	08.12.2024		and other continuous numerics, onaionges and insights	paper	excluded	to healthcare		
		Rafael Meneses SantosIsrael Meneses SantosMethanias Colaço JúniorManoel Mendonça	Evaluating a LSTM Neural Network and a Word2vec Model in the Classification of Self-admitted Technical Debts and Their	Conference		keyword mismatch: not directly related	2021	http://ink.springer.com/chapter/10.1007/978-3- 030-75418-1_25
Springer	08.12.2024	Arthur-Jozsef MolnarAlexandra NeamtuSimona	Types in Code Comments Evaluation of Software Product Quality Metrics	paper	excluded	to healthcare keyword mismatch:	2020	http://link.springer.com/chapter/10.1007/978-3-
		Motogna	Evaluation of Gortware Product Quality Metrics	Conference		not directly related	2020	030-40223-5_8
Springer	08.12.2024	Konstantinos Manikas	Evolving Software Products, the Design of a Water-Related	paper	excluded	to healthcare keyword mismatch:	2017	http://ink.springer.com/chapter/10.1007/978-3-
Springer	08 12 2024		Modeling Software Ecosystem	Conference	excluded	not directly related		319-69191-6_14
ophiligei	00.12.2024	Paraskevi SmiariStamatia BibiDaniel Feitosa	Examining the Reusability of Smart Home Applications: A	paper	excluded	keyword mismatch:	2019	http://ink.springer.com/chapter/10.1007/978-3- 030-22888-0_16
Springer	08.12.2024		Case Study on Eclipse Smart Home	Conference paper	excluded	not directly related to healthcare		MAR LEMMA V_10
		Davide CalvaresiGiovanni CiattoAmro NaiiarRevhan AvdoğanLeon Van der TorreAndrea	Expectation: Personalized Explainable Artificial Intelligence for Decentralized Agents with Heterogeneous Knowledge	Conference		keyword mismatch: not directly related	2021	http://link.springer.com/chapter/10.1007/978-3- 030-82017-6_20
Springer	08.12.2024	OmiciniMichael Schumacher		paper	excluded	to healthcare	0040	http://link.com/opension/chapter/40.4007/078.0
		Bjorn JonanssonBierta DelialiisiPien Walraven	Start-Ups	Conference		not directly related	2016	319-47009-2_11
Springer	08.12.2024	Ismail AlsharifAdel Khelifi	Exploring the Opportunities and Challenges of Open Source	paper	excluded	to healthcare keyword mismatch:	2019	http://link.springer.com/chapter/10.1007/978-3-
Coringor	09 12 2024		Software and Its Economic Impact on the Cybersecurity	Conference	oveluded	not directly related		030-01659-3_16
Springer	00.12.2024	Kejiang YeYangyang LiuGuoyao XuCheng-Zhong	Fault Injection and Detection for Artificial Intelligence	paper	excluded	keyword mismatch:	2018	http://link.springer.com/chapter/10.1007/978-3- 210-04205-7_8
Springer	08.12.2024	Xu	Applications in Container-Based Clouds	Conference paper	excluded	not directly related to healthcare		010-0-200-7_0
		Shirin TavaraAlexander SchliepDebabrota Basu	Federated Learning of Oligonucleotide Drug Molecule Thermodynamics with Differentially Private ADMM-Based SVM	Conference		keyword mismatch: not directly related	2021	http://link.springer.com/chapter/10.1007/978-3- 030-93733-1_34
Springer	08.12.2024	Relf KlammaMuhammad Abdub Arifin	Gamification of Web-Based Learning Services	paper	excluded	to healthcare	2017	http://ink.springer.com/chapter/10.1007/978-3-
			autimoution of these subside counting contracts	Conference		not directly related	2011	319-66733-1_5
Springer	00.12.2024	Sajitha KrishnanJ. AmudhaSushma Tejwani	Gaze Fusion-Deep Neural Network Model for Glaucoma	paper	excluded	keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978-
Springer	08.12.2024		Detection	Conference paper	excluded	to healthcare		<u></u>
		Elaine M. RaybournKillian Muollo	Guidelines for Practicing Responsible Innovation in HPC: A Sociatechnical Approach	Conference		keyword mismatch: not directly related	2023	http://ink.springer.com/chapter/10.1007/978-3- 031-34668-2_8
Springer	08.12.2024			paper	excluded	to healthcare		
		Nada AlmasriLuay TahatLaila Al Terkawai	How Can Technology Support Education in War – WarAware Education Platform for Syria	Conference		not directly related	2018	030-02131-3_39
Springer	08.12.2024	Luiz Philipe Serrano Alvesloor Scaliante WieseAna	How to Find My Task? Chatbot to Assist Newcomers in	paper	excluded	to healthcare keyword mismatch:	2022	http://ink.springer.com/chapter/10.1007/978-3-
Carianas	08 42 2024	Paula ChavesIgor Steinmacher	Choosing Tasks in OSS Projects	Conference	eveluded	not directly related		030-94890-0_6
Springer	00.12.2024	Dennis LawoThomas NeiferMargarita EsauGunnar	Human-Centred Digital Sovereignty: Explorative Conceptual	paper	excluded	keyword mismatch:	2023	http://link.springer.com/chapter/10.1007/978-3-
Springer	08.12.2024	Stevens	Model and Ways Forward	Conference paper	excluded	to healthcare		001-1000-0-0
		Mehmet S. AktasMustafa Kapdan	Implementation of Analytical Hierarchy Process in Detecting Structural Code Clones	Conference		keyword mismatch: not directly related	2017	http://ink.springer.com/chapter/10.1007/978-3- 319-62404-4_48
Springer	08.12.2024			paper	excluded	to healthcare	0040	http://figh.appingar.appi/chapter/10.1007/078-0
		PradoWanderley Lopes de SouzaSissi Marilia dos	Ontologies	Conference		not directly related	2018	319-77028-4_37
Springer	08.12.2024	Santos Forghieri PereiraLuís Ferreira Pires Takuva NakataTasuku WatanabeSinan	Integrating Uni-messe and FIWARE for Low-Code	paper	excluded	to healthcare keyword mismatch:	2022	http://ink.springer.com/chapter/10.1007/978-3-
Springer	08 12 2024	ChenMasahide Nakamura	Development of Complex Context-Aware Applications	Conference	excluded	not directly related		031-05463-1_14
opingoi	00.12.2024	Manan TrivediRaghunath Nambiar	Lessons Learned: Performance Tuning for Hadoop Systems	paper	CACIDADA	keyword mismatch:	2017	http://link.springer.com/chapter/10.1007/978-3- 319-54334-5-9
Springer	08.12.2024			Conference paper	excluded	to healthcare		
		Ali ParsaiAlessandro MurgiaSerge Demeyer	LittleDarwin: A Feature-Rich and Extensible Mutation Testing Framework for Large and Complex Java Systems	Conference		keyword mismatch: not directly related	2017	http://link.springer.com/chapter/10.1007/978-3- 319-68972-2_10
Springer	08.12.2024	Paul AlgarAriana Polyviou	Management of Multi-cloud Computing	paper	excluded	to healthcare	2017	http://link.springer.com/chapter/10.1007/978-3-
Caria	00.40.000		management of manifolda computing	Conference		not directly related	2017	319-70305-3_8
Springer	08.12.2024	Sapir GershovFadi MahameedAeyal RazShlomi	More Than Meets the Eye: Analyzing Anesthesiologists' Visual	paper	excluded	keyword mismatch:	2023	http://link.springer.com/chapter/10.1007/978-3-
Springer	08.12.2024	Laufer	Attention in the Operating Room Using Deep Learning Models	Conference paper	excluded	not directly related to healthcare		ua 1-46005-0_22
- F901		Paulo Rupino CunhaJacinto Estima	Navigating the Landscape of AI Ethics and Responsibility	Conformer		keyword mismatch:	2023	http://link.springer.com/chapter/10.1007/978-3- 031-49008-8_8
Springer	08.12.2024			paper	excluded	to healthcare		
		Gennady DikAlexander BogdanovNadezhda ShchegolevaJasur KiyamovAleksandr Dik	New Security Challenges of Internet of Things	Conference		keyword mismatch: not directly related	2023	http://ink.springer.com/chapter/10.1007/978-3- 031-37120-2_20
Springer	08.12.2024	Havder K. FatlawiAttila Kiss	On Robustness of Adaptive Random Forest Classifier on	paper	excluded	to healthcare keyword mismatch:	2020	http://ink.springer.com/chapter/10.1007/978-3-
Parinac-	09 10 0004		Biomedical Data Stream	Conference	eveluter	not directly related	-020	030-41964-6_29
opninger	00.12.2024	Stefano Ceri	On the Big Impact of "Big Computer Science"	hahai	excluded	keyword mismatch:	2017	http://link.springer.com/chapter/10.1007/978-3-
Springer	08.12.2024			Conference paper	excluded	not directly related to healthcare		M10-30/30-0_4

Search venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
		Georg MacherRumy NarayanNikolina DragicevicTiina LeinoOmar Veledar	Open Innovation Cultures	Conference		keyword mismatch: not directly related	2023	http://ink.springer.com/chapter/10.1007/978-3- 031-42310-9_20
Springer	08.12.2024	Joao GiaoJoao SarraipaRicardo Jardim-Gonçalves	Open Modular Components in the Industry Using vf-OS	paper	excluded	to healthcare keyword mismatch:	2019	http://link.springer.com/chapter/10.1007/978-3- 030-17771-3-20
Springer	08.12.2024	Muungsoo Kim lashuqang Las Japanii. Japa	Components	Conterence paper	excluded	to healthcare	2010	http://ink.springer.com/chapter/10.1007/978-3-
Springer	08.12.2024	wyungsoo rumaaenyeong Leebongpii seong	Concept, Comparison and Challenges	Conference paper	excluded	not directly related to healthcare	2013	030-24311-1_8
		Hazar Y. HmoudZainah QasemDao's HajawiJumana Ziad Al. Zoubi	Open Source Software Governance: Towards a Link to the Business World	Conference		keyword mismatch: not directly related	2022	http://link.springer.com/chapter/10.1007/978-3- 031-17968-6_11
Springer	08.12.2024	Aqeel KazmiZeeshan JanAchille ZappaMartin Serrano	Overcoming the Heterogeneity in the Internet of Things for Smart Cities	Conference	excluded	to healthcare keyword mismatch: not directly related	2017	http://link.springer.com/chapter/10.1007/976-3- 319-56877-5_2
Springer	08.12.2024	Nicole ShadowenThomas LodatoDaria Loi	Participatory Governance in Smart Cities: Future Scenarios	paper	excluded	to healthcare keyword mismatch:	2020	http://ink.springer.com/chapter/10.1007/978-3-
Springer	08.12.2024	Kan, AsañaCarla Astheff Sizzalla Caslha Marcela	and Opportunities	Conference paper	excluded	not directly related to healthcare	2020	http://link.springer.com/chapter/10.1007/978-9-
Springer	08.12.2024	Galheigolsabela CanutoDouglas de OliveiraDaniel de Oliveira	Phylogenetic Trees in Santos Dumont Supercomputer: A Practical Approach	Conference paper	excluded	not directly related to healthcare	2020	030-41005-6_31
		Mutian YangJingzheng WuShouling JiTianyue LuoYanjun Wu	Pre-Patch: Find Hidden Threats in Open Software Based on Machine Learning Method	Conference		keyword mismatch: not directly related	2018	http://link.springer.com/chapter/10.1007/978-3- 319-94472-2_4
Springer	08.12.2024	José NevesAlmeida DiasAna MoraisFrancisca FonsecaPatrícia LoretoVictor AlvesIsabel AraújoJoana MachadoBruno FernandesJorge RibeiroCesar AnalideFilipa FerrazJoão	Predicative Vagueness in Lung Metastases in Soft Tissue Sarcoma Screening	Conference	excluded	keyword mismatch: not directly related	2018	http://link.springer.com/chapter/10.1007/978-3- 030-05918-7-8
Springer	08.12.2024	NevesHenrique Vicente Mary AsanteCarsten MapleGregory Epiphaniou	Prevention of IoT-Enabled Crime Using Home Routers	paper	excluded	to healthcare keyword mismatch:	2022	http://ink.springer.com/chapter/10.1007/978-3-
Springer	08.12.2024	lürgen Neises Adrien Resse Joan-Rantiste Douguier	(PITCHR)	Conference paper	excluded	not directly related to healthcare	2021	http://ink.springer.com/chapter/10.1007/978-3-
Springer	08.12.2024	Jurgen Neises-Aunen besseJean-baptiste Houquier	Intelligence	Conference paper	excluded	not directly related to healthcare	2021	030-69781-5_1
		Petri OksaPekka Loula	Private Cloud Deployment Model in Open-Source Mobile Robots Ecosystem	Conference		keyword mismatch: not directly related	2016	http://ink.springer.com/chapter/10.1007/978-3- 319-40379-3_24
Springer	08.12.2024	Yuanfang ChiWei DaiYuan FanJun RuanKai HwangWei Cai	Private Cloud in 6G Networks: A Study from the Total Cost of Ownership Perspective	Conference	excluded	to healthcare keyword mismatch: not directly related	2020	http://link.springer.com/chapter/10.1007/978-3- 030-63941-9_30
Springer	08.12.2024	Elia KouzariLazaros Sotiriadisloannis Stamelos	Process Mining for Process Conformance Checking in an OSS	paper	excluded	to healthcare keyword mismatch:	2018	http://link.springer.com/chapter/10.1007/978-3-
Springer	08.12.2024		Project: An Empirical Research	Conference paper	excluded	not directly related to healthcare		<u>319-92375-8 7</u>
Springer	08 12 2024	Fangyu GaiBaosheng WangWenping DengWei Peng	Proof of Reputation: A Reputation-Based Consensus Protocol for Peer-to-Peer Network	Conference	excluded	keyword mismatch: not directly related to healthcare	2018	http://ink.springer.com/chapter/10.1007/978-3- 319-91458-9_41
		António CorreiaHugo ParedesBenjamim Fonseca	Reframing Taxonomy Development in Collaborative Computing Research: A Review and Synthesis of CSCW	Conference		keyword mismatch: not directly related	2018	http://link.springer.com/chapter/10.1007/978-3- 319-99504-5_5
Springer	08.12.2024	Eunice CarrasquinhaJoão SantinhaAlexander MonoclinMeria Lieitskiva Ioana Bibairo Eátima	Literature 2003–2010 Regularization Techniques in Radiomics: A Case Study on the Prediction of pCR in Breast Turnours and the Avilla	paper	excluded	to healthcare	2020	http://link.springer.com/chapter/10.1007/978-3- 030-63061-4_24
Springer	08.12.2024	CardosoCelso MatosLeonardo VanneschiNickolas Papanikolaou	Fredetion of port in breast remotis and the Axing	Conference paper	excluded	not directly related to healthcare		
		Ivo Emanuilov	Security Through Transparency and Openness in Computer Design	Conference		keyword mismatch: not directly related	2021	http://link.springer.com/chapter/10.1007/978-3- 030-68887-5_6
Springer	08.12.2024	Binod Kumar AdhikariWan Li ZuoRamesh Maharianl in Guo	Sensitive Data Detection Using NN and KNN from Big Data	Conference	excluded	to healthcare keyword mismatch: not directly related	2018	http://link.springer.com/chapter/10.1007/978-3- 030-05063-4_49
Springer	08.12.2024	Prof. Dr. Jorge Marx GómezAndreas SolsbachDr.	Smart Cities/Smart Regions - Technische, wirtschaftliche und	paper	excluded	to healthcare keyword mismatch:	2019	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	Thomas KlenkeProf. Dr. Volker Wohlgemuth	gesellschaftliche Innovationen	Conference paper	excluded	not directly related to healthcare	0000	608-2021U-5
Springer	08.12.2024	Matthias BoekerCise Midogiu	Soccer Athlete Data visualization and Analysis with an Interactive Dashboard	Conference paper	excluded	not directly related to healthcare	2023	031-27077-2_44
C	09 40 0004	Olena LiashenkoTetyana KravetsKateryna Krytsun	Software Packages for Econometrics: Financial Time Series Modeling	Conference		keyword mismatch: not directly related	2018	http://ink.springer.com/chapter/10.1007/978-3- 319-76168-8_9
Springer	00.12.2024	Lukas SommerCristian AxenieAndreas Koch	SPNC: Fast Sum-Product Network Inference	Conference	excluded	keyword mismatch: not directly related	2021	http://link.springer.com/chapter/10.1007/978-3- 030-93736-2_31
Springer	08.12.2024	Osvaldo GervasiDamiano PerriMarco	Strategies for the Digitalization of Cultural Heritage	paper	excluded	to healthcare keyword mismatch:	2022	http://ink.springer.com/chapter/10.1007/978-3-
Springer	08.12.2024	SimonettiSergio Tasso Georne ValencaCarina Alves	Strategies to Factor Software Ecosystem Partnerships – $\Delta$	Conterence paper	excluded	to healthcare keyword mismatch:	2018	http://link.springer.com/chapter/10.1007/978-3-
Springer	08.12.2024	doorgo valonyadanna vivos	System Dynamics Analysis	Conference paper	excluded	not directly related to healthcare	2010	319-93375-7_10
Carianaa	00 40 0004	Dmitriy Gakh	Street Addressing System as an Essential Component in Sustainable Development and Realization of Smart City	Conference	eveluded	keyword mismatch: not directly related	2021	http://ink.springer.com/chapter/10.1007/978-3- 030-71846-6_7
Springer	06.12.2024	Sathish A. P. KumarNorman AhmedAnastasios Bikos	SWIoTA: Anomaly Detection for Distributed Ledger Technology-Based Internet of Things (IOTA) Using Silding	Conference	excluded	keyword mismatch: not directly related	2022	http://link.springer.com/chapter/10.1007/978-3- 031-18872-5_11
Springer	08.12.2024	Alinafe KaliwoChomora Mikeka	Window (SW) Technique Tackling the Issues of Powering Mobile IoT Sensors and	paper	excluded	to healthcare	2018	http://link.springer.com/chapter/10.1007/978-3- 319-98827-6.21
Springer	08.12.2024		Systems Using Off-Grid Sources of Energy, the Case of the Real-Time Web-Based Temperature Monitoring System in Malawi	Conference paper	excluded	not directly related to healthcare		
		Simos GerasimouDimitris KolovosRichard PaigeMichael Standish	Technical Obsolescence Management Strategies for Safety- Related Software for Airborne Systems	Conference		keyword mismatch: not directly related	2018	http://ink.springer.com/chapter/10.1007/978-3- 319-74730-9_34
Springer	08.12.2024	Antonio Cerone	Ten Years of DataMod: The Synergy of Data-Driven and	Conference	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978-3- 031-16011-0_2
Springer	08.12.2024	Hubert GaravelMaurice H. ter BeekJaco van de Pol	The 2020 Expert Survey on Formal Methods	paper	excluded	to healthcare keyword mismatch:	2020	http://link.springer.com/chapter/10.1007/978-3-
Springer	08.12.2024			Conference paper	excluded	not directly related to healthcare		030-58298-2_1
Springer	08.12 2024	miniadis SiavvasUmitrios TsoukalasCharalampos MarantosAngeliki-Agathi TsintziraMarija JankovicDimitrios SoudrisAlexander ChatzigeorgiouDionvsios Kehaaias	וחפ אטעגע Hattorm for Embedded Software Quality Improvement - Preliminary Overview	Conference	excluded	keyword mismatch: not directly related to healthcare	2020	1000-58811-3-73
- p901		Luís RosaFábio SilvaCesar Analide	TMSA: Participatory Sensing Based on Mobile Phones in Urban Spaces	Conference		keyword mismatch: not directly related	2020	http://link.springer.com/chapter/10.1007/978-3- 030-62362-3_23
Springer	08.12.2024	Hasan MahmudA. K. M. Najmul IslamBilal	Toward a GDPR Compliant Blockchain Governance	paper	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978-3- 031-15342-6-36
Springer	08.12.2024	rvaqviMatti Mantymäki Terry AmoreseGenparo CordascoGiovanni D'	rramework	conterence paper	excluded	not directly related to healthcare	2022	http://link.springer.com/chapter/10.1007/978-3-
		AngeloMaria Stella de BiaseMichele Di GiovanniAnna EspositoClaudia GrecoStefano	Related Data	Conference		keyword mismatch: not directly related		031-24801-6_27
Springer	08.12.2024	MarroneFiammetta MarulliLaura Verde Giovanna BrocciaVincenzo CianciaDiego	Towards a GUI for Declarative Medical Image Analysis:	paper	excluded	to healthcare keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978-3- 031-06388-6_14
Springer	08.12.2024	Latendivileke massifik	Cognitive and Memory Load Issues	paper	excluded	to healthcare		

venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
		Isabelle FriesMaximilian GreinerManfred	Towards a Layer Model for Digital Sovereignty: A Holistic			keyword mismatch:	2023	http://link.springer.com/chapter/10.1007/978-3-
		HofmeierRazvan HresticUlrike LechnerThomas	Approach	Conference		not directly related		031-35190-7_9
Springer	08.12.2024	Wendeborn		paper	excluded	to healthcare		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
		Gabor LekoPeter Balazs	Transmission Based Adaptive Automatic Tube Voltage Selection for Computed Tomography	Conference		keyword mismatch:	2020	http://ink.springer.com/chapter/10.1007/978-3- 030-51002-2_14
Springer	08 12 2024		Selection for Computed fornography	conterence	evoluded	to healthcare		
Opiniger	00.12.2024	Miloš KravčíkKatervna NeulingerRalf Klamma	Using Personal Learning Environments to Support Workplace	paper	excluded	keyword mismatch:	2016	http://link.springer.com/chapter/10.1007/978-3-
		,	Learning in Small Companies	Conference		not directly related		319-47440-3_33
Springer	08.12.2024			paper	excluded	to healthcare		
		Mercedes Arguello CasteleiroDmitry TsarkovBijan	Using Semantic Web Technologies to Underpin the SNOMED			keyword mismatch:	2017	http://ink.springer.com/chapter/10.1007/978-3- 319-71078-5_20
Contractor	00 40 0004	ParsiaUlrike Sattler	CT Query Language	Conference	augh i da d	not directly related		
Springer	08.12.2024	Carlos E. Budde	Using Statistical Model Checking for Cybersecurity Analysis	paper	excluded	keyword mismatch:	2023	http://link.springer.com/chapter/10.1007/978-3-
		Carlos L. Dudde	Using Statistical Model Checking for Cybersecurity Analysis	Conference		not directly related	2020	031-36096-1_2
Springer	08.12.2024			paper	excluded	to healthcare		
		Nadim KobeissiGeorgio NicolasMukesh Tiwari	Verifpal: Cryptographic Protocol Analysis for the Real World			keyword mismatch:	2020	http://link.springer.com/chapter/10.1007/978-3-
				Conference		not directly related		030-65277-7_8
Springer	08.12.2024	Valéria la serre Oscaria a Davia da babistilada a	Million Country Oriented Country time Martin the InT. A Line Count	paper	excluded	to healthcare	0010	http://ink.spripper.com/chapter/10.1007/078-9-
		Valerie IssamyGeorgios Bouloukakisivikolaos GeorgantasFrancoise SailhanGéraldine Texier	in the Context of Urban Mobile Crowdsensing	Conference		not directly related	2018	319-99819-0_1
Springer	08 12 2024	Georganitasi tançoise Galinai Geraidine Texiel	In the context of orbait mobile crowdaenang	paper	excluded	to healthcare		
opinigoi	CONTRIECT	Dr. Shahram Latifi	ITNG 2021 18th International Conference on Information	Conference	ontonauou	keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
			Technology-New Generations	paper		not directly related		030-70416-2
Springer	08.12.2024				excluded	to OS		
						keyword mismatch:		
Springer	08 12 2024	Ulf GerhardtBüdiger BreitschwerdtOliver Thomas	Engineering sustainable mHealth: the role of Action Research	Article	excluded	to OS	2017	http://link.springer.com/article/10.1007/s00146- 015-0640-5
Opiniger	00.12.2024		Engineering dotainable nin toath, the role of Potent hostaren	744010	excluded	keyword mismatch:	2017	010-0040-0
			A novel agent-based, evolutionary model for expressing the			not directly related		http://link.springer.com/article/10.1007/s10489-
Springer	06.12.2024	Alex DoboliSimona Doboli	dynamics of creative open-problem solving in small groups	Article	excluded	to OS	2021	020-01919-6
						keyword mismatch:		
Continuer	00 40 0004	Amina AmaraMohamed Ali Hadj TalebMohamed	Multilingual topic modeling for tracking COVID-19 trends	Article	a secolar adaptation of	not directly related	0001	http://link.springer.com/article/10.1007/s10489-
opringer	08.12.2024	Toon De PessemierKris VanheckeAnises	based on Facebook data analySIS	Article	excluded	keyword mismatch:	2021	020-02033-3
		AllStephanie Van HoveLieven De MarezLuc	Recipe recommendations for individual users and groups in a			not directly related		hite (Bell and an and a start to see a start
Springer	08.12.2024	MartensWout JosephDavid Plets	cooking assistance app	Article	excluded	to OS	2023	023-04909-6
			GRANDPA: GeneRAtive network sampling using degree and			keyword mismatch:		
		Carly A. BobakYifan ZhaoJoshua J. LevyA. James	property augmentation applied to the analysis of partially			not directly related		http://link.springer.com/article/10.1007/s41109-
Springer	08.12.2024	O'Malley	contidential healthcare networks	Article	excluded	to OS	2023	023-00548-5
		Wamanrao A.; Raibagkar R.L.	A novel lot architecture based healthcare monitoring system	Article		keyword mismatch:	2020	s2.0-
Sconus	01 07 2024				excluded	to OS		85085012965&partneriD=40&md5=f551f2b99d8f3 ae3fc5ef1a9d709ac01
Scopus	01.07.2024	Privadarshini R.; Barik R.K.; Panigrahi C.; Dubey	An investigation into the efficacy of deep learning tools for big	Article	excluded	keyword mismatch:	2018	https://www.scopus.com/inward/record.uri?eid=2-
		H.; Mishra B.K.	data analysis in health care			not directly related		s2.0-85046906775&doi=10.4018%2fJJGHPC. 20180701018.partner/D=408.pd5=c9b307d79821
Scopus	01.07.2024				excluded	to OS		5015dc7711f08766ee0f
		Sengupta D.; Karnat A.M.; Smit Q.; Jayawardhana	Piezoresistive 3D graphene-PDMS spongy pressure sensors	Article		keyword mismatch:	2022	https://www.scopus.com/inward/record.uri?eid=2- c2.0.951254494918.doi=10.1099%.2f2059.9595%
		B.; Kottapalli A.G.P.	for IoT enabled wearables and smart products			not directly related		2fac4d0e&partnerID=40&md5=0eb786440d6f708
Scopus	27.06.2024	Codegooon D - Dovi C - Adithua C V -	DeluEffNetV/4: A CNN based colorectel polym detection in	Article	excluded	to US	2022	77a10d4c0bd865e3c https://www.scopus.com/inword/racord.uri?aid#?-
		Vivekanandhan S.	colonoscopy images	Article		not directly related	2023	s2.0-85147427802&doi=10.1177%
Scopus	27.06.2024				excluded	to OS		2/09544119221149233&partnerID=40&md5=110d 04e9d4bccaa63ea2b2a6f8e4c418
000000		Hallberg D.; Salimi N.	Qualitative and quantitative analysis of definitions of e-health	Article			2020	https://www.scopus.com/inward/record.uri?eid=2-
			and m-health			keyword mismatch:		s2.0-85086867826&doi=10.4258%2fhir, 2020.26.2.119
Sconus	01 07 2024				evoluded	to OS		&partnerID=40&md5=6f91cdcf6dd95b62eafd8001
Scopus	01.07.2024	Von Glinski A.: Erieler S.: Elia C.: Patchana T.:	Risk Factors Associated with 90-day Readmissions Following	Article	excluded	keyword mismatch:	2021	https://www.scopus.com/inward/record.uri?eid=2-
		Takayanagi A.; Arvind V.; Pierre C.; Ishak B.;	Odontoid Fractures: A Nationwide Readmissions Database	744010		not directly related	LOLI	s2.0-85111951351&doi=10.1097%2fBRS.
Scopus	30.06.2024	Chapman J.R.; Oskouian R.J.	Study		excluded	to OS		bd030e9f67937190e3d5af135f
		Lucas G.; Colson S.; Boyer L.; Inthavong K.; Haller	Risk factors for burnout and depression in healthcare workers:	Article		keysyard miamataby	2022	https://www.scopus.com/inward/record.uri?eid=2-
		P.H.; Lancon C.; Auquier P.; Gentile S.; Fond G.	The national AMADEUS study protocol; [Facteurs de risque de			reyword mismatch:		2021.06.001
Scopus	30.06.2024		étude nationale AMADEUS1		excluded	to OS		&partneriD=40&md5=43735a311795892c9ea792f e0679ab99
000000	50.00.2021	Davido B.; De Truchis P.; Lawrence C.; Annane D.;	SARS-CoV-2 reinfections among hospital staff in the greater	Article	Choladoa	keyword mismatch:	2021	https://www.scopus.com/inward/record.uri?eid=2-
		Domart-Rancon M.; Gault E.; Saleh-Mghir A.;	Paris area			not directly related		s2.0-85107391235&doi=10.1093%2fitm% 2ftaab058&partnerID=40&md5=8a97372ed7f8475
Scopus	01.07.2024	Delarocque-Astagneau E.; Gautier S.			excluded	to OS		d697fe110dbbe4711
				Article		keyword mismatch:		
Soringer	07 12 2024	Sajid IqbalWasif AltafMuhammad AslamWaqar MahmoodMuhammad Usman Chani Khan	Application of intelligent agents in health-care; review		ovaludad	not directly related	2016	http://link.springer.com/article/10.1007/s10462-
Springer	07.12.2024	Manihoodividhammad Osman Gham Kham	Application of intelligent agents in realth-care, review	Article	excluded	keyword mismatch:	2010	016-9457-y
		Vankamamidi S. NareshM. ThamaraiV. V. L.	Privacy-preserving deep learning in medical informatics;	7 a doite		not directly related		
Springer	08.12.2024	Divakar Allavarpu	applications, challenges, and solutions		excluded	to OS	2023	023-10556-7
		Qi LangChongli ZhongZhiyun LiangYizhou		Article		keyword mismatch:		
		ZhangBaokang WuFeng XuLing CongShuodong	Six application scenarios of artificial intelligence in the precise			not directly related		http://link.springer.com/article/10.1007/s10462-
Springer	08.12.2024	WuYu Tian	diagnosis and treatment of liver cancer	A -11-1-	excluded	to OS	2021	021-10023-1
		Cigdem AvciBedir Tekinerdoganloannis N	Software architectures for big data: a systematic literature	Article		keyword mismatch:		
Springer	08.12.2024	Athanasiadis	review		excluded	to OS	2020	http://link.springer.com/article/10.1186/s41044- 020-00045-1
F 9 or		Koon-Pong WongMarvin BergsneiderThomas C.	A semi-automated workflow solution for multimodal	Article		keyword mismatch:		
		GlennVladimir KepeJorge R. BarrioDavid A.	neuroimaging: application to patients with traumatic brain			not directly related	a -	http://link.springer.com/article/10.1007/s40708-
Springer	06.12.2024	HovdaPaul M. VespaSung-Cheng Huang	injury	Articl-	excluded	to OS	2016	015-0026-y
			Diabetes prediction in basilthears sustants using machine	Arucie		net directly related		
Springer	07.12.2024	N. YuvaraiK. R. SriPreethaa	learning algorithms on Hadoop cluster		excluded	to OS	2019	http://link.springer.com/article/10.1007/s10586- 017-1532-x
Shunder	J		g agenance an riddoop oldeler	Article	shouded	keyword mismatch:	_010	<u></u>
			Hybrid fragmentation of medical images' attributes for			not directly related		http://ink.springer.com/article/t0-4007is10599
Springer	08.12.2024	Ali Asghar Safaei	multidimensional indexing		excluded	to OS	2022	021-03356-7
		N. CheeneChellei Icaart Oballast O'	Light unlight where a shirt to be a set of the	Article		keyword mismatch:		
Springer	08 13 2024	N. SneenaSheibi JosephShailesh SivanBharat	Lignt-weight privacy enabled topology establishment and		ovoluted	not directly related	2000	http://link.springer.com/article/10.1007/s10586-
opninger	00.12.2024	Erukala Suresh BabuR, V. Ram Naresh VadavA	MediBlocks: secure exchanging of electronic health records	Article	excluded	keyword mismatch:	2023	022-03815-9
		Kousar NikhathSoumva Ranjan NavakWaleed	(EHRs) using trust-based blockchain network with privacy			not directly related		http://link.engineer.enginter.tr/
Springer	08.12.2024	Alnumay	concerns		excluded	to OS	2023	022-03652-w
				Article		keyword mismatch:		
Orada	07 40 000	Serge Sonfack SounchioBernard Kamsu-	Construction of a base ontology to represent accident			not directly related	0000	http://link.springer.com/article/10.1007/s10111-
Springer	07.12.2024	FoguernLaurent Geneste	expertise knowledge	Conference	excluded	to US	2023	023-00724-8 http://link.apringer.com/book/10.1007/078-3-
		Anban PiliayEogar JembereAurona J. Gerber	Artificial Intelligence Hesearch	naner		net directly related	2023	031-49002-6
Springer	07,12,2024			hahai	excluded	to OS		
- 5		Edgar JembereAurona J. GerberSerestina	Artificial Intelligence Research	Conference	Jacob	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
		ViririAnban Pillay		paper		not directly related		030-95070-5
Springer	07.12.2024				excluded	to OS		
		Mohammed SalemJuan Julián MereloPatrick	Artificial Intelligence: Theories and Applications	Conference		keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-28540-0
Springer	07 10 0004	SiarryHochdi Bachir BouladjraMohamed		paper	ovoluded	not directly related		
Springer	07.12.2024	Claudine GebinBruno WacogneAlevandro	Biomedical Engineering Systems and Taphaologies	Conference	excluded	keyword miematob	2022	http://link.springer.com/book/10.1007/078-3-
		DouplikRonny LorenzBethany BrackenCátia	conservation of the second state of the second	paper		not directly related	2022	031-20664-1
Springer	07.12.2024	PesquitaAna FredHugo Gamboa			excluded	to OS		
		Dr. Hong-Ning DaiProf. Dr. Xuanzhe LiuDaniel	Blockchain and Trustworthy Systems	Conference		keyword mismatch:	2021	http://ink.springer.com/book/10.1007/978-981-
		Xiapu LuoJiang XiaoXiangping Chen		paper		not directly related		16-/393-3
Springer	07.12.2024	Marcala Najau Care Dural Error Ottable I.I	Claud Computing, Dia Data & Examples, Table	Conferre	excluded	to OS	0000	http://link.spripper.com/book/40.4007.070.0
		Marcelo NaloutEnzo RucciFranco ChichizolaLaura	Cioud Computing, Big Data & Emerging Topics	Conterence		keyword mismatch:	2023	031-40942-4
Springer	07,12,2024	De clasti		Paper	excluded	to OS		
F		Nirbhay ChaubeySatyen ParikhKiran Amin	Computing Science, Communication and Security	Conference		keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3-
				paper		not directly related		030-76776-1
Springer	07.12.2024				excluded	to OS		

venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
		Alfredo CuzzocreaOleg GusikhinSlimane HammoudiChristoph Quix	Data Management Technologies and Applications	Conference paper		keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-37890-4
Springer	07.12.2024	Dr. Maria ShehadeTheopisti Stylianou-Lambert	Emerging Technologies and the Digital Transformation of Museums and Heritage Sites	Conference paper	excluded	to OS keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-83647-4
Springer	07.12.2024	Prof. Valentina E. BalasDr. G. R. SinhaDr. Basant	Emerging Technologies in Computer Engineering: Cognitive	Conference	excluded	to OS keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3-
Springer	07.12.2024	Agarwal Iarun Kumar SharmaPankaj DadheechMehul Mahrishi Rajan AliProf. Dr. Hermann KajndlProf. Leszek A	Computing and Intelligent Io1	Conference	excluded	to OS keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	Maciaszek	Evaluation of Novel Approaches to contivate Engineering	paper	excluded	not directly related to OS	2022	030-96648-5
Springer	08.12.2024	Tran Khanh DangJosef Kung lai M. Chung	Future Data and Security Engineering. Big Data, Security and Privacy, Smart City and Industry 4.0 Applications	Conference paper	excluded	not directly related to OS	2023	nttp://ink.springer.com/bdok/10.1007/978-981- 99-8296-7
Springer	09 12 2024	George WellsMonelo NxoziBobby Tait	ICT Education	Conference paper	ovaludad	keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-92858-2
opinger	00.12.2024	Edwige PissalouxProf. George Angelos PapadopoulosAchilleas AchilleosRamiro Velázquez	ICT for Health, Accessibility and Wellbeing	Conference paper	excluded	keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-94209-0
Springer	08.12.2024	Andrzej DziechWim MeesMarcin Niemiec	Multimedia Communications, Services and Security	Conference paper	excluded	keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-20215-5
Springer	08.12.2024	Łukasz Tomczyk	New Media Pedagogy: Research Trends, Methodological Challennes and Successful Implementations	Conference	excluded	to OS keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-44581-1
Springer	08.12.2024	Bernabé DorronsoroMario PavoneAmir NakibEl-	Optimization and Learning	Conference	excluded	to OS keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	Ghazali Talbi Ana I. PereiraAndrej KoširFlorbela P.	Optimization, Learning Algorithms and Applications	paper Conference	excluded	to OS keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	FernandesMaria F. PachecoJoão P. TeixeiraRui P. Lopes	Dennet Transfa in Januara Danasanian and Dellara Dennerilian	paper	excluded	not directly related to OS	0000	031-23236-7
Springer	08.12.2024	KC Santoshayush GoyalDjamlia AduadaAaisha MakkarYao-Yi ChiangSatish K Singh	Recent Irends in image Processing and Pattern Recognition	paper	excluded	not directly related to OS	2023	031-23599-3
Springer	09 12 2024	Selma BoumerdassiMounir GhoghoÉric Renault	Smart and Sustainable Agriculture	Conference paper	ovaludad	keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-88259-4
opinger	00.12.2024	Antonia MoropoulouAndreas GeorgopoulosMarinos IoannidesAnastasios	Transdisciplinary Multispectral Modeling and Cooperation for the Preservation of Cultural Heritage	Conference paper	excluded	keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-42300-0
Springer	08.12.2024	DoulamisKyriakos LampropoulosAlfredo Ronchi Li CuiXiaolan Xie	Wireless Sensor Networks	Conference paper	excluded	to OS keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-981- 16-8174-5
Springer	08.12.2024				excluded	to OS keyword mismatch:		
Springer	08.12.2024	Vasilis VlachokyriakosClara CrivellaroHara KoukiChristos GiovanopoulosPatrick Olivier Madeleine RenviPetra GaugischAlexandra	Research with a Solidarity Clinic: Design Implications for CSCW Healthcare Service Design	Article	excluded	not directly related to OS keyword mismatch:	2021	http://link.springer.com/article/10.1007/s10606- 021-09404-5
Springer	08.12.2024	HunckStefan StrunckChristophe KunzeFrank Teuteberg	Uncovering the Complexity of Care Networks – Towards a Taxonomy of Collaboration Complexity in Homecare	Article	excluded	not directly related to OS	2022	http://link.springer.com/article/10.1007/s10606- 022-09433-8
Springer	08.12.2024	Hrudaya Kumar TripathySushruta MishraShubham SumanAnand NayyarKshira Sagar Sahoo	Smart COVID-shield: an IoT driven reliable and automated prototype model for COVID-19 symptoms tracking	Article	excluded	not directly related to OS	2022	http://link.springer.com/article/10.1007/s00607- 021-01039-0
ACM Digital Library	30.11.2024	Stan Swanborn and Ivano Malavolta	Robot runner: a tool for automatically executing experiments on robotics software	Conference paper	excluded	keyword mismatch: not directly related to healthcare	2021	https://doi.org/10.1109/ICSE-Companion52605, 2021.00029
ACM Digital	28.05.2024	Dominik Renzel, István Koren, Ralf Klamma, and	Preparing research projects for sustainable software	Conference	evoluted	keyword mismatch: not directly related	2017	
ACM Digital	20.03.2024	Jenny T. Liang, Maryam Arab, Minhyuk Ko, Amy J.	A Qualitative Study on the Implementation Design Decisions of	Conference	excluded	keyword mismatch: not directly related	2017	https://doi.org/10.1109/ICSE-SEIS.2017.4
Library	28.10.2024	Ko, and Thomas D. LaToza Simon J. E. Taylor, Anastasia Anagnostou, Adedeji Fabiyi, Christine Currie, Thomas Monks, Roberto	Developers	paper	excluded	to healthcare keyword mismatch: not directly related	2023	https://doi.org/10.1109/ICSE48619.2023.00047
Library	22.06.2024	Barbera, and Bruce Becker	simulation	paper	excluded	to healthcare keyword mismatch:	2017	https://ieeexplore.ieee.org/document/8247813
ACM Digital Library	23.05.2024	Niina Maarit Novak and A. Min Tjoa	ICT as an Enabler for a Society where No One is Left Behind	paper	excluded	to healthcare keyword mismatch:	2018	https://doi.org/10.1145/3282373.3282381
ACM Digital Library	11.11.2024	Tiago Espinha Gasiba, Ulrike Lechner, Maria Pinto- Albuquerque, and Daniel Mendez	Is secure coding education in the industry needed? an investigation through a large scale survey	Conference paper	excluded	not directly related to healthcare	2021	https://doi.org/10.1109/ICSE-SEET52601, 2021.00034
ACM Digital Library	28.05.2024	Per Runeson	Open collaborative data: using OSS principles to share data in SW engineering	Conference paper	excluded	not directly related to healthcare	2019	https://doi.org/10.1109/ICSE-NIER.2019.00015
ACM Digital Library	30.11.2024	Sean McGuire, Erin Schultz, Bimpe Ayoola, and Paul Ralph	Sustainability is Stratified: Toward a Better Theory of Sustainable Software Engineering	Conference paper	excluded	keyword mismatch: not directly related to healthcare	2023	https://doi.org/10.1109/ICSE48619.2023.00169
ACM Digital	00.44.0004	Robert W. Brennan, Peter Goldsmith, and Nancy	Participatory Simulation to Support Transactional Curriculum	Conference	and a dealer of	keyword mismatch: not directly related	0000	
ACM Digital	29.11.2024	Juliana Bowles, Giovanna Broccia, and Mirco	DataMod2020: 9th International Symposium "From Data to	Conference paper	excluded	keyword mismatch: not directly related	2023	https://ieeexplore.ieee.org/document/10015321
Library	08.11.2024	Nanni Boris Velichkov, Kristina Ivanova, Valeri Hristov, Ivan Borisov, Alexander Paychev, Ivan Kovchev	Models and Back"	Conference	excluded	to OS keyword mismatch:	2020	https://doi.org/10.1145/3340531.3414073
Library	04.11.2024	and Svetla Boytcheva	Corpus Generation	paper Conference	excluded	to OS keyword mismatch:	2021	https://doi.org/10.1145/3439133.3439141
ACM Digital Library	10.11.2024	Samuel Marcos, Juan A. Juanes, and Michael Walters	Impact of state-of-the-art technologies on medical training processes and clinical practice	paper Conference	excluded	to OS keyword mismatch:	2021	https://doi.org/10.1145/3486011.3486459
ACM Digital Library	09.11.2024	Khuloud Abou Amsha, Erik Gronvall, and Joanna Saad-Sulonen	Emergent Collaborations Outside of Organizational Frameworks: Exploring Relevant Concepts	paper	excluded	not directly related to OS	2023	https://doi.org/10.1145/3593743.3593778
ACM Digital Library	30.11.2024	Klymenko, Andreas Biesdorf, Uwe Hohenstein, and Florian Matthes	Uncertainty expressions in software architecture group decision making: explorative study	paper	excluded	not directly related to OS	2018	https://doi.org/10.1145/3241403.3241447
ACM Digital	30.11.2024	Diva Smriti, Ji Youn Shin, Munif Mujib, Meghan Colosimo, Tsui-Sui Kao, Jake Williams, and Jina Huh-Yoo	TAMICA: Tailorable Autonomous Motivational Interviewing Conversational Agent	Conference paper	excluded	keyword mismatch: not directly related to OS	2021	https://doi.org/10.1145/3421037.3421083
2.5rary	00.11.2024	Badr A.; Elgazzar K.	A Framework for Real-time Remote ECG Monitoring and Diagnoses	Conference paper	CAGINGO	keyword mismatch: not directly related	2022	https://www.scopus.com/inward/record.uri?eid=2- s2.0-85147552015&doi=10.1109% 2fICCSPA5580.2022.10019050
Scopus	05.12.2024	Van Der Burgt O.; Castermans L.	Big Data Driven Reliability Growth for Healthcare Systems	Conference	excluded	to OS	2021	apartnertII=40&md5=4c44335a625f2b25ae22cee 5d5c4c06d https://www.scopus.com/inward/record.uri?eid=2- s2_0-65123050154&doi=10.1109%2/IRAM%48/097
Scopus	05.12.2024			hahai	excluded	not directly related to OS		2021.9605713 &partneriD=40&md5=a8f48912d6d954d9fbd510b 00&cfb9d6
		Rombosia K.; Oele E.; Rangara N.; Mwaura J.; Mitto B.; Ondura E.; Onyango D.; Akoth C.	EFFECT of the DISTANCES of PUBLIC HEALTH FACILITIES from the NEAREST MAJOR ROADS on SKILLED DELIVERIES CONDUCTED in KISUMU COUNTY, KENYA	Conference paper		keyword mismatch: not directly related	2019	https://www.scopus.com/inward/record.uri?eid=2- s2.0-86074340447&doi=10.5194%2fisprs- archivos-XLII-4-W14-203- 2019&partnerID=40&md5=2a6565baa56bfca5269
Scopus	06.12.2024	Yüksekogul N.; Schreiweis B.; Weiss N.; Aguduri L. S.; Brandner A.; Bronsch T.; Pensold P.; Stein K.E.;	Patient portal integration - A native IHE connector implementation for PEHR	Conference paper	excluded	to OS keyword mismatch:	2017	8e333a31a78d8 https://www.scopus.com/inward/record.uri?eid=2- s2.0.85020374523&doi=10.3233%2/978-1-61499- 679.1
Scopus	06.12.2024	Helmer A.; Bergh B.; Heinze O.			excluded	not directly related to OS		172&partnerID=40&md5=98c364aab7bd3373dd8 3b384f0d67b9d

Search	Query date	Author	Title	Type	Status	Reason	Year	Article URI
Venue	Query dute	Alfaify L.; Alnajem N.; Alanzi H.; Almutiri R.; Alotaibi A.; Alhazri N.; Alqahtani A.	Performance Evaluation of SDN Controllers: RYU and POX for WBAN-based Healthcare Applications	Conference paper	otatao	keyword mismatch: not directly related	2023	https://www.scopus.com/inward/record.uri?eid=2- s2.0-85189761078&doi=10.1109% 2/ICMERALDA60125.2023.10458183 &nather/D=40&md52/2023.10458183
Scopus	06.12.2024	Gartner D.; Williams E.M.; Harper P.R.	Prescriptive Healthcare Analytics: A Tutorial on Discrete	Conference	excluded	to OS keyword mismatch:	2022	a1964/295 https://www.scopus.com/inward/record.uri?eid=2- s2.0-85139026541&doi=10.1109%2tICHI54592.
Scopus	06.12.2024	Serbanati A : Morena M : Lancia L	Protocol adapter: A reusable solution to interoperability and	Conference	excluded	not directly related to OS keyword mismatch:	2016	2022.00111 &partnerID=40&md5=02bb195b013ba5445b8ae8 bb39ef348c https://www.scopus.com/inward/record.uri?eid=2-
Scopus	06.12.2024	Constant PL, Morona M., Eanola E.	integration issues in mHealth data-collection systems	paper	excluded	not directly related to OS	2010	s2.0-84969265170&doi=10.5220% 2f0005845605500560&partnerID=40&md5=2eeea f00c028c5df20327e6b28c64172
		Grammatikakis M.D.; Koumarelis A.; Mouzakitis A.	Software Architecture of a User-Level GNU/Linux Driver for a Complex E-Health Biosensor	Conterence paper		keyword mismatch: not directly related	2021	https://www.scopus.com/inwardrecord.ur/reid=z- s2.0-85101758278&doi=10.1007%2f978-3-030- 66729- 0_12&partnerID=40&md5=ce43393ade16daddaf5
Scopus	06.12.2024	Zamfir M.; Preda Ş.; Florian V.; Militaru G.; Tigănoaia B.	Towards a Platform for Clinical Big Data Analysis	Conference paper	excluded	to OS keyword mismatch: not directly related	2017	Aeffce94f3681 https://www.scopus.com/inward/record.uri?eid=2- s2.0- 850486208108.partner/D=408.md5=439c68cfcrd64
Scopus	06.12.2024	Ghaleb M.; Azzedin F.	Towards designing a federated database framework for disease outbreak notification systems	Conference paper	excluded	to OS keyword mismatch:	2016	2578698f77db73bda69a https://www.scopus.com/inward/record.uri?eid=2- s2.0-84964861770&doi=10.1109%2fOSSCOM. 0016.7372001
Scopus	06.12.2024	Gligorić N.; Dordevic M.; Martín D.S.; Jokić S.;	Web ontology design for data and services in CLIMOS project	Conference	excluded	not directly related to OS	2023	Roartner1D=40&md5=c647ace3d9c805c0338b6f3 f72f0944 https://www.scopus.com/inward/record.uri?eid=2-
Scopus	06.12.2024	Jokić I.		paper	excluded	not directly related to OS		s2.0-85165413702800=10.1109% 2flcETRAN59631.2023.10192131 &partnerID=40&md5=4dd051839cdbe34b5f0e3de ab23cf36f
Scopus	06 12 2024	Beller S.E.; Monatesti S.J.; Bachman P.	Whole Person Integrated Care (WPIC): A healthcare transformation strategy supported by a novel spreadsheet- based software framework	Conference paper	aveluded	keyword mismatch: not directly related	2017	https://www.scopus.com/inward/racord.uri?eid=2- s2.0-850477490348doi=10.1109%2fGHTC. 2017.8239223 &partnerID=40&md5=2027e32a214ce85738cfa14 b245-72-44
ACM Digital	00.12.2024	Allegra Crahay, Federico Chiarelli, Miguel Alvarez Rodriguez, Ghita Ennadif, Solène Vossot, Ruben	The Berlin Declaration monitoring mechanism: An assessment framework supporting the implementation of a political	Conference paper	excluded	keyword mismatch: not directly related		<u>1224077.044</u>
Library	30.11.2024	Narzul, Esther Bleys, and Victoria Kalogirou	declaration	Conference paper	excluded	to OS keyword mismatch: not directly related	2022	https://doi.org/10.1145/3560107.3560305
Library	04.11.2024	Ruochen Liao	Attention-based View of Online Information Dissemination Making Cyberspace Towards Sustainability A Scientometric	Conference	excluded	to OS keyword mismatch:	2018	https://doi.org/10.1145/3209626.3209628
ACM Digital Library	11.11.2024	Zijia Wang, Han-Teng Liao, Jiacheng Lou, and Yu Liu	Review for a Cyberspace that Enables Green and Digital Transformation	paper Conference	excluded	not directly related to OS keyword mismatch:	2021	https://doi.org/10.1145/3444370.3444603
ACM Digital Library	04.11.2024	Thomas Hanne, Phillip Gachnang, Stella Gatziu Grivas, Ilyas Kirecci, and Paul Schmitter	Artificial Intelligence and Machine Learning for Maturity Evaluation and Model Validation	paper	excluded	not directly related to OS	2022	https://doi.org/10.1145/3556089.3556102
ACM Digital Library	27.10.2024	Michael Galliakis, Christos Skourlas, Eleni Galiotou, and Ioannis Voyiatzis	A low-cost smart home for the assistance of elderly persons and patients	paper	excluded	not directly related to OS	2018	https://doi.org/10.1145/3291533.3291578
ACM Digital Library	11.11.2024	Philip Kaiser, André Langer, and Martin Gaedke	MPCC: Generic Secure Multi-Party Computation in Centralized Cloud-based Environments	conterence paper	excluded	keyword mismatch: not directly related to OS	2020	https://doi.org/10.1145/3372454.3372467
ACM Digital	27 10 2024	Verayuth Lertnattee, Perayot Pamonsinlapatham, and Woarawut Oniam	A Course Design and a Flexible Platform for Learning Programming Languages in Health Informatics Courses: Flexible Platform for Learning Programming Languages	Conference paper	excluded	keyword mismatch: not directly related to OS	2023	https://doi.org/10.1145/3588730.3588808
ACM Digital	24.40.2024	Mamatas Resum and Salma Sabanavia	*All for One and One for All": Systematic Data Collection and	Conference paper		keyword mismatch: not directly related	2047	
ACM Digital	24.10.2024	Jarkko Hyysalo, Gavin Harper, Jaakko Sauvola, Anja Keskinarkaus, Ilkka Juuso, Miikka Salminen,	Sharing to Advance Socially Assistive Robots	Conference paper	excluded	keyword mismatch: not directly related	2017	https://doi.org/10.1145/3029796.3029803
Library	08.11.2024	and Juha Partala Sinha A.; Bhargavi M.; Singh N.K.; Garg N.; Pal S.; Verma A.	Defining an architecture for evolving environments Comparative Analysis of Machine Learning and Data Mining based Multi-Models for Diabetes Risk Prediction	Conference paper	excluded	to OS keyword mismatch: not directly related	2017 2022	https://doi.org/10.1145/3019612.3019902 https://www.scopus.com/inward/record.uri?eid=2- s2.0-8516475625&doi:10.1109% 28CWITE57052.2022.10176232
Scopus	05.12.2024	Krishna Sathya A.C.; Mastan Rao P.; Babu S.	Development of a Covid-19 Information Dashboard to Access the Health Care Resources and Requirements Online	Conference paper	excluded	to OS keyword mismatch:	2022	ApartnerIJ=40&md5=9821e92e42b86c361c2Uta6 a66508264 https://www.scopus.com/inward/record.uri?eid=2- s2.0-85146928865&doi=10.1109%2fICAN56228, 2022 10007167
Scopus	05.12.2024	Schönbeck N.; Hussein Y.; Haack A.; Schmager A.;	Evaluating REDCap as the Central Data Collection Tool for the	Conference	excluded	to OS keyword mismatch:	2023	ApartneriD=40&md5=d9c7ca4702aa7887f9584cc daffd5b67 https://www.scopus.com/inward/record.uri?eid=2-
Scopus	06.12.2024	Harney U.; Trübe L.; Ückert F.; Gottfried K.	Hamburg City Health Study Mixed-Mode Pandemic Modeling	paper Conference	excluded	not directly related to OS	2019	s2.0-85170628760&doi=10.3233% 2fSHTI230693&partnerID=40&md5=f6fe0b13b74e e6381f0a8c548db13ba5 https://www.scopus.com/inward/record.uri?eid=2-
Scopus	06.12.2024		minos moso i sinsonno mosoning	paper	excluded	keyword mismatch: not directly related to OS	20.0	s2.0-850811228318doi=10.1109%2fWSC40007, 2019.9004801 &partnerID=40&md5=6622dfed8a925b4be56210t 96bfd2870
Coopus	00.12.2021	Tran Q.N.H.; Nguyen N.Q.; Amodeo L.; Yalaoui F.; Chehade H.	Multidimensional Assignment Problem in the medical sector: A case study	Conference paper	CABIDACA	keyword mismatch: not directly related	2022	https://www.scopus.com/inward/record_uri?eid=2- s2.0.85144569083&doi=10.1016%2fj.ifacol. 2022.09.416 &nartnerID=40&md5=a605ca75c3719cceca98d81
Scopus	06.12.2024	Kaniz Fatema Madhobi, Methun Kamruzzaman, Ananth Kalvanaraman, Eric Lofaren, Rebekab	A Visual Analytics Framework for Analysis of Patiant	Conference	excluded	to OS keyword mismatch: not directly related		82c09430e
ACM Digital Library	04.11.2024	Manuf Raiyanaraman, cric Lorgren, Rebekan Moehring, and Bala Krishnamoorthy Sun RT.; Hu HC.; Chang HL.	A visual Analysics Framework for Analysis of Patient Trajectories A design science approach to improve adherence on exercise	Conference	excluded	to OS keyword mismatch:	2019 2016	https://doi.org/10.1145/3307339.3342143 https://www.scopus.com/inward/record.uri?eid=2- s2.0-
Scopus	05.12.2024	Martinez J.; Ruiz A.; Garzo A.; Keller T.;	Modelling the Component-based Architecture and Safety	Conference	excluded	to OS	2021	85021740463&partnerID=40&md5=fcf1955d6237 9d32fde57dd8ea71a036 https://www.scopus.com/inward/record.uri?eid=2- e2.0.85129872927&di=10.11098/292655553
Scopus	06.12.2024	Radermacher A.; Ionetta S.	Contracts of ArmAssist in Papyrus for Robotics	paper	excluded	not directly related to OS		2021.00009 &partnerID=40&md5=118e292cebbb2a019389363 cb93c3820
		Rodrigo Dantas da Silva, Jean Jar Pereira de Araújo, Álvaro Ferreira Pires de Paiva, Ricardo Alexsandro de Medeiros Valentim, Karilany Dantas		paper		keyword mismatch:		
ACM Digital Library	25.10.2024	Coutinho, Jailton Carlos de Paiva, Azim Roussanaly, and Anne Boyer	A big data architecture to a multiple purpose in healthcare surveillance: the Brazilian syphilis case	Conference	excluded	not directly related to OS keyword mismatch:	2021	https://doi.org/10.1145/3401895.3402092
ACM Digital Library	09.11.2024	F. Fathi, N. Abghour, and M. Ouzzif	From Big data platforms to smarter solution, with intelligent learning: [PAV] 4 - Pave the way for Intelligence	paper	excluded	not directly related to OS	2017	https://doi.org/10.1145/3141128.3141143
ACM Digital Library	30.11.2024	Anna-Lena Theus	Sciences can Promote more Equitable and Just Algorithmic System Design	paper	excluded	not directly related to OS	2023	https://doi.org/10.1145/3593013.3594022
ACM Digital Library	25.10.2024	Yiyang Wang	A Comparison of Machine Learning Algorithms in Blood Glucose Prediction for People with Type 1 Diabetes	Conference paper	excluded	keyword mismatch: not directly related to OS	2021	https://doi.org/10.1145/3500931.3500993
ACM Digital	08 11 2024	Praowpan Tansitoong	Choice variation on electronic health database: a case study of medical decision mapping from healthcare scheme of Thailand	Conference paper	excluded	keyword mismatch: not directly related to QS	2022	https://doi.org/10.1145/3545720.3545740
ACM Digital	30 44 000 1	Suzana Brown and Eshaem Hussein	South-South Cooperation: Can Rwanda's Community Health	Conference paper	oveluded	keyword mismatch: not directly related	2040	
ACM Digital	30.11.2024		Convoc Carri From Bangiaucoril IIIIO Laby	Conference paper	excluded	keyword mismatch: not directly related	2010	naps.vdoi.org/10.1145/2909609.2909619
Library ACM Digital	11.11.2024	Juan Antonio Juanes and Pablo Ruisoto	Latest technology trends in health sciences education TAF: Temporal Analysis Framework for Handling Data velocity	Conference paper	excluded	to OS keyword mismatch: not directly related	2016	https://doi.org/10.1145/3012430.3012556
Library	30.11.2024	C. S. Sindhu and Nagaratna P. Hegde Baniecki H.; Sobieski B.; Bombiński P.; Szatkowski P. Biecek P.	in Healthcare Analytics Hospital Length of Stay Prediction Based on Multi-modal Data Towards Trustworthy Human-Al Collaboration in Badiamics	Conference	excluded	to OS keyword mismatch:	2016 2023	https://doi.org/10.1145/3010089.3010095 https://www.scopus.com/inward/record.uri?eid=2- s2.0-85164037083&doi=10.1007%2(978-3-031-
Scopus	06.12.2024	Kowalavicz C - Pirrona 1- Huarte M	Implementation model using a hippografic protocol in mobile	Conference	excluded	not directly related to OS	2017	34344- 5_9&partnerID=40&md5=fe8746b8279028bca741 6e934ef0ec87 https://www.scopus.com/inward/record.uri2e <sup>14</sup> -2-
Scopus	06.12.2024	Nonaionicz G., Fillone G., Fühlte M.	terminals with NFC technology	paper	excluded	keyword mismatch: not directly related to OS	2017	s2.0-85025835880&doi=10.1109%2/ICCDCS. 2017.7959706 &partnerID=40&md5=667acdeb87649545fd438c6 d3tbc9201
Coopus	00.12.2024				skolaued			444444201
Search	Quany data	Author	Title	Tune	Statue	Passon	Vear	Addate UDI
------------------------	------------	---	---	---------------------	----------	--	--------------	---
venue	Query date	Author Ramzan Ali A.B.; Alagumariappan P.; Vijayalakshmi S.; Gughan N.M.; Bhaskar K.B.	Analysis of Secure Transfer of Healthcare Data using Fog Computing	Conference paper	Status	keyword mismatch: not directly related	2020	Article URL https://www.scopus.com/inward/record.uri?eid=2 s2.0-85094203249&doi=10.1109% 2/ICOSEC49089.2020.9215405 Enathwell.bad&md558089.901c070131475c012
Scopus	05.12.2024	Lam F.S.; Chow B.Y.	Disaster response network analysis in rural Temerloh, Pahang communities during the Malaysia 2020-2021 flood	Conference paper	excluded	to OS keyword mismatch: not directly related	2022	apartnernu==4ukmids=eudeatan cu/rusa i=v3cu/z 824c3062/ https://www.scopus.com/inward/record.uri?eid=2 s2.0=85146870775&doi=10.1051%2fe3scoff% 2/202234705003&nathertD=40&Mid5=s56c76e5
Scopus	06.12.2024	Farooq E.; Nawaz Ui Ghani M.A.; Naseer Z.; Iqbal S.	Privacy Policies' Readability Analysis of Contemporary Free Healthcare Apps	Conference paper	excluded	to OS keyword mismatch:	2020	3a6a02caa8bff70831b4dca https://www.scopus.com/inward/record.uri?eid=2 s2.0-85101451694&doi=10.1109% 2ftCOSST51357_2020.9332991
Scopus	06.12.2024	Menger V.; Spruit M.; De Bruin J.; Kelder T.; Scheepers F.	Supporting reuse of EHR data in healthcare organizations: The CARED research infrastructure framework	Conference paper	excluded	to OS keyword mismatch: not directly related	2019	&partnerID=40&md5=1054014b08ceaca201674 92ad5259e3 st0.e5004639109&doi=10.5220% 92002342000410058.costeadD=408.md5=25c0 92002342000410058.costeadD=408.md5=25c0
Scopus ACM Digital	06.12.2024	Bill Tomlinson, M. Six Silberman, Andrew W. Torrance, Kurt Squire, Paramdeep S. Atwal, Ameya		Conference paper	excluded	to OS keyword mismatch: not directly related		1ea6e9f49579b8ea770c1e667cc
Library	28.10.2024	N. Mandalik, Sahil Railkar, and Rebecca W. Black Conte R.; Sansone F.; Grande A.; Tonacci A.; Napoli F.; Pala A.P.; Raciti M.; Landi P.	A Participatory Simulation of the Accountable Capitalism Act Development of an integrated ICT system for data production, standardization and elaboration in health & care	Conference paper	excluded	to OS keyword mismatch: not directly related	2020 2017	https://doi.org/10.1145/3313831.3376326 https://www.scopus.com/inward/record.uri?eid=2 s2.0-85028570562&doi=10.1109%2/EHB, 2017.7995426
Scopus	05.12.2024	Sethi T.; Nagori A.; Bhatnagar A.; Gupta P.; Fletcher R.; Lodha R.	Validating the tele-diagnostic potential of affordable thermography in a big-data data-enabled ICU	Conference paper	excluded	to OS keyword mismatch: not directly related	2017	Apartneril ==4U&mds=097.40cscbe9U89478bbs.3581 557fb525b https://www.scopus.com/inward/record.uri?eid=2 s2.0.85019711127&doi=10.1145%2005219, 3055234&partnerID=40&md5=9873914fe068400
Scopus	06.12.2024		Function the Dhish: The Effect of Developing Techniques in	Conference	excluded	to OS keyword mismatch:		d1524fde975bf5cc
ACM Digital Library	09.11.2024	Michael Koddebusch Regulski P.; Wendykier P.; Kantiem K.; Murdzek W.	Phishing E-Mails Advanced methods of visual analysis and visualization of various aspects of the COVID-19 outbreak in Poland	Conference	excluded	to OS keyword mismatch:	2022 2021	https://doi.org/10.1145/3543434.3543476 https://www.scopus.com/inward/record_uri?eid=2 s2.0=851168978718doi=10.1016%2fj.procs. 2021.09.195
Scopus	05.12.2024	Alaiad A.; Najadat H.; Al-Mnayyis N.; Khalil A.	Associative Classification of the Jordanian Hospitals Efficiency Based on DEA	Conference	excluded	not directly related to OS keyword mismatch:	2018	Apartner/ID=40&md5=7b8e857ea7019ef7149b8 c2769794 https://www.scopus.com/inward/record.uri?eid=2 s2.0-85052776602&doi=10.1109%2/IC8DAA,
Scopus	05.12.2024	Maudail K - Mahashwari D - Darakh H P - Davadkar	Cloud based secure smarteard healthcare monitoring and	Conforance	excluded	not directly related to OS	2017	2018.8629579 &partnerID=40&md5=52d2098568f0e3cc1229df 6c3ca55cc bttps://www.scopus.com/inward/record_ui?eid=2
Scopus	05.12.2024	K.	cloue-based secure smartcard nearthcare monitoring and tracking system	paper	excluded	keyword mismatch: not directly related to OS	2017	2.0-850405411758doi=10.1109%28CECCT_ 2017.8117869 8partner10=40&md5=27a51de860af032d37c531 0446e3398
Scopus	05 12 2024	Mansoor H.; Gerych W.; Buquicchio L.; Chandrasekaran K.; Agu E.; Rundensteiner E.	ComEX: Identifying mislabeled human behavioral context data using visual analytics	Conference paper	excluded	keyword mismatch: not directly related to OS	2019	https://www.scopus.com/inward/record.uri?eid=2 s2.0-850726860968doi=10.1109%2fCOMPSAC 2019.10212 &partnerID=40&md5=2480ad94b1a3b2fe630e3 27dd3acf57
0	00.10.0001	Payne M.; Pooke F.; Geoffrey Chase J.; Campbell J.; Holder-Pearson L.; Knopp J.	The separation of insulin pump hardware and software - A novel and low-cost approach to insulin pump design	Conference paper		keyword mismatch: not directly related	2021	https://www.scopus.com/inward/record.uri?eid=2 s2.0-85120667040&doi=10.1016%2fj.ifacol, 2021.10.306 &partner/D=40&md5=ad2bf29865b118cff8738e5
ACM Digital	06.12.2024	Ava Scott, Carolin Stellmacher, Jasmin Niess, and	Vicious or Virtuous Cycle? The Privacy Implications of Active	Conference paper	excluded	to OS keyword mismatch: not directly related		322d5d88
Library	30.11.2024	Maheux MA.; Caya C.; Letourneau D.; Michaud F.	T-Top, a SAR Experimental Platform	Conference paper	excluded	keyword mismatch: not directly related	2022	https://doi.org/10.1145/3529190.3534732 https://www.scopus.com/inward/record.uri?eid=2 s2.0-85130936073&doi=10.1109%2/HRI53351. 2022.9889346 Anather/10-4108md5ccr6d/08ach244a63a23387
Scopus	06.12.2024	Iulia Valtanuta Distor A Diamama Manjam	Conception Enciel European Date: Commutational and	Conference	excluded	to OS keyword mismatch:		b1c30c9d49
ACM Digital Library	09.11.2024	Julija Valtonyte, Pieter A. Blomsma, Maryam Alimardani, and Max M. Louwerse Joy A.; Hafsiya T.H.; King G.	Generating Facial Expression Data: Computational and Experimental Evidence A Review on Glucose Monitoring Using Enabling Technologies of Internet of Things	Conference paper	excluded	to OS keyword mismatch:	2019 2021	https://doi.org/10.1145/3308532.3329443 https://www.scopus.com/inward/record.uri?eid=2 s2.0-8510802407684.doi=10.11093 g1CocCS81409.0201.0442042
Scopus	05.12.2024	Chan J.; Glenn A.; Itani M.; Mancl L.R.; Gallagher F · Bly R · Patel S · Gollakota S	Wireless earbuds for low-cost hearing screening	Conference	excluded	not directly related to OS keyword mismatch:	2023	Apartner/D=40&md5=147/5aa423282(275a906c 66daf0411 https://www.scopus.com/inward/record.uri?eid=2 2.0-85169419374&doi=10.1145%2(2581791.
Scopus	06.12.2024	Kelvin K.F. Tsoi, Janet Y.H. Wong, Michael P.F.		Conference	excluded	to OS		3596856&partnerID=40&md5=8f5835e12ec91al 09df22cee14cd2a7
ACM Digital Library	29.11.2024	Wong, Gary K.S. Leung, Baker K.K. Bat, Felix C.H. Chan, Yong-Hong Kuo, Herman H.M. Lo, and Helen M.L. Meng	Personal Wearable Devices to Measure Heart Rate Variability: A Framework of Cloud Platform for Public Health Research	paper	excluded	keyword mismatch: not directly related to OS	2017	https://doi.org/10.1145/3079452.3079453
ACM Digital Library	09.11.2024	Miguel Ehécatl Morales-Trujillo and Gabriel Alberto García-Mireles	Evolving with patterns: a 31-month startup experience report	paper	excluded	not directly related to OS	2019	https://doi.org/10.1145/3338906.3340447
Scopus	05.12.2024	Rahman H.; Slepian M.; Mitra A.	A novel big-data processing framwork for healthcare applications: Big-data-healthcare-in-a-box	paper	excluded	keyword mismatch: not directly related to OS	2016	https://www.scopus.com/invaro/record_un/reid=z s2.0=85015221932&doi=10.1109%2fBigData. 2016.7841018 &partner/D=40&md5=b2254f3f7f30c6fe8e3243fs a70fcf9
Scopus	06.12.2024	Nicolai M.; Pascarella L.; Palomba F.; Bacchelli A.	Healthcare android apps: A tale of the customers' perspective	Conference paper	excluded	keyword mismatch: not directly related to OS	2019	https://www.scopus.com/inward/record.uri?eid=2 s2.0.85076463724&doi=10.1145%2f3340496. 33427558.partner/D=40&md5=df5de32a214292f 8d18b840f145a3c4
ACM Digital Library	08.11.2024	Keh-Kim Kee and Boung-Yew Lau Simon	Cloud-Based IoT Solution for Predictive Modeling of Ship Fuel Consumption	Conference paper	excluded	keyword mismatch: not directly related to OS	2019	https://doi.org/10.1145/3316615.3316710
ACM Digital Library	11.11.2024	Borhane Blili-Hamelin and Leif Hancox-Li	Making Intelligence: Ethical Values in IQ and ML Benchmarks	Conference paper	excluded	keyword mismatch: not directly related to OS	2023	https://doi.org/10.1145/3593013.3593996
ACM Digital Library	30.11.2024	Jinfeng Lin, Yalin Liu, and Jane Cleland-Huang	Supporting Program Comprehension through Fast Query response in Large-Scale Systems	paper	excluded	not directly related to OS	2020	https://doi.org/10.1145/3387904.3389260
Scopus	06.12.2024	Dayai M.; Singn N.	Indian Health Care Analysis using Big Data Programming Tool	paper	excluded	keyword mismatch: not directly related to OS	2016	https://www.scopus.com/invariorecord.unreed=_ s2.0-84986557074&doi=10.1016%2fj.procs. 2016.06.101 &partner/D=40&md5=36274c84bb5b3ebaf405d 90dfs2154
Connue	05 10 0001	Antunes M.; Silva C.; Barranca J.	A Telemedicine Application Using WebRTC	Conference paper	oveluded	keyword mismatch: not directly related	2016	https://www.scopus.com/inward/record.uri?eid=2 s2.0-85005923057&doi=10.1016%2fj.procs. 2016.09.177 &nartnerID=40&md5=1bf086f3b5f391431a886b
Scopus	05.12.2024	Senbagavalli M.; Singh S.K.	Improving Patient Health in Smart Healthcare Monitoring Systems using IoT	Conference paper	excluded	keyword mismatch: not directly related	2022	10020083 https://www.scopus.com/inward/record.uri?eld=2 s2.0-85154041797&doi=10.1109% 2/INCOFT5651.2022.10094409 &nartner/10=408md5=55ee2dd4bb5a25da64de7
Scopus	06.12.2024	Sheshasaayee A.; Meenakshi V.	Ischemic heart disease deduction using doppler effect spectrogram	Conference paper	excluded	to OS keyword mismatch: not directly related	2019	5149883a18 https://www.scopus.com/inward/record.uri?eid=2 s2.0-85059588263&doi=10.1007%2f978-981-13 1742-
Scopus	06.12.2024			Conference	excluded	to OS keyword mismatch:		2_14&partnerID=40&md5=08d3cd86be6fcdcfe7 8ae20573b9637
ACM Digital Library	08.11.2024	Junmei Zhong, Chuangui Gao, and Xiu Yi Kubota A.; Peterson E.I.C.; Rajendren V.; Kress- Gazit H.; Bick I. D.	Categorization of Patient Disease into ICD-10 with NLP and SVM for Chinese Electronic Health Record Analysis JESSIE: Synthesizing social robot behaviors for personalized neurorababilitation and beyond	paper Conference	excluded	not directly related to OS keyword mismatch:	2018 2020	https://doi.org/10.1145/3268866.3268877 https://www.scopus.com/inward/record.uri?eid=2 s2.0-85082003605&doi=10.1145%2f3319502.
Scopus	06.12.2024	Torfi A.; Fox E.A.	CorGAN: Correlation-capturing convolutional generative adversarial networks for generating synthetic healthcare	Conference	excluded	to OS keyword mismatch: not directly related	2020	3374836&partnertD=40&md5=acc81096aba37f/ 4aa5a6610270d372 https://www.scopus.com/inward/record.uri?eid=2 s2.0- 85102409855&partnertD=40&pvi5=fdack9ac500
Scopus ACM Digital	05.12.2024	R. A. Jeewantha, Malka N. Halgamuge, Azeem	records Classification Performance Analysis in Medical Science: Using	Conference paper	excluded	to OS keyword mismatch: not directly related		d1c049a58e6c9647dfdc
Library ACM Digital	08.11.2024	Mohammad, and Gullu Ekici	Kidney Disease Data Is Yelp Actually Cleaning Up the Restaurant Industry? A Re-	Conference paper	excluded	to OS keyword mismatch: not directly related	2017	https://doi.org/10.1145/3152723.3152724
Library	11.11.2024	Kristen M. Altenburger and Daniel E. Ho	Analysis on the Relative Usefulness of Consumer Reviews		excluded	to OS keyword mismatch:	2019	https://doi.org/10.1145/3308558.3313683
Springer	06.12.2024	SabithaTanupriya ChoudhurySaurabh Mukherjee	nearest neighbor and Naive Bayes classification techniques	Article	excluded	to OS	2016	http://link.springer.com/article/10.1007/s40012- 016-0100-5

Search venue	Query date	Author	Title	Type	Status	Reason	Year	Article URL
						keyword mismatch:		
Springer	08.12.2024	Ritika ChadhaShubhankar Mayank	Prediction of heart disease using data mining techniques	Article	excluded	to OS	2016	http://link.springer.com/article/10.1007/s40012- 016-0121-0
		Martin FrankaMarcal Cladbach7iad SchiliElarana	See DS Desearch on Seelable Privacy preserving Deserved			keyword mismatch:		
Springer	08.12.2024	RohdeErhard Rahm	Linkage	Article	excluded	to OS	2019	http://link.springer.com/article/10.1007/s13222- 019-00305-y
			Human-Curated Validation of Machine Learning Algorithms for			keyword mismatch:		
Springer	08.12.2024	Magnus Boman	Human-Curated validation of Machine Learning Algorithms for Health Data	Article	excluded	to OS	2023	http://link.springer.com/article/10.1007/s44206- 023-00076-w
						keyword mismatch:		
Springer	07.12.2024	Hui GuoOzgur KataliAnne-Liz JeukengLaure WilliamsMunindar P. Singh	Corba: crowdsourcing to obtain requirements from regulations and breaches	Article	excluded	to OS	2020	http://link.springer.com/article/10.1007/s10664- 019-09753-2
			To be an active law unless has shell as to block the second stars have			keyword mismatch:		
Springer	08.12.2024	Jason KingJon StallingsMaria RiazLaurie Williams	events – a controlled experiment	Article	excluded	to OS	2017	http://link.springer.com/article/10.1007/s10664- 016-9449-1
		Assistables Usebabase Asso Uses Otto				keyword mismatch:		
Springer	08.12.2024	OsenEgil Tennfjord Mikalsen	Predictive digital twin for offshore wind farms	Article	excluded	to OS	2023	http://link.springer.com/article/10.1186/s42162- 023-00257-4
						keyword mismatch:		
Springer	08.12.2024	Koen Bruynseels	COVID-19: the role of data positionality	Article	excluded	to OS	2021	http://link.springer.com/article/10.1007/s10676- 020-09565-9
						keyword mismatch:		
Springer	07.12.2024	Michał LupaMonika Chuchrowojciech SarlejKatarzyna Adamek	Emergency ambulance speed characteristics: a case study of Lesser Poland voivodeship, southern Poland	Article	excluded	to OS	2021	http://link.springer.com/article/10.1007/s10707- 021-00447-w
						keyword mismatch:		
Springer	08.12.2024	Gang Luo	building with big clinical data	Article	excluded	to OS	2016	http://link.springer.com/article/10.1186/s13755- 016-0018-1
-10						keyword mismatch:		
Springer	08.12.2024	Xuegiang ZengGang Luo	Progressive sampling-based Bayesian optimization for efficient and automatic machine learning model selection	Article	excluded	not directly related to OS	2017	http://link.springer.com/article/10.1007/s13755- 017-0023-z
						keyword mismatch:		
Springer	08.12.2024	Jing ZhangJianhua LiYi ZhuYu FuLixia Chen	Thyroidkeeper: a healthcare management system for patients with thyroid diseases	Article	excluded	not directly related to OS	2023	http://link.springer.com/article/10.1007/s13755- 023-00251-w
opinigoi	CONTRIBUTION OF	DL. Popa; ML. Mocanu; RT. Popa; LF.	Augmented and mixed reality for collaborative cardiac planning		011010000	keyword mismatch:	2020	10.1109/ICSTCC50638.2020.9259677
IEEE Xplore	30.11.2024	Barbulescu; L. N. Barbulescu; C. Clobirca	using 3D and 4D medical imaging data	Conference paper	excluded	not directly related to OS		
ILLE Aploto	OU. THEORY	T. N. Nguyen; L. M. Nguyen; T. Pham; M. Dinh; U.	Big Data for Healthcare: Using Entity-Attribute-Value (EAV)	paper.	UNUIDUUU	keyword mismatch:	2023	10.1109/ICSSE58758.2023.10227230
IFFF Xplore	05.12.2024	S. Khwakhali; Q. Tran	Model to Build a National Platform for Disability Management	Conference	excluded	not directly related to OS		
ILLE Aplois	00.12.2024	A. Kotiyal; A. Gupta; K. Agarwal; V. Kansal; G.	Design and Development of Effective IoT-Enabled Multi Health	paper	excluded	keyword mismatch:	2022	10.1109/CISCT55310.2022.10046471
IFFF Xnlore	05 12 2024	Dhasmana	Factor Monitoring System	Conference	evoluded	not directly related to OS		
ILLE Aploie	00.12.2024	S. Pandey; S. Mishra; R. K. Jain	IoT Interface Device for Sensing Arsenic in Contaminated	paper	excluded	keyword mismatch:	2023	10.1109/REEDCON57544.2023.10151287
IEEE Yolore	05 12 2024		Water	Conference	excluded	not directly related		
ILLE Aplois	03.12.2024	L. Kamulegeya; C. Chikwata; W. Lubega; D.	Lessons Learnt from Initial Deployments of Rocket Health	papor	excluded	keyword mismatch:	2021	
IEEE Valass	05 12 2024	Musinguzi; J. M. Bwanika	Telemedicine Service to Deliver Last Mile Medical Services in Uganda	Conference	oveluded	not directly related		
ILLL Aploie	03.12.2024	G. Li; Y. Liu; H. Cai; H. Zhao	Research on Mobile Medical Data Acquisition and Application	paper	excluded	keyword mismatch:	2018	10.1109/ICRIS.2018.00101
IEEE Valora	05 12 2024		Based on Visualization	Conference	ovoludod	not directly related		
IEEE APIOIO	05.12.2024	S. I. Khan; A. S. M. L. Hoque	Similarity analysis of patients' data: Bangladesh perspective	paper	excluded	keyword mismatch:	2016	10.1109/MEDITEC.2016.7835390
	05 40 0004			Conference		not directly related		
IEEE Aplore	05.12.2024	F. Yang; J. Li	TestIME: A Tool for Testing Chinese Input Method Engines in	paper	excluded	keyword mismatch:	2019	10.1109/ICHI.2019.8904549
IEEE Valara	05 10 0004		Electronic Medical Record Entry Task	Conference	ovoludod	not directly related		
IEEE Aplote	05.12.2024	A. Yovera-Loayza; R. Fernandez-Nevado; P.	Architectures for Blockchain in the Management of Medical	paper	excinded	keyword mismatch:	2019	10.1109/INTERCON.2019.8853614
	20.44.2024	Shiguihara-Juárez	Records: A Comparison	Conference	a secolar da al	not directly related		
IEEE Aplore	30.11.2024	J. M. Johnson; T. M. Khoshgoftaar	Cost-Sensitive Ensemble Learning for Highly Imbalanced	paper	excluded	keyword mismatch:	2022	10.1109/ICMLA55696.2022.00225
	05 10 0004		Classification	Conference	avaludad.	not directly related		
IEEE Apiore	05.12.2024	Z. Yanan	Design and Implementation of Smart College Psychological	paper	excluded	keyword mismatch:	2021	10.1109/ICITBS53129.2021.00028
IEEE Valasa	05 10 0004		Service System for Community Based on C/S Architecture	Conference	ovoludod	not directly related		
IEEE Aploie	05.12.2024	S. M; S. H. A; K. A. Shastry; R. R. D; H. V; L. K	Interoperability in Blockchain based Healthcare	paper	excluded	keyword mismatch:	2022	10.1109/MysuruCon55714.2022.9972496
	05 40 0004			Conference		not directly related		
IEEE Aplote	05.12.2024	S. Narain Gupta; R. Bhardwaj; R. Dev; A.	IoT Based Real Time Health Score Indicator	рары	excluded	keyword mismatch:	2023	10.1109/AISC56616.2023.10085632
IEEE Valara	05 40 0004	Kushwaha; G. Gupta; S. Kumari; U. Prajapati		Conference	avaludad	not directly related		
IEEE Apiore	05.12.2024	M. A. Makarem; M. A. Razaz	Role of Big Data Analytics in the Internet of Things: A	paper	excluded	keyword mismatch:	2023	10.1109/eSmarTA59349.2023.10293480
	05 40 0004		Comprehensive Survey	Conference		not directly related		
IEEE Xpiore	05.12.2024	G. Donkal; G. K. Verma	Securing Big Data Ecosystem with NSGA-II and Gradient	paper	excluded	keyword mismatch:	2018	10.1109/ICCONS.2018.8663120
			Boosted Trees Based NIDS Using Spark	Conference		not directly related		
ELE Xplore	05.12.2024	P. Kadam; P. Padalkar; A. Mohite; S. Mirajgave: S.	An Autonomous Assistance Robot for Multi-Purpose Medical	paper	excluded	keyword mismatch:	2023	10.1109/ICCECE51049.2023.10085585
IEEE V-	20 44 000 -	Gudadhe	Applications Using ROS	Conference	and it	not directly related		
ILLE Aplore	30.11.2024	P. Jain; A. Agarwal; R. Behara; C. Beachle	Developing HPCC Based COPD Readmission Risk Analysis	haher	excinded	keyword mismatch:	2018	10.1109/CSCI46756.2018.00209
IEEE V-	05 40 000 -		Platform	Conference	over the state	not directly related		
IEEE Xplore	05.12.2024	L. E. Mendoza; L. Rivas; C. Ganvini	Factors for the development of mHealth apps : Process	paper	excluded	keyword mismatch:	2020	10.23919/CISTI49556.2020.9140914
			perspective	Conference		not directly related		
IEEE Xplore	05.12.2024	M. K. Nalini; D. K; E. D; R. Gnanavel; D. Vinod	Implementation of Indexing Techniques to Prevent Data	paper	excluded	to US keyword mismatch:	2022	10.1109/ACCAI53970.2022.9752554
			Leakage and Duplication in Internet	Conference		not directly related		
IEEE Xplore	05.12.2024	T. Kumari; R. Kumari; N. Devi; B. Sharma	Personalized Healthcare Monitoring System	paper	excluded	to OS keyword mismatch:	2021	10.1109/ICOEI51242.2021.9453056
				Conference		not directly related		
IEEE Xplore	05.12.2024	H. Tobita: M. Matsumoto: Y. Maida: Y. Eguchi: K.	SinkNet: Interactive Sink to Detect Living Habits for Healthcare	paper	excluded	to OS keyword mismatch:	2018	10.1109/AIVR.2018.00020
		Terada	and Quality of Life Using Private Networks	Conference		not directly related		
IEEE Xplore	05.12.2024	A. Vii: S. Saini: R. Bathla	Big Data in Healthcare: Technologies, Need, Advantages, and	paper	excluded	to OS keyword mismatch:	2020	10.1109/ICRITO48877.2020.9197775
			Disadvantages	Conference		not directly related	2020	
IEEE Xplore	05.12.2024	J. Baranda: J. Manques-Bafalluv	End-to-End Network Service Orchestration in Heterogeneous	paper	excluded	to OS keyword mismatch:	2022	10.1109/NOMS54207.2022.9789813
			Domains for Next-Generation Mobile Networks	Conference		not directly related	2.722	
IEEE Xplore	05.12.2024	N & Neumann: E & C, de Oliveira: L C	Evoloration of Al-based innovation opportunities within the	paper	excluded	to OS	2010	10.1109/SHIRCON48091 2019 9024850
		Machicao; O. C. Velásquez	"Pastoral da Criança" maternal-child healthcare model	Conference		not directly related	2019	
IEEE Xplore	05.12.2024	A H Albarbi	Health informatics e-learning object repository Hil OP	paper	excluded	to OS	2017	10.1109/ICIHT 2017 7899138
		A. H. AHRIW	real mornaus elearning object repository HLOK	Conference		not directly related	2017	
IEEE Xplore	05.12.2024	C. Vuonalanati: A. Ilanakusti: S. Kadasi: P.	Stratification of albeit Mathematical Optimization and Artificial	paper	excluded	to OS	2020	10.1109/BigData50022.2020.0378431
		Vuppalapati; J. Vuppalapati; S. Kedari	Intelligent (AI) Driven, High-Risk Elderly Outpatients for priority	0(-		keyword mismatch:	2020	
IEEE Xplore	05.12.2024		nouse call visits - a framework to transform healthcare services from reactive to preventive	paper	excluded	to OS		
4		F. A. A. Nuaimi; J. Zeddoug; A. N. Belkacem	Real-time Control of UGV Robot in Gazebo Simulator using	Oracla		keyword mismatch:	2022	10.1109/BIBM55620.2022.9995623
IEEE Xplore	05.12.2024		r soo-based brain-Computer intellace	paper	excluded	to OS		

venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
		A. Reyes–Ascanio; A. Colmenar–Santos; C. Pérez–Molina; E. Rosales–Asensio	Application of Microgrids to Improve Resilience with a Focus on Healthcare	Conference		keyword mismatch: not directly related	2021	10.1109/ACEEE51855.2021.9575058
IEEE Xplore	30.11.2024			paper	excluded	to OS		
		S. Stephens; J. D. Applen	Rhetorical dimensions of social network analysis visualization for public health	Conference		keyword mismatch: not directly related	2016	10.1109/IPCC.2016.7740480
IEEE Xplore	05.12.2024			paper	excluded	to OS		10 1100000 1 1 00 0010 0010 7001700
		D. G. Korzun; I. V. Galov; A. A. Lomov	Smart space deployment in wireless and mobile settings of the Internet of Things	Conference		keyword mismatch: not directly related	2016	10.1109/IDAACS-SWS.2016.7805793
IEEE Xplore	05.12.2024	O Mariaci, E. Zambaralli, A. Zamidi, Oscarda	Disk Desiliation on a One-instance DOO Assibilitations Descentions	paper	excluded	to OS	0040	10 1100/CBMC 2010 00020
		S. Manani; F. Zambonelli; A. Tenyi; T. Cano; J. Roca	Interoperability and Collaboration	Conference		not directly related	2019	10.1109/CBW0.2019.00009
IEEE Xplore	05.12.2024	T Tojogući C. Muroli	Semantia inference mothed using antelesion	paper	excluded	to OS	2016	10 1100/05525 2016 7880858
		T. Tejaświ, G. Muran	Semanuc interence metriod using ontologies	Conference		not directly related	2010	10.1103/02010.2010.703/000
IEEE Xplore	05.12.2024	O Rep Dali: V Sellami: S. Zhukay: H yen	Ultraconsitive and low cost incole for goit applying using	paper	excluded	to OS	2022	10 1109/SENSORS52175 2022 9967198
		Seggern; N. Schäfer; B. Latsch; G. M. Sessler; P.	piezoelectrets	Conference		not directly related	2022	
IEEE Xplore	05.12.2024	Beckerle; M. Kupnik I. P. Dharma Yudha: A. Y. Ridwan: L.	Developing Sustainable Solid Medical Waste Management	paper	excluded	to OS keyword mismatch:	2022	10.1109/IColCT55009.2022.9914904
		Septianingrum	Based on Enterprise Resource Planning in General Hospital	Conference		not directly related		
IEEE Xplore	05.12.2024	R. Walid; K. P. Joshi; S. G. Choi	Secure Cloud EHR with Semantic Access Control, Searchable	paper	excluded	to US keyword mismatch:	2021	10.1109/ICDH52753.2021.00016
IFFF V-I	05 40 0004		Encryption and Attribute Revocation	Conference	and a dead	not directly related		
IEEE Xpiore	05.12.2024	S. Kale; S. Mane; P. Patil	Wearable biomedical parameter monitoring system: A review	paper	excluded	keyword mismatch:	2017	10.1109/ICECA.2017.8203611
IEEE Voloro	05 12 2024			Conference	oveluded	not directly related		
ILLL Aploie	05.12.2024	S. Del Din; A. Hickey; S. Woodman; H. Hiden; R.	Accelerometer-based gait assessment: Pragmatic deployment	ραροι	excluded	keyword mismatch:	2016	10.1109/SSP.2016.7551794
IFFF Xplore	30 11 2024	Morris; P. Watson; K. Nazarpour; M. Catt; L. Rochester; A. Godfrey	on an international scale	Conference paper	excluded	not directly related to OS		
		S. I. Khan; A. S. M. L. Hoque	An analysis of the problems for Health Data integration in			keyword mismatch:	2016	10.1109/ICISET.2016.7856517
IEEE Xplore	30.11.2024		Bangladesh	Conference paper	excluded	to OS		
		J. R. Cordeiro; O. Postolache	Big Data Storage for a Health Predictive System	0		keyword mismatch:	2018	10.1109/ISSI.2018.8538285
IEEE Xplore	05.12.2024			paper	excluded	to OS		
		T. Narmadha; M. Kalaiarasi; M. Meenakshi	Lightweight secure ECG transmission in wireless body area networks — PRESENT cipher based implementation	Conference		keyword mismatch:	2017	10.1109/ICCSP.2017.8286538
IEEE Xplore	05.12.2024		networks - Freezerr eigher based imperioritation	paper	excluded	to OS		
		P. Ndayizigamiye; L. Hangulu; O. Akintola	A design of a mobile health intervention to enhance home- carers' disposal of medical waste in South Africa	Conference		keyword mismatch: not directly related	2017	10.1109/GHTC.2017.8239241
IEEE Xplore	30.11.2024			paper	excluded	to OS		
		Z. Liang; M. A. Chapa-Martell	Combining Resampling and Machine Learning to Improve Sleep-Wake Detection of Fitbit Wristbands	Conference		keyword mismatch: not directly related	2019	10.1109/ICHI.2019.8904753
IEEE Xplore	05.12.2024	N. Defendé D. Kaslé	Int for COVID 40 Indees Control Dreventing Court Detection	paper	excluded	to OS	0004	10 110084101 02704 2021 060000
		N. Petrovic; D. Kocic	Air Quality Control and Contact Tracing	Conference		keyword mismatch: not directly related	2021	10.110s/miEL52/94.2021.9569099
IEEE Xplore	05.12.2024	S Hariai S K Khatri	An Intelligent Clinical Decision Support System Read on	paper	excluded	to OS	2010	10 1109/AICAI 2019 8701237
		S. Haja, S. K. Klatt	Artificial Neural Network for Early Diagnosis of Cardiovascular	Conference		not directly related	2013	
IEEE Xplore	30.11.2024	A. Rabioui	Diseases in Rural Areas Manaoing Patient Medical Record using Blockchain in	paper	excluded	to OS keyword mismatch:	2020	10.1109/Morgeo49228.2020.9121901
			Developing Countries: Challenges and Security Issues	Conference		not directly related		
IEEE Xplore	05.12.2024	S. S. Sadavarte; E. Bodanese	Pregnancy Companion Chatbot Using Alexa and Amazon Web	paper	excluded	to OS keyword mismatch:	2019	10.1109/PuneCon46936.2019.9105762
IEEE Valara	05 10 0004		Services	Conference	evoluded	not directly related		
IEEE Apiole	05.12.2024	S. Marathe; D. Zeeshan; T. Thomas; S. Vidhya	A Wireless Patient Monitoring System using Integrated ECG	paper	excluded	keyword mismatch:	2019	10.1109/VITECoN.2019.8899541
IEEE Xplore	30.11.2024		Sensor	Conference paper	excluded	not directly related to OS		
		K. N. Baluprithviraj; R. Harini; M. M. Janarthanan;	Design and Development of Smart Saline Level Indicator for	0		keyword mismatch:	2022	10.1109/ICAIS53314.2022.9742794
IEEE Xplore	05.12.2024	0. 000000000	Treatmoare using for	paper	excluded	to OS		
		M. Mazorchuck; V. Dobriak; D. Chumachenko	Web-Application Development for Tasks of Prediction in Medical Domain	Conference		keyword mismatch: not directly related	2018	10.1109/STC-CSIT.2018.8526684
IEEE Xplore	05.12.2024			paper	excluded	to OS		
		R. Salla; R. Pahle; A. Amresh	A complex systems framework approach towards multidisciplinary tumor boards	Conference		keyword mismatch: not directly related	2016	10.1109/HIC.2016.//9/724
IEEE Xplore	30.11.2024	K.G. Dinesh: K. Arumugarai: K.D. Santhosh: V	Prediction of Cardiovascular Disease Lleing Machine Learning	paper	excluded	to OS	2018	10.1109/CCTCT.2018.8550857
		Mareeswari	Algorithms	Conference		not directly related	2010	
IEEE Xplore	05.12.2024	Z. H. Shaik: A. Hussain Shaik: J. H. Shaik	Electronic Health Records (EHR) Management in Hospitals	paper	excluded	to OS keyword mismatch:	2023	10.1109/ICECCME57830.2023.10252523
			using Blockchain Technology	Conference		not directly related		
IEEE Xplore	05.12.2024	L. Lévêque; W. Zhang; C. Cavaro-Ménard; P. Le	Study of Video Quality Assessment for Telesurgery	paper	excluded	to OS keyword mismatch:	2017	10.1109/ACCESS.2017.2704285
IEEE Valara	00.07.0004	Callet; H. Liu		Antinio	evel-ideal	not directly related		
IEEE Apiore	02.07.2024	Denis CavallucciPavel LivotovStelian Brad	Towards AI-Aided Invention and Innovation	Article	excluded	keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
Springer	08.12 2024			Conference paper	excluded	not directly related to OS		901-12032-0
- p				1 81		keyword mismatch:		
Springer	07.12.2024	Carsten Dochow	Editorial	Article	excluded	not directly related to OS	2022	http://link.springer.com/article/10.1365/s43439- 022-00066-7
	/	Senuri De SilvaSanuwani DayarathnaGangani	Computational Decision Support Support for ADLD			keyword mismatch:		
Springer	07.12.2024	JayarathnaAnne M. P. Michalek	Identification	Article	excluded	to OS	2021	http://link.springer.com/article/10.1007/s11633- 020-1252-1
		Johann M. KrausLudwig LausserPeter KuhnFranz JobstMichaela BockCarolin HalankaMichael	Big data and precision medicine: challenges and strategies			keyword mismatch: not directly related		
Springer	07.12.2024	HummelPeter HeuschmannHans A. Kestler	with healthcare data	Article	excluded	to OS	2018	http://link.springer.com/article/10.1007/s41060- 018-0095-0
		Ricardo Cruz-CorreiaDuarte FerreiraGustavo				keyword mismatch: not directly related		
Springer	08.12.2024	BacelarPedro MarquesPriscila Maranhão	Personalised medicine challenges: quality of data	Article	excluded	to OS	2018	018-0127-9
		Sourabh ShastriParamjit KourSachin KumarKuljeet	A nested stacking ensemble model for predicting districts with			not directly related		http://ink.epsinger.com/article/10.1007/e41870-
Springer	06.12.2024	SinghAnand SharmaVibhakar Mansotra	high and low maternal mortality ratio (MMR) in India	Article	excluded	to OS	2021	020-00560-3
		Odu Nkiruka BridgetRajesh PrasadClement	Drug resistant tuberculosis classification using logistic			not directly related		http://ink.springer.com/article/10.1007/s41870-
Springer	07.12.2024	OnimeAdamu Abubakar Ali	regression	Article	excluded	to OS keyword mismatch:	2021	020-00592-9
			A market and the second se			not directly related		http://link.springer.com/article/10.1007/s13735-
Springer	06.12.2024	o. ouganyadeviv. SeethalakshmiK. Balasamy	A review on deep learning in medical image analysis	Article	excluded	keyword mismatch:	2022	021-00218-1
Paring -	00 10 0004	Shahidul Jelam KhanAhu Paued Md Latiful Harris	SICE: an improved missing data imputation technique	Article	ovolutert	not directly related	2020	http://link.springer.com/article/10.1186/s40537-
opringer	08.12.2024	Sharingal Islam Khariyou Sayed Md Latitul Hoque	orde, an improved missing data imputation technique	Articie	excluded	keyword mismatch:	2020	<u>uzu-uu313-w</u>
Springer	08.12 2024	Saurabh DevSriniyas SampalliQiang Ye	MDA: message digest-based authentication for mobile cloud computing	Article	excluded	not directly related to OS	2016	http://link.springer.com/article/10.1186/s13677- 016-0068-6
obuni8qi	5011E-5024	open note campanetary re			Shoraded	keyword mismatch:	2010	<u>2122202020</u>
Springer	07.12.2024	Qian HeEmmanuel O. Agu	Context-Aware Probabilistic Models for Predicting Future Sedentary Behaviors of Smartphone Users	Article	excluded	not directly related to OS	2022	http://link.springer.com/article/10.1007/s41666- 021-00107-6
			Al in Medicine, Could-10 and Environmentational Ones A			keyword mismatch:		
Springer	07.12.2024	Daniel Sonntag	Agreement	Article	excluded	to OS	2020	http://link.springer.com/article/10.1007/s13218- 020-00661-y

Search	Query date	Author	Title	Type	Status	Reason	Year	Article URI
Vende	dutry dute	Prof. Dr. Joaquim FilipeMichał ŚmiałekAlexander BrodskySlimane Hammoudi	Enterprise Information Systems	Conference paper	otatas	keyword mismatch: not directly related	2021	http://ink.springer.com/book/10.1007/978-3- 030-75418-1
Springer	08.12.2024	Xiaofeng WangAntonio MartiniDr. Anh Nguyen- DucDr. Viktoria Stray	Software Business	Conference paper	excluded	keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-91983-2
Springer	08.12.2024	Robert A. SottilareJessica Schwarz	Adaptive Instructional Systems	Conference	excluded	to OS keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-05887-5
Springer	06.12.2024				excluded	to OS		
Springer	07 12 2024	Siva Teja KakiletiMaria GabraniGeetha ManjunathMichal Rosen-ZviNathaniel BramanRobert G. SchwartzAlejandro F. FrangiPau- Choo ChungChristopher WeightVekataraman Jacadish	Artificial Intelligence over Infrared Images for Medical Applications and Medical Image Assisted Biomarker Discovery	Conference paper	evoluded	keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-19660-7
Springer	07.12.2024	Max BramerFrederic Stahl	Artificial Intelligence XXXIX	Conference paper	excluded	keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-21441-7
Springer	07.12.2024	Věra KůrkováProf. Yannis ManolopoulosBarbara HammerLazaros IliadisIlias Maglogiannis	Artificial Neural Networks and Machine Learning – ICANN 2018	Conference paper	excluded	keyword mismatch: not directly related	2018	http://link.springer.com/book/10.1007/978-3- 030-01424-7
Springer	07.12.2024	Dylan D. SchmorrowCali M. Fidopiastis	Augmented Cognition	Conference paper	excluded	to OS keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-35017-7
Springer	07.12.2024	Prof. Lucio Tommaso De PaolisPasquale ArpaiaPatrick Bourdot	Augmented Reality, Virtual Reality, and Computer Graphics	Conference paper	excluded	to OS keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-87595-4
Springer	07.12.2024	José Manuel Ferrández Vicente José Ramón	Bio-inspired Systems and Applications: from Robotics to	Conference	excluded	to OS keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
Springer	07.12.2024	Álvarez-SánchezFélix de la Paz LópezHojjat Adeli	Ambient Intelligence	paper	excluded	not directly related to OS	0001	031-06527-9
Springer	07.12.2024	HerreraProf. Héctor Pomares	Bioengineering and Biomedical Signal and Image Processing	paper	excluded	not directly related to OS	2021	030-88163-4
Springer	07.12.2024	Alexander HuntVasiliki VouloutsiKenneth MosesRoger QuinnAnna MuraTony PrescottPaul F. M. J. Verschure	Biomimetic and Biohybrid Systems	Conference paper	excluded	keyword mismatch: not directly related to OS	2022	http://link.springer.com/book/10.1007/978-3- 031-20470-8
Springer	07 12 2024	Fabian MederAlexander HuntLaura MargheriAnna MuraBarbara Mazzolai	Biomimetic and Biohybrid Systems	Conference paper	oveluded	keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-39504-8
Springer	07.12.2024	Richard Chen LiSimon K. S. CheungPeter H. F.	Blended Learning: Engaging Students in the New Normal Era	Conference	excluded	keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-08939-8
Springer	07.12.2024	Dr. Mufti MahmudProf. M Shamim KaiserStefano	Brain Informatics	Conference	excluded	to OS keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3- 030-86993-9
Springer	07.12.2024	Feng LiuYu ZhangHongzhi KuaiEmily P.	Brain Informatics	paper Conference	excluded	to OS keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
Springer	07.12.2024	StephenHongjun Wang Prof. Mufti MahmudJing HeStefano	Brain Informatics	paper Conference	excluded	not directly related to OS keyword mismatch:	2022	031-43075-6 http://ink.springer.com/book/10.1007/978-3-
Springer	07.12.2024	VassanelliAndré van ZundertProf. Dr. Ning Zhong	Olisical Image Record Procedures Estrates of Allis Medical	paper	excluded	not directly related to OS	2022	031-15037-1
Carianaa	07.40.0004	Steran Wesargestner Puyol AntonJonn S. H. Baxter/Marius ErdtKlaus Drechsler/Cristina Oyarzun LauraMoti FreimanYufei ChenIslem RekikRoy EaglesonAasa FeragenAndrew P. KingVeronika ChepIyginaMelani Ganz-BenjaminsenEnzo EasentABea ClandenDenis Marusefilm Beharana	Clinical imaging-based Procedures, Partness of Al In Medical Imaging, and Ethical and Philosophical Issues in Medical Imaging	paper		keyword mismatch: not directly related	2023	1002/2018-8920100000-10-1007/01/0-2- 031-45249-9
Springer	07.12.2024	Klaus MiesenbergerGeorgios Kouroupetroglou	Computers Helping People with Special Needs	Conference paper	excluded	keyword mismatch: not directly related	2018	http://link.springer.com/book/10.1007/978-3- 319-94277-3
Springer	07.12.2024	Alexander MaedcheJan vom BrockeAlan Hevner	Designing the Digital Transformation	Conference paper	excluded	to OS keyword mismatch: not directly related	2017	http://link.springer.com/book/10.1007/978-3- 319-59144-5
Springer	07.12.2024	Leyla BilgeLorenzo CavallaroGiancarlo PellegrinoNuno Neves	Detection of Intrusions and Malware, and Vulnerability Assessment	Conference paper	excluded	to OS keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-80825-9
Springer	07.12.2024	Dr. Vincent G. Duffy	Digital Human Modeling and Applications in Health, Safety, Economics and Bisk Management AL Product and Service	Conference	excluded	to OS keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3- 030-77820-0
Springer	07.12.2024	Assoc. Prof. Vincent G. Duffy	Digital Human Modeling and Applications in Health, Safety,	Conference	excluded	to OS keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3- 031-05990-5
Springer	07.12.2024	Vincent G. Duffy	Ergonomics and Hisk Management. Anthropometry, Human Behavior, and Communication Digital Human Modeling and Applications in Health, Safety,	Conference	excluded	to OS keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
Springer	07.12.2024	Dr. Vincent G. Duffy	Ergonomics and Risk Management. Health, Operations Management, and Design Digital Human Modeling and Applications in Health, Safety	paper	excluded	not directly related to OS keyword mismatch:	2021	031-06018-2 http://link.springer.com/book/10.1007/978-3-
Springer	07.12.2024	Dr. Whiten G. Dany	Ergonomics and Risk Management. Human Body, Motion and Behavior	paper	excluded	not directly related to OS	2021	030-77817-0
		Shadi AlbarqouniSpyridon BakasSophia BanoM. Jorge CardosoBishesh KhanalBennett LandmanXiaoxiao LiChen QinIslem RekikNicola	Distributed, Collaborative, and Federated Learning, and Affordable AI and Healthcare for Resource Diverse Global Health	Conference paper		keyword mismatch: not directly related	2022	http://ink.springer.com/book/10.1007/978-3- 031-18523-6
springer	07.12.2024	Lucio Tommaso De PaolisPasquale ArpaiaMarco Sacco	Extended Reality	Conference paper	excluded	keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-15553-6
Springer	08.12.2024	Prof. Dr. Marieke HuismanCorina PäsäreanuNaijun Zhan	Formal Methods	Conference paper	excluded	to OS keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-90870-6
Springer	08.12.2024	Alessandro CimattiLaura Titolo	Formal Methods for Industrial Critical Systems	Conference	excluded	to OS keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-43681-9
Springer	08.12.2024	Kirstin PetersAssoc. Prof. Tim A. C. Willemse	Formal Techniques for Distributed Objects, Components, and	Conference	excluded	to OS keyword mismatch:	2021	http://link.springer.com/book/10.1007/978-3- 030-78089-0
Springer	08.12.2024	Guy-Vincent JourdanLaurent MounierCarlisle	Foundations and Practice of Security	paper Conference	excluded	to OS keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	AdamsFlorence SèdesJoaquin Garcia-Alfaro Dr. Daniel B. EnnisDr. Luigi E. PerottiVicky Y. Wang	Functional Imaging and Modeling of the Heart	paper Conference	excluded	not directly related to OS keyword mismatch:	2021	us1-su122-3 http://ink.springer.com/book/10.1007/978-3-
Springer	08.12.2024	Oliviar Barnard Patrick Classes No	Functional Imaging and Madalina of the Unert	paper	excluded	not directly related to OS	2000	030-78710-3
Springer	08.12.2024	Olivier BernaroPatrick ClarysseNicolas DuchateauJacques OhayonMagalie Viallon	runctional imaging and Modeling of the Heart	conterence paper	excluded	not directly related to OS	2023	11.35302-4
Springer	08.12.2024	Agma TrainaHua WangYong ZhangSiuly SiulyRui ZhouLu Chen	Health Information Science	Conference paper	excluded	keyword mismatch: not directly related to OS	2022	http://link.springer.com/book/10.1007/978-3- 031-20627-6
Springer	08 12 2024	Qin GaoJia Zhou	Human Aspects of IT for the Aged Population	Conference paper	aveluded	keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-34917-1
opringer	00.12.2024	Prof. Hugo Sanjurjo GonzálezIker Pastor LópezPablo García BringasHéctor QuintiánEmilio	Hybrid Artificial Intelligent Systems	Conference paper	excluded	keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-86271-8
Springer	08.12.2024	Gorchado Isaac SserwangaAnne GouldingHeather Moulaison-SandyJia Tina DuAntónio Lucas	Information for a Better World: Normality, Virtuality, Physicality, Inclusivity	Conference paper	excluded	to OS keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-28032-0
Springer	08.12.2024	SoaresViviane HessamiRebecca D. Frank			excluded	to OS		

venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
Carianaa	00.40.0004	Malte Smits	Information for a Better World: Shaping the Global Future	Conference paper		keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 030-96957-8
Springer	08.12.2024	Prof. Dr. Ute SchmidFranziska KlüglDiedrich Wolter	KI 2020: Advances in Artificial Intelligence	Conference paper	excluded	keyword mismatch: not directly related	2020	http://link.springer.com/book/10.1007/978-3- 030-58285-2
Springer	08.12.2024	Dion H. GohShu-Jiun ChenSuppawong Tuarob	Leveraging Generative Intelligence in Digital Libraries: Towards Human-Machine Collaboration	Conference paper	excluded	to OS keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-981- 99-8085-7
Springer	08.12.2024	Nandinee HaqPatricia JohnsonAndreas MaierTobias WürfJaeiun Yoo	Machine Learning for Medical Image Reconstruction	Conference	excluded	to OS keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-88552-6
Springer	08.12.2024	Chunfeng LianXiaohuan Caolslem RekikDr. Xuanang XuDr. Pingkun Yan	Machine Learning in Medical Imaging	Conference	excluded	to OS keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-87589-3
Springer	08.12.2024	Alejandro F. FrangiJulia A. SchnabelChristos DavatzikosCarlos Alberola-LópezGabor Fichtinger	Medical Image Computing and Computer Assisted	Conference	excluded	to OS keyword mismatch: not directly related	2018	http://link.springer.com/book/10.1007/978-3- 030-00934-2
Springer	08.12.2024	Dinggang ShenProf. Tianming LiuDr. Terry M.	Medical Image Computing and Computer Assisted	Conference	excluded	to OS keyword mismatch:	2019	http://link.springer.com/book/10.1007/978-3- 030-32251-9
Springer	08.12.2024	ZhouProf. Dr. Pew-Thian YapAli Khan Prof. Anne L. MartelPurang AbolmaesumiDanail	Medical Image Computing and Computer Assisted	Conference	excluded	to OS keyword mismatch:	2020	http://ink.springer.com/book/10.1007/978-3-
Springer	08.12.2024	StoyanovDiana MateusMaria A. ZuluagaS. Kevin ZhouDaniel RacoceanuProf. Leo Joskowicz Prof. Anne L. MartelPurang AbolmaesumiDanail	Intervention – MICCAI 2020 Medical Image Computing and Computer Assisted	paper Conference	excluded	not directly related to OS keyword mismatch:	2020	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	StoyanovDiana MateusMaria A. ZuluagaS. Kevin ZhouDaniel RacoceanuProf. Leo Joskowicz Prof. Anne L. MartelPurang AbolmaesumiDanail	Intervention – MICCAI 2020 Medical Image Computing and Computer Assisted	paper Conference	excluded	not directly related to OS keyword mismatch:	2020	030-59722-1 http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	StoyanovDiana MateusMaria A. ZuluagaS. Kevin ZhouDaniel RacoceanuProf. Leo Joskowicz	Intervention – MICCAI 2020	paper	excluded	not directly related to OS	2021	030-59719-1
Springer	08.12.2024	CattinStéphane CotinNicolas PadoyProf. Stefanie SpeidelYefeng ZhengCaroline Essert	Intervention – MICCAI 2021	paper	excluded	not directly related to OS	2021	030-87240-3
Springer	08.12.2024	Linwei WangQi DouP. Thomas FletcherStefanie SpeidelShuo Li	Medical Image Computing and Computer Assisted Intervention – MICCAI 2022	Conference paper	excluded	keyword mismatch: not directly related to OS	2022	http://ink.springer.com/book/10.1007/978-3- 031-16434-7
Springer	08.12.2024	Hayit GreenspanAnant MadabhushiParvin MousaviSeptimiu SalcudeanJames DuncanTanveer Sveda-MahmoodRussell Taylor	Medical Image Computing and Computer Assisted Intervention – MICCAI 2023	Conference paper	excluded	keyword mismatch: not directly related to OS	2023	http://link.springer.com/book/10.1007/978-3- 031-43993-3
		M. Emre CelebiMd Sirajus SalekinHyunwoo KimShadi AlbarqouniCatarina BarataAllan HalpernPhilipp TschandlMarc CombaliaYuan	Medical Image Computing and Computer Assisted Intervention – MICCAI 2023 Workshops	Conference paper			2023	http://link.springer.com/book/10.1007/978-3- 031-47401-9
		LiuGhada ZamzmiJoshua LevyHuzefa RangwalaAnnika ReinkeDiya WynnBennett LandmanWon-Ki. JeongYiging ShenZhongying				keyword mismatch:		
Springer	08.12.2024	DengSpyridon BakasXiaoxiao LiChen QinNicola RiekeHolger RothDaguang Xu			excluded	not directly related to OS		
Springer	08.12.2024	Guang YangAngelica Aviles-RiveroMichael RobertsCarola-Bibiane Schönlieb	Medical Image Understanding and Analysis	Conference paper	excluded	keyword mismatch: not directly related to OS	2022	nttp://ink.springer.com/dook/10.1007/978-3- 031-12053-4
Springer	08.12.2024	Duc-Tien Dang-NguyenCathal GurrinMartha LarsonAlan F. SmeatonStevan RudinacMinh-Son DaoChristoph TrattnerPhoebe Chen	MultiMedia Modeling	Conference paper	excluded	keyword mismatch: not directly related to OS	2023	http://link.springer.com/book/10.1007/978-3- 031-27077-2
Springer	08 12 2024	Tanveer Syeda-MahmoodXiang LiDr. Anant MadabhushiHayit GreenspanQuanzheng LiRichard LeahyBin DongHongzhi Wang	Multimodal Learning for Clinical Decision Support	Conference paper	excluded	keyword mismatch: not directly related to OS	2021	http://link.springer.com/book/10.1007/978-3- 030-89847-2
Springer	08 12 2024	Xiang LiJinglei LvYuankai HuoBin DongRichard M. LeahyQuanzheng Li	Multiscale Multimodal Medical Imaging	Conference paper	excluded	keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-18814-5
Opringer	00.12.2024	Dr. Teddy MantoroMinho LeeMedia Anugerah AyuDr. Kok Wai WongDr. Achmad Nizar Hidayanto	Neural Information Processing	Conference paper	excluded	keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-92185-9
Springer	08.12.2024	Hamdi AloulouProf. Bessam AbdulrazakAntoine de Marassé-EnoufMounir Mokhtari	Participative Urban Health and Healthy Aging in the Age of Al	Conference paper	excluded	keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-09593-1
Springer	08.12.2024	Raghunath NambiarMeikel Poess	Performance Evaluation and Benchmarking	Conference paper	excluded	keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-29576-8
Springer	08.12.2024	Islem RekikEhsan AdeliSang Hyun ParkCelia CintasGhada Zamzmi	Predictive Intelligence in Medicine	Conference paper	excluded	to OS keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-46005-0
Springer	08.12.2024	Xinxing XuXiaomeng LiDwarikanath MahapatraLi ChenaCaroline PetitieanHuazhu Fu	Resource-Efficient Medical Image Analysis	Conference	excluded	to OS keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-16876-5
Springer	08.12.2024	Frank StajanoVashek MatyášBruce	Security Protocols XXVIII	Conference	excluded	to OS keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-43033-6
Springer	08.12.2024	Mads HaahrAlberto Rojas-SalazarStefan Göbel	Serious Games	Conference	excluded	to OS keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	Christian WachingerBeatriz PaniaguaShireen	Shape in Medical Imaging	paper Conference	excluded	not directly related to OS keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024	ElhabianJianning LiJan Egger Bernhard KainzAlison NobleJulia SchnabelBishesh	Simplifying Medical Ultrasound	paper Conference	excluded	not directly related to OS keyword mismatch:	2023	031-46914-5 http://ink.springer.com/book/10.1007/978-3-
Springer	08.12.2024	KhanalJohanna Paula MüllerThomas Day	Statistical Atlases and Computational Models of the Heart	paper	excluded	not directly related to OS keyword mismatch:	2022	031-44521-7 http://ink.springer.com/book/10.1007/978-3-
Springer	08.12.2024	IslaMaxime SermesantAvan SuinesianutraDr. Oscar CamaraKarim LekadirAlistair Young	Multi-Disease, Multi-View, and Multi-Center Right Ventricular Segmentation in Cardiac MRI Challenge	paper	excluded	not directly related to OS	20000	030-93722-5
Springer	08.12.2024	Secar Gamaracsurer Proyol-AntonChen QinMaxime SermesantAvan SuinesiaputraShuo WangAlistair Young	Regular and CMRxMotion Challenge Papers	paper	excluded	not directly related to OS	2022	031-23443-9
Springer	08.12.2024	Kamil EkšteinFrantišek PártlMiloslav Konopík	Text, Speech, and Dialogue	Conference paper	excluded	keyword mismatch: not directly related to OS	2021	http://iiik.springer.com/book/10.1007/978-3- 030-83527-9
Springer	08.12.2024	Savvas PapagiannidisEleftherios AlamanosSuraksha GuptaYogesh K. DwivediMatti Mäntymäkillias O. Pappas	The Role of Digital Technologies in Shaping the Post- Pandemic World	Conference paper	excluded	keyword mismatch: not directly related to OS	2022	http://link.springer.com/book/10.1007/978-3- 031-15342-6
Springer	08 12 2024	Gobinda ChowdhuryJulie McLeodProf. Val GilletPeter Willett	Transforming Digital Worlds	Conference paper	excluded	keyword mismatch: not directly related to OS	2018	http://link.springer.com/book/10.1007/978-3- 319-78105-1
opiniger	JU. 12.2U24	Carole H. SudreRoxane LicandroChristian BaumgartnerProf. Dr. Andrew MelbourneDr. Adrian DalcaDr. Jana HutterRyutaro TannoEsra Abaci TurkKoen Van LeemputJordina Torrents	Uncertainty for Safe Utilization of Machine Learning in Medical Imaging, and Perinatal Imaging, Placental and Preterm Image Analysis	Conference paper	aveluded	keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-87735-4
Springer	08.12.2024	barrenawilliam M. WellsChristopher Macgowan Gabriel ZachmannMariano Alcañiz RayaPatrick BourdotMaud MarchalJeanine StefanucciXubo	Virtual Reality and Mixed Reality	Conference paper	excluded	to US keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 031-16234-3
Springer	08.12.2024	Yang Jessie Y. C. ChenGino Fragomeni	Virtual, Augmented and Mixed Reality	Conference	excluded	to OS keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-35634-6
Springer	08.12.2024	Rasika DhavseVinay KumarSalvatore Monteleone	Emerging Technology Trends in Electronics, Communication	Conference	excluded	to OS keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-981- 19-6737-5
Springer	07.12.2024		and NetWorking	paper	excluded	not directly related to OS		

venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
		Pradeep Kumar SinghSławomir T. WierzchońJitender Kumar ChhabraSudeep Tanwar	Futuristic Trends in Networks and Computing Technologies	Conference paper		keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-981- 19-5037-7
Springer	08.12.2024	Sarika JainSven GroppeNandana	Proceedings of the International Health Informatics	Conference	excluded	to OS keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-981- 19-9090-8
Springer	08.12.2024	Velantias Emilia DelasVilau Pheekas ComusiApond	Intelligent Commuting and Naturalian	Casterence	excluded	to OS	0000	http://link.epringer.com/book/10.1007/078-091-
Springer	08 12 2024	Khandare	intelligent Computing and Networking	paper	evoluded	not directly related	2023	<u>99-0071-8</u>
opiniger	00.12.2024	Dr. Nancy L. BlackDr. W. Patrick NeumannDr. Ian Nov	Proceedings of the 21st Congress of the International Eroonomics Association (IEA 2021)	Conference	excluded	keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-74608-7
Springer	08.12.2024	Masood Ur RehmanAhmed Zoha	Body Area Networks, Smart IoT and Big Data for Intelligent	Conference	excluded	to OS keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
Springer	07.12.2024		Health Management	paper	excluded	not directly related to OS		030-95593-9
		Mohammad R. KhosraviQiang HeProf. Haipeng Dai	Cloud Computing	Conference paper		keyword mismatch: not directly related	2022	http://link.springer.com/book/10.1007/978-3- 030-99191-3
Springer	07.12.2024	Dr. Pavel GladyshevProf. Sanjay GoelJoshua	Digital Forensics and Cyber Crime	Conference	excluded	to OS keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
Springer	07.12.2024	JamesDr. George MarkowskyDaryl Johnson	Entry Assess Fachlass for Historiteus and Intelligent	paper	excluded	to OS	0004	http://link.apringer.com/book/10.1007/078-2-
Springer	08 12 2024	Dragan PerakovicDr. Lucia Knapcikova	Infrastructures	paper	ovaludad	not directly related	2021	030-78459-1
Springer	06.12.2024	Nguyen-Son VoHoai-An Tran	Industrial Networks and Intelligent Systems	Conference	excluded	keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-3- 031-47359-3
Springer	08.12.2024	Shuihua WangZheng ZhangYuan Xu	IoT and Big Data Technologies for Health Care	Conference	excluded	to OS keyword mismatch:	2022	http://link.springer.com/book/10.1007/978-3-
Springer	08.12.2024		······································	paper	excluded	not directly related to OS		030-94182-6
		Prof. Dr. Sara PaivaSérgio Ivan LopesRafik ZitouniNishu GuptaSérgio F. LopesTakuro	Science and Technologies for Smart Cities	Conference paper		keyword mismatch: not directly related	2021	http://link.springer.com/book/10.1007/978-3- 030-76063-2
Springer	08.12.2024	Yonezawa Der-Jiunn DengHan-Chieh ChaoJyh-Cheng Chen	Smart Grid and Internet of Things	Conference	excluded	to OS keyword mismatch:	2023	http://link.springer.com/book/10.1007/978-3- 031-31275-5
Springer	08.12.2024	han Misuel Dise Draf. Dr. Queenne Onincente Draf.	Smort Objects and Technologies for Social Coord	paper	excluded	to OS	0004	http://link.apringer.com/book/10.1007/078-2-
Springer	08 12 2024	Eftim ZdravevskiPetre Lameski	Smart Objects and Technologies for Social Good	paper	evoluded	not directly related	2021	030-91421-9
opiniger	00.12.2024	António CunhaNuno M. GarciaJorge Marx GómezSandra Pereira	Wireless Mobile Communication and Healthcare	Conference	excluded	keyword mismatch: not directly related	2023	http://ink.springer.com/book/10.1007/978-3- 031-32029-3
Springer	08.12.2024	Xinbo GaoAbbas JamalipourLei Guo	Wireless Mobile Communication and Healthcare	Conference	excluded	to OS keyword mismatch:	2022	http://ink.springer.com/book/10.1007/978-3-
Springer	08.12.2024			paper	excluded	not directly related to OS		031-06368-8
		Mei-Ting SuMei-Ling ChiangChia-Hsuan TsaiChi- Wei LinRong-Xuan LiuYong-Ting JuangHsin-Hao	An acupoint health care system with real-time acupoint			keyword mismatch: not directly related		http://link.springer.com/article/10.1007/s00530-
Springer	07.12.2024	Chen	localization and visualization in augmented reality	Article	excluded	to OS keyword mismatch:	2023	023-01104-y
Springer	08.12.2024	Arshia RehmanSaeeda NazImran Razzak	trends, challenges and opportunities	Article	excluded	to OS	2022	http://link.springer.com/article/10.1007/s00530- 020-00736-8
Springer	07.12.2024	Sekyoung YoumChanggyun KimSeunghyun ChoiYong-Shin Kang	Development of a methodology to predict and monitor emergency situations of the elderly based on object detection	Article	excluded	not directly related to OS	2019	http://link.springer.com/article/10.1007/s11042- 018-6660-7
		Hamed JelodarYongli WangChi YuanXia	Latent Dirichlet allocation (LDA) and topic modeling: models,			keyword mismatch: not directly related		http://link.springer.com/article/10.1007/s11042.
Springer	08.12.2024	FengXiahui JiangYanchao LiLiang Zhao	applications, a survey	Article	excluded	to OS keyword mismatch:	2019	018-6894-4
Springer	08.12.2024	Priya AroraAshutosh MishraAvleen Malhi	Machine learning Ensemble for the Parkinson's disease using protein sequences	Article	excluded	not directly related to OS	2022	http://link.springer.com/article/10.1007/s11042- 022-12960-7
Coringer	08 12 2024	Márcio Renê Brandão SoussaValter de	Modeling elderly behavioral patterns in single-person	Article	oveluded	not directly related	2021	http://link.springer.com/article/10.1007/s11042-
Springer	00.12.2024	Pablo CaballeroGuadalupe OrtizAlfonso Garcia-de-	Paving the way to collaborative context-aware mobile applications: a case study on preventing worsening of allergy	Article	excluded	keyword mismatch: not directly related	2021	021-10635-5
Springer	08.12.2024	PradoJuan Boubeta-Puig	symptoms	Article	excluded	to OS keyword mismatch:	2021	http://link.springer.com/article/10.1007/s11042- 021-10759-6
Springer	08.12.2024	Dharmendra DangiDheeraj K. DixitAmit Bhagat	Sentiment analysis of COVID-19 social media data through machine learning	Article	excluded	not directly related to OS	2022	http://link.springer.com/article/10.1007/s11042- 022-13492-w
		Steven A. NiedererMichael S. SacksMark				keyword mismatch: not directly related		http://link.springer.com/article/10.1038/s43588-
Springer	08.12.2024	GirolamiKaren Willcox	Scaling digital twins from the artisanal to the industrial	Article	excluded	to OS keyword mismatch:	2021	021-00072-5
Springer	06.12.2024	Gang Luo	algorithms and hyper-parameter values	Article	excluded	to OS	2016	http://link.springer.com/article/10.1007/s13721- 016-0125-6
Springer	08.12.2024	Zhichao ZhangAbbas Z. Kouzani	Implementation of DNNs on IoT devices	Article	excluded	not directly related to OS	2020	http://link.springer.com/article/10.1007/s00521- 019-04550-w
		Xiang WuYongting ZhangAming WangMinyu	MNSSp3: Medical big data privacy protection platform based			keyword mismatch: not directly related		http://link.soringer.com/article/10.1007/s00521-
Springer	08.12.2024	ShiHuanhuan WangLian Liu	on Internet of things	Article	excluded	to OS keyword mismatch:	2022	020-04873-z
Springer	08.12.2024	Sina DamiMahtab Yahaghizadeh	Predicting cardiovascular events with deep learning approach in the context of the internet of things	Article	excluded	to OS	2021	http://ink.springer.com/article/10.1007/s00521- 020-05542-x
Springer	08 12 2024	R V Guadannini A SantanaR S Neves	Inference of pressure ulcer stage and texture on an image training set	Article	eveluded	not directly related	2016	http://link.springer.com/article/10.
opiniger	00.12.2024	A. V. KhvostikovA. S. KrylovI. A. MikhailovP. G.	Visualization of Whole Slide Histological Images with	10000	CACIDGOG	keyword mismatch: not directly related	2010	
Springer	08.12.2024	Malkov R. Moguel-SánchezC. S. Sergio Martínez-	Automatic Tissue Type Recognition	Article	excluded	to OS keyword mismatch:	2022	1134/S1054661822030208
Springer	07.12.2024	PalaciosJ. O. Ocharán-HernándezX. LimónA. J. Sánchez-García	Bots in Software Development: A Systematic Literature Review and Thematic Analysis	Article	excluded	not directly related to OS	2023	http://link.springer.com/article/10, 1134/S0361768823080145
			Understanding requirement prioritization artifacts: a			keyword mismatch: not directly related		http://link.springer.com/article/10.1007/s00766-
Springer	08.12.2024	Rahul Thakurta	systematic mapping study	Article	excluded	to US keyword mismatch:	2017	016-0253-7
Springer	07.12.2024	Aparicio	patents for EU countries Inequality and collaboration patterns in Canadian	Article	excluded	to OS keyword mismatch:	2023	http://link.springer.com/article/10.1007/s11192- 023-04699-1
Springer	08.12.2024	Gita GhiasiMatthew HarshAndrea Schiffauerova	nanotechnology: implications for pro-poor and gender- inclusive policy	Article	excluded	not directly related to OS	2018	http://link.springer.com/article/10.1007/s11192- 018-2701-2
		Tripti SwarnkarSrikanta PatnaikPabitra MitraSanjay MisraManohar Mishra	Ambient Intelligence in Health Care	Conference paper		keyword mismatch: not directly related	2023	http://link.springer.com/book/10.1007/978-981- 19-6068-0
Springer	07.12.2024	Angelo CroattiSara MontagnaCarolina	An Internet of Medical Things System to Increase Continuous		excluded	to OS keyword mismatch:		
Springer	07.12.2024	LombardiGianfranco ParatiMartino F. PengoAlessandro Silvani	Positive Airway Pressure Usage in Patients with Sleep- Disordered Breathing	Article	excluded	not directly related to OS	2021	http://link.springer.com/article/10.1007/s42979- 021-00522-x
Springer	08 12 2024	John T. HancockTaghi M. Khoshgoffaar	Hyperparameter Tuning for Medicare Fraud Detection in Big	Article	evolution	not directly related	2022	http://link.springer.com/article/10.1007/s42979-
Shundar	JU. 12.2UZ4	Yogesh KumarApeksha KoulSukhpreet KaurYi-	Machine Learning and Deep Learning Based Time Series Prediction and Forecasting of Ten Nations' COVID-19		avenued	keyword mismatch: not directly related	LUEE	
Springer	08.12.2024	Chen Hu	Pandemic	Article	excluded	to OS	2022	022-01493-3

Search	Query date	Author	Title	Type	Status	Reason	Year	Article IIRI
Tende	query dute		Machine Learning Workflow to Explain Black-Box Models for	1300	otatao	keyword mismatch:	rear	AILUG UNE
Springer	08.12.2024	Louise BlochChristoph M. Friedrich	Early Alzheimer's Disease Classification Evaluated for Multiple Datasets	Article	excluded	not directly related to OS	2022	http://link.springer.com/article/10.1007/s42979- 022-01371-v
		Anna Ivan Tihariy Nicolaas Dasilya				keyword mismatch:		
Springer	08.12.2024	Anca Jurcut I iberiu NiculceaPasika RanaweeraNhien-An Le-Khac	Security Considerations for Internet of Things: A Survey	Article	excluded	to OS	2020	http://link.springer.com/article/10.1007/s42979- 020-00201-3
		Tauraad Avaula ShaikhTahaayaa Daasad DasShahir	A data contribution intelligent and extended reality			keyword mismatch:		
Springer	06.12.2024	Sofi	technology in smart healthcare systems	Article	excluded	to OS	2022	http://link.springer.com/article/10.1007/s13278- 022-00888-7
			31 ConOnt: a three-level ontology for context modelling in			keyword mismatch:		
Springer	06.12.2024	Oscar CabreraXavier FranchJordi Marco	context-aware computing	Article	excluded	to OS	2019	http://link.springer.com/article/10.1007/s10270- 017-0611-z
		Giancarlo GuizzardiTiago Prince SalesJoão Paulo	Automated conceptual model clustering: a relator-centric			keyword mismatch: not directly related		
Springer	07.12.2024	A. AlmeidaGeert Poels	approach	Article	excluded	to OS	2022	http://ink.springer.com/article/10.1007/s10270- 021-0019-5
		Antônio Márcio Tavares ThomeRafael Garcia BarbastefanoLuiz Felipe ScavardaJoão Carlos	Industrial Engineering and Operations Management	Conference paper		keyword mismatch: not directly related	2021	030-78570-3
Springer	08.12.2024	Gonçalves dos ReisMarlene Paula Castro Amorim			excluded	to OS		
			A secured internet of robotic things (IoRT) for long-term care			not directly related		http://link.springer.com/article/10.1007/s11227-
Springer	06.12.2024	Shih-Hao ChangChih-Hsien HsiaWei-Zhi Hong	services in a smart building	Article	excluded	to OS keyword mismatch:	2023	022-04845-1
		Peyman TirandaziSeyed Mojtaba Hosseini	A review of studies on internet of everything as an enabler of			not directly related		http://link.springer.com/article/10.1007/s11227-
Springer	06.12.2024	BamakanAref Toghroljerdi	neuromarketing methods and techniques	Article	excluded	to OS keyword mismatch:	2023	022-04988-1
		Lingjun ZhaoChunhua SuZeyang DaiHuakun	Indoor device-free passive localization with DCNN for			not directly related		http://link.springer.com/article/10.1007/s11227-
Springer	08.12.2024	HuangShuxue DingXinyi HuangZhaoyang Han	Integration of improved YOLOv5 for face mask detector and	Article	excluded	to OS keyword mismatch:	2020	019-03110-2
		Thi Next Discover the Next Sector	auto-labeling to generate dataset for fighting against COVID-	Autola I.		not directly related	0000	http://link.springer.com/article/10.1007/s11227-
Springer	08.12.2024	I ni-Ngot Phamviet-Hoan NguyenJun-Ho Hun	OrthopedVR: clinical assessment and pre-operative planning	Article	excluded	keyword mismatch:	2023	022-04979-2
Coringor	09 12 2024	David SibrinaSarath BethapudiGeorge Alex	of paediatric patients with lower limb rotational abnormalities	Article	ovoludod	not directly related	2023	http://link.springer.com/article/10.1007/s00371-
ophinger	00.12.2024	Notificita	in virtual reality	ALLOIG	excluded	keyword mismatch:	2020	023-02949-0
Springer	08 12 2024	Alexia Dini KounoudesGeorgia M. Kapitsakiloannis Katakis	Enhancing user awareness on inferences obtained from fitness trackers data	Article	evoluded	not directly related	2023	http://link.springer.com/article/10.1007/s11257-
opninger	00.12.2024	TOTO TO TOTO	inchere data	/ a dolo	excluded	keyword mismatch:	LOLO	022-03353-0
Springer	08.12.2024	Kiran FahdSitalakshmi Venkatraman	Visualizing risk factors of dementia from scholarly literature using knowledge maps and next-generation data models	Article	excluded	not directly related to OS	2021	http://link.springer.com/article/10.1186/s42492- 021-00085-x
		Mauro Del Rio, Vittorio Meloni, Francesca Frexia,		Conference		keyword mismatch:		<u></u>
ACM Digital Library	04.11.2024	Francesco Cabras, Hoberto Tumbarello, Sabrina Montis, Andrea Marini, and Gianluigi Zanetti	Augmented Reality for Supporting Real Time Telementoring: an Exploratory Study Applied to Ultrasonography	paper	excluded	to OS	2018	https://doi.org/10.1145/3239438.3239444
			Tracing Contacts With Mobile Phones to Curb the Pandemic:	Conference		keyword mismatch:		
ACM Digital Library	30.11.2024	Steven Arzt, Andreas Poller, and Gisela Vallejo	Official German Contact-Tracing App	рарен	excluded	to OS	2021	https://doi.org/10.1145/3411763.3451631
		Fabian Okeke, Lucas Nene, Anne Muthee, Stephen	Opportunities and challenges in connecting care registered to	Conference		keyword mismatch:		
Library	29.11.2024	Dell	the community health feedback loop	paper	excluded	to OS	2019	https://doi.org/10.1145/3287098.3287111
		Andres Felipe FuentesDiego Fernando BoteroCristhian Torres Bamirez	A Covid-19 Vaccination Tracking and Control Platform in Santiago de Cali	Conference		keyword mismatch: not directly related	2022	http://link.springer.com/chapter/10.1007/978-3- 030-96753-6_13
Springer	06.12.2024			paper	excluded	to OS		
		Jianqiao JinKunyang WangLei RenZhihui QianWei LiangWei ChenXuewei LuDi Zhao	A New 3D Printed Passive Flexible Prosthesis Based on the Human Foot	Conference		keyword mismatch: not directly related	2022	http://link.springer.com/chapter/10.1007/978-3- 031-13841-6_60
Springer	06.12.2024			paper	excluded	to OS		http:///init.apr/2007.070.0
		David Ben-TovimMariusz BajgerShaowen Qin	A Novel Application of a Mutual Information Measure for Analysing Temporal Changes in Healthcare Network Graphs	Conference		not directly related	2023	031-46671-7_7
Springer	06.12.2024	Podriguo P. TohomaGoorgios TravollasMarios	A Bilot mHoalth Project for Monitoring Vital Body Signals and	paper	excluded	to OS	2022	http://link.springer.com/chapter/10.1007/978-3-
		NestorosAnastasis C. Polycarpou	Skin Conditions	Conference		not directly related	2022	031-06368-8_18
Springer	06.12.2024	Mario CiampiMario SicuranzaAngelo	A Technological Framework for EHR Interoperability	paper	excluded	to OS keyword mismatch:	2017	http://link.springer.com/chapter/10.1007/978-3-
		EspositoRoberto GuarasciGiuseppe De Pietro	Experiences from Italy	Conference		not directly related	2011	319-62704-5_6
Springer	06.12.2024	Kibriva InamdarOormila Ramanandan Kottavi	A Web Architecture for E-Health Applications Supporting the	paper	excluded	to OS keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978-3-
		PilapprathodiJopaul JohnMarkus WolffMarcel	Efficient Multipath Transport of Medical Images	Conference		not directly related		031-06668-9_11
Springer	06.12.2024	Nuria García-SantaKendrick Cetina	AGG: An Automated Genogram Generator by Discovering	paper	excluded	keyword mismatch:	2023	http://link.springer.com/chapter/10.1007/978-3-
Coringor	06 12 2024		Information in Clinical Texts	Conference	ovoludod	not directly related		031-26422-1_38
ophngei	00.12.2024			Conference	excluded	keyword mismatch:		
ACM Digital	04 11 2024	Laura Lucaj, Patrick van der Smagt, and Djalel Benbouzid	Al Regulation Is (not) All You Need	paper	excluded	not directly related to OS	2023	https://doi.org/10.1145/2502012.2504070
				Conference		keyword mismatch:		Inspectation of the transfer to the control of the total of
ACM Digital Library	08.11.2024	Darren Edge, Jonathan Larson, and Christopher White	Bringing AI to BI: Enabling Visual Analytics of Unstructured Data in a Modern Business Intelligence Platform	paper	excluded	not directly related to OS	2018	https://doi.org/10.1145/3170427.3174367
				Conference		keyword mismatch:		
ACM Digital Library	30.11.2024	Ashwed Patil	The role of ICTs in refugee lives	paper	excluded	to OS	2019	https://doi.org/10.1145/3287098.3287144
		Tolulope Peter OyekanmiNhat NguyenVangalur	A Context-Aware Healthcare Architecture for the Elderly	Conference		keyword mismatch:	2016	http://link.springer.com/chapter/10.1007/978-3- 319-29236-6_3
Springer	06.12.2024	riugai		paper	excluded	to OS		
		Jorgina AvilesKathleen ColfordMegan HallMassimo MarenzanaAlena UusSharon GilesPhilippa	A rast Anatomical and Quantitative MRI Fetal Exam at Low Field			keyword mismatch:	2022	031-17117-8_2
		BridgenMary A. RutherfordShaihan J. MalikJoseph		Conference		not directly related		
Springer	06.12.2024	Mohammad R. H. MullickMarcel GroßmannUdo R.	A Feasibility Study of a Lightweight Fog Computing	paper	excluded	keyword mismatch:	2023	http://link.springer.com/chapter/10.1007/978-3-
Coringor	06 12 2024	Krieger	Architecture Integrating Blockchain Technology for Smart E-	Conference	ovoludod	not directly related		031-40852-6_14
ophinger	00.12.2024		Tourin Approvisions	Conference	excluded	keyword mismatch:		
ACM Digital	08 11 2024	Marten Sigwart, Michael Borkowski, Marco Peise, Stefan Schulte, and Stefan Tai	Blockchain-based Data Provenance for the Internet of Things	paper	excluded	not directly related to OS	2019	https://doi.org/10.1145/3365871.3365886
		Paolo AbbondanzaSilvio GiancolaRemo SalaMarco	Accuracy of the Microsoft Kinect System in the Identification			keyword mismatch:	2017	http://ink.springer.com/chapter/10.1007/978-3- 319-58877-3_37
Springer	06.12.2024	Tarabini	of the Body Posture	paper	excluded	to OS		
		Ibtihel SelmiNadia KabachiSana Ben Abdalah Ben	Adaptive Agent-Based Architecture for Health Data Integration	Conforman		keyword mismatch:	2020	http://link.springer.com/chapter/10.1007/978-3- 030-45989-5_18
Springer	06.12.2024	Lamineriajer Baazaoui Zghai		paper	excluded	to OS		
ACM Disital		Wubing Chen, Zhiving Xu, Shuvu Shi, Yang Zhao,		Conference		keyword mismatch:		
Library	04.11.2024	and Jun Zhao	A Survey of Blockchain Applications in Different Domains		excluded	to OS	2018	https://doi.org/10.1145/3301403.3301407
ACM Digital		Yutao Zhang, Robert Chen, Jie Tang, Walter F.	LEAP: Learning to Prescribe Effective and Safe Treatment	Conference paper		keyword mismatch: not directly related		
Library	11.11.2024	Stewart, and Jimeng Sun	Combinations for Multimorbidity		excluded	to OS	2017	https://doi.org/10.1145/3097983.3098109
			Presenting Oblighteria Original Thinking Truncis Data 1	Casta		keyword mismatch:		net/publication/357827199_Promoting_children's_ critical thinking towards robotics through robot
ACM Digital Library	29.11.2024	Maria Luce Lupetti and Maarten Van Mechelen	through Robot Deception	paper	excluded	to OS	2022	deception/link/622b31c5a39db062db93778c/down load
ACM District		Alistair Sutcliffe, Pete Sawyer, Wei Liu, and Mally	Investigating the potential impact of values on requirements	Conference		keyword mismatch:		
Library	11.11.2024	Bencomo	and software engineering	paper	excluded	to OS	2021	https://doi.org/10.1109/ICSE-SEIS52602, 2021.00013
		Debora C. EngelmannLucca Dornelles CezarAlison R. PanissonRafael H. Bordini	A Conversational Agent to Support Hospital Bed Allocation	Conference		keyword mismatch: not directly related	2021	nttp://ink.springer.com/chapter/10.1007/978-3- 030-91702-9_1
Springer	06.12.2024	Mathaua HS Musin, Jack Meda M. Devid, S. 1		paper	excluded	to OS		
ACM Digital Library	29.11.2024	Braga, Fernanda Campos, and Victor Stroele	Pragmatic interoperability in IoT: a systematic mapping study	paper	excluded	english	2019	https://doi.org/10.1145/3323503.3349561

Search	Query date	Author	Title	Type	Statue	Reason	Vear	Article IIDI
venue	Query uate	Débora C. Engelmann, Alison R. Panisson, Renata	The	туре	Status	Reason	real	Article URL
ACM Digital		Vieira, Jomi Fred Hübner, Viviana Mascardi, and	MAIDS - A Framework for the Development of Multi-Agent	Conference				
Library	11.11.2024	Rafael H. Bordini	Intentional Dialogue Systems	paper	excluded	type: no access	2023	https://dl.acm.org/doi/10.5555/3545946.3598764
		Steranos StavrotneodorosNikolaos KaklanisKonstantinos VotisDimitrios Tzovaras	A Smart-Home Io I Infrastructure for the Support of Independent Living of Older Adults	Conference		reyword mismatch:	2018	319-92016-0_22
Springer	06.12.2024		independent Enring of order Addite	paper	excluded	to OS		
ACM Digital		Alexa Meda dan						
Library	21.06.2024	Alex vakulov	Protecting Life-Saving Medical Devices from Cyberattack	Article	excluded	tocus not on US keyword mismatch:	2023	https://doi.org/10.1145/3631535
ACM Digital						not directly related		
Library	22.06.2024	Irving Wladawsky-Berger	Digital Economy: The Evolution of the Digital Economy	Article	excluded	to healthcare	2023	https://doi.org/10.1145/3587258
		William D. Tucker	Amplifying Positive Deviance with ICT	Conference		keyword mismatch:	2017	http://ink.springer.com/chapter/10.1007/978-3- 319-59111-7_18
Springer	07.12.2024			paper	excluded	to OS		
-pringer		Bernardo SantosImran Qayyrm KhanBruno	Anomaly Detection in Cellular IoT with Machine Learning	p op or		keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978-3-
		DzogovicBoning FengVan Thuan DoNiels		Conference		not directly related		030-91421-9_5
Springer	07.12.2024	JacotThanh Van Do	Appending to the Out Ecoling: How Intermittent Eastern Chasses	paper	excluded	to OS	2021	http://link.springer.com/chapter/10.1007/978-3-
		Tryeyoung Hyuseoyeon Hong	Information Tab Interfaces for Information Acquisition	Conference		not directly related	2021	030-71292-1_23
Springer	07.12.2024			paper	excluded	to OS		
		Sofia ScatagliniMarcello FuscaGiuseppe	Application of Wearable Monitoring System in Tourette	Conference		keyword mismatch:	2017	http://ink.springer.com/chapter/10.1007/978-3- 319-58877-3_13
Springer	07.12.2024	Andreonimauro Porta	Syndrome Assessment	paper	excluded	to OS		
opinigoi	OTTELEDE T	Vladimir F. KhoroshevskyVladimir F. Efimenkolrina	Artificial Intelligence, Biotechnology and Medicine: Reality,	F-F	Choladea	keyword mismatch:	2020	http://link.springer.com/chapter/10.1007/978-3-
		V. Efimenko	Myths and Trends	Conference		not directly related		030-59535-7_31
Springer	07.12.2024	Salvador CarriónÁlvaro Lónez-Chilet Javier	Automatic Detection of Epileptic Seizures with Recurrent and	paper	excluded	to US keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978-3-
		Martínez-BerniaJoan Coll-AlonsoDaniel Chorro-	Convolutional Neural Networks	Conference		not directly related	2022	031-13321-3_46
Springer	07.12.2024	JuanJon Ander Gómez		paper	excluded	to OS		
		Bryce A. BeslerAndrew S. MichalskiNils D.	Automatic Full Femur Segmentation from Computed	0		keyword mismatch:	2018	http://ink.springer.com/chapter/10.1007/978-3- 319-74113-0 11
Springer	07.12.2024	ForkertSteven K. Boyd	tomography Datasets Using an Atlas-Based Approach	Daper	excluded	to OS		
opinigoi	OTTENEOR T	Affan YasinLin LiuZhanqiang CaoJianmin	Big Data Services Requirements Analysis	F-F		keyword mismatch:	2018	http://link.springer.com/chapter/10.1007/978-
		WangYingbo LiuTan Sheau Ling		Conference		not directly related		<u>981-10-7796-8_1</u>
Springer	07.12.2024	Veravuth LertnatteePeravot Pamonsinlanatham	Blended Learning for Improving Elevibility of Learning	paper	excluded	to US keyword mismatch:	2017	http://link.springer.com/chapter/10.1007/978-3-
		verayuti termateererayot ramonsmapatham	Structure Query Language (SQL)	Conference		not directly related	2017	319-59360-9_30
Springer	07.12.2024			paper	excluded	to OS		
		Puneet GoswamiVictor Hugo C. de	Blockchain Based B-Health Prototype for Secure Healthcare	Conformer		keyword mismatch:	2023	http://link.springer.com/chapter/10.1007/978-3- 031-28350-5_6
Springer	07 12 2024	AlbuquerqueLakshita Aggarwai	Iransactions	Daper	evoluded	to OS		
opiniger	07.12.2024	Simon KrejciMarten SigwartStefan Schulte	Blockchain- and IPFS-Based Data Distribution for the Internet	paper	excluded	keyword mismatch:	2020	http://link.springer.com/chapter/10.1007/978-3-
			of Things	Conference		not directly related		030-44769-4_14
Springer	07.12.2024	Mine Li Al Celusi Metthews Key Asses Mana avia D	PCNOlaudi Olaud Castand Wissland Badu Castan Data	paper	excluded	to OS	0000	http://link.epringer.com/chapter/10.1007/978-3-
		Prabhakaran	Collection, Streaming, and Analytics System	Conference		not directly related	2020	030-64991-3_5
Springer	07.12.2024		obligation, outcarning, and relaying of option	paper	excluded	to OS		
		Kashish Ara ShakilMansaf Alam	Cloud Computing in Bioinformatics and Big Data Analytics:			keyword mismatch:	2018	http://link.springer.com/chapter/10.1007/978- 981-10-6620-7_60
Cosinger	07 40 0004		Current Status and Future Research	Conference	avaludad	not directly related		361-10-0020-1-00
Springer	07.12.2024	Solomon Olalekan OvenugaLalit GargAmit Kumar	Cloud-Based Clinical Decision Support System	paper	excluded	keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978-3-
		BhardwajDivya Prakash Shrivastava		Conference		not directly related		030-67187-7_24
Springer	07.12.2024			paper	excluded	to OS		
		Ahlam MallakJeonghwan GwakJong-In SongSang- Woong Lee	Computational and Comparative Study on Multiple Kernel	Conference		keyword mismatch:	2016	http://ink.springer.com/chapter/10.1007/978-3- 319-46909-6_30
Springer	07.12.2024	Woong Lee	Alzheimer's Disease	paper	excluded	to OS		
-1		Shimelis Assefa	Data Management Strategy for AI Deployment in Ethiopian			keyword mismatch:	2023	http://link.springer.com/chapter/10.1007/978-3-
. ·			Healthcare System	Conference		not directly related		031-31327-1_3
Springer	07.12.2024	Andreas E. SchützTohias FertigKristin Weber	Defining a Core Ontology for Medical Devices in Germany to	paper	excluded	to US keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978-3-
		Andreas E. Schutz robias Perugkhsun Weber	Ensure Semantic Interoperability	Conference		not directly related	2021	030-68527-0_25
Springer	07.12.2024			paper	excluded	to OS		
		Ar Kar KyawZhuang TianBrian Cusack	Design and Evaluation for Digital Forensic Ready Wireless	0		keyword mismatch:	2020	http://ink.springer.com/chapter/10.1007/978-3- 030-42029-1 9
Springer	07 12 2024		Medical Systems	paper	excluded	to OS		
opinigoi	OTTILLOL T	Frédéric VoisinArnaud BidottiFrance Mourey	Designing Soundscapes for Alzheimer's Disease Care, with	holeo.	entituded	keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978-3-
			Preliminary Clinical Observations	Conference		not directly related		030-70210-6_34
Springer	07.12.2024	Thuong-Cang PhanAnh-Cang PhanKhac-Tuong	Detection and Segmentation of Brain Tumors on 3D MB	paper	excluded	to US keyword mismatch:	2023	http://link.springer.com/chapter/10.1007/978-
		NguyenHo-Dat Tran	Images Using 3D U-Net	Conference		not directly related	2020	981-99-8296-7_38
Springer	07.12.2024			paper	excluded	to OS		
		Salvador CarriónÁlvaro López-ChiletJavier	Detection of Pulmonary Conditions Using the DeepHealth	0		keyword mismatch:	2022	http://ink.springer.com/chapter/10.1007/978-3- 031-13321-3.49
Springer	07.12.2024	JuanJon Ander Gómez	Framework	paper	excluded	to OS		
opgo.		Robert KrimmerStefan DedovicCarsten	Developing Cross-border E-Governance: Exploring	P of a		keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978-3-
0		SchmidtAndreea-Ancuta Corici	Interoperability and Cross-border Integration	Conference		not directly related		030-82824-0.9
Springer	07.12.2024	Jenneh Lawson	Developing Cryptography Key Management Plans for	paper	excluded	to US keyword mismatch:	2020	http://link.springer.com/chapter/10.1007/978-3-
			Healthcare Institutions	Conference		not directly related	2020	030-43020-7_82
Springer	07.12.2024			paper	excluded	to OS		
		Hopert BrayLuke MacDougallCody BlankenshinKimberly MitchellEengoei VugoSuluio	Development and Assessment of a Friendly Robot to Ease	Conference		Keyword mismatch:	2022	031-24670-8_34
Springer	07.12.2024	Cerel-SuhlXiaopeng Zhao		paper	excluded	to OS		
		Tommaso Di NotoChirine AtatEduardo Gamito	Diagnostic Surveillance of High-Grade Gliomas: Towards			keyword mismatch:	2021	http://ink.springer.com/chapter/10.1007/978-3-020.99729-1.30
Springer	07 12 2024	TelgaMonika HegiAndreas HottingerMeritxell Bach	Automated Change Detection Using Radiology Report	Conference	excluded	not directly related		300 307 00-1_00
opninger	07.12.2024	John SoldatosNikos KefalakisGeorgios	Digital Platform and Operator 4.0 Services for Manufacturing	Paper	excinded	keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978-3-
		MakantasisAngelo MarguglioOscar Lazaro	Repurposing During COVID19	Conference		not directly related		030-85910-7_33
Springer	07.12.2024			paper	excluded	to OS		
		Surieen KaurSandeep Sharma	Edge-Assisted IoT Architecture: A Case of Air Pollution Monitoring Frameworks	Conference		keyword mismatch:	2022	031-09469-9_21
Springer	07.12.2024		Montoling Planetic Inc	paper	excluded	to OS		
		Alba Catalina Morales TiradoEnrico DagaEnrico	Effective Use of Personal Health Records to Support	_		keyword mismatch:	2020	http://link.springer.com/chapter/10.1007/978-3- 020-61244-2_4
Coringer	07 40 2024	Motta	Emergency Services	Conference	avaludad	not directly related		030-01244-3_4
Springer	07.12.2024	Marietheres DietzLudwig EnglbrechtGünther	ENHANCING INDUSTRIAL CONTROL SYSTEM FORENSICS	рары	excluded	keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978-3-
		Pernul	USING REPLICATION-BASED DIGITAL TWINS	Conference		not directly related		030-88381-2_2
Springer	08.12.2024			paper	excluded	to OS		
		A. KavithaS. Pravin KumarG. DarsanaG. Sudhir	Ennancing Visualization of Surgical Tool Through Integrated Motion Tracking System	Conference		keyword mismatch:	2023	031-43404-4_26
Springer	08.12.2024		meter intening cyclent	paper	excluded	to OS		
		Telmo SilvaMartinho Mota	Evaluating iTV Companion Application with Seniors			keyword mismatch:	2020	http://link.springer.com/chapter/10.1007/978-3- 030-52677-1_9
Corio	00 10 0001			Conference	owebut (	not directly related		waw-we0//-1_8
opninger	00.12.2024	Amrit ChhetriMari IversenJens KaasbollChino	Evaluating mHealth Apps Using Affordances: Case of	paper	excluded	keyword mismatch	2019	http://link.springer.com/chapter/10.1007/978-3-
		Kanjo	CommCare Versus DHIS2 Tracker	Conference		not directly related		030-18400-1_51
Springer	08.12.2024	Conjunito Adou Outbolo Mantaña-David France Mil	Evolution of Dispeture Obseif-star Matheda (a. C. 1)	paper	excluded	to OS	0047	http://link.enringer.com/chanter/10.1007/070.0
		casimiro Aday Curbelo MontanezPaul FergusAbir HussainDhiya Al-JumeilyMehmet Tevfik	Evaluation of Phenotype Glassification Methods for Obesity Using Direct to Consumer Genetic Data	Conference		net directly related	2017	319-63312-1_31
Springer	08.12.2024	DorakRosni Abdullah	and allock a containing contain party	paper	excluded	to OS		
		Mateusz GołoszDariusz Mrozek	Exploration of Data from Smart Bands in the Cloud and on the			keyword mismatch:	2019	http://link.springer.com/chapter/10.1007/978-3- 030-22744-9_47
Springer	08 12 2024		Edge – The Impact on the Data Storage Space	Conference	excluded	not directly related to OS		
opringer	00.12.2024	Olga FajardaAlina TrifanMichel Van	Exploring the Value of Electronic Health Records from Multiple	խորժյ	excluded	keyword mismatch	2019	http://link.springer.com/chapter/10.1007/978-3-
		SpeybroeckPeter R. RijnbeekJosé Luís Oliveira	Datasets	Conference		not directly related		030-29196-9_19
Springer	08.12.2024			paper	excluded	to OS		

Search								
venue	Query date	Author Haisaun HanBing-bing Han	Title East Integration System of English Online Learning Resources	Туре	Status	Reason	Year	Article URL http://ink.spripper.com/chapter/10.1007/978-3-
		har-yun hanoling-bing han	Based on Multi Sensor Network	Conference		not directly related	2022	030-94182-6_5
Springer	08.12.2024			paper	excluded	to OS		
		N. KeerthanaK. LokeshwaranP. S. KavinkumarT.	Fingerprint Integrated E-Health Monitoring System Using IoT			keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978- 981-16-5048-2 12
Springer	08 12 2024	KaviyanM. ManimaranS. Elango		Conference	excluded	to OS		<u></u>
opiniger	00.12.2024	Priya RanjanKumar Dron ShrivastavSatya	HRIDAI: A Tale of Two Categories of ECGs	paper	excluded	keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978-
		VadlamaniRajiv Janardhanan		Conference		not directly related		981-16-0425-6_18
Springer	08.12.2024			paper	excluded	to OS	0040	http:///init.com/com/chapter/10.1007/070.0
		Koray Inçkimenmet S. Aktaş	Improving Awareness in Ambient-Assisted Living Systems: Consolidated Data Stream Processing	Conference		keyword mismatch: not directly related	2016	319-51234-1_14
Springer	08.12.2024		Consolidated Data Otican Processing	paper	excluded	to OS		
		Elaine ByrneJørn A. BraaJohn LewisSam	Including the Excluded: Achieving Representation Through			keyword mismatch:	2022	http://ink.springer.com/chapter/10.1007/978-3-
0	00 40 0004	BouphaChansally Phommavong	Strengthening Health Information Systems in Lao PDR	Conference	and a start	not directly related		031-19429-0_11
Springer	08.12.2024	Kamila Misieiuk lennifer SciannaBogers KaliisaKarl	Incorporating Sentiment Analysis with Epistemic Network	paper	excluded	to US keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978-3-
		VachuskaDavid Williamson Shaffer	Analysis to Enhance Discourse Analysis of Twitter Data	Conference		not directly related	LOLI	030-67788-6_26
Springer	08.12.2024			paper	excluded	to OS		
		Ahmad TahaYao GeWilliam TaylorAhmed ZobaKbaled AssalebKamrap ArshadOammer H	Indoor Activity Position and Direction Detection Using Software Defined Radios	Conference		keyword mismatch:	2022	http://ink.springer.com/chapter/10.1007/978-3- 030-95593-9_2
Springer	08.12.2024	AbbasiMuhammad Ali Imran	Software Denned Habios	paper	excluded	to OS		
		Suzana BrownFaheem Hussain	Information Ecology as a Framework for South-South			keyword mismatch:	2017	http://ink.springer.com/chapter/10.1007/978-3-
0			Cooperation: Case Studies of Rwanda and Bangladesh ICT-	Conference	and a start	not directly related		319-39111-7_05
Springer	00.12.2024	Kevin MaxiVillie Morocho	Integrating Medical Information Software Using Health Level	paper	excluded	keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978-3-
			Seven and FHIR: A Case Study	Conference		not directly related		030-99170-8 7
Springer	08.12.2024			paper	excluded	to OS		
		Qing-bang ZengWen-da Xie	Intelligent Encrypted Storage Method for Medical Health Database Based on Internet of Things	Conference		keyword mismatch:	2022	http://ink.springer.com/chapter/10.1007/978-3- 030-94182-6_16
Springer	08.12.2024		Database based on internet of Things	paper	excluded	to OS		
-p		Sushunova G. MartinezKarla A. Badillo-	Investigation of Multimodal Mobile Applications for Improving	p op o		keyword mismatch:	2016	http://link.springer.com/chapter/10.1007/978-3-
		UrquiolaRebecca A. LeisJamie ChavezTiffany	Mental Health	Conference		not directly related		319-39952-2_32
Springer	08.12.2024	Green Iravis Clements	Kushkalla: A Web-Based Blatform to Improve Eurotional	paper	excluded	to OS	2017	http://link.spripper.com/chapter/10.1007/978-3-
		LunaChristian MerchánMaría C. CuencaGloria M.	Movement Rehabilitation	Conference		not directly related	2017	319-67283-0_15
Springer	08.12.2024	Díaz		paper	excluded	to OS		
		Eric Adu-GyamfiPetter Nielsen	Leveraging Software Platform Capabilities to Support HIV			keyword mismatch:	2017	http://ink.springer.com/chapter/10.1007/978-3- 319-59111-7_4
Paringer	00 10 0004		(ART) Treatment Adherence Management: A Case from Sierra	Conference	ovoludod	not directly related		313-33111-7_4
Springer	06.12.2024	Hamza Ben BoummaneCherki Daoui	Leone	paper	excluded	keyword mismatch:	2023	http://link.springer.com/chapter/10.1007/978-3-
			Robot in Hospital Environment	Conference		not directly related	2020	031-37872-0_12
Springer	08.12.2024			paper	excluded	to OS		
		Catherine TryfonaGiles OatleyAna CalderonSimon	M-Health Solutions to Support the National Health Service in the Diagnosis and Manitaring of Autism Spectrum Diagnders in	Conference		keyword mismatch:	2016	http://ink.springer.com/chapter/10.1007/978-3- 319-40238-3_24
Springer	08.12.2024	mone	Young Children	paper	excluded	to OS		
opgei		Ivan ErmilovAxel-Cyrille Ngonga NgomoAad	Managing Lifecycle of Big Data Applications	p op o			2017	http://link.springer.com/chapter/10.1007/978-3-
		VerstedenHajira JabeenGezim SejdiuGiorgos				keyword mismatch:		319-69548-8_18
Paringer	09 40 0004	ArgyriouLuigi SelmiJürgen JakobitschJens		Conference	ovoludod	not directly related		
Springer	00.12.2024	Mohit KumarBernhard MoserLukas	Membership-Mappings for Data Representation Learning: A	paper	excluded	keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978-3-
		FischerBernhard Freudenthaler	Bregman Divergence Based Conditionally Deep Autoencoder	Conference		not directly related		030-87101-7_14
Springer	08.12.2024			paper	excluded	to OS		hite Web and any february 0.007/070.0
		Bastian BierMathias Unberath Iobias Geimer, Jennifer MaierGarov GoldMarc	Motion Compensation Using Range Imaging in C-Arm Cone- Ream CT	Conference		keyword mismatch: not directly related	2017	319-60964-5_49
Springer	08.12.2024	LevenstonRebecca FahrigAndreas Maier	boarron	paper	excluded	to OS		
		Jasmin HenzeSalima HoutaRainer	Multimodal Detection of Tonic-Clonic Seizures Based on 3D			keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978-3- 030-69763-2_27
Paringer	09 40 0004	SurgesJohannes KreuzerPinar Bisgin	Acceleration and Heart Rate Data from an In-Ear Sensor	Conference	ovoludod	not directly related		Manager Marz_ar
Springer	00.12.2024	Margarita ZaleshinaAlexander Zaleshin	Multiscale Integration for Pattern Recognition in Neuroimaging	paper	excluded	keyword mismatch:	2016	http://link.springer.com/chapter/10.1007/978-3-
				Conference		not directly related		319-51469-7_35
Springer	08.12.2024			paper	excluded	to OS		
		David Ben-TovimMariusz BajgerViet Duong	Network Graph Analysis of Hospital and Health Services	Conforance		keyword mismatch:	2022	http://ink.springer.com/chapter/10.1007/978-3- 030-95405-5_3
Springer	08.12.2024	Bushaowen gin	Functional Structures	paper	excluded	to OS		
		Vladimir V. GodinAnna E. TerekhovaSofia V.	New Areas of Blockchain Technology Applications During the			keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978-3-
		Kalamagina	COVID-19 Pandemic	Conference		not directly related		030-95494-9_16
Springer	08.12.2024	Peter LinTerenz EscartinMatthew NoMenovuan	Novel Imaging Biomarkers to Evaluate Heart Dysfunction Post-	paper	excluded	to US keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978-3-
		LiMelissa LarsenJennifer Barryldan	chemotherapy: A Preclinical MRI Feasibility Study	Conference		not directly related	2022	030-93722-5 4
Springer	08.12.2024	RoifmanMihaela Pop		paper	excluded	to OS		
		Andreas MaierSeung Hee YangFarhad	Offer Proprietary Algorithms Still Protection of Intellectual	0		keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978-3- 658-36932-3 70
Springer	08 12 2024	Malekinikesn MuthukrishnanReza Forghani	Property in the Age of Machine Learning?	conference	excluded	to OS		
opinigoi	00.12.2024	Hiranya G. KumarAnthony R. LawnB.	Opti-Speech-VMT: Implementation and Evaluation	perpet.	excideda	keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978-3-
		PrabhakaranWilliam F. Katz		Conference		not directly related		030-95593-9_19
Springer	08.12.2024	Zhongije WangDianhui ChuViaofai Vu	Personal Service Eco-Environment (DSE9): A Llear-Centric	paper	excluded	to US keyword miematch:	2016	http://link.springer.com/chapter/10.1007/978-3-
		zionglie wangblannu onuxiaolei xu	Services Computing Paradigm	Conference		not directly related	2010	319-32689-4 11
Springer	08.12.2024			paper	excluded	to OS		
		Julia KeizerNienke Beerlage-de JongNashwan Al	Persuading from the Start: Participatory Development of	Conference		keyword mismatch:	2020	http://ink.springer.com/chapter/10.1007/978-3- 030-45712-9_9
Springer	08.12.2024	Nalemia, E. W. G. van Gement-Pijnen	Healthcare	paper	excluded	to OS		
- p901		Diana AlOmariMohammad Mehedi Hassan	Predicting Telecommunication Customer Churn Using Data			keyword mismatch:	2016	http://link.springer.com/chapter/10.1007/978-3-
			Mining Techniques	Conference		not directly related		<u>319-45940-0_15</u>
Springer	08.12.2024	Chomundanual Artimeter Director	Description of COL Integration Attacks in Mich. Acceleration	paper	excluded	to OS	0015	http://link.springer.com/abaster/10.1007/070.0
		onamunoeswan Arumugamvarsha Bhargavi DwarakanathanS, GnanamarvVishalroi Nataraian	Frediction of SQL Injection Attacks in Web Applications				2019	030-24305-0_37
		NeyveliRohit Kanakuppaliyalil				keyword mismatch:		
		RameshYeshwanthraa KandhavelSadhanandhan		Conference		not directly related		
Springer	08.12.2024	Balakrishnan	Promises and Broblems in the Adaptics, 10, 100, 11	paper	excluded	to OS	0005	http://link.springer.com/abaster/10.1002/020.0
		marco nunselereva Poli	Identity Management from a Consumer Perspective	Conference		not directly related	2023	031-31971-6_8
Springer	08.12.2024		ional, management nem a concenter i erepeente	paper	excluded	to OS		
-		Patrick GuilbaudMichael J. Hirsch	Promoting Equity and Achievement in Real-Time Learning			keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978-3- 031.05887-5 12
Caringer	00 40 0004		(PEARL): Towards a Framework for the Formation, Creation,	Conference	a valuada d	not directly related		001-00007-0_12
opringer	00.12.2024	Sarathkumar RangarajanHuai LiuHua WangChuan-	Scalable Architecture for Personalized Healthcare Service	haher	excluded	keyword mismatch	2018	http://link.springer.com/chapter/10.1007/978-3-
		Long Wang	Recommendation Using Big Data Lake	Conference		not directly related		319-76587-7_5
Springer	08.12.2024			paper	excluded	to OS		
		Boning FengVan Thuan DoNiels JacotBernardo	Secure 5G Network Slicing for Elderly Care	Conference		keyword mismatch:	2019	030-27192-3_16
Springer	08.12.2024	Do		paper	excluded	to OS		
		Enes Yigitbaslvan JovanovikjGregor Engels	Simplifying Robot Programming Using Augmented Reality and			keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978-3-
			End-User Development	Conference		not directly related		usu-85623-6_36
Springer	08.12.2024	Chiara LosquadroSilvia ConforteMaurizio	Small and Lame Bile Ducts Intrahenatic Cholongioganicant	paper	excluded	to OS	2024	http://link.springer.com/chapter/10.1007/978-9-
		SchmidGaetano GiuntaMarco RengoVincenzo	Classification: A Preliminary Feature-Based Study			keyword mismatch:	2021	030-89128-2_23
		CardinaleGuido CarpinoAndrea LaghiAna	,	Conference		not directly related		
Springer	08.12.2024	LleoRiccardo MugliaEzio LanzaGuido Torzilli		paper	excluded	to OS	0.7	hate (Med. and and
		Patrick GuilbaudEric BubarElizabeth Langran	STEM Excellence and Equity in K-12 Settings: Use of Augmented Reality-Based Educational Experiences to	Conference		keyword mismatch:	2021	nttp://ink.springer.com/chapter/10.1007/978-3- 030-78645-8_6
Springer	08.12.2024		Promote Academic Achievement and Learner Success	paper	excluded	to OS		
		Fausto MilletariVighnesh BirodkarMichal Sofka	Straight to the Point: Reinforcement Learning for User			keyword mismatch:	2019	http://link.springer.com/chapter/10.1007/978-3-
			Guidance in Ultrasound	Conference		not directly related		030-32875-7_1
Springer	08.12.2024			paper	excluded	to OS		

Search	Querv date	Author	Title	Type	Status	Reason	Year	Article URL
Tende	autory auto	Guy C. HembroffDaniel BoyleTimothy Van Wagner	The Design of a Holistic mHealth Community Library Model and Its Impact on Empowering Rural America	Conference	otatao	keyword mismatch: not directly related	2020	http://ink.springer.com/chapter/10.1007/978-3- 030-49289-2_8
Springer	08.12.2024	Min ZhangTianshi WuXitong GuoXiaoxiao	The Effects of the Externality of Public Goods on Doctor's	paper	excluded	to OS keyword mismatch:	2017	http://ink.springer.com/chapter/10.1007/978-3-
Springer	08.12.2024	LiuWeiwei Sun	Private Benefit: Evidence from Online Health Community	Conference paper	excluded	not directly related to OS		319-67964-8_14
		Ivan SmirnovMaksim StankevichYulia KuznetsovaMargarita SuvorovaDaniil LarionovElena NikitinaMikhail SavelovOleg	TITANIS: A Tool for Intelligent Text Analysis in Social Media	Conference		keyword mismatch: not directly related	2021	http://ink.springer.com/chapter/10.1007/976-3- 030-86855-0_16
Springer	08.12.2024	Grigoriev Alina TrifanJosé Luís Oliveira	Towards a More Reproducible Biomedical Research Environment: Endorsement and Adoption of the FAIR	paper Conference	excluded	to OS keyword mismatch: not directly related	2020	http://link.springer.com/chapter/10.1007/978-3- 030-46970-2_22
Springer	08.12.2024	André OliveiraFilipe PortelaManuel Filipe	Principles Towards an Intelligent System for Monitoring Health	Conference	excluded	to US keyword mismatch:	2016	http://link.springer.com/chapter/10.1007/978-3- 319-42089-9-45
Springer	08.12.2024	Klaus KadesJonas SchererMaximilian ZenkMarius	Towards Real-World Federated Learning in Medical Image	paper	excluded	to OS keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978-3- 021-18522-6-12
Springer	08.12.2024	KempfKlaus Maier-Hein	Analysis Using Kaapana	Conference paper	excluded	not directly related to OS	0000	http://link.covinger.com/objects/10.1007/078-2
Springer	08.12.2024	Akio Sashimamitsuru Kawamoto	Using Wearable Device	Conference paper	excluded	not directly related to OS	2020	030-43887-6_5
Springer	08 12 2024	Ricardo Colomo-PalaciosLuis Omar Colombo- MendozaRafael Valencia-García	Towards Supporting International Standard-Based Software Engineering Approaches Using Semantic Web Technologies: A Systematic Literature Baview	Conference	excluded	keyword mismatch: not directly related	2016	http://link.springer.com/chapter/10.1007/978-3- 319-48024-4_14
Springer	09 12 2024	Carina AlvesJoyce OliveiraSlinger Jansen	Understanding Governance Mechanisms and Health in Software Ecosystems: A Systematic Literature Review	Conference	oveluded	keyword mismatch: not directly related	2018	http://link.springer.com/chapter/10.1007/978-3- 319-93375-7_24
Springer	00.12.2024	Magnus Li	Usability Problems and Obstacles to Addressing Them in Health Information Software Implementations	Conference	excluded	keyword mismatch: not directly related	2019	http://link.springer.com/chapter/10.1007/978-3- 030-19115-3_20
Springer	08.12.2024	Paul AlparThomas DriebePeter Schleussner	Using 3D-Technology to Support Facial Treatment	paper	excluded	to OS keyword mismatch:	2022	http://link.springer.com/chapter/10.1007/978-3- 030-95947-0_34
Springer	08.12.2024	Louise BlochChristoph M. Friedrich	Lising Revesian Ontimization to Effectively Tune Random	paper	excluded	to OS	2021	http://ink.springer.com/chapter/10.1007/978-3-
Springer	08.12.2024		Forest and XGBoost Hyperparameters for Early Alzheimer's Disease Diagnosis	Conference paper	excluded	not directly related to OS	2021	030-70569-5_18
		Susana BulinasAna Paula ClaudioMana Beatriz CarmoJoão BalsaAfonso CavacoAnabela MendesIsa FélixNuno PimentaMara Pereira	Type 2 Diabetes: First Prototype	Conference		keyword mismatch: not directly related	2019	030-16028-9_21
Springer	08.12.2024	Guerreiro Marc VilaMaria-Ribera SanchoErnest Teniente	XYZ Monitor: IoT Monitoring of Infrastructures Using	paper	excluded	to OS keyword mismatch:	2021	http://link.springer.com/chapter/10.1007/978-3- 030-76352-7_43
Springer	08.12.2024		Microservices	paper	excluded	to OS		3000 / 3000a / _ TX
ACM Digital	22.06.2024	Michael & Cusumano	Constative All as a New Innovation Platform	Articlo	evoluded	not directly related	2023	https://doi.org/10.1115/2015050
Library	22.00.2024	Prof. Dr. Thomas TolxdorffProf. Dr. Thomas M. DesernoProf. Dr. Heinz HandelsProf. Dr. Andreas	Bildverarbeitung für die Medizin 2020	Alloce	excluded	to nearricare	2020	http://link.springer.com/book/10.1007/978-3- 658-29267-6
Springer	07.12.2024	MaierDr. Klaus H. Maier-HeinProf. Dr. Christoph Palm		Conference paper	excluded	language: not english		
		Prof. Dr. Klaus Maier-HeinProf. Dr. Thomas M. DesernoProf. Dr. Heinz HandelsProf. Dr. Andreas	Bildverarbeitung für die Medizin 2022				2022	http://ink.springer.com/book/10.1007/978-3- 658-36932-3
Springer	07.12.2024	MaierProf. Dr. Christoph PalmProf. Dr. Thomas Tolxdorff	Dikhararhaituna für dia Madizia 2022	Conference paper	excluded	language: not english	2022	http://link.spripger.com/book/10.1007/978-3-
Springer	07.12.2024	MaierKlaus Maier-HeinChristoph PalmThomas Tolxdorff	bildverarbeitung iur die medizin 2023	Conference paper	excluded	language: not english	2023	658-41657-7
Springer	08.12.2024		Mitteilungen der Gesellschaft für Informatik 240. Folge	Article	excluded	language: not english	2016	http://link.springer.com/article/10.1007/s00287- 016-0976-3
Springer	08.12.2024	Dr. Jan MesserschmidtProf. Dr. Paul MolitorProf. Dr. Jürgen Steimle	Festschrift zum 90. Geburtstag von Prof. Dr. Dr. h.c. mult. Günter Hotz	Conference paper	excluded	language: not english	2022	http://link.springer.com/book/10.1007/978-3- 658-37822-6
		Claus-Peter SchnorrHermann WalterVolker ClausWolffried StuckyHans-Peter BlattOtto SpaniolGerd KaufholzRainer KempWolfgang PaulEberhard BertschHerbert KoppManfred	Wissenschaftliche Kinder von Günter Hotz				2022	http://ink.springer.com/chapter/10.1007/978-3- 658-37822-6_1
Springer	08.12.2024	StadelUlrich SchmittWolfgang WeidnerWolfgang GräberDung Huynh		Conference paper	excluded	language: not english		
		Devi R.; Mehrotra D.; Lamine S.B.A.B.; Baazaoui- Zghal H.	A RDF model-based predictive model on dengue patients' diagnostic PDF reports	Article			2023	https://www.scopus.com/inwardirecord.un/eid=2 s2.0-85161873364&doi=10.1504%2fJEH, 2023.130513 Asstration 20.2020 and 20.002 and 20.0
Scopus	28.06.2024	Wolf H.; Bruin C.; Dobbe J.G.G.; Gordijn S.J.;	Computerized fetal cardiotocography analysis in early preterm	Article	excluded	type: no access	2019	apartneriD=auands=918070650e7677387600 9d1fb184ed https://www.scopus.com/inward/record.uri?eid=2 s2.0-85064898573&doi=10.1515%2fjpm-2018-
Scopus	01.07.2024	Cui M.; Baek SS.; Crespo R.G.; Premalatha R.	applications Internet of things-based cloud computing platform for analyzing	Article	excluded	type: no access	2021	0412&partnerID=40&md5=49ber5a9ef6e203ae8 c5db704631edb https://www.scopus.com/inward/record.uri?eid=2 20.951187381388.edu=10.32338.2014C
Scopus	01.07.2024	West 8 H	the physical health condition	A-17-1-	excluded	type: no access	0000	213003&partnerID=40&md5=a802ee33107fd536 64f4ee22a4e27f1
		WOOD R.W.	Modelling the impact of COVID-19 on elective waiting times	Arucie			2022	s2.0-85086330420&doi=10.1080%2117477778. 2020.1764876 &partner/D=408md5=3340bas5ed2afdfdb78d7b
Scopus	29.06.2024	Do S.	The future of artificial intelligence for physicians	Article	excluded	type: no access	2016	4ef672025 https://www.scopus.com/inward/record.uri?eid=2 s2.0-84975887165&doi=10.5124%2fjkma.
Scopus	02.07.2024				excluded	type: no access		2016;59:6;410 &partneriD=40&md5=bbaac33efe7b6fe8bfcbe74 33c12230
		Skripcak T.; Just U.; Simon M.; Buttner D.; Luhr A.; Baumann M.; Krause M.	Toward Distributed Conduction of Large-Scale Studies in Radiation Therapy and Oncology: Open-Source System Integration Approach	Article			2016	https://www.scopus.com/inward/record.uri?eid=2 s2.0-84986550138&doi=10.1109%2fJBHI. 2015.2450833 Bender=10_f08_mdf=0fdt 44047000000055
Scopus	02.07.2024	Akhshirsh G.; Azeez B.; Bezenchek A.; Fanti I.;	A workflow to develop and implement an E-Health information	Article	excluded	type: no access	2022	SpartnerID=40&md5=9(4)4(1/90/224280(085c) bbb/039bf https://www.scopus.com/inward/record.uri?eid=2
Scopus	28.06.2024	Salih S.O.; Basa F.B.; Malizia A.; Moramarco S.; Gialloreti L.E.	system in war-torn countries: A case study in Iraqi Kurdistan	Conformação	excluded	type: no access	2010	8514250453&partnerID=40&md5=3e13c61d0cr 4ff3115647152c0d90065
Scopus	06.12.2024	Moreno-Conde J., Avarez-Romero C., Suarez- Mejías C.; Martínez-Maestre M.Á.; Silvan-Alfaro J. M.; Parra-Calderón C.L.	prescription of genetic tests in the gynecological cancer risk	paper	excluded	type: no access	2019	s2.0-85071516656&doi=10.3233% 2fSHTi190314&partner/D=40&md5=4fa387956c ee2361587d4a29ff9e4e
		Tyndall T.; Tyndall A.	FHIR Healthcare directories: Adopting shared interfaces to achieve interoperable medical device data integration	paper			2018	s2.0-85049038956&doi=10.3233%2f978-1-6149 868-6-
Scopus	06.12.2024	Gil E.B.; Caldas R.; Rodrigues A.; Da Silva G.L.G.; Rodrigues G.N.; Pelliccione P.	Body Sensor Network: A Self-Adaptive System Exemplar in the Healthcare Domain	Conference paper	excluded	type: no access	2021	To repartment/~worlddar/bser/symmologics eca3c98e4188 https://www.scopus.com/inward/record_uri?eid=2 s2.0-851135462758doi=10.1109% g9ECAMSE3265.2002
Scopus	05.12.2024				excluded	type: no access		255EAM/SD1201.2021.0003/ &partnerID=40&md5=d5d7616d89397192b0c61: 3a440d8090
		Rinner C.; Duttschmid G.	Bridging the gap between HL/ CDA and HL7 FHIR: A JSOn based mapping	conterence paper			2016	1008.patherid 240.000 and 240.0000 and 240.000 and 240.000 and 240.000 and 240.000 and 240.000 and 240.0000 and 240.00000 and 240.0000 and 240.00000 and 240.0000 and 240
Scopus	05.12.2024	P. Datta; S. Roy; A. Kumar Rana; P. Kumar Sinha	Low-cost Robot for smart healthcare services in Hospital	Conference	excluded	type: no access	2023	8b2d7ae31e5 10.1109/SILCON59133.2023.10404435
IEEE Xplore	05.12.2024			paper	excluded	type: no access keyword mismatch:		
ACM Digital Library	22.06.2024	Aaditeshwar Seth	what's missing in the ACM Code of Ethics and Professional Conduct	Article	excluded	to healthcare keyword mismatch:	2023	https://doi.org/10.1145/3588003
ACM Digital Library	22.06.2024	Ajay Deshpande	Careers in computer science: a three-tiered architecture	Article	excluded	not directly related to OS	2020	https://doi.org/10.1145/3416057
Soopur	06 12 2024	barakat A.N.; Ambark T.M.; Bozed K.A.	remote Healthcare Monitoring System using IoT platform	conference paper	oveluted	type: not reliable	2021	rsups//www.scopus.com/inward/record.uri?eid=2 s2.0-65122041389&doi=10.1145%273492547, 3492628&partnerID=40&md5=46cd8a8dac3c49 Olade 45: 204-52%24.5
Joopus	00.12.2024				enoluded	-ypor not reliable		21212407 200 17 00 12

Search venue	Query date	Author	Title	Туре	Status	Reason	Year	Article URL
IEEE Xplore	05.12.2024	D. Sivabalaselvamani; D. Selvakarthi; L. Rahunathan; S. N. Eswari; M. Pavithraa; M. Sridhar	Investigation On Heart Disease Using Machine Learning Algorithms	Conference paper	excluded	type: not reliable	2021	10.1109/ICCCI50826.2021.9402390
Springer	08.12.2024	Liping WeiYishuo Yang	RETRACTED ARTICLE: Development trend of sharing economy in big data era based on duplication dynamic evolution game theory	Article	excluded	type: retracted article	2019	http://link.springer.com/article/10.1007/s10586- 018-1832-9
Springer	08.12.2024	Rana M. Amir LatifKhalid HussainN. Z. JhanjhiAnand NayyarOsama Rizwan	RETRACTED ARTICLE: A remix IDE: smart contract-based framework for the healthcare sector by using Blockchain technology	Article	excluded	type: retracted article	2022	http://link.springer.com/article/10.1007/s11042- 020-10087-1
Springer	08.12.2024	Victor Chang	RETRACTED ARTICLE: An overview, examples, and impacts offered by Emerging Services and Analytics in Cloud Computing virtual reality	Article	excluded	type: retracted article	2018	http://link.springer.com/article/10.1007/s00521- 017-3000-1
Springer	08.12.2024	Amir JavadpourGuojun WangSamira RezaeiKuan- Ching Li	RETRACTED ARTICLE: Detecting straggler MapReduce tasks in big data processing infrastructure by neural network	Article	excluded	type: retracted article	2020	http://link.springer.com/article/10.1007/s11227- 019-03136-6
Springer	08.12.2024	S. HemavathiB. Latha	RETRACTED ARTICLE: FRHO: Fuzzy rule-based hybrid optimization for optimal cluster head selection and enhancing quality of service in wireless sensor network	Article	excluded	type: retracted article	2023	http://link.springer.com/article/10.1007/s11227- 023-05106-5

## **Appendix C**

## **Open-Source Foundations and Open-Source Projects in Healthcare Literature Review Protocol V1**

(Guideline: Kitchenham, 2004)

#### 1. Purpose of the Review

#### 1.1 Research Objective

The aim of this literature review is to provide an overview of the state of the art in open-source software (OSS) research within the healthcare sector. It seeks to identify and analyze examples of OSS foundations and projects, as well as to gather detailed information about their practical implementations and trends in the related literature.

#### **Expected outcomes are:**

- A comprehensive list of published research on OSS in healthcare, including details such as author, title, year, publication type (journal or conference paper), research topic, topic keywords, journal, and search venue
- Identification and classification of open-source foundations relevant to the healthcare domain
- Analysis of open-source projects, highlighting their roles, applications, and impacts in the healthcare sector

#### 1.2 Research Questions

**RQ1**: What are the research trends regarding open-source foundations and projects in healthcare between 2016 and 2023?

**RQ2:** Which open-source foundations have been explored in the literature?

RQ3: Which open-source projects have been studied in the literature?

#### 2. Search Strategy

#### 2.1 Search Process

The literature review will be conducted by searching the defined search term across selected electronic databases and libraries. The search will be restricted to papers published in English to ensure consistency and accessibility.

- Search Term: "open source" AND "healthcare"
- Databases: Google Scholar, ACM Digital Library, IEEE Xplore and Scopus

The search will target titles, abstracts, and keywords within the specified databases.

#### 2.2 Documenting the Search

The following information will be gathered during the search process:

- Name of the Search Venue
- Name of the Authors
- Title of the Paper
- Year of Publication
- Type of Publication (e.g., journal article, conference paper)
- URL

Table 1 is how our table is planned to be structured to document the key information from the literature search process.

Table 1: Literature Search Documentation

Search Venue	Authors	Title	Year of Publication	Туре	URL

#### **3. Selection Process**

After collecting the potentially relevant papers, we will evaluate these to determine if we should include them in the review. Papers that do not meet the defined criteria will be excluded, and the reasons for exclusion will be documented in a list of "excluded papers". The exclusion reason could be added to the list (Table 2).

#### **Potential Exclusion Criteria**:

- Papers not written in English
- Duplicate papers
- Studies that are not peer-reviewed research publications
- Papers that do not focus on open-source software (OSS) in healthcare

#### 4. Data Extraction Process

The data extraction process will focus on gathering detailed information from the selected studies to answer the research questions. The following information will be systematically extracted from each included paper:

- Date of Query
- A summary of the paper's content
- Key topics addressed in the study
- Publication Venue
- Mentioned open-source projects
- Mentioned open-source foundations

The extracted data will be documented in a list of "included papers" (Table 2).

Search Venue	Query Date	Authors	Title	Туре	Publication Venue	Year of Publication	Sum mary	Topic keyword	URL

Table 2: Example of Extracted Data from Selected Studies

### References

Kitchenham, B. (2004). Procedures for performing systematic reviews. Keele, UK, Keele University, 33(2004), 1-26.

# **Open-Source Foundations and Open-Source Projects in Healthcare Literature Review Protocol V2**

After conducting the initial search, we revised and improved our protocol to refine the process for the second search.

#### **Updated Search Strategy**

- Search Term: "open source software" AND "healthcare"
- Databases: ACM Digital Library, IEEE Xplore, Scopus, Springer

To improve the relevance of our results, we refined our search strategy by using the search term "open source software" AND "healthcare", as the broader term led to an overwhelming number of irrelevant results.

Additionally, we decided to exclude Google Scholar from the list of databases, as it included too many irrelevant papers. Instead, we included Springer in our search to ensure a broader and more focused coverage of relevant publications.